NUREG/CR-5813 EGG-2664 Vol. 1

isk Analysis System (IRRAS) ersion 4.0

ference Manual

ared by Russell, M. K. McKay, M. B. Sattison, N. L. Skinner, Wood, D. M. Rasmuson

to National Engineering Laboratory kG Idaho, Inc.

ared for Nuclear Regulatory Commission

202110290 920131 PDR NUREG CR-5813 R PDR

NUREG/CR-5813, Vol. 1 INTEGRATED RELIABILITY AND RISK ANALYSIS JANUARY 1992 SYSTEM (IRRAS) VERSION 4.0

NUREG/CR-5813 EGG-2664 Vol. 1 RG

Integrated Reliability and Risk Analysis System (IRRAS) Version 4.0

Reference Manual

Manuscript Completed: December 1991 Date Published: January 1992

Prepared by K. D. Russell, M. K. McKay, M. B. Sattison, N. L. Skinner, S. T. Wood, D. M. Rasmuson*

Idaho National Engineering Laboratory Managed by the U.S. Department of Energy

EG&G Idaho, Inc. Idaho Falls, ID 83415

Prepared for Division of Safety Issu. Jution Office of Nuclear Regulatory Research U.S. Nuclear Regulatory Commission Washington, DC 20555 NRC FIN L1429 Under DOE Contract No. DE-AC07-76ID01570

*U.S. Nuclear Regulatory Commission Washington, DC 20555



AVAILABILITY NOTICE

Availability of Reference Materials Cited in NRC Publications

Most documents olted in NRC publications will be available from one of the following sources:

- 1. The NRC Public Document Room, 2120 L Street, NW., Lower Level, Washington, DC 20555
- The Superintendent of Documents, U.S. Government Printing Office, P.O. Box 37082, Washington, DC 20013-7082
- 3. The National Technical Information Service, Springfield, VA 22161

Although the listing that follows represents the majority of documents cited in NRC publications, it is not intended to be exhaustive.

Referenced documents available for inspection and copying for a fee from the NRC Public Document Room includs NRC correspondence and internal NRC memoranda: NRC bulletins, circulars, information notices, inspection and investigation notices, inspection and incenses event reports; vendor reports and correspondence; Commission papers; and applicant and licenses documents and correspondence.

The following documents in the NUREG series are available for purchase from the GPO Sales Program: formal NRC staff and contractor reports. NRC-sponsored conference proceedings, international agreement reports, grant publications, and NRC booklets and brochures. Also available are regulatory guides, NRC regulations in the Code of Federal Regulations, and Nuclear Regulatory Commission Issuances.

Documents available from the National Technical Information Service include NUREG-series reports and technical reports prepared by other Federal agencies and reports prepared by the Atomic Energy Commission, forerunner agency to the Nuclear Regulatory Commission.

Documents available from public and special technical libraries include all open literature items, such as hooks, journal articlas, and transactions. *Federal Register* notices, Federal and State legislation, and congressional reports can usually be obtained from these libraries.

Documents such as theses, dissertations, foreign reports and translations, and non-NRC conference proceedings are available for purchase from the organization sponsoring the publication cited.

Single copies of NRC draft reports are available free, to the extent of supply, upon written request to the Office of Administration, Distribution and Mail Services Section, U.S. Nuclear Regulatory Commission, Washington, DC 20555.

Copir's of industry code, and standards used in a substantive manner in the NRC regulatory process are maintained at the NRC Library, 7920 Norfolk Avenue, Bethesda, Maryland, for use by the public. Codes and standards are usually copyrighted and may be purchased from the originating organization or. If they are American National Standards, from the American National Standards Institute, 1430 Broadway, New York, NY 10018.

DISCLAIMER NOTICE

This report was prepared as an account of work sponsored by an agency of the United States Government. Neither the United States Government nor any agency thereof, or any of their employees, makes any warranty, expressed or implied, or assumes any legal liability of responsibility for any third party's use, or the results of such use, of any information, apparatus, product or process disclosed in this report, or represents that its use by such third party would not infringe privately owned rights.



ABSTRACT

The Integrated Reliability and Risk Analysis System (IRRAS) is a state-of-the-art, microcomputer-based probabilistic risk assessment (PRA) model development and analysis tool to address key nuclear plant safety issues. IRRAS is an integrated software tool that gives the user the ability to create and analyze fault trees and accident sequences using a microcomputer. This program provides functions that range from graphical fault tree construction to cut set generation and quantification.

Version 1.0 of the IRRAS program was released in February of 1987. Since that time, many user comments and enhancements have been incorporated into the program providing a much more powerful and user-friendly system. This version has been designated IRRAS 4.0 and is the subject of this Reference Manual. Version 4.0 of IRRAS provides the same capabilities as Version 1.0 and adds a relational data base facility for managing the data, improved functionality, and improved algorithm performance.

Fin No. L1429 - SARA, IRRAS, and MAR-D Maintenance and User Support

Previous Reports in the Series

K.D. Russell et al., Integrated Reliability and Risk Analysis System (IRRAS) Version 2.5 Reference Manual, NUREG/CR-5300, EGG-2613, March 1991.

K.D. Russell, M.B. Sattison, D.M. Rasmuson, Integrated Reliability and Risk Analysis System (IRRAS) - Version 2.0 User's Guide, NUREG/CR-5111, EGG-2535, manuscript completed March 1980, published June 1990

K.D. Russell, D.M. Snider, M.B. Sattison, H.D. Stewart, S.D. Matthews, K.L. Wagner, Integrated Reliability and Risk Analysis System (IRRAS) User's Guide - Version 1.0 (DRAFT), NUREG/CR-4844, EGG-2495, June 1987





CONTENTS

ABSTRACT	ij
FOREWORD	Ŷ
ACKNOWLEDGMENTS	ii
EXECUTIVE SUMMARY	1
1. INTRODUCTION	1
2. GETTING TO KNOW IRRAS 4.0	1
2.1 IRRAS 4.0 Modules	1
2.2 IRRAS 4.0 Data Base Concepts	3
2.3 Using IRRAS for the First Time	5
2.4 Getting Around Inside IRRAS 4.0	7
2.5 Invcking IRRAS	8
2.6 Hot Keys	8
2.7 Menu Options Naming Conventions	8
2.8 Other Things You Should Know	0
3. SELECT FAMILY	1
3.1 Exit	2
3.2 Select 3-	2
3.3 Family Copy	2
3.4 Conv.	2
4 GENERATE EVENT DATA	1
4.1 Evit	1
4.9 A.H	1
The Aud an a construction of the second seco	R.

v



4.3	Modify				i.		4	i. La constante da la constante da							Kk	4-3
4.4	Delete .		0.8.90.000	- 34 - 34 -	1.11	6 × 4			a a ja	1.1		$\leq \cdot \leq \lambda$	4 1 A 1 6	$< \pm$	$i \ll i$	4-3
4.5	Probabili	tv														4.4
4.5	riopaoni	9	4.4.4.4.4.4.4	1.1.15	5.3	5 16 14	8.9	XXX		(X X)			8. j. 18		1 H K	14-14
	4.5.1	Exit														4.5
	4.5.2	Probability										1.1.1		4.4.1		4-5
	453	Reset Probability	to Base			5-4-4	-X. 8	1.1.1	1.11.1			43.4	4.4.4		e (e)e	4.10
	4.00.00	Reser Freedounty	no trase		91 N.	97 K. 34	3. K.	a	8-36-3 -			* * *	* 8. 8	14	×	4-10
4.6	Class			a a				i.					* * *	2.4	e si	4-11
4.7	Histogram	ns	101 A X X -	ch is	1.	e a g	$\kappa_{i,k}$			((A)	×. i	<pre>x</pre>	1.1	$q \rightarrow 1$	÷ 4	4-12
	4.7.1	Exit		l é a	× × -		e^{-x}	6 9 X	1.1.	de co		e x 4	х. ж.Э		÷	4-14
	4.7.2	Add Histograms	dares	i i i	28.	i e a	4.4					1.		1.5	÷	4-14
	4.7.3	Modify Histogra	ms	e a la	2.4	i a a			S - 4 - 4		e p la	i e e	A a a	4.4		4-17
	4.7.4	Delete Histogram	ns								e se se	r i s			i	4-19
4.8	Generate	Changes		Ar	× × 1	1.1	1.12	e. 11. 14.				e é a			х. ж	4-20
4.9	Report C	hanges	44.644	3.5	(\cdot, \cdot)	1 5 8	$\boldsymbol{\lambda}_{i} \rightarrow \boldsymbol{\lambda}_{i}$	4 12 14	24.9	2.4.2	0.00	1.47	÷ 4, 4	4.4	1	4-22
	4.0.1	Parts														1 22
	4.9.1	EXIL	1.1.1.1.1.1	(tex	11	1 1 3	$X \rightarrow 0$	× 8.4	4.4.7	13.14.1		4. A. K.	1.4.4	8.00	1. X.	4-22
	4.9.2	Unaffected Even	ts (Pase)	24.7	1.	e ej a	$\mathbb{R}^{ X }$	1, 16, 16	3 X I	-8.8	1.1.1	12.4	1.73	1.1	5 (K -	4-22
	4.9.3	AFfected Events	(Current	2 .	1.1	5.8.8	β_{i},β_{i}	i, ese	1.8.9		6.8.8	0.68	121	* *	έg.	4-23
	4.9.4	All Lyents (Base	, Current	9 .	8.8	133	1.1	12.7	199	1.4	8.4	13.5	144	1.1	13.	4-24
4.10	Base Ca	se Update			÷	÷.,		. 3	١.							4-24
BUILD	FAULT T	REES		à.		di k	1.5	i i i		de se o	en de	i. Na s			i ka	5-1
5.1	Build Gra	aphic Trees	41144	6.8	6-6	6 x 7	$\vec{s} \neq \vec{s}$		14	4.4.5	i, k	5 K. F	4.4.4	5.4	і. 1. н. н	5-2
	5.1.1	EXIT	1.1.1.1.1.1.1.1	4.4	2.4	t fille	$\epsilon_{\rm c}$	64.5	126	4.4.9	1.1	1.1	1.1.1		1.1.1	5-2
	5.1.2	Move	an kan s	16-16	si di s	e e de	8.3	8 (18) (18)	10 M A	R DC 3	4.4	e e les	8.5.8	1.50	é a á	5-2
	5.1.3	SHOW	والم فراج و	÷х	1.1	e é é	11	6.54	$s \approx 0$	$(-, \cdot)$	4.4	e x Fr	1.1.2	8.4.3	1.1	5-2
	5.1.4	BILD	1444	3.5		i ka	< x	Če A	(, k, i)	$\vec{x} \in \vec{x}$	e Kiste	1.1	1.1.2	(χ,χ)	16.8	5-2
	5.1.5	EDIT	11000	1.1	8.6.5	1.1	$\leq - \leq$	1.1	$i \rightarrow i$		114	e e i	1.4.5	4.4		5-8
	5.1.6	TEXT	ha dha y	la a	6	(and	$\hat{\mathbf{x}} \in$	1.11	2.5.4	a si	dia.	Èn. A	1.1.3			5-18
	5.1.7	VIEW		1.8	s a s		Âх.	121	x s is	1.63	* 4	e ka	× 11. k	$\sim 1e$	4.4	5-22
	5.1.8	FILE		1.1	1.1		6. N.	1.5.5	k a 4			1.1	1.1.1	1.1		5-24
	5.1.9	NAME												\bar{x} is		5-27
	5.1.10) Text	dia and	1.0	1		4.8	e a la	h na là	3.8.5	-6 X	1.1		x 9.		5-27
	5.1.11	cntr/left/right .		lew.	1.1		1.4	8 . N. A	1.5		e na Sa					5-29
	5.1.12	Text Size (0.50)		1.1	in i	5.4	die a				c é e				5-29
	5.1.13	Fill	*****	1 1 1 1 7 1			11						1	4.1	÷., ;	5-29
	5.1.14	Line		i e		e é a	66			- * * 2			1.4.2	+ 1		5-29

5

5.1. 5.1. 5.1.	15 type 5- 16 Mult 5- 17 Scroll Color Bar († 4) 5-	29 30 36
5.2 Plot Tr	ees	10
5.2. 5.2. 5.2. 5.2.	1 Plotter 5-3 2 Hpgl Files 5-3 3 Raster File 5-3 4 Low Resolution Raster 5-3	11
5.3 Load G	raphic Trees	13
5.3. 5.3.1	1 Exit	33
5.4 Alpha te	o Graphics	4
5.4.1 5.4.2	Exit	4
5.5 Graphic	al Pager	5
5.5.1 5.5.2	1 Exit	5
5.6 eXtract	Fault Trees	7
5.6.1 5.6.2 5.6.3 5.6.4	Exit 5-3 2 eXtract Trees 5-3 3 Clear Extracted Trees 5-3 4 Display Extracted Trees 5-3	7 7 8 8
5.7 Define I	Plotter Pens	0
.0 FAULT TREE	ANALYSIS	1
6.1 Exit		1
6.2 Analyze	Systems	1
6.2.1 6.2.2 6.2.3 6.2.4 6.2.5	Exit 6- Generate Cut Sets 6- Cut Set Update 6- Quantification 6- Uncertainty Analysis 6-	33688
6.3 Display	Results	1

	6.3.1	Exit																		1	6-13	
	622	Cutente																		1.2	6.12	
	6.3.6	Cuiseis	* * * * *		14 14 1			ALX.	× 7		1.20						63			1.1	2 1 6	
	0.3.3	Importan	ce	16.16.14	N N	e kar		15 A	1.1	6 H H		4.16	1.1				10.1			1.5	0-18	
	6.3.4	Uncertain	nty		1.11							(1, 2)		11							6-20	
6.4	Cut Set E	iditor	l'abra a	d da	× 8.	i a a						A 16								. – ł	6-20	
	641	Exit																		. 1	6-23	
	643	Add			2.3			2.12				11									6.23	
	643	Mailler	14 3 3 8 9 9	1.11		1.8.9			1.1		1.15.13			5 (6) 3		1.10.1			5.5.5	1.1	6.96	
	0.4.3	Moulty	A & X & Y	3,8,8	e le		\rightarrow \propto		× ×	1.6.1	1.1.1	2.5	× 9.		6.85	x - x - x				6.3	0-20	
	0.4.4	Delete .	A 3-3-5-5	(X,Y)	-10-1			[0, -1]	2.2		0.00	× 18	(d_{i},\ldots,d_{i-1})			X (X)				11	0-20	
	6.4.5	Locate .	$k \in [k, k] \times [k]$		$\mathcal{K} = \mathcal{K}$	ex.		5 x.	× 4.1		ene la	$E \ge 0$	$ \hat{u}_{i} ^{2} \leq u_{i} ^{2}$					έ÷,	6.913	1.1	6-28	
	6.4.6	Next	12.2.2.2.2	1.1.1.1	+10				4.3	e e e		$ A \geq 0$	3.8		e e e			14	è e s	1.1	6-29	
	6.4.7	Previous		is is is	11	da i		2.4		i e e	d de la	11			e sé		181	e el	ĥe,	. 1	6-29	
	5 . 8	Search .		25.				1								1.1				. 14	6-29	
	64	Ontions																			6-30	
	6 4 10) Invart E	and the second						1								1				6.31	
	0.4.10	Disen E	-vent	1.1	11.5		n an an T	9, K	8.8		5. 16. 19		8.87	8.4.					11		6 21	
	0.4.17	Replace	Events	1.1.1	1.0	12			11		1.15.13	+ +	9.3	1	÷ ×			r	140	1.1	0-31	
	6.4.12	2 Copy C	utset ar	id Re	plac	ęЕ	vent	S .	$\hat{x} \neq$	E F.	6.4.9			ξ÷.	81 K.		5.85		8.85	1.1	0-31	
	6.4.13	3 Undo .	4.8.8.4.4	$ k_{i} ^{2}$		1.63			$\hat{A} = \hat{A}$	1.10			(n, n)							1.1	6-33	
	6.4.14	Find an	d Delet	e Cu	t Set	i ya	i s d			i di		1.1	1.5	ŝ.		e a j			÷ 11.	1.1	6-33	
6.5	Logic Ed	litor														l u i					6-33	
4.195	molet too																					
	100	Train																			6.22	
	0.5.1	EXIL		199	14	1.1	1.1	1.1	11	4. 9. 3	19.9	10.8	3.3	. 1	12	11. A.			1		6.25	
	6.5.2	Logic 1.0	litor .	1.8.9	10.00	2.3.2	en k	ä a	81		6.819		1.1	1.1	e de	893	12	12	1,1	1.7	0-33	
	6.5.3	Sub-Syst	em	e i c i o	in pr	к я, с		$\times 1$	1.16		e tij	14	$\dot{v} \neq i$	ŧ E	14	e e s		1.5	6.4	5.7	6-41	
CREA'	FE EVEN	T TREES	1				1.	s k		1	2.5-1	i si x	4.4	a ai	1.1		1.	i.		a di	7-1	
7 1	Freit																				7-2	
1.1	Acres 1.2	*****							12	22			14				н.					
	Sume 1	and therein																			7.9	
1.4	Create E	vent Trees	S. e. s. e.					8.16	(k, k)	12	99	1	11				11			12	1.0	
		-																				
	7.2.1	EXIT .							$\tilde{v} \sim t_{i}^{\prime}$	14	114.1	e fer	$\mathbf{x} \neq \mathbf{x}$						1.1		7-2	
	7.2.2	MOVE (•+) .							λę.			(\mathbf{x}, \mathbf{x})	8.8					8.8	6.6	7-2	
	7.2.3	Show .							1.5		de la composición de La composición de la c		a d						10	1.1	7-2	
	7.2.4	EDIT .											2.1		ä.				5	de la	7-2	
	725	TOPS														1.1					7-6	
	736	ENDE	1.1.1.1			1.1			1.1	10			13	1.3	1	191					7-11	
	1.4.0	ENDS .	1 2 1 4					11			177										7.16	
	1.2.1	TEAT .	$\xi = \xi - \xi$	6 S. K. J	1.1			[1,1]	10	1.1		1.1	1.1	4.4	X. X.			1.1	8. 8	×	9 00	
	7.2.8	VIEW .	$A_{-}(\xi) = K_{-}(\xi)$							×. 11		2.1	4.4						1.1		1-23	
	7.2.9	FILE .								÷ k	6.4.3	11	\mathbf{x} +	8.4		1.5			8 5		7-25	
	7.2.10	0 TRAN								14	1.6.			1.1				x. 11	1.1		7-27	
	7.2.1	1 Text									11.	1.	14	- x				1.11	11		7-29	
	721	2 cntr/lef	Vrght							3				1				Ξ.	11		7-29	
	7 7 1	3 Toxt Si	28																		7-30	
	1 . Ar . 3 .	1 1 10/01 131	100 C 1					12. 8			A. 18. 1	1 1 13					1.18	1.1				
	19 19 4	4 Linn																			7.30	
	6.4 6.5 CREAT 7.1 7.2	6.3.1 6.3.2 6.3.3 6.3.4 6.4 Cut Set F 6.4.1 6.4.2 6.4.3 6.4.4 6.4.5 6.4.6 6.4.7 6.4.1 6.5.2 6.5.3 CREATE EVEN 7.2.1 7.2.3 7.2.4 7.2.5 7.2.6 7.2.7 7.2.8 7.2.9 7.2.1 7.2.1 7.2.1 7.2.1 7.2.1 7.2.1 7.2.1 7.2.1	6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importan 6.3.4 Uncertain 6.4 Cut Set Editor 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous 1.8 Search 6.4.7 Previous 1.8 Search 6.4.10 Insert E 6.4.10 Insert E 6.4.11 Replace 6.4.12 Copy C 6.4.13 Undo 6.4.14 Find an 6.5 Logic Editor 6.5.1 Exit 6.5.2 Logic 2d 6.5.3 Sub-Syst CREATE EVENT TREES 7.1 Exit 7.2 Create Event Trees 7.1 Exit 7.2 Create Event Trees 7.2.1 EXIT 7.2.3 Show 7.2.4 EDIT 7.2.5 TOPS 7.2.6 ENDS 7.2.6 ENDS 7.2.10 TRAN 7.2.11 Text 7.2.10 TRAN 7.2.11 Text 7.2.12 cntr/lef	 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous . 3 Search 6.4.7 Previous . 3 Search 6.4.10 Insert Event 6.4.10 Insert Event 6.4.12 Copy Cutset at 6.4.13 Undo 6.4.14 Find and Delet 6.5.1 Exit 6.5.2 Logic Editor 6.5.3 Sub-System CREATE EVENT TREES 7.1 Exit 7.2 Create Event Trees 7.2.1 EXIT 7.2.2 MOVE (↔) 7.2.3 Show 7.2.4 EDIT 7.2.5 TOPS 7.2.6 ENDS 7.2.7 TEXT 7.2.9 FILE 7.2.10 TRAN 7.2.11 Text 7.2.12 cntr/left/rght 	 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4 Cut Set Editor 6.4.2 Add 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous 8 Search 6.4.10 Insert Event 6.4.10 Insert Event 6.4.11 Replace Events 6.4.12 Copy Cutset and Re 6.4.13 Undo 6.4.14 Find and Delete Cu 6.5 Logic Editor 6.5.1 Exit 6.5.2 Logic Editor 6.5.3 Sub-System CREATE EVENT TREES 7.1 Exit 7.2 Create Event Trees 7.2 Create Event Trees 7.2 Show 7.2.4 EDIT 7.2.5 TOPS 7.2.6 ENDS 7.2.7 TEXT 7.2.8 VIEW 7.2.9 FILE 7.2.10 TRAN 7.2.11 Text 7.2.12 cntr/left/rght 	 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous 6.4.0 Prions 6.4.10 Insert Event 6.4.10 Insert Event 6.4.11 Replace Events 6.4.12 Copy Cutset and Replac 6.4.13 Undo 6.4.14 Find and Delete Cut Set 6.5.1 Exit 6.5.2 Logic Editor 6.5.3 Sub-System CREATE EVENT TREES 7.1 Exit 7.2 Create Event Trees 7.1 Exit 7.2 Create Event Trees 7.2 Create Event Trees 7.4 EDIT 7.2.5 TOPS 7.2.6 ENDS 7.2.7 TEXT 7.2.8 VIEW 7.2.1 Text 7.2.10 TRAN 7.2.11 Text 7.2.12 cntr/left/rght 	 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous . · 3 Search 6.4.0 Insert Event 6.4.10 Insert Event 6.4.10 Insert Event 6.4.11 Replace Events 6.4.12 Copy Cutset and Replace E 6.4.13 Undo 6.4.14 Find and Delete Cut S^At 6.5 Logic Editor 6.5.1 Exit 6.5.2 Logic Editor 6.5.3 Sub-System CREATE EVENT TREES 7.1 Exit 7.2 Create Event Trees 7.2.1 EXIT 7.2.3 Show 7.2.4 EDIT 7.2.5 TOPS 7.2.6 ENDS 7.2.7 TEXT 7.2.9 FILE 7.2.10 TRAN 7.2.11 Text 7.2.12 cntr/left/rght	 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous . · 8 Search 6.4.7 Previous . · 8 Search 6.4.9 Options 6.4.10 Insert Event 6.4.12 Copy Cutset and Replace Event 6.4.14 Find and Delete Cut Set 6.5.1 Exit 6.5.2 Logic Editor 6.5.3 Sub-System CREATE EVENT TREES 7.1 Exit 7.2 Create Event Trees 7.2.1 EXIT 7.2.2 MOVE (↔) 7.2.3 Show 7.2.4 EDIT 7.2.5 TOPS 7.2.6 ENDS 7.2.7 TEXT 7.2.9 FILE 7.2.10 TRAN 7.2.11 Text 7.2.12 cntr/left/rght 	 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Modify 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous . 3 Search 6.4.0 Insert Event 6.4.10 Insert Event 6.4.11 Replace Events 6.4.12 Copy Cutset and Replace Events 6.4.13 Undo 6.4.14 Find and Delete Cut Set 6.5.1 Exit 6.5.2 Logic Editor 6.5.3 Sub-System CREATE EVENT TREES 7.1 Exit 7.2 Create Event Trees 7.2.1 EXIT 7.2.2 MOVE (↔) 7.2.3 Show 7.2.4 EDIT 7.2.5 TOPS 7.2.6 ENDS 7.2.7 TEXT 7.2.8 VIEW 7.2.10 TRAN 7.2.11 Text 7.2.12 entr/left/rght 	 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous 3 Search 6.4.7 Previous 3 Search 6.4.9 Options 6.4.10 Insert Event 6.4.11 Replace Events 6.4.12 Copy Cutset and Replace Events 6.4.13 Undo 6.4.14 Find and Delete Cut Sot 6.5.1 Exit 6.5.2 Logic Editor 6.5.3 Sub-System CREATE EVENT TREES 7.1 Exit 7.2 Create Event Trees 7.2.1 EXIT 7.2.3 Show 7.2.4 EDIT 7.2.5 TOPS 7.2.6 ENDS 7.2.7 TEXT 7.2.8 VIEW 7.2.9 FILE 7.2.10 TRAN 7.2.11 Text 7.2.12 cntr/left/rgbt 	 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous . 3 Search 6.4.7 Previous . 3 Search 6.4.10 Insert Event 6.4.11 Replace Events 6.4.12 Copy Cutset and Replace Events 6.4.13 Undo 6.4.14 Find and Delete Cut Srt 6.5 Logic Editor 6.5.1 Exit 6.5.2 Logic Editor 6.5.3 Sub-System CREATE EVENT TREES 7.1 Exit 7.2 Create Event Trees 7.2.1 EXIT 7.2.2 MOVE (↔) 7.2.3 Show 7.2.4 EDIT 7.2.5 TOPS 7.2.6 ENDS 7.2.7 TEXT 7.28 VIEW 7.2.9 FILE 7.2.10 TRAN 7.2.11 Text 7.2.12 cntr/left/rgbt 	6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous . 3 Search 6.4.7 Previous . 3 Search 6.4.10 Insert Event 6.4.10 Insert Event 6.4.12 Copy Cutset and Replace Events 6.4.13 Undo 6.4.14 Find and Delete Cut Sat 6.5.1 Exit 6.5.2 Logic Editor 6.5.3 Sub-System CREATE EVENT TREES 7.1 Exit 7.2 Create Event Trees 7.1 Exit 7.2 Create Event Trees 7.2.1 EXIT 7.2.2 MOVE (↔) 7.2.3 Show 7.2.4 EDIT 7.2.5 TOPS 7.2.6 ENDS 7.2.1 Texit 7.2.10 TRAN 7.2.11 Texit 7.2.12 cntr/left/rgbt	 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous . 3 Search 6.4.0 Ditons 6.4.10 Insert Event 6.4.11 Replace Events 6.4.12 Copy Cutset and Replace Events 6.4.14 Find and Delete Cut S^at 6.5 Logic Editor 6.5.2 Logic Editor 6.5.3 Sub-System CREATE EVENT TREES 7.1 Exit 7.2 Create Event Trees <li< td=""><td>6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Moolfy 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous </td><td> 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous . *8 Search 6.4.7 Previous . *8 Search 6.4.7 Previous . *8 Search 6.4.10 Insert Event 6.4.11 Replace Events 6.4.12 Copy Cutset and Replace Events 6.4.13 Undo 6.4.14 Find and Delete Cut Set 6.5 Logic Editor 6.5.2 Logic Editor 6.5.3 Sub-System CREATE EVENT TREES 7.1 Exit 7.2 Create Event Trees 7.2 Create Event Trees 7.2 EXIT 7.2 Solve (↔) 7.2.3 Show 7.2.4 EDIT 7.2.6 ENDS 7.2.7 TEXT 7.2.8 VIEW 7.2.10 TRAN 7.2.11 Text 7.2.12 cutr/leftrght </td><td> 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.10 Insert Event 6.4.12 Copy Cutset and Replace Events 6.4.13 Undo 6.4.14 Find and Delete Cut Set 6.5 Logic Editor 6.5.1 Exit 6.5.2 Logic Éditor 6.5.3 Sub-System CREATE EVENT TREES 7.1 Exit 7.2 Create Event Trees 7.2.1 EXIT 7.2.5 TOPS 7.2.6 ENDS 7.2.9 FILE 7.2.10 TRAN 7.2.11 Text 7.2.12 cntr/left/rght </td><td> 6.3.1 Exit</td><td> 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous . 3 Search 6.4.10 Insert Event 6.4.11 Replace Events 6.4.12 Copy Cutset and Replace Events 6.4.13 Undo 6.4.14 Find and Delete Cut Snt 6.5.1 Exit 6.5.2 Logic £ditor 6.5.3 Sub-System CREATE EVENT TREES 7.2.1 EXIT 7.2.2 MOVE (→) 7.2.3 Show 7.2.4 EDIT 7.2.5 TOPS 7.2.6 ENDS 7.2.10 TRAN 7.2.12 cutr/lef/rgbt </td><td> 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4 Cut Set Editor 6.4 Delete 6.4.3 Moolify 6.4.4 Delete 6.4.5 Locate 6.4.0 Options 6.4.10 Insert Event 6.4.11 Replace Events 6.4.12 Copy Cutset and Replace Events 6.4.13 Undo 6.4.14 Find and Delete Cut Set 6.5 Logic Editor 6.5 Logic Editor 6.5 Logic Editor 7.1 Exit 7.2 Create Event Trees 7.1 Exit 7.2 Create Event Trees 7.2.1 EXIT 7.2.5 TOPS 7.2.6 ENDS 7.2.9 FILE 7.2.10 TRAN 7.2.10 TRAN 7.2.12 cntt/left/rght </td><td> 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.3 Modify 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous . 3 Search 6.4.10 Insert Event 6.4.11 Replace Events 6.4.10 Insert Event 6.4.11 Replace Events 6.4.12 Copy Cutset and Replace Events 6.4.13 Undo 6.4.14 Find and Delete Cut Set 6.5 Logic Editor 6.5 Logic Editor 6.5 Logic Editor 7.1 Exit 7.2 Create Event Trees 7.2.1 EXIT 7.2.6 TOPS 7.2.6 ENDS 7.2.7 DPS 7.2.8 VIEW 7.2.9 FILE 7.2.10 TRAN 7.2.12 cutr/left/rght </td><td> 6.3.1 Exit 6.3.2 Cutasts 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous 6.4.0 Insert Event 6.4.10 Insert Event 6.4.10 Insert Event 6.4.11 Replace Events 6.4.12 Copy Cutset and Replace Events 6.4.13 Undo 6.4.14 Find and Delete Cut S^{nt} 6.5 Logic Editor 6.5.1 Exit 6.5.2 Logic 2ditor 6.5.3 Sub-System CREATE EVENT TREES 7.1 Exit 7.2 Create Event Trees 7.2 FORS 7.2 Create Event Trees 7.2 EXIT 7.2 S VIEW 7.2.6 ENDS 7.2.7 ITEXT 7.2.9 FILE 7.2.10 TRAN 7.2.11 Text 7.2.12 cutr/lefl/rght </td><td>6.3.1 Exit 6-13 6.3.2 Cutsets 6-13 6.3.3 Importance 6-13 6.3.4 Uncertainty 6-20 6.4 Cut Set Editor 6-20 6.4.1 Exit 6-23 6.4.2 Add 6-23 6.4.3 Modify 6-26 6.4.4 Delete 6-26 6.4.5 Locate 6-28 6.4.6 Next 6-29 6.4.7 Previous 6-29 6.4.0 Ditons 6-30 6.4.10 Insert Event 6-31 6.4.10 Insert Event 6-31 6.4.11 Replace Events 6-33 6.4.12 Copy Cutset and Replace Events 6-33 6.5.1 Logic Editor 6-33 6.5.1 Logic Editor 6-33 6.5.2 Logic Editor 6-33 6.5.3 Sub-System 6-41 CREATE EVENT TREES 7-1 7.1 Exit 7-2 7.2 Create Event Trees 7-2 7.2 A MOVE (\leftarrow) 7-2 7.2 S Now 7-2 7.2 S Now 7-2 7.2 S Now 7-2 7.2 Greate Event Trees 7-2</td></li<>	6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Moolfy 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous	 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous . *8 Search 6.4.7 Previous . *8 Search 6.4.7 Previous . *8 Search 6.4.10 Insert Event 6.4.11 Replace Events 6.4.12 Copy Cutset and Replace Events 6.4.13 Undo 6.4.14 Find and Delete Cut Set 6.5 Logic Editor 6.5.2 Logic Editor 6.5.3 Sub-System CREATE EVENT TREES 7.1 Exit 7.2 Create Event Trees 7.2 Create Event Trees 7.2 EXIT 7.2 Solve (↔) 7.2.3 Show 7.2.4 EDIT 7.2.6 ENDS 7.2.7 TEXT 7.2.8 VIEW 7.2.10 TRAN 7.2.11 Text 7.2.12 cutr/leftrght 	 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.10 Insert Event 6.4.12 Copy Cutset and Replace Events 6.4.13 Undo 6.4.14 Find and Delete Cut Set 6.5 Logic Editor 6.5.1 Exit 6.5.2 Logic Éditor 6.5.3 Sub-System CREATE EVENT TREES 7.1 Exit 7.2 Create Event Trees 7.2.1 EXIT 7.2.5 TOPS 7.2.6 ENDS 7.2.9 FILE 7.2.10 TRAN 7.2.11 Text 7.2.12 cntr/left/rght 	 6.3.1 Exit	 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous . 3 Search 6.4.10 Insert Event 6.4.11 Replace Events 6.4.12 Copy Cutset and Replace Events 6.4.13 Undo 6.4.14 Find and Delete Cut Snt 6.5.1 Exit 6.5.2 Logic £ditor 6.5.3 Sub-System CREATE EVENT TREES 7.2.1 EXIT 7.2.2 MOVE (→) 7.2.3 Show 7.2.4 EDIT 7.2.5 TOPS 7.2.6 ENDS 7.2.10 TRAN 7.2.12 cutr/lef/rgbt 	 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4 Cut Set Editor 6.4 Delete 6.4.3 Moolify 6.4.4 Delete 6.4.5 Locate 6.4.0 Options 6.4.10 Insert Event 6.4.11 Replace Events 6.4.12 Copy Cutset and Replace Events 6.4.13 Undo 6.4.14 Find and Delete Cut Set 6.5 Logic Editor 6.5 Logic Editor 6.5 Logic Editor 7.1 Exit 7.2 Create Event Trees 7.1 Exit 7.2 Create Event Trees 7.2.1 EXIT 7.2.5 TOPS 7.2.6 ENDS 7.2.9 FILE 7.2.10 TRAN 7.2.10 TRAN 7.2.12 cntt/left/rght 	 6.3.1 Exit 6.3.2 Cutsets 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.3 Modify 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous . 3 Search 6.4.10 Insert Event 6.4.11 Replace Events 6.4.10 Insert Event 6.4.11 Replace Events 6.4.12 Copy Cutset and Replace Events 6.4.13 Undo 6.4.14 Find and Delete Cut Set 6.5 Logic Editor 6.5 Logic Editor 6.5 Logic Editor 7.1 Exit 7.2 Create Event Trees 7.2.1 EXIT 7.2.6 TOPS 7.2.6 ENDS 7.2.7 DPS 7.2.8 VIEW 7.2.9 FILE 7.2.10 TRAN 7.2.12 cutr/left/rght 	 6.3.1 Exit 6.3.2 Cutasts 6.3.3 Importance 6.3.4 Uncertainty 6.4 Cut Set Editor 6.4.1 Exit 6.4.2 Add 6.4.3 Modify 6.4.4 Delete 6.4.5 Locate 6.4.6 Next 6.4.7 Previous 6.4.0 Insert Event 6.4.10 Insert Event 6.4.10 Insert Event 6.4.11 Replace Events 6.4.12 Copy Cutset and Replace Events 6.4.13 Undo 6.4.14 Find and Delete Cut S^{nt} 6.5 Logic Editor 6.5.1 Exit 6.5.2 Logic 2ditor 6.5.3 Sub-System CREATE EVENT TREES 7.1 Exit 7.2 Create Event Trees 7.2 FORS 7.2 Create Event Trees 7.2 EXIT 7.2 S VIEW 7.2.6 ENDS 7.2.7 ITEXT 7.2.9 FILE 7.2.10 TRAN 7.2.11 Text 7.2.12 cutr/lefl/rght 	6.3.1 Exit 6-13 6.3.2 Cutsets 6-13 6.3.3 Importance 6-13 6.3.4 Uncertainty 6-20 6.4 Cut Set Editor 6-20 6.4.1 Exit 6-23 6.4.2 Add 6-23 6.4.3 Modify 6-26 6.4.4 Delete 6-26 6.4.5 Locate 6-28 6.4.6 Next 6-29 6.4.7 Previous 6-29 6.4.0 Ditons 6-30 6.4.10 Insert Event 6-31 6.4.10 Insert Event 6-31 6.4.11 Replace Events 6-33 6.4.12 Copy Cutset and Replace Events 6-33 6.5.1 Logic Editor 6-33 6.5.1 Logic Editor 6-33 6.5.2 Logic Editor 6-33 6.5.3 Sub-System 6-41 CREATE EVENT TREES 7-1 7.1 Exit 7-2 7.2 Create Event Trees 7-2 7.2 A MOVE (\leftarrow) 7-2 7.2 S Now 7-2 7.2 S Now 7-2 7.2 S Now 7-2 7.2 Greate Event Trees 7-2

7.

.





		7.2.15	1	4	(Co	lut)		1				4					а.			a 4									÷					7-3	\$0
7	3.1	ink Ever	nt '	Tre	es	àr a	N K	e ie			÷.,	. 9						ł	÷	÷			+		6.6	÷,	e s				A -			7-3	30
		7.3.1 7.3.2 7.3.3 7.3.4	E) Gi Li Si	kit ensi ink eque	rate Edi	Sei ito: e Ec	que lito	inci	65	1.1 1 1 1 1	4 4 4			6 X 6 3 7 4 7 4	* * * *					14 1 14 1 14	, , , , , , , , , , , , , , , , , , ,		- 14 - 14 - 15 - 15				1 1 1 1 1 2 1		大 方 子 声	10 · 10 关 关 法 法	大 大 大			7-: 7-: 7-	31 31 33 37
1	1.4 1	lot Tree	s.			i. Ly	i. 4	÷		.,	÷					×.		х О		5		х			, ,			i, s		a s	i it	<i>y</i> -		7-	40
1.3	7.5 (Oraphics	Ĺ	oad	× +		; 3 ×		<u>.</u> 1	1	- a.	× 1	4		- 10	e.						13		X	0	-		X =			i a	×.	21	7-	40
		7.5.1 7.5.2	EL	xit oad		5 x 7 3	а ж н я		к ус 1 - 4	1 17 - 19 14 - 19			(8) (8		() ()	8	4 X	4 A		- A - A	a a a a	і н. 1 м.	т. У 1	с в С В	a 1 4 1	1 1 1 1	X (R	a a 3, 3	і.а. і ж	x J X			k K	7-	40 40
	7.6	eXtract E	Eve	ent '	Tre	es .	1.0		i i	4	8 18						1.3			×,			÷		ł	5.5	×	a, 1	6. X	a.	X Y	x		7	41
		7.6.1 7.6.2 7.6.3 7.6.4	E C D	Exit Xtra Clea Disp	act r Ei lay	Tre xtra Ex	es icte trac	d 1 ctea	Гre d 1	es Te	es	2 - 2 - 2 -		4 4 2 2				4 1 1 1			1 1 1		4.) 3. 3.	4 4 7 7 7 7 4 7	х 	* * K 5 K 5		4 1 4 1 4 1		4 14 14	* *	1 K. 1 K. 1 K. 1 K. 1 K.	3. 	7777	-41 -41 -43
	7.7	Define P	Not	tter	Per	ns		r. A	16 J				, - , -7	s.	i.				5	ę.				× 0		×				i, i je	4			7	-43
0 AN	ALY	ZE SEQ	วบ	EN	CE	S						i. N			4.1	i E six	х.	r je	r.	ų, s		а. э	. 10		: ()	k				ç	ş.,	1	1	4	8-1
	8.1	Exit .				. 7	i.	i.	-	é a		Ofe	i i	ų		Εß		e lije	i,	,	R.	e i	k		e s	ž		i,		e 96	je i	i X A	ŝ	ю I.	8-1
	8.2	Analyze	S	equi	enci	cs.	h			4				,	ł		,	i e		i i s			6.3	×	×,×	4		à	R.	1.1	i i i i	4	e e	a' i	8-1
		8.2.1 8.2.2 8.2.3 8.2.4 8.2.4 8.2.4 8.2.4 8.2.4	1 1 2 0 3 1 4 1 5 1 6	Exit Gen Cut Qua Unc Spli	set Set untif cert it F	te C t Ug fical aint ract	Cut oda tior y A ion	Se te	ts dys	sis														* * * *			电子关注 计计计	1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1	" 年 子 子 子 子 法 法						8-3 8-3 8-5 8-5 8-5 8-16
	8.3	Display	R	esu	lts				i e	19 1	à	i.	×.	1				×		3		-3		ł	d,	i.	÷	ş. 3	÷	K.	e à	1	4.1		8-18
	8.4	Cut Set	E	dito	n .		Y S	i,	с. к. н.			e, a	1		,	×	5	÷		¥		4	ċ	i,	÷	r a		d a		al I	i a	×	- 14		8-19
	8.5	Link E	vei	nt T	ree	s .		į.			k					4	i a		1 12		8		1		×.			A.)		e.	y " 4		+ +		8-19
9.0 M	IODI	FY DAT	ΓĂ	BA	SE						1			×.			κ, 1					6 x						к			ă. ș		1 3 A		9-1
	9.1	Exit .							х. қ				5.3				i i														х з		1		9-2



-

and all a same

9

110

*

January 1992

8,

0.2	Esmiler																												
7.4	ramity .	5 8 9 8	4.3.3	1.1	823	$3 \cdot 1$	8.1	- 10	8.3	6.90	8.3	1	3.15	8.3			8.5		κ'ς.	1.1	16.1		1.1	÷ 4			8.81	. 9	1.2
	9.2.1	Exi	£	di 11			53																					0	1.5
	9.2.2	Add	100																						1.1		6.1	1.6	1.0
	023	Ma	Alfo	100		1.1							1			3.15						12		18				1.7	- 4
	034	Dal	ally	1643	63.4	9.10	3.13	4	1	14	÷ ×		K. K			4.4	6.54	Υ.		4.3			8.3	÷ 8. j	j i		$< \gamma_{\rm c}$. 9	-3
	9.6.4	Dei	ete .	(Also		(X, X)	1.1	×.	8.8	×		1.	ŧά	2.1			5.2			4.14		c ai	≤ 1			÷		. 9	1-4
	9.2.5	Tex	$\mathbf{k} > \mathbf{k}$		6.8.3	6.0			i.	÷.			6.8			a di	14		i.	a A	ж. 1	Č.e.	167. N		я н	4		. 9	1.5
9.3	Event Tr	rees .	5.1.1																									ö	1
												1					1.1	11		1.3	2.1		2.1			1.0	2.18.1	< . 17	-0
	0.2.1	Ente	1.00																									1	
	2.2.1	EXI	1. 1. 1.	2.53	5 A X	2.5	1,3	3		×	1.1	× 1	6.8	÷	10	×	4.15	-	X	1.14	4.1		2.1	÷.	1.3	÷	$k \neq 1$. 9	-7
	9.3.2	Add	1.38	$(\mathbf{x}, \mathbf{x}, \mathbf{x})$	$t \neq \overline{n}$	$t \in \mathcal{T}$	2.9	×.	6.3	\mathcal{F}	- 4	4.1	6.8	$\in \mathbb{R}$	(\mathbf{x})		x si		ιń.	e, k	1.3	- 36	10 ⁻¹ 10		÷.		k a i	. 9	-7
	9.3.3	Mot	lify	inc.	Ge k	1.1	× 10	×.	l. k	×	e a	×		i. a			а к											. 9	-7
	9.3.4	Del	ete .	1.16				. 1																				0	-8
	9.3.5	Tex	6																							2		6	0
	036	Sam	annoi				ð *				× - 4	2.1		2.2		× ×,	1.1	19		1.1	4.9		2,1	÷		a,	ter i	. 7	-9
	0.2.0	Deg	acity a	25 11		11		Ċ.,	1.4	8.1	9 ÷		0.8				8 X.	8.0	18	1.1		- 16	8.4		14	÷	13.	. 9	-9
	9.3.1	Base	/ Cas	se U	pdat	е.,	1.1	ы.		(0, 0)	44	8.3	1.1	8.3	÷.	1.1	έx		h.	1. N	4.5			\mathbf{x}_{i}	8. X		6.4	9.	13
	9.3.8	Clea	ir Ai	terna	ate (ase	£,	ς.	a e	х.	ā ie	al i		1.1			a 1	1.1		÷	i a	- w.					11	9.	13
9.4	Systems	1.1.1	1.1	1.1																								0.	1.6
				212																	1.1		4.15	1.1	1.1	4	٠.		12
	0.4.1	Timbe																										1.1	
	9.4.1	EXI	. X . K	(\mathbf{x}_{i})			13	6.1	1. 7.	81		1.4	18	-		ŧ X	ξi, έ	8.9		1.2	10	1.4.1	έx.	4.1	r aj		6 N. 1	9-1	15
	9.4.2	Add		X_{i},X_{i},λ		(1, 0)	i x	÷	e k	16.5	1.12	4.4	- la.,	1.8	18-19	- 6		1.1		ιų.	8.9		÷ 3		i k		i ni	9-1	15
	9.4.3	Mod	lify	1.1	4.3	di ka	÷	x's	e k		é e			×		i k -		* *		i de	x à		. i.			1	1.	9-1	16
	9.4.4	Dele	ste .	1.1.1	1.1	1.1												ù.,										9.1	17
	945	Text																							1			0.1	17
	946	Baco	C'as	0.11	adat.	111		1		1.1		1.5		8. X				2.1		1	2.1		e	2	e ng	1			1.14
	0.4.7	Dasc	1. as	e uj	puan	9.1		1	1	1	6.96	8.9		1	1.1	1	12	8.9	1.1	1.0	ń, r		8.79	1.1	E DE	1	1.1	9.1	17
	9.4.7	Ciea	r All	terna	ne c	ase	1	12		18. 1	r eff	1.13		414	1.1		e si	8.0		1.15	1.1		κ, π	(\mathbf{r}_{i})	Ēк	÷.	Ġr.	9-1	17
9.5	End State	es .		111	14		l k					11			x-14				÷.,					5.5	i.	λ.		9-1	18
	951	Exit																										0.1	0
	052	Add			15			S		1	1			13				2.3	11		6.2	Υ.	15	70	18	1	1	9-1	0
	2.3.6	Muu	in the	8 K 4	1.1	26	12.		196	1.1		* *	11.1	115	23	11	637	11	1		1.7	1	14	8.0	2	20	1.	9-1	8
	9.3.3	Mod	ity	8.8.9	$\{V_{i},V_{i}\}$	1.1	6. 16	i, i		8.9		8.8	10.0	6.6	19		ČR.	674	84		$i \in$		1.1	8.1	÷.	1	1	9-1	9
	9.5.4	Dele	te .	1.1.1	$\mathbf{x} \in \mathbf{r}$	ene b	6.76) i i	4	1.1	d.	1.1	Q. 1	÷.,		4.1		4.9	Ъ.,	1	é n		2.4	÷ is	i.s	a is	6	9-1	2
	9.5.5	Text		24.4	1.8					a la			1.5			Υ.	2				5.5			1	1			9-2	20
9.6	Basic Eve	ents																										0.0	m
1.16	APROTE APTE	enta	1.1.1			199		1.1		3.4	1	1.12	11	1	1.1	11		1.15	15.13	2	6.9	22	1	2.2	1	8.8	110	7-6	20
	0.4.1	-																											
	9.6.1	Exit		$x_i \in \mathcal{X}$	$q_{i}(q_{i})$			ų,	×.	÷, i	8	йцж	19.13	e sel	1.18	$\hat{\lambda} \geq$	c de l	6.14	1.1	18	6.4	4.1	14	1.14	ίę.	4.4	18.	3-2	20
	9.6.2	Add		1 1 1	1.1					ä. 1	÷	1.1		6.90	1.4	1.4		1.4		÷	1.1		ć.				1	9-2	20
	9.6.1	Mod	ify				1					7.5	1						1									9.2	22
	9.6.4	Dele	te																									0.0	13
	965	Rem	me	Umu	and I	Eve	nt	11								1			2.3		1.1	1	. *	1	1	8.1		0 0	12
	" dia	nom	0101	onu	sent 1		125	1.3	1	X. 4	4	4. A			1.1	1.1	1	1.1	4		8.1	1	1	2.3		3.7	18	2.7	3
9.7	Attributes	8	6. K. K.	1.5.4	1.1	6 9 9	1	1.9		5.3	ł.	1.15			1,1	8.18		1.0	1.1	х	i ŝ		×.	1.1	1			9-2	13
	9.7.1	Exit		1.10	1 2									1						7								9.2	6
																1									1		113		

January 1992



X

	9.7.2 9.7.3 9.7.4	Add Modif Delet	(y	cala cala a cala a			2 -				ते हे ते के ते के		-		2 - 1 2 - 1 2 - 1	4 4 4					* *					9-26 9-27 9-27
9.8 G	aTes		i. Le ri	i de la composición de la comp	3 X 3								÷				0			1.4			÷ 8			9-27
	9.8.1 9.8.2 9.8.3 9.8.4	Exit Add Modi Delet	fy .	2 4 <u>8</u> 5 8 4 8 8 8 4 8 8 8 4 9 8 8 4 9 1				 	r r r r r r	4 3. 4 4 4 4		4 1 4 1 4 1		2 4 2 3 2 3			2 4 2 4 2 5 2 5			1 X 1 X 2 X	3. 1		4 X X X X X X X	¥ 14 14	к. К. 3	9-27 9-27 9-29 9-29
9.9 G	iraphics		44.8											a a	ł,		i Kij	1						- 4		9-30
	9.9.1 9.9.2 9.9.3 9.9.4	Exit Fault Even P&IE	Tree t Tree) Gra	Grap Grap phics	hics				* * * * * *	8 8 8 9 8 9		с ж 6 ж 6 ж 8 ж	 			4 4 4 4 4 9	2 A 1				6 B.	4 8 4 8	* *	1 年 1 年 1 年 1 年 1 年 1 年	* * *	9-33 9-33 9-34 9-35
9.10	Histogra	ims .		$\lambda \ll 1$	k je se	- x	e e	F	4 H	8.9	e se	6. X				4 (j					÷			ş		9-36
	9.10.1 9.10.2 9.10.3 9.10.4	Exit Add Mod Del	l Hist dify I ete H	ograr listog	ns tram: ams		""天天"于汉	* * *		4 14 14		e é e é i e e	 					, x. , y. , y.	2 2 2 2 2 2	4 4 4	 	4 4 4 4 4 4	а с л с л е к			9-36 9-37 9-40 9-41
REPOR	T ON D	ATA	BAS	Е.,		е 1. ж	98	× A		į				i.	x à	÷.)	Q.				1.1	4	1	1		10-1
10.1	Exit	• • •	Ş.,	11		4.3	8 Z	÷ e	te j	2		сэ. К.Э.			8 - A	3.0					ń.		1 1 - 10 1	- 	e se	10-2
10.2	Family	ė.		x x 4	i e i	i i	i u				1	×. 1	2.2	k	4 8	1	È si		-	1	e ș		6 B.		4	10-2
	10.2.1 10.2.5 10.2.3	Exi Sur Ter	t nmary ct	y								9 8 9 8 8 9	et Al P		43 44 44				1			* *		1		10-2 10-2 10-3
10.3	Basic E	vent		i.	k lit, s	e e	1 91 10	ł		1		4.2	d.	i,		Ŕ		. 1		8.3	4	4	ֈ		k x	10-3
	10.3. 10.3. 10.3. 10.3.	1 Bai 2 Bai 3 Bai 4 Bai	tic Ev sic Ev sic Ev sic Ev	vent C vent P vent L vent C	Overv Proba Jnces Cross	view bili rtain Re	ty nty fer	Rej Re	por por ce l	rt Rej	por	t		5 4 3 4 3 4 3 5			9.8 9.8 9.8 9.8				*				14 13 14 14	10-4 10-5 10-6 10-6
10.4	Attribu	tes .	* 1.8												1	ŝ	1		i R		ł,		¥. 1	×.		10-8
10.5	SYstem	1			-			e e		i le			Ċ.	n R y	1 1 1			x 0	- 1	16 - A		1.5		4	× a	10-8
	10.5. 10.5.	1 Sy 2 Sy	stem stem	Sumr Logic	nary Rej	Reg	por	t .	6 38 8 9			n A K		* *					6 x 6 X	4		5 - 5 3 - 5				10-8 10-9

10.0

0 01

January 1992

.

xi

	10.5.2	Cut Sate Danort																		0.10
	10.0.0	Con Seis Repon	0.000	19.14				1.2						1.4.		× ÷		4.16	×	01.0
	10.3.4	System Importa	nce .,	×. ×	10.00		\times		资本						Ċ?				4, 8	10-12
	10.5.5	X-Reference	1.24.28	9.18		≤ 1							χ×.							10-14
10.6	EVent Tr	00																		10.15
		<u></u>																		10.15
	10.01	a second second																		
	10.0.1	Logic/Rules	20.00				1	$(0, -1)^{-1}$	10.14	X = 0	h = h	× 3	8 X.	4.1	××	8. e		4.3		10-15
	10.6.2	Initiating Event	Report	18.9	1.1		8.9	1.1	$\hat{X} \ge$	$+ - \epsilon$	$3 \rightarrow 1$	$x_{i} \in \hat{x}$	$i \in i$	1.1	÷ ×	$\dot{\kappa}$ +	$i \in \mathcal{I}$	$3 \rightarrow$		10-18
	10 6.3	System		4.4		4.3					2.2		x e	4.4		1.1	x	1.1		10-17
10.7	Samance																			10-17
10.1	sequence	in a sea a se la se a se L										10 X	3. 2	8.8	2.16	2.1	2.10		1.15	10.17
		100 C																		i cani
	10.7.1	Sum. Iry	$(k,k) \in \mathcal{K}^{(1)}$	1.1	1.1					9.8	(1, 1)	$\tilde{n} \to 0$	$\theta \rightarrow$		$t \ge 1$	$\lambda_{i} \in$		3.3		1 -17
	10.7.2	Logic	1.000						4.4			$\times +$	s' x		1.3	1.1			(x ¹	10-18
	10.7.3	Cut Sets												1.						10-18
	1074	Importance																		10-19
	10010	importance	11.1										A A			1.1				1.00 1.00
10.0																				10.00
10.8	ENd Stat	CS CARANA CA	13.2.2.1.							13.3		1.1	(\cdot, \cdot)		4.8	17	1.3		6.6	10-22
	10.8.1	Brief Summary	ha na si	140		e a c						1.1				in a				10-22
	10.8.2	Uncertainty																		10-22
		sometrianity																		
10.0	11																			10.00
10.9	User Inn	19. 计实际运行系统	化生产剂	1.1	(x)			1.2.1	6.8.3	1.0.0		1.1		4.5	1.1		19	1.4.2	4 E.	10-22
UTILIT	Y OP110	NS	en a a n													Ala	14	1.1		11-1
11.1	Exit																			11.1
4.4.1.4	KAR	e concerne p	C (4) (5) (5) (5) (5)			11				8.76.1		1.00		19.3	2.1		10			4.4 . 4
																				11.11
11.2	D. nne (onstants	C + X + Y	$i \in I$	2.2	$+$ \times	$\xi \in \mathcal{X}$	1.4				00	14.1	(X .)	16.1	6. 31	ίų.		(0,1,1)	11-1
11.3	Load M.	AR-D Data File					1.											14		11-10
	11.2.1	Land Warifland	ine .																	11.12
	11.2.1	Load vernicat	1011	1.1			1.1	5.16	1		ş er			e e	۰.	1.1.1	8 (B) 	1.1	1.5	11-13
	11.3.2	Family	12324		4.3	6.1	+ +				2.4	131			6.8	14	4.15	1.4	14.16	11-14
	11.3.3	Attributes .	415×314			$\in \mathcal{X}$					a k.	1.16		6.87		ьk,		1.1	1.1	11-14
	11.3.4	Basic Event .	a a a a a											e a l		i.	i.			11-15
	11.3.5	SYstem																		11-15
	1136	EVent Tree																		11-15
	11.2.1	Ebidatata	12.512	11	2.1	7.1					1.1					1.1				11.15
	11.3.7	Endstate .	13353	1.1	8,1			25.3		2.8	8.8	830	8.8	1,81	14	1.1	1.4	8.8	≤ 1	11-13
	11.3.8	Sequence	* * * * *	4.4	$^{+}$	1.1		3° λ	$\in \mathcal{X}$		$h_{i} \in \mathbb{R}$		(\cdot, \cdot)	s, c	έx.	1.1		$ \xi_{-} \in [0,\infty)$	(0, 1)	1.
	11.3.9	Gate	$x^{'} x = \hat{x}^{'} \in X$		1. 1	$\sim x$	1.4	ж 8		$\mathbf{X} \rightarrow \mathbf{X}$	dia.	8. 10	e k	69	1.1		v e	$(4 \rightarrow 0)$	3.3	11-15
	11.3.1	0 Change Sets													i			10.11	1.4	11-16
11.4	Extract	MAR D Data E	iles																	11.16
11.4	EXHIBIT	man-D Data P	ues	1.4	2.4	1	4.14		3.1	1.1			4.8	- 1		1.1	14 14	1. 1		11-10
																				12.20
	11.4.	Family	$A_{i} = \frac{1}{2} (A_{i} - A_{i}) A_{i}$													$X_{i} \mid X$			1.1	11-16
	11.4.1	Attributes													6.16	1.16				11-16
	11.4	Basic Event																		11-16
		and the second se																		

.

đ

11.

6



.

January 1992

xii

-		
1		
100		
All h		
AND	2	
AND DESCRIPTION OF A DE	Ŀ.	
100000000000000000000000000000000000000	в	
 NAMESARE 	,	
16280		

.

.

.

N

T)

41

A

A

	1.4.4	SYstem																				-1	1-16
	11.4.5	EVent Tree				N 8																1	1-17
	11.46	ENdstate																				1	1-17
	11.4.7	Sequence																				Ŧ.	1.17
	11.4.8	Gate																	1.14			1	1-17
	11,4.9	Change Sets	a r																			1	1-17
11.5	Version	1.0/4.0 Interface	х э.																		* *	1	1-20
	11.5.1	Exit																				1	1-20
	11.5.2	Convert 1.0 to 4.0 .		i. A I			ć e															1	1-20
	11.5.3	Convert 4.0 to 1.0 .	÷;	k is				1											k g	Č.,		1	1-24
11.6	Recover	Data Base		5.3									y .	-			a a				*	1	1-29
	11.6.1	Recover Everything .	ж. ж.	x i										e k						t		j,	1-29
	11.6.2	Data Base Recovery	÷ i			i.				с.;-;-;-;-;-;-;-;-;-;-;-;-;-;-;-;-;-;-;-	k i					e de	2					1	1-30
	11.6.3	System Logic Events					È.				x 1.							1.1		с. а	(1-30
	11.6.4	System Cut Set Event	ŝ			6.8	i.							с. ж.								13	1-31
	11.6.5	Sequence Cut Se' Eve	ints								4.4	1.1				. x						. 1	1-32
	11.6.6	Fault Tree Sub Trees			a l		ie ie	1. 1		ж. н.	x	d.		ćя								1	1-33
	11.6.7	Sequence Logic Syste	ms	ψ,		a sé					ял			9 R		6.3		1.1		4.4		1	1-33
	11.6.8	Sequence ENd States		ι.					6	é e	4.4	8.1						e's			-	, I	1-34
	11.6.9	Event Tree Sub Trees		£.							i a	1	e e	× 4	4.3	÷ F	÷	4,3	Р.	χ.,	3.	Ļ	1-35
PPENDIX	A - HAR	DWARE REQUIREM	EN	ГS	À١	ND	IN	151	ΓA	LL	ΑΊ	10	N	PR	0	CE	ED	U	RE	S	k		A-1
PPENDIX	B - DAT	A INTERCHANGE FO	RN	1.4	TS								i.	1.4		i,		2.4		÷	6-8		B-1

18 4 . 11.74

1.0

.

.

FOREWORD

The U. S. Nuclear Regulatory Commission has developed a powerful suite of computer programs for the performance of Probabilistic Risk Assessments (PRAs). This suite of programs allows an analyst to perform many of the functions necessary to create, quantify, and evaluate the risk associated with a facility or process being analyzed. These programs include software to define the data base structure, to create, analyze, and quantify the data, and to display results and perform sensitivity analysis. The programs included in this suite are as follows; Models And Results Database (MAR-D) software, Integrated Reliability and Risk Analysis System (IRRAS) software, Systems Analysis and Risk Assessment (SARA) software, and Fault tree, Event tree, and P&ID (FEP) graphical editor software. Each of these programs perform a specific function in taking a PRA from the conceptual state all the way to publication.

MAR-D is a program that is used primarily for PRA data loading. This program defines a common relational database structure that is used by the entire suite of programs. This structure allows all of the software to access and manipulate data created by other software in the system without performing a lengthy conversion. Hence, data created by IRRAS is immediately available to SARA for sensitivity analysis. The MAR-D program also provides the facilities for loading and unloading of PRA data from the relational database structure used to store the data to an ASCII format for interchange with other PRA software not included in NRC's suite of programs. This feature allows for compatibility with previously developed software systems and allows for maximum data interchange. Elements of this software are included with both IRRAS and SARA to allow these programs to load ano unload data in the MAR-D format. Normally, the entire MAR-D software is used only by those performing a data loading function and is not required by the end user. Documentation for MAR-D, Version 4.0 is in draft form and will be finalized.

IRRAS is a program developed for the purpose of performing those functions necessary to create and analyze a complete PRA. This program includes functions to allow the user to create event trees and fault trees, to define accident sequences and basic event failure data, to solve system and accident sequence fault trees, to quantify cut sets, and to perform uncertainty analysis on the results. Also included in this program are features to allow the analyst to generate reports and displays that can be used to document the results of an analysis. Since this software is a very detailed technical tool, the user of this program should be familiar with PRA concepts and the methods used to perform these analyses. Although IRRAS has been designed to be user friendly and makes the process of performing a Pi-A easier, the complexity of this type of analysis requires a user with a more detailed understanding of PRA concepts than is required by other tools in this suite. The IRRAS 4.0 tutorial, Volume 2, is in preparation.



SARA is a program that allows the user to review the results of a PRA and to perform limited sensitivity analysis on these results. It is limited primarily to the extent that changes in the plant model can be accommodated by using the cut set editor. If other than simple changes are being simulated, then IRRAS should be used so that new cut sets can be accurately generated. This tool is intended to be used by a less technical oriented user and does not require the level of understanding of PRA concepts required by IRRAS. With this program a user can review the information generated by a PRA analyst and compare the results to those generated by making limited modifications to the data in the PRA. Also included in this program is the ability to graphical display the information stored in the MAR-D database. This information includes event trees, fault trees, P&IDs and uncertainty distributions. The user of this program can gain a better understanding of the results of a PRA without getting into the details of the construction and analysis work behind the PRA. The SARA reference manual and tutorial are available as NUREG/CR-5303, Volumes 1 and 2, res. setilely.

FEP is a program developed to provide a common access to the suite of graphical tools developed for performing risk assessment. These tools include the graphical event tree, fault tree, and P&ID editors. The event tree and fault tree editors are available through IRRAS; however, the P&ID editor is only accessible through FEP. The event tree editor allows the analyst to construct and modify graphical event trees. The fault tree editor allows the user to construct and modify graphical fault trees. The P&ID editor allows the user to construct and modify graphical fault trees. The P&ID editor allows the user to construct and modify plant drawings. These drawings can then be used to document the modeling used in a PRA. These editors are an integral part of a PRA. With the FEP tool, the user need not be concerned with the complexity of the IRRAS program if the need is only to generate one of these graphical displays. Documentation for FEP, Version 4.0 is in draft form and will be finalized.

ACKNOWLEDGMENTS

The IRRAS project would not have enjoyed the success it has without the efforts of some other key contributors. These individuals provided a major contribution to the development and success of the IRRAS system.

- Howard D. Stewart for his support in designing the overall IRRAS concept. and for his software support in implementing these concepts.
- Dale M. Snider and Kurt L. Wagner for their contribution to the graphical concepts and software development in IRRAS 1.0.
- Scott D. Matthews and Cory Atwood for their support on the uncertainty analysis.

Appreciation is also expressed to all the various organizations and individuals who provided valuable comments and reviews of the preliminary versions of iRRAS.

INTEGRATED RELIABILITY AND RISK ANALYSIS SYSTEM (IRRAS) VERSION 4.0 VOLUME 1 - REFERENCE MANUAL

EXECUTIVE SUMMARY

The Integrated Reliability and Risk Analysis System (IRRAS) is being developed at the Idaho National Engineering Laboratory (INEL) as the U.S. Nuclear Regulatory Commission's (NRC's) state-of-the-art microcomputer-based probabilistic risk assessment (PRA) model development and analysis tool to address key nuclear plant safety issues.

IRRAS is an integrated PRA software tool that gives the user the ability to create and analyze fault trees and accident sequences using a microcomputer. This program provides functions that range from graphical fault tree and event tree construction to cut set generation and quantification.

The INEL's role in the IRRAS program is that of software developer and interface to the user community, including training and technology transfer. Version 1.0 of the IRRAS program was released in February of 1987 to prove the concept of performing this kind of analysis on microcomputers. That version contained many of the basic features needed for fault tree analysis and was well received by the PRA community. Since the release of Version 1.0, many user comments and enhancements have been incorporated into the program, providing a much more powerful and user-friendly system. This version has been designated IRRAS 4.0. Previous versions of IRRAS include versions 2.0 and 2.5.

IRRAS has all the capabilities and functions required to create, modify, reduce, and analyze fault tree models used in the analysis of complex systems and processes. IRRAS uses advanced graphic and analytical techniques to achieve the greatest possible realization of the potential the microcomputer. Version 4.0 of IRRAS provides all of the same capabilities as Version 1.0 and adds a relational data base facility for managing the data, improved functionality, and improved algorithm performance.



1. INTRODUCTION

The Integrated Reliability and Risk Analysis System (IRRAS) software development project was started as a result of a recognized need for microcomputer-based software to aid the probabilistic risk assessment (PRA) analyst. The nitial scope of the project was to provide a software package that could demonstrate the feasibility of using the microcomputer as a workstation for performing PRA analyses. This package did not necessarily need to perform all of the functions required; however, it did need to provide certain essential functions such as fault tree construction, failure data input, cut set generation, and cut set quantification. The result of this software development project was IRRAS 1.0. That version of the software was released in February of 1987 and contained only the essential concepts mentioned above.

IRRAS 1.0 was an immediate success and clearly demonstrated not only the tremendous need but also the feasibility of performing this work on a microcomputer. As a result of this success, IRRAS 2.0 development was begun. This package was designed to be a comprehensive PRA analysis package and included all the functions necessary for a PRA analyst to perform his or her work. The areas that were not treated in version 1.0 were addressed, and a complete, integrated package was developed. Because IRRAS version 2.0 was a complete rewrite from version 1.0, a thorough test plan was necessary. The major features of IRRAS 2.0 along with an Alpha test were completed in early March of 1988. Following the Alpha test, approximately 15 sites were selected from among the sites currently using IRRAS 1.0, and were sent a Beta test version of IRRAS 2.0. In May of 1988, the Beta test was completed and work began on fixing any bugs found. In addition, any desired new features that could reasonably be incorporated into version 2.0 were included. IRRAS 2.0 was released in June 1990 and work began on the development of IRRAS 2.5.

IRRAS version 2.5 was an integrated PRA software tool that gave the user the ability to create and analyze fault trees and event trees using a personal computer (PC). This program provided functions for fault tree and event tree construction and analysis. The fault tree functions ranged from graphical fault tree construction to fault tree cut set generation and quantification. The event tree functions included graphical event tree construction, the linking of fault trees, defining accident sequences, generating accident sequence cut sets, and quantifying them.

IRRAS version 4.0 contains many significant enhancements over previous versions. This version provides much more powerful cut set generation algorithms. These algorithms are more than a thousand times faster than previous versions of IRRAS. Problems that took hours to solve can now be solved in seconds using IRRAS 4.0. Other enhancements provided in this version include the ability to use the system fault tree logic to solve accident sequences and the addition of flag sets to automatically prune the sequence logic. Many of the operations in IRRAS have also been streamlined and simplified to provide an even more powerful tool for the PRA analyst. This version has undergone a rigorous testing program to ensure reliability and useability. Overall, IRRAS 4.0 continues to provide more powerful tools for the PRA analyst.

The objective of the IRRAS development is to provide the U.S. Nuclear Regulatory Commission (NRC) with a PC-based tool that can be used by NRC staff to evaluate incidents in *r* timely manner. IRRAS 1.0 was released in February 1987. Since that time, it has been distributed to over 400 users in the United States. These users include NRC staff, national laboratories, contractors, vendors, utilities, architectural engineering firms, consultants, universities, other government agencies, and their



contractors. IRRAS is being used as an analysis tool in the resolution of generic issues, inspection activities, and other NRC programs.

At the center of the PRA is the fault tree model. This model, along with component reliability data, provides the basis for risk analysis in most PRA methods. Prior to the development of IRRAS, an analyst would generate models using an alphanumeric text editor. The information would be input in a format compatible with the analysis software, such as SETS, used to process the information. When changes to the model were needed, the analyst would go back to the text file and modify the card images to reflect the changes in the logic model. The analyst would then resubmit them to generate the new cut sets. This process is promoted errors and difficult for the analyst to visualize. IRRAS helps eliminate this problem.

IRRAS automates the model creation, manipulation, modification, and quantification processes. Designed for the IBM-PC and compatibles, IRRAS is readily accessible and portable. Taking advantage of new state-of-the-art algorithms, IRRAS is quite fast and powerful.

IRRAS simplifies the analysis process and automates the construction of input to the analysis software. The analyst can graphically construct and modify fault trees. IRRAS gives the users better visualization of the fault tree and simplifies the construction and maintenance. The program supports all of the basic constructs involved in fault tree construction, including NOT gates. Once the fault tree is constructed, the program automatically generates the alphanumeric input for the analysis software. The component reliability information is then easily input into the IRRAS data base using specially designed menus and screens.

After constructing the system fault tree models, the analyst processes these models with an integrated fault tree analysis package. This package includes the capability to read a fault tree and failure rate data associated with the basic events. The program then generates the minimal cut sets of the fault tree and quantifies the fault tree top event probability using the minimal cut set upper bound. Importance measures for both cut sets and basic events are calculated. The results are documented in various reports generated by the program. The user may select truncation of cut sets by size and/or probability and specify the gate where reduction is to begin. The user may perform some cut set level analysis by using a cut set editor to modify the cut sets, save the new cut sets, and recalculate the minimal cut set upper bound and the new importance measures.

The analyst defines accident sequences in terms of the systems. IRRAS 4.0 has the ability to link fault trees according to the accident sequence logic to create core damage sequence cut sets. These accident sequence cut sets are then quantified to determine the accident sequence a frequency. Importance measures are also calculated.

Many features of mainframe codes have been incorporated into IRRAS 4.0. Improved fault tree reduction techniques, such as identification of independent subtrees and coalescing gates, help to speed up the analysis and make it possible to process larger logic models. Error-checking routines have been added to ensure that the input can be processed when it is time to analyze the fault trees.

In IRRAS 4.0, the graphical fault tree logic can be directly generated from alphanumeric input. This allows the user to read mainframe code input files, such as SETS, and generate the fault tree graphics. The logic models are then easily modified for re-analysis.





IRRAS 4.0 includes fault tree, event tree and cut set editors to improve the analysis capabilities without requiring complete regeneration and reduction of the fault trees. Basic event or initiating event frequencies are easily changed. Cut sets are easily modified with the cut set editor to add recovery actions, or cut sets may be deleted if desired. These changes can be saved in the data base and quantified as desired.

The sections that follow provide the user with the detailed information needed to become familiar with IRRAS 4.0. Sections 3 through 11 present a step-by-step approach to using the basic capabilities of IRRAS. Appendix A provides the specifics of hardware requirements and installation. Appendix B contains data interchange formats.

NOTE:

It should be noted that this document assumes the reader is familiar with PRA methods and terminology. This manual does not instruct the reader in PRA methodology, but merely serves as a reference manual for the IRRAS user.



2. GETTING TO KNOW IRRAS 4.0

The purpose of this section is to give you, the user, an overview of what IRRAS 4.0 can do. We begin with brief discussions of the IRRAS 4.0 modules. (Each module roughly corresponds to a major function or feature.) Next, we explain a little about the IRRAS 4.0 data base so you can begin to learn how IRRAS 4.0 works. Then, we give some pointers on how to best get around inside the program with some special keys. Finally, we provide some insight as to how IRRAS 4.0 works, the usual order of operating IRRAS, and things you must do to prevent unpredictable results.

2.1 IRRAS 4.0 Modules

We have structured IRRAS 4.0 so that the various functions are contained in individual modules or program units. Each module is activated by the use of a command from a menu on the screen. The main modules are SELECT Family, GENERATE Event Data, BUILD Fault Trees, FAULT Tree Analysis, CREATE Event Trees, ANALYZE Sequences, MODIFY Data Base, REPORT on Data Base, and UTILITY Options. Each module is described below.

The SELECT Family module is used to select the family data set you wish to work with and provides the capability of copying raw (MAR-D) data files or IRRAS data base files into a specific family.

The GENERATE Event Data module is used to create a working copy of user event data for use in analysis. This data can then be modified using change sets or used to update the base case data.

The BUILD Fault Trees module is used to build and edit fault tree models and output them to several hard copy hardware devices. Fault trees can be drawn using the keyboard or a mouse. If you use the keyboard, you'll soon see the advantages of getting a mouse. We have used pop-up menu technology in this module to make tree drawing as sim, le as possible.

The FAULT Tree Analysis module is used to quantify the fault tree. Here you determine the minimal cut sets, calculate the probability of failure of the top event of the fault tree, calculate the importance measures of the basic events and the cut sets, and perform uncertainty analysis on the top event failure probability. Sensitivity analysis can be done by changing the failure probability of basic events and requantifying.

The *CREATE Event Trees* module is used to build, edit, and link event trees in a fashion similar to the way the fault trees are constructed in IRRAS 4.0. This greatly simplifies and automates the event tree construction and analysis process.

The ANALYZE Sequences module is a powerful sequence analyzer. This module generates the cut sets for an accident sequence by combining the cut sets or fault trees for the top events that make up the sequence. Comparisons are made betwhen the cut set lists for successful and failed fault trees to eliminate impossible failure combinations (similar to the delete term option in SETS). Once the sequence cut sets have been generated, you can quantify the sequence, calculate importance measures, and perform uncertainty analysis. The results can be displayed on the screen, sent to a printer, or written to a file for later use.



The MODIFY Data Base module provides access to the IRRAS 4.0 relational data base that is created, in part, from use of the other modules. Figure 1 shows the structure of this data base. All information is grouped into families. A family can be a facility, a group o. systems, or any other logical group for which fault trees or sequences are desired. You define the families and assign descriptions in this module. Fault trees are assigned to families in the BUILD Fault Trees module. Sequences are created and assigned descriptions in the MODIFY Data Base module. Basic events are automatically extracted from the fault trees assigned to the family, but you must assign reliability data and descriptions in the MODIFY Data Base module. Additions, deletions, and modifications to the data base can be made here. Section 2.2 provides more detailed information on the data base.



Figure 1. IRRAS 2.6 data base structure.

The *REPORT on Data Base* module lets you see what is in the IRRAS 4.0 data base. Reports are available on families, basic events, attributes, systems, event trees, sequences, end states, and user information.

The last module, UTILITY Options, performs program and computer-related tasks such as setting new machine constants after a hardware configuration change or interfacing with SETS or the PRA data base. The Utility option consists of five functions: Define Constants, Load MAR-D Data Files, Extract MAR-D Data Files, Version 1.0/4.0 Interface, and Recover Data Base. The first option, Define Constants, is used to indicate system hardware configuration, set uncertainty and cut set defaults, and set event tree and fault tree graphics defaults. The second option, Load MAR-D data Files, allows you to load raw files into IRRAS. Extract MAR-D Data Files allows you to output the IRRAS data base records into files that conform to MAR-D data format. The Version 1.0/4.0 Interface allows you to convert data base files using IRRAS Version 1.0 software to IRRAS Version 4.0, or from Version 4.0 to Version 1.0.

0

Finally, the Recover Data Base option allows you to rebuild the index files to each relation in the data base.

2.2 IRRAS 4.0 Data Base Concepts

Before you can use IRRAS 4.0 effecti ely, you need a good understanding of the IRRAS 4.0 data base structure and concepts. The data base provides an effective way to manage the inputs and outputs of the various modules.

The IRRAS 4.0 data base is divided into *families*. A family is the first thing that must be created when using IRRAS 4.0. As delivered, IRRAS 4.0 provides an initial family called DEMO. (Once you are comfortable with IRRAS 4.0, you may want to clear this family from the data base.) A *family* is any logical grouping of fault trees and sequences with their associated basic events, cut sets, reliability data, and descriptions. Access to any portion of the data base is attained through the appropriate family.

A fault tree is automatically assigned to the current family selected when the BUILD Fault Trees module is exited. The basic events used by the fault tree are checked against the events already assigned to the family. Only new basic events are added to the family list. Thus, for a given family there is only one list that contains all the basic events for all the fault trees in the family. This ensures that the reliability data used for a basic event in one tree are identical to those used for the same event in another tree in the family.

The family relation maintains data common to all fault trees in the family, including

- family name and description,
- basic event names and attributes,
- · event-to-fault tree association,
- event-to-sequence association,
- · fault tree change status information, and
- sequence change status information.

'c*.

0

'n

NOTE:

Change status is a flag set by the system indicating one of the following:

- = flags the system or sequence as having cut sets that must be recalculated
- = flags the system or sequence as having cut sets that must be requantified

 flags the system or sequence as needing uncertainty distributions recalculated.

These are discussed in detail in Section 6.

The fault tree relation maintains data for the following fault tree attributes:



- · fault tree name and description,
- · fault tree logic,
- graphic representation,
- · minimal cut sets,
- · top event probability,
- · importance measures, and
- · uncertainty analysis results.

Similarly, the relation defined for an accident sequence includes:

- name and description,
- logic (combination of successful and failed systems),
- · minimal cut sets,
- sequence probability,
- · importance measures, and
- · uncertainty analysis results.

The event tree relation includes:

- · event tree name and description,
- · graphic representation, and
- initiating event.

A relation is also maintained to track event changes relating to event class and event probability.

Unrelated to trees and/or families is the user/system data.

The relations are permanent relations used to maintain the IRRAS 4.0 data base. For each relation, three files exist:

- *.DAT,
- · *.IDX, and
- *.BLK.

The *.DAT file contains the actual data and information about which data constitutes a "key." The *.IDX file contains indices into the data for keyed fields. Finally, the *.BLK file contains variable length data (s.) as cut sets) that are accessible through pointers contained in the *.DAT file.

2.3 Using IRRAS for the First Time

The installation procedure for loading IRRAS 4.0 into your computer is found in Appendix A. The loading process creates three new directories, if they do not already exist. These are \PRADATA.B1, \PRATOOLS.B1, and \HALO88. The PRATOOLS.B1 directory contains the program, the \PRADATA.B1\DEMO directory contains sample data, and the HALO88 directory contains the graphics package. These directories take approximately 6000K of memory on the hard disk when first loaded. Of course, storage requirements go up as you use the program and generate more data.

IRRAS 4.0 uses approximately 550K of internal memory when executed, therefore, resident programs such as Sidekick must be unloaded before using IRRAS 4.0 on a 640K machine.

The following procedure assumes your system is running DOS 3.3 or newer. If your DOS version is older than 3.3, you must upgrade to use IPRAS 4.0.

To start IRRAS 4.0, simply change directories to PRADATA.B1 (CD\PRADATA.B1), type IRRAS, and press < Enter >. (The < > designates pressing the indicated key.) This will execute IRRAS 4.0 with the keyboard as the graphics input device unless you have a mouse and have loaded your mouse driver. If so, then you may use eith input device interchangeably.

On the very first entry into II ^AS 4.0 after installation, the IRRAS Constants screen will appear (see Figure 2). With this screen, you tell the program the configuration of your hardware and the default values desired for various functions within the program. This screen, once completed, will not appear on subsequent entries into IRRAS. If you desire to change a default or your hardware configuration has changed, this screen can be called up from the UTILITY Options menu (se. Section 11).

When this form is filled out properly, press < Enter > to place the information in the data base. Another defaults/constants screen will appear. This screen, shown in Figure 3, establishes the defaults for the fault tree graphics. Once this is filled in properly, press < Enter >, and a third and final screen (Figure 4) will appear with defaults for the event tree editor. Fill this screen in as required and press < Enter >.



permission and User Information
User Name EG&G Idelo, Inc. (IWEL) Monitor type (O-CGA, 1-EGA, 2-VGA, 3-VGA+, 4-8514A) (O-IBM, 1-PA-A-ISE, 2-ATL, 3-TECMAR, 4-TSENG, 5-VIDEO7, PLASMA) O Printer type (O-Other, 1-Epwon, 2-HP Laser) * ster type (O-None, 1-Com1, 2-Com2, 3-Ctom3) O alternative start sport (O-None, 1-Com1, 2-Com2, 3-Ctom3) * alternative start sport (O-None, 1-Com1, 2-Com2, 3-Ctom3) * ster type (O-None, 1-Com1, 2-Com2, 3-Ctom3) * ster type (Talternate N=primary) * ster type (
Uncertainty
Random insolver seed 0 Sample size 1000
Cutoff by size? (Y/N) Y Size cutoff 6 Cutoff by provability? (Y/N) Y Probability cutoff 1.000C-015 Mission time (hours) 2.400E+001 File Locations
Ocratch drive and directory Halo drive and directory path \HALO88\

Figure 2 IRRAS Constants form for example.



Figure 3. IRRAS Event tree graphics information.

L	C	0 1	n 6	t	8	n	1	8		J									
Event	Tre	e 1	Gre 5	ph 6	10			01	-ma 10	iti II 1		12		1		15	11		1
Main Menu Text Cold	P		1	ψ.	÷.,			2.	. 5		1				2	1	a.	15	1
Mair Menu Backgrour	nd Co	010	6.	έ.		2.	2.5				1		÷.	2	÷.	0.1	1	3	1
2nd Level Menu Text	Col	or	1.1	21									۰.		Ξ.		2	14	1
2nd Level Menu Back	arou	ind	De	10	6	1.1	11					1	1	3	31	1.		5	
3rd Level Menu Text	Co.	ar		11					1	1.1		1	£.,	2	2	2	č.	12	1
3rd Level Menu Back	ano	mei	è.	in.	2							17	1	1	0	2	2	0	1
Cursor Color					÷.,	1				10	10	1	1	2			7	15	1
line Color			÷.	11	1	80 - 2 -	1		12		19	1			1		1	10	
Text Color	1.00	1.0	12	٩.,		11	ŧ.,			1.17	. 7	. *	12	2	1	2		10	1
Text Delaht	1.1.1	5.75	1		1.1	1	8. T	11	80 ja	1.1		1			1	- 11	2	14	
Nide Test	2.1	1.1	ð.,	3	12	۰.,	12	1.1	1.1	1.1	1	12	14		1	3	10	.00	
nide lext	1.57	-	n10	le,	N	-	di	on	1	111	de	P. 4	1	1	*		Χ.	N	1
Text Justification	<r l<="" td=""><td></td><td>1.61</td><td>τ,</td><td>C</td><td>-</td><td>C</td><td>en</td><td>ter</td><td>14</td><td>R</td><td>8</td><td>1.11</td><td>gh</td><td>65</td><td></td><td>×</td><td>1</td><td></td></r>		1.61	τ,	C	-	C	en	ter	14	R	8	1.11	gh	65		×	1	
Main Menu Side	12	*	let	τ,	R	-	r	19	h¢:	۶.,		14		$\cdot \in$		16	×	- k	
File compacting	< 4	*	pac	ck,	N	-	d	on	12	pe	tck	э.,	14	18			4	N	

Change any of the constant values and press the <Enter> key.



2.4 Getting Around Inside IRRAS 4.0

Option selection from any menu with a vertical list of options can be done in several ways. The up and down arrow keys can be used to move a highlight through the options. When the highlight is on the desired option, press < Enter > to select the option. The < Space bar > will also move the Fighlight through the options. Another method of selecting an option is to type in the first or second letter of the option and press < Enter >. On menus that have horizontal lists of options, the highlighted letter of the option must be typed in the option field followed with < Enter >.

Certain po, the program have help screens to aid the user. These help screens are accessed by pressing the $\langle F1 \rangle$ key. The $\langle Esc \rangle$ key is used to cancel or abort a function.

Pressing < Alt> H displays the Key Function help screen. This screen is shown in Figure 5. This screen identifies the functions of some special keys for use on fill-in-the-blank type forms used in IRRAS 4.0.

unction metp
restore field-Ctl/R clear formCtl/Home clear fieldCtl/End copy screenCtl/K B&W / colorCtl/B deleteDel

Figure 5. Key Functions Help screen.

The carriage return $\langle CR \rangle$ or $\langle Enter \rangle$ key sends a completed screen to the program for execution. The arrow keys and $\langle Tab \rangle$ key move the cursor around the fields on a screen. The $\langle Home \rangle$ key moves the cursor to the home field on the screen. The $\langle Ins \rangle$ key inserts characters in the text in a field. The $\langle F10 \rangle$ key terminates a help screen. $\langle Ctrl \rangle$ R restores a field to its original setting. $\langle Ctrl \rangle \langle Home \rangle$ clears all entries in all fields on the screen. $\langle Ctrl \rangle \langle End \rangle$ clears the selected field. $\langle Ctrl \rangle$ B switches the display between black and white and color. $\langle Ctrl \rangle$ K copies the screen into a file on the disk named SCLEEN.CPY. Multiple uses of $\langle Ctrl \rangle$ K will cause each screen to be appended to the bottom of the SCREEN.CPY file

2.5 Invoking IRRAS

As mentioned earlier, to invoke IRRAS type IRRAS at the C:\PRADATA.B1 prompt. When you invoke this option, the main menu will be displayed (see Figure 6). Each menu option is discussed in the following sections (Sections 3 through 11).

2.6 Hot Keys

Hot keys allow you to directly transfer to any main menu option from any IRRAS module. To view the help screen for the hot key options, press <Fi> trom the main menu. Figure 7 will be displayed. This is a help screen only, and does not have to appear in order for the hot keys to be active. From most screens in IRRAS, you simply press the <Alt> key combination that corresponent to the desired main menu option. You will be immediately transferred into the corresponding module.

2.7 Menu Options Naming Conventions

Throughout IRRAS, a consistent naming convention is used on all menus. On each menu (like the Create Event tree menus), some options appear in all upper-case letters, while for other options only the first letter is uppercase. When an option appears in all upper-case letters, an associated pop-up menu will be displayed listing the suboptions available when the option is invoked. If the option is not all uppercase, then no pop-up menu is associated and therefore no suboptions are available with the option.





ų,

.

.



Figure 7. Help screen for hot keys

ø

0

- --- Press <Enter> to return -

2.8 Other Things You Should Know

IRRAS 4.0 uses a relational data base to store and retrieve information. Certain information records are automatically added to the data base depending on the functions performed by the user. IRRAS contains nine relations or tables of information. Each relation stores specific PRA information. The first relation is the "Family" relation. This relation contains general information about the PRA. Normally, a user will need to create a family relation for each PRA they are performing. All the information in a family is stored together in a subdirectory. The family relation also contains cross reference information for each basic event, system, sequence, and event tree in the family. The cross reference information is automatically updated each time one of these pieces of data is modified. When IRRAS is initially installed, only one family exists. This family is the "DEMO" family. The first thing the user will need to do is create a new family where the results of the PRA can be stored.

Once the 'Family'' is created, the user is ready to add information to the data base. Typically the next step would be to use the Event Tree editor to generate the event trees for the Family. When the user creates event trees using this editor, IRRAS automatically adds certain information to the data base. When the event tree is saved, an entry is created in the Graphics relation that contains the graphical information associated with the event tree. If the event tree contains an initiating event as the first top event in the tree, then an entry is also created in the Event Tree relation containing the name of the event tree, the initiating event for the event tree, and other descriptive data. The entry in the Event Tree relation is only created if the user has specified an initiating event. The reason for this is that if you have specified an initiating event, IRRAS assumes that you will be solving this event tree. Next, IRRAS creates an entry in the "System" relation for each top event defined in the event tree. By default, IRRAS assumes that each top event is represented by a system with the same name. At the time the system entry is created, IRRAS also creates a basic event with the same name as the system. This event is used when the system ' gic is not expanded during fault tree analysis or when the logic has not yet been developed.

The next step in the PRA analysis is to create the fault trees. If the graphical fault tree editor is used to create the fault trees, then IRRAS creates an entry in the Graphics relation for each fault tree created. An entry is not automatically created in the System relation for each fault tree. IRRAS only creates an entry in the System relation if it knows that the user will be solving this fault tree; otherwise, it assumes that it is only a subtree. If the user has not created any event trees with a top event having the same name as a fault tree, and the fault tree is to be solved, then the user must manually add the system name to the data base, using the Modify Data Base option.

IRRAS requires that certain conventions be used for naming fault trees. If a fault tree is to be used as a top event in an event tree, or referenced in a sequence, then the name of the fault tree and the name of the top gate in the fault tree must be the same as the name of the top event in the event tree and system name in the System relation. A good procedure to follow is to always name the top gate of a fault tree the same as the fault tree.

IRRAS also adds all the basic event references in the fault tree to the Basic Event relation. If a description is provided for a Basic Event, then this description is also added to the data base. All other informatica for the basic event is left unchanged or defaulted. IRRAS als preates an entry in the Gate relation for each gate defined in the fault tree. The gate type and description is also stored in the data base. Finally, IRRAS creates an entry in the Basic Event relation and the Graphics relation for each

transfer gate defined in the fault tree. This is done to allow the user to specify whether a transfer gate is to be expanded or whether the basic event representing the transfer gate is to be used.

Once the fault trees are defined, the user will need to enter the failure data for each basic event in the data base. This can be done with either the Modify Database option or with the Generate Failure Data and Base Case update options. If the Modify Database option is used to change the default failure data, then the user must be sure to use the Generate Failure Data option to generate the current failure data to be used in the remainder of the analysis. Any time changes are made to the basic event failure data, the user MUST use the Generate option before the changes can be used. If it appears that changes made to basic event failure data are not being used properly by IRRAS, the user should check to ensure that the proper data were generated.

The Sequence relation contains information about the sequences for an Event Tree. The user normally generates the sequence entries using the Link Event Tree option. When the user selects this option, IRRAS generates a sequence entry for each valid sequence in the event tree. Any sequences that may $\frac{1}{2}$ we been generated previously for an event tree are automatically removed from the data base. Each sequence contains an endstate name that is created in the Endstate relation, if it does not already exist. Changing the name of an endstate in a sequence only changes the name of or that sequence and does not change the information in the Endstate relation or the Event Tree for that sequence. When a sequence is solved, the cut sets generated are stored in the sequence relation.

The other relations in the IRRAS data base are the Gate relation and the UserInfo relation. The Gate relation stores the name, type, and description of each gate contained in the fault trees for this data base. This information can be edited. However, the changes are only reflected in the graphical fault trees if the user performs an alpha to graphics conversion on the fault trees that use the gates.

The Userinfo relation contains basic operating constants for IRRAS. The user may change these constants to reflect the desired defaults to be used. One of the pieces of information stored in the Userinfo relation is a scratch drive and directory. IRRAS uses this information to locate scratch files during fault tree and sequence solutions. If the user has additional extended memory, a virtual disk drive may be created and the scratch path set to that drive. Doing this will improve the performance of IRRAS by a factor of from 2 to 5 times depending upon the problem being solved.

As the user adds information to the IRRAS data base, a cross reference map is maintained for each basic event, system, fault tree, and event tree in the data base. This map is automatically updated each time one of the specified pieces of information is added or changed. When the user decides to delete a piece of information from the data base, IRRAS checks these cross reference maps to ensure the piece of information being deleted is not used anywhere in the data base. If it is used, IRRAS will not allow the information to be deleted until those i cords referencing it are changed or deleted. For instance, if a basic event is to be deleted, the user must ensure that the event is not used by a fault tree logic, system cut sets (base or current), event tree logic, sequence logic, or sequence cut sets (base or current). In the Modify Data Base module, IRRAS will display a "-" character in front of any event which it determines is not used by anything in the data base. When deleting fault tree or event tree subtrees that are not used, the user will not find them in the System or Event Tree relation lists. Since they are subtrees, they are only found in the Graphics relation. To delete them, the user must use the Modify Data Base module and select the Graphics relation to be edited.



The information stored in the IRRAS data base is buffered in and out of memory during processing. When the information is modified, IRRAS may not write the information out to the hard disk immediately. If the user does not exit IRKAS normally by using the "EXIT" option from the main menu, then the information that was changed may be lost. The user should always exit IRRAS before turning off or rebooting the computer. If a power failure or abnormal program abort occurs, the user should enter (RRAS, select the Utilities menu, and from that menu select the Rebuild option. The user should then rebuild everything in the data base before continuing to use the program on this data. Failure to do this may cause unpredictable results. If a data base is generating results that the user cannot explain, the data base family in question should be rebuilt and the results checked again. If the results still cannot be explained, then a bug may be causing the error and should be reported immediately.





3. SELECT FAMILY

This option allows you to select the family data set you wish to work with and provides the capability of copying raw (MAR-D) data files or IRRAS data base files into a specific family. A family is a group of models, such as those for a single plant, unit, or facility. When the family is successfully selected, the name is shown in the upper left corner of the screen.

The current directory is the current family unless you select another family. IRRAS retains the last family you selected when you exited the program so when you enter IRRAS again the last family selected is the current family. The Select Family screen (Figure 8) lists all families in the IRRAS data base. The select family function provides four options: Exit, Select, Family Copy, and Copy. In addition, three function keys are available:

Displays on-line help messages.

- <Esc> Exits the Select Family module and returns you to the IRRAS main menu.
- <F1>

<F5>

Allows you to locate a specified family. When you press $\langle F5 \rangle$ a blank line will appear on the screen. Enter all or part of the family name you wish to locate and press $\langle Enter \rangle$. This feature will place the highlight on the located name. If the requested name is not found, then the next name in alphabetical order will be highlighted. This feature is especially useful when there are several screens of families to display.

	Option [5] Fri	· / Calant / I	amilu Canu / Dam	
PRADATA SURRY	e — Directory — PRADATA SURRY S	SURRY UNIT 1	- Description -	
	<esc></esc>	<f1></f1>	<f5></f5>	

Figure 8. Select family menu.



Typing $\langle E \rangle$ in the option field and pressing $\langle Enter \rangle$, or pressing $\langle Esc \rangle$ will return you to the IRRAS main menu.

3.2 Select

This option is used to select the family data files that will be accessed during subsequent IRRAS functions. To invoke the option, type $\langle S \rangle$ in the option field, highlight a family, and press \langle Enter \rangle . If a family is not highlighted, the message **Position the cursor over the family to select** will be displayed. When a family is highlighted and selected, you will be returned to the IRRAS main menu where the selected family name will appear at the top of the menu. For later functions, the selected family name will appear in the box in the upper left corner of the screen. If for any reason the family will be retained, and you will be given another chance to select a family. If the highlighted family's data version does not match the current software version, the version update screen appears (Figure 9), and you will be asked if you want to rebuild the data. To select the family, the data must be rebuilt, so enter a $\langle Y \rangle$ to rebuild, and then select the desired family of you type $\langle N \rangle$, that family's data will not be rebuilt, that family will not be selected, the former selected family will be retained, and the message **Unable to** select family of you type $\langle N \rangle$, that family's data will not be rebuilt, that family will not be selected.



Figure 9. Version conflict warning message.

3.3 Family Copy

This option provides the means of copying data base files between families. The family you are copying to should be empty. This option will overwrite all existing files. To invoke this option, type $\langle F \rangle$ in the option field, highlight a family, and press $\langle Enter \rangle$. If no family has been highlighted, the message **Position the cu- or over the family to copy from** appears at the bottom of the screen. If this message appears, highlight a family and press $\langle Enter \rangle$. The message **Position the cursor over the family to copy to** will then appear. Again, highlight a family and press $\langle Enter \rangle$. All family data




base files will be copied from the first family highlighted to the second family highlighted. When the files are copied, the message Family successfully copied appears.

3.4 Copy

This option provides the means of copying any file (raw data and/or a MAR-D file) into any family. To invoke this option, type $\langle C \rangle$ in the option field, highlight the family to copy to, and press $\langle Enter \rangle$. If no family is highlighted, the message **Position the cursor over the family to copy to** is displayed at the bottom of the screen. If this message appears, highlight a family and press $\langle Enter \rangle$. A new screen File Copy (shown in Figure 10), requesting the path and file name of the source data being copied into the selected family is displayed. Specify the entire path of the data to be copied and press $\langle Enter \rangle$ (For example, A:*.* or D:\RAWDATA\DEMO*.*). A confirmation message **File(s)** successfully copied is displayed when the files have been copied.

Entering an invalid path, a nonexistent file name, or pressing < Enter > without specifying a path results in the data not being located, and displays an error message **Unable to locate requested file(s)**. To return to the Select Family screen without copying a file, press < Esc >. The message **Copy attempt terminated at users's request** appears and the Select Family screen is redisplayed.

If you want to copy to a family that does not yet exist, you must first add the f mily. To add a family, see the Modify Data Base option. After you have added the family, you may return to the Select Family option and copy your data base into the new family.







4. GENERATE EVENT DATA

The Generate Event Data module is used to create a working copy of user event data for use in analysis. This data can then be modified using change sets or used to update the base case data. This option allows you to add, modify, and delete change sets. Change sets contain information about the probability/class changes that are to be applied to basic events during system or sequence analysis. Within a change set you may modify elected event probabilities, and reset those event probabilities to the base case values. In addition, you may perform a base case update that takes all current basic event data and makes it the new base case data. The original base case data are overwritten in this process. You may also add, modify, or delete user-defined histograms.

The event data can be manipulated to examine the changes in the probabilities of plant accidents and accident sequence failures based on basic events. Events can be added, modified, and deleted. The probability of events occurring can be recalculated. You may generate change sets to be applied to the basic events for later propagation through the sequence cut sets. You may update the base case that will replace the present base case data with current data.

From the Generate Event Data option, the following operations may be performed: add, modify, or delete change sets; create and reset individual event probability changes; add a class probability change to a group of events; add, modify, or delete user-defined histograms; update the base case event data with the current case event data; generate now current case event data using the temporary modifications made to the marked change sets; and create reports that reflect the event modifications that currently exist within the data base.

Highlight Generate Event Data on the main menu or type a < G > to select this option and press < Enter >. The Change Sets screen shown in Figure 11 will be displayed. A change set is a set of sensitivity data modifications to be applied to the basic events. Change sets modify the current case basic event data. A change set consists of probability and/or class changes for a group of basic events. As shown in Figure 11, ten options are available. Each of these options is discussed in the following paragraphs

4.1 Exit

Type an $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key to return to the IRRAS main menu.

4.2 Add

The Add function is used to add a change set name and description to the data base. You supply a name (up to 16 characters) and a description (up to 60 characters). The current date is supplied by the system, however, you may change it if you wish, by typing over the default date.

Ξ.

Option	A Exit / A Histograms	dd / Modify / Generate Ch	/ Delete / Probab anges / Report Change	ility / Clase s / Base Case Update
# AAA OEP P&ID	N a m c		Description	
here's surround		<f1></f1>	<\$2>	<f3></f3>



To invoke the add function, type an $\langle A \rangle$ in the option field and press $\langle Enter \rangle$. The Add Change Set screen shown in Figure 12 will be displayed. This screen is displayed with the $\langle A \rangle$ (Add) option as the default entry in the option field. After entering the change set name (required) and description (optional) and pressing $\langle Enter \rangle$, the message New change record added is displayed at the bottom of the screen and you are returned to the previous screen (Figure 13), where the newly added change set will be included. If the name you entered already exists, the message Duplicate change record name - not added is displayed at the bottom of the screen.

	Option A Exit / Add	
Name HP-1 Description	annan, dan menerikan di dipat ana di kalendar kalendar kalendar kalendar dan kalendar dan kalendar dan kalendar	Date 1991/07/19

Figure 12. Add a change set.



Figure 13. Change set screen after an Add.

To exit the Add Change Set screen without entering a new change set, press < Esc> or move the cursor to the option field by pressing < Home>, type an < E> (for Exit), and press < Enter>.

4.3 Modify

This option allows you to modify the name, description, and/or date for an existing change set. To invoke this function, type an $\langle M \rangle$ in the option field, highlight the change set you want to modify, and press $\langle Enter \rangle$. The Modify Change Set screen (shown in Figure 14) appears. If you did not highlight the change set to modify before pressing $\langle Enter \rangle$, the message **Record must be highlighted first** appears.

Edit the name, description, and/or date and press < Enter> to execute the change. Use the <Tab> key to move the cursor from field to field. A confirmation message, **Record modif.** d, is displayed. To return to the Change Sets screen without modifying the change set, press < Esc> or move the cursor to the option field by pressing < Home>, typing an < E>, and pressing < Enter>.

4.4 Delete

This function is used to remove a change set from the data base. To invoke this function, type a < D > in the option field, highlight the change set you want to delete, and press < Enter >. The Delete Change Set screen (see Figure 15) is displayed with an "E" (Exit) in the option field. To delete the change set, type a < D > and press < Enter >. You will be returned to the Change Sets screen, and the message Change record deleted is displayed.



	Option M Exit / Mod	i fy
Name P&1D Description	na canada na mangana na I	Date 1991/01/07
	REAL PACTOR AND	NAMES AND ADDRESS OF A STATE OF A DRESS OF A
Modify : chang	ge set.	NANT WARE REAL STOCK MICH. AND A MARKET FRANCISC REPORTED AND ADDRESS TAKE
Modify - chang	ge set.	San turnen auf til som vit Skrendel i fræmsen ter Sandel verkenne i Lan
Modify : chang	je set.	SANT GANNE GANT SKIDDA Y SSAMONI I GANNAR GANSKID OG SAND GAN AND SAND CANTHER SJANSKID KANNAR GAN SAND SAND GAN AND SAND SAND SAND SAND SAND SAND SAN
Modify : chang	ge set. Delete Change	
Modify : chang	ge sel. Delete Change	
Modify : chang	ge sel. Delete Change Opticn E Exit / Oc	cenner an constant constant and constant and constant and constant and constant and constant

Figure 15. Delete a change set.

If you did not indicate the change set you wanted to delete before pressing < Enter>, the message **Record must be highlighted first** appears.

4.5 Probability

This function allows you the flexibility to experiment with setting different basic event failure and uncertainty data. These data values may be set for a single event or for a specified group of events. The function also provides a reset option to set data values back to the base case values. To invoke the Probability function, type a $\langle P \rangle$ in the option field, highlight a change set, and press $\langle Enter \rangle$. If you press $\langle Enter \rangle$ without highlighting a change set, the message A line must be highlighted will be displayed. If you receive this message, highlight an event and press $\langle Enter \rangle$. The Select Event screen (see Figure 16) appears with $\langle P \rangle$ (Probability) as the default option. The change set name is shown

in the upper right corner in the Change Set box. Three options are available: Exit, Probability, and Reset probability to base.

Opt	ion P Exit	/ Probability	/ Reset proba	bility to base	
y*==P==Ceremon N 1	a m e	De	scripti	0 11	and an owner of the second second
ACC-CK	-FT-CV107 CHE	CK VALVE CV107	FAILS TO OPEN		
ACC-CKY	-FT-CV109 CHE	CK VALVE CV109	FAILS TO OPEN		이 가지 않는 것이 같이 많이 많이 많이 많이 했다.
ACC - CKN	-FT-CV128 CHE	CK VALVE CV128	FAILS TO OPEN		121212
ACC-CK\	-FT CV130 CHE	CK VALVE JV130	FAILS TO OPEN		1. S.
ACC-CKV		CK VALVE CV145	FAILS TO OPEN		
ACC-CKV	-FT-CV147 CHE	CK VALVE CV147	FAILS TO OPEN		
ACC-MON	-PG-1865A ACC	MCTOR OPERATE	D VALVE 1865A	PLUGGED	
ACC-MON	-PU-10035 ALL	MUTUR OPERATE	D VALVE 10058	FLUGGED	
ACC-MUN	-PG-10030 ALC	MUTUR UPERATE	D VALVE 1003L	PLOUGED	A
AUF DAL	-5: INI 400	1 NO 13 111 0	USWURK FAILURE		
<esc></esc>	<f1></f1>	×F2>	×F3×	<f4></f4>	<f5></f5>
Pula	Hala	Mark Event	Close All	Hark Danas	Locate

Figure 16. Select an event for probability change.

4.5.1 Exit

This option terminates the probability function and returns you to the Change Set screen (Figure 11). To invoke this option, type an $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key

4.5.2 Probabinty

This function allows you to modify the current uncertainty and failure data values. This current data can then be applied to the specified events for use during event tree analysis. The changes applied to basic event data may be for a single event or a group of marked events.

To invoke this function, type a < P > in the option field, highlight an event or mark a group of events (using the <FC > and/or <F4 > function keys) and press < Enter > (see Figure 17). If only a single event has been selected, then the Event Probability Changes screen (for a single event) will be displayed (Figure 18). The display is divided into four data areas: Event Attributes, Uncertainty Data, Failure Data, Process Flag. The Event Attributes data display is for information only; no changes may be made to the data fields in this display. The Uncertainty Data, Failure Data areas display both the base case and current data values. You may change only the current data values. The Process Flag area is a one-character field that specifies if certain processes should take special note of the selected event. One of the following upper-case values can be input: "S' (sensitivity analysis), "X' (do not expand transfers),



'Y' (never expand transfers), and 'I' (always expand transfers).



Figure 17. Events marked for modification.



Figure 18. Changes to an event probability.



When the Process Flag field is blank, the transfer associated with this event is expanded for failure references. For success references, the transfer is also expanded; however, the cut sets generated are removed from the failure cut sets using cut set matching. An "X" tells IRRAS that the basic event is to be used for failure references, but success references are to be treated the same as if the flag was blank. A "Y" indicates that a transfer is to be replaced with its basic event for failed references and the complement of the event is to be used for success references. An "I" causes IRRAS to treat the transfer as independent. Logic below this transfer is expanded for failure references, and for success references the complement of the logic is used.

If a group of events were marked (see Figure 17) using the $\langle F2 \rangle$ and $\langle F4 \rangle$ function keys, and the cursor was in the tion field, then upon pressing $\langle Enter \rangle$ the Events Probability Changes screen (for a group of events) will be displayed (see Figure 19). This display is divided into three areas: Uncertainty Data, Failure Data, and Process Flag. As before, both the base case and current data values are shown. You may only change the current values. The changes entered on this screen will be applied to all marked events.

Distribution Type	Calculation Type
Value meromanication Class	Probability
Flag	MISSION TIME www

Figure 19. Event change screen for marked events.

The probability changes made are reflected on the Setect Event display by showing a "P" to the left of the affected events, Probability changes have a higher priority than class changes. When both are applied to an event, the probability change will be used during fault tree/sequence analysis. This is indicated by flagging the affected events with a "P" and "c" as shown in Figure 20.

On the Event Probability Changes display, help screens are supplied for all data entry fields. Pressing < F1 > when the cursor is in any data entry field will cause a help screen to be displayed. Most of the help forms contain a data entry field at the bottom of the display. This field may be used to enter the desired data value, which will then be transferred into the proper data field upon exiting the help screen. Highlighting an item in the help form list will set the correct value in the data entry field.



	Option P E	kit / Probabi	lity / Rese	t probaril	ity to base	
* ParCommon	Name		Descri	ption	-	and the second second
* P c ACC	-CKV-FT-LV107	CHECK VALVE C	V107 FAILS	TO OPEN		
ACC	-CKV-FT-CV109	CHECK VALVE C	V109 FAILS	TO OPEN		
* P c ACC	-CKV-FT-CV128	CHECK VALVE C	V128 FAILS	TO OPEN		
ACC	-CKV-FT-CV130	CHECK VALVE C	V130 FAILS	TO OPEN		
C ACC	CKV-FT-CV145	CHECK VALVE C	V145 FAILS	TO OPEN		
* F c ACC	-CKV-FT-CV147	CHECK VALVE C	V147 FAILS	TO OPEN		
ACC	-MOV-PG-1865A	ACC MOTOR OPE	RATED VALVE	1865A PLL	JGGED	
* P c ACC	MOV-PG-18658	ACC MOTOR OPE	RATED VALVE	18658 PL	JGGED	
* P c ACC	-MOV-PG-1865C	ACC MOTOR UPE	RATED VALVE	1865C PLL	JGGED	
* P C ACP	BAC-ST-1H1	BOV AC BUS 1	41 BUSHORK	FAILJRE		
non-server a server server		And the second		entropic planette		in the second second second
(Far)	6815	<f2></f2>	< 63		<845	< 65.2
-For-	and the second se	March Prove		100 m		

Figure 20. Event probability and class changes flagged.

Currently, there r > seven predefined distribution types available. The predefined distribution types are normal, lognormal, beta, gamma, chi-squared, exponential, and uniform. In addition to these predefined distribution types, user-defined histograms may be used. The default distribution type is the lognormal. Figure 21 shows the harmonic associated with the distribution type field on the Event Probability Changes screen. From this help screen, help forms associated with each of the seven predefined distribution types are available. To view these distribution help forms, press the < F1 > key while the cursor is positioned on the desired distribution type.

Correlation classes are used to account for data dependencies among like events in the data base. Correlation classes consist of four character upper case values. A blank correlation class indicates that there are no data dependencies. When running the uncertainty analyses, the same sample value will be used for all basic events with the same correlation class.

In the Failure Data box the calculation type is a numerical reference to the calculation method to be used. There are 13 calculation types numbered 1 through 9, T, F, I, and S. When the cursor is positioned in the Calculation Type field, press < F1 > for an explanation of these calculations. The help screen is shown in Figure 22.



1				k
Ø.				
	p.	13		
	胞	2		
-				

j

L	Event	Probability Changes	
Distri		Uncertainty Distribution Types	
Value	Type	Distribution Values	E
Flag .	LNBGCEU	Log Normal, Error Factor Normal, Standard Dev. Beta, b of Beta(a,b) Gamma, u of Gamma(a) Chi-Squared, Degrees of Freedom Exponential, none Uniform, Upper End Pt.	E · · · · · · · · · · · · · · · · · · ·
Note	н	Histogram, Histogram Number	ted.





Figure 22. Failure data calculation types (probability changes).

1.

The following symbols are used in the equations for calculating failure probability

- P = failure probability of the basic event,
- p = failure probability,
- bp = base case failure probability.
- L = failure rate per hour, input as lambda,
- t_m = mission time expressed in hours, input as a default, and
- T = average time to repair expressed in hours, input as tau.

An equation for each calculation type follows:

Calculation Type 1P = p.Calculation Type 2 $P = L * \subseteq$.Calculation Type 3 $P = 1 - Exp (-L * t_{o})$.Calculation Type 4 $P = L * Min (t_{m}, T)$.Calculation Type 5 $P = ([L * T] / [1 + [L * T]]) * (1 - EXP [-(L + 1 / T) * t_{m}])$.Calculation Type 6 $\tau = L * (T/2)$.Calculation Type 7P = 1 + (EXP[-L * T] - 1) / (L * T).Calculation Type 8P = bp + p.Calculation Type 9P = bp * p.Calculation Type 7P = 1.0 (House event - failed).Calculation Type 1P = 0.0 (ignore event).

<u>Calculation Type S</u> P == 0.0 (find a system with the same name and use its current mincut upperbound as the probability)

4.5.3 Reset Probability to Base

This option allows you to reset the probability changes (not class changes) for a selected event back to the base case values. You may reset probability changes for a single event, a group of events, or all events.

To reset the probability change for a single event, type an $\langle R \rangle$ (Reset probabilities) in the option field, highlight the desired event name and then press $\langle Enter \rangle$. The message "Reset **HIGHLIGHTED probability change?** (Y/N)" is displayed at the bottom of the screen. To reset the probability back to the base case value enter $\langle Y \rangle$ for yes. To terminate the reset operation enter

<N> for no.

To reset the probability changes for a group of events, mark the desired events using the F2 and/or F4 keys, type an $\langle R \rangle$ in the option field and press $\langle Enter \rangle$. The message "Reset ALL marked probability changes? (Y/N)" is displayed at the bottom of the screen. Press $\langle Y \rangle$ to complete the reset operation, or $\langle N \rangle$ to terminate the reset operation.

T reset all event probabilities to the base case values, clear all existing marks with the F3 key, type an $\langle R \rangle$ in the option field and press $\langle Enter \rangle$. The message "Reset ALI "obability changes? (Y/N)" is displayed at the bottom of the screen. Press $\langle Y \rangle$ to perform the resc, operation, or $\langle N \rangle$ to terminate the reset operation.

4.6 Class

This option allows you to change event data parameters for a specified grouping of events. To invoke this option, enter a $\langle C \rangle$ in the option field, highlight the desired change set, and press $\langle Enter \rangle$. If you do not highlight a change set before pressing $\langle Enter \rangle$, the message **An event change must be highlighted** will be displayed at the bottom of the screen. The screen display for this option is shown in Figure 23. All data fields in the four data areas are data entry fields. The event class is defined by entoring data in the Event Attributes data fields. The more of these fields that are filled in the finer the class definition becomes.



SURRY	Class	Change	OEP
Names <p>OEP*</p>	Comp Id Sys Tr	tributes ain Type F/Mode	Location Init?
Clas: Attributes	12345 NNNNN	6 7 8 9 10 11 12 N N N N N N N	13 14 15 16 N N N N
Distribution T (press <f1> fo Value Correlation Cl</f1>	inty Data ypa r list of types) ass=> rs ilag	Failure D Calculation Type - (press <f1> for li Probability Lambda Tau</f1>	ata



The class changes that you request are reflected on the Select Event display by showing a "C" to the left of the affected events (Figure 24) Probability changes have a higher priority than class changes. When both are applied to an event, the probability change will be used during event tree



analysis. This is indicated by flagging the affected events with ... "P" and "c" as shown in Figure 24.

	Opti	on [P] E	xit / Prob	ability	/ Reset #	probabi	lity to base	e
r*PC				- D 2 5	crip	tio	[]	
Pc	ACC-CKV-	FT-CV107	CHECK VALVI	E CV107	FAILS TO	OPEN		
PC	ACC-CKV-	FT-CV109	CHECK VALVI	E CV109	FAILS TO	OPEN		1.1.1.1.1.1
PC	ACC-CKV-	FT-CV128	CHECK VALVI	E CV128	FAILS TO	OPEN		101111
P P	ACC-CKV-	FT-CV145	CHECK VALV	CV165	FALLS TO	OPEN		1.111.111
1 6	ACC-CKV-	FT-CV1+7	CHECK VALVI	E CV147	FAILS TO	OPEN		- C. C. M
	ACC-MOV-	PG-1865A	ACC MOTOR (PERATED	VALVE 18	R65A PL	UGGED	11110
	ALC-MOV-	PG-18658	ACC MOTOR	OPERATED	VALVE 18	8658 PL	UGGED	2151.53
1.1	ACC-MOV-	PG-1865C	ACC MOTOR (OPERATED	VALVE 1	865C PLI	UGGED	
	ACP-BAC-	ST-1H1	480V AC BU	5 1H1 BL	ISWORK FA	LURE		
heuroscom	interaction of the set	al main consideritors a sur com		and very commutation is		encomercia y con		and the second
<esc< td=""><td>2</td><td>×F12</td><td><f2:< td=""><td></td><td><f3></f3></td><td></td><td><\$42</td><td><15></td></f2:<></td></esc<>	2	×F12	<f2:< td=""><td></td><td><f3></f3></td><td></td><td><\$42</td><td><15></td></f2:<>		<f3></f3>		<\$42	<15>

Figure 24. Class and probability changes applied to select events.

On the Class Change display, help screens are supplied for all data entry fields. Pressing $\langle F1 \rangle$ when the cursor is in any data entry field will cause a help screen to be displayed. Figure 25 and Figure 26 show the help screens for Distribution Type and Calculation Type, respectively. Most of the help forms in this situation contain a data entry field at the bottom of the display. This field may be used to enter the desired data value, which will then be transferred into the proper data field upon exiting the help screen. Highlighting an item in the help form list will set the correct value in the data entry field.

4.7 Histograms

This option allows you to create, modify, and delete user-defined histograms. This is a useful option for allowing you to input your own uncertainty distribution for a variable that can not be expressed with one of the predefined distribution types. The Edit Histograms screen (Figure 27) displays the names of all currently existing histograms. A unique number is associated with each histogram and is listed on the Edit Histograms screen to the left of the histogram name. The format type is also provided. There are two format types: percentage format and range format. If the histogram was entered in a percent format, a "P" will be displayed in the type field. If the histogram was entered in a range format, an "R" will appear in the type field. To activate this option, type an $\langle H \rangle$ (Histograms) in the option field of the Change Sets screen and press $\langle Enter \rangle$. Figure 27 will be displayed.













Figure 27. Edit histograms menu.

4.7.1 Exit

r

To return to the Change Sets screen, type an $\langle E \rangle$ (Exit) in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

4.7.2 Add Histograms

This option allows you to create a user-defined distribution type. To add a new histogram to the data base, type an <A> (Add) in the option field and press < Enter>. At this point, Figure 28 is displayed and you are given the choice of adding the histogram data in either a percentage format or in a range format.

If you wish to add a percentage histogram to the data b_{k-e} , enter a $\langle P \rangle$ (Percentage) in the option field. Entering a $\langle P \rangle$ in the option field will cause a Percentage Format Histogram screen to appear. On this screen you should type in a name and a description for the new histogram. Enter the percentages for the histogram along with the corresponding probabilities. Figure 29 demonstrates how to enter a percentage histogram, given that 15% of the data points have a probability of 0.04, 46% of the data points have a probability of 0.02, and the remaining 3% of the data points have a probability of 0.8. The sum of the percentages entered must total 100% in order for the histogram to be accepted as a valid percentage histogram (Figure 30). In the upper right-hand area of the screen is a box that shows the current sum of the percentages that have been input and the remaining percentage needed to reach the 100% total.







Figure 28. Select percentage or range format for the histogram.

1000000 - 100000						
	-	Option A	Exit / Add	Perce Perce	nt Sum 97 nt Left 3	.00 .00
Name	PERCENT	HIST				
peace ip.	ion percent	age histogram	example			
Area	Probability	Percent	Area	Probability	Percent	
	4.000E-002	15.00	11	Frank		
2	1.20CE-001	46.00	12	******F****		
3	2.000E-002	36.00	13		******	
4		******	14	****** E ****	*******	
5	******E****	******	15	******E****	*******	
6	*****E****	******	16			
7	* ···· E · · · ·	******	17	· · · · · · · E · · · ·	******	
8	*****E****	*******	18	E		
ş	E	******	19	*****E****	******	
10	E	*****	20	*****E****		
	-		And and in the Arrist of the A			-





SURKT		Add His	togra	m	
	C 10. Lat Di Li restanza i Conservativa di	Option (A) E	xit / Add Format -	Percent Percent	t Sum 97.0 t Left 3.0
Descript	ion percents	s: e histogram ex	ample		
	Lander La				
Area	Probability	+ rcent	Area	Probability	Percent
	4.0008-002	10	11	ALLEY PL. A	
6	1.2008-001	40.00	12	A Y & A Y & B & Y & Y & Y	
5	5.0006-005	36.00	13	A A A A A A A A A A A A A	1 4 1 4 7 5 5
	8.000E-001	3.00	14	+++++E++++	***
5	*****E****		15	(Y, X, Y,	******
6		(X, Y, X, Y, X, Y,	16	$(-1,+1,+1) \geq (1,+1) \prod_{i=1}^{n} (1,+1,+1) \sum_{i=1}^{n} (1,+1,+1) \sum_{i=1}^{n} (1,+1,+1) \sum_{i=1}^{n} (1,+1) \sum$	$(A \to A) \gg (A \to A) \gg A$
7	*****		17	REAL PRESS	
8	E	ANDERSONA	18	Series Brins -	******
0	******	*****	19	LANGE AND	

Figure 30. Add the remaining percent to create a valid histogram.

Another way to input a histogram is to use the range format. To add a range histogram to the data base, enter an $\langle R \rangle$ (Range) in the option field of the Add Histograms screen. This will bring up a Range Format Histogram screen (^Cigure 31). On this screen, type in a name and a description for the range histogram. Then, enter the starting probability point, the ending probability point, and the height associated with the first bin of the histogram. Next, for each successive bin of the histogram, an ending probability point and a height should be entered. There is a maximum of 20 bins allowed for each range histogram. Figure 31 is an example of inputting a range histogram whose data points lie on the closed interval of 0.0 and 1.6. The height associated with the data points on the sub-interval of 0.0 to 0.2 is 10.0 (Bin 1), the height for the sub-interval of 0.2 to 0.6 is 70.0 (Bin 2), the height for the sub-interval of 0.6 to 0.8 is 20.0 (Ein 3), and the height for the last sub-interval of 0.8 to 1.0 is 5.0 (Bin 4).

The code calculates the midpoint of each bin, finds the area of each bin, and normalizes each area so the sum of the areas equals 1.0. The midpoint is the probability for each bin and the normalized area corresponds to the percent in the Percent Histogram format. The basic event mean probability should correspond to the mean of the histogram.

Once you have typed in the histogram data, enter as $\langle A \rangle$ in the option field and press $\langle E_{h}$ to save the newly created histogram. However, if you wish to exit the Add process without saving the new histogram, enter an $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.



SURRY	Add His	1091	a m	
	Option A E	xit / Ack	d	
PROPERTY AND ADDRESS OF THE REAL PROPERTY AND ADDRESS A	excession Range F	OLUIGI New	name and the factor of the second second	and the second state of the
Nume PANGS	ENIST			
Description examp	ple of a range hist	ogram		
Starting Prob +0.00	00E+000			
Din End Droh	Reinht	Rin	Fed Prob	He isht
1 2.000F-0	01 1.000F+001	11	Arra Fines	ANALAN EXANT
2 6.000F-0	01 7.000F+001	12	AXXAANETTAX	****** Erers
3 8,000E-00	01 2.000E+001	13	******E****	
4 1,000E+00	00 5.000E+000	14	******E****	
5Er.	an arrente Erren	15		TRAFTEREZE
6E	en ananan Ersen	16	******	
7Ein	en vannenBrerei	17	$x, w, w, w, w, w \in \bigcup_{i=1}^{n} W_i (w_i, w_i, w_i, w_i, w_i, w_i, w_i, w_i, $	$(a,b,a) \in (a,b,b) \in \mathbb{R}^{n} \setminus \{a,b,c\}$
8		18	$(-1) \leq (1+1) \leq \left \sum_{i=1}^{n} (1+i) + \sum_{i=1$	$X^{-1}(X,Y) = X^{-1}(Y,Y) = $
Q	$\psi = \psi_{1}$, $-\psi_{2} = \psi_{2} + \psi_{2} + \psi_{3} + \psi_{3} = 0$	19	$- + \times $	$\label{eq:alpha} V = V + V + V + V + V + V + V + V + V +$

Figure 31. Add a range bistogram.

4.7.3 Modify Histograms

To modify a currently existing histogram, type an < M > (Modify) in the option field of the Edit Histograms screen, highlight the desired histogram, and press < Enter >.

If you selected to modify a percentage histogram, then the histogram data will be displayed in percentage format. You may make modifications to the histogram's name, description, or any of the probabilities or percentages. Remember, the percentages must total 100% before it will be accepted as a valid percentage histogram. Figure 32 demonstrates the screen for modifying histograms in percentage format.

If you selected to modify a range histogram, then the histogram data will be displayed in range format. You may change the histogram's name, description, and any of the bin's starting points, ending points, or probabilities. Figure 33 demonstrates the screen for modifying histograms in range format.

If you wish to save the modifications made to the selected histogram, type an <M> (Modify) in the option field of the Modify Histograms screen and press < Enter >. If you wish to exit this screen without modifying the histogram, enter an <E> in the option field and press < Enter >, or press the <Esc > key.



BURN I		Modify	Histogra	m	
ne ser la real reality		Option [M] Percentage	Exit / Modify e Format	Percen Percen	t Sum 100.00 t Left .00
Descript	ion this is	CENTIEST an example of i	a percentage hi	stogram	
Area	Probability	Percent	Area Prob	ability	Percent
- 1 -	4.000E-002	15.00	11	v = E = -1	
5	1.200E-001	46.00	12	E	
3	2.000E-002	36.00	13	E	A 199 1 199 1 199 1
	B.000E-001	3.00	14 - 14 - 1 × + 1	· · E · · · ·	
4			15	e · E · · · ·	
45	$+++++== \sum_{i=1}^{n-1} a_i + a_i + a_i$				
4 5 6	$\label{eq:production} \begin{array}{c} c + c + c + c \\ c + c + c + c + c \end{array} \overset{\text{\mathbf{F}}}{\underset{\text{\mathbf{F}}}{\overset{\mathbf{F}}}{\overset{\text{\mathbf{F}}}{\overset{{$\mathbf{F}}$}}{\overset{{$\mathbf{F}}$}{\overset{{$\mathbf{F}}$}\\{\mathbf{F}}\\{\mathbf{F}}\\{&\\{\mathbf{F}}\\{&\\{\mathbf{F}}\\{&\\{\mathbf{F}}\\{&\\{\mathbf{F}}\\{&\\{\mathbf{F}}\\{&\\{&\\{&\\{}\end{array}{}}}{\overset{{$\mathbf{F}}$}{\overset{{$\mathbf{F}}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}}{\overset{{$\mathbf{F}}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{\overset{{$\mathbf{F}}}{{\mathbf	******	16	· Eners	
4567	$\begin{array}{c} c & c & c & c & c & c & c & c & c & c$	******	16 ****	- E	******
45678	$\begin{array}{c} + + + + + + - + \sum\limits_{i=1}^{n} \sum\limits_{j=1}^{n} \sum\limits_{i=1}^{n} \sum\limits_{i=1}^{n} \sum\limits_{j=1}^{n} \sum\limits_{j=1}^{n} \sum\limits_{i=1}^{n} \sum\limits_{j=1}^{n} \sum\limits_{j=1}^{n} \sum\limits_{i=1}^{n} \sum\limits_{j=1}^{n} \sum \sum\limits_{j=$	******	16 17 18	- E	******
456789	$\begin{array}{c} c + c + c + c = \sum_{i=1}^{n} c_{i+1} + c + c \\ c + c + c + c = \sum_{i=1}^{n} c_{i+1} + c + c \\ c + c + c + c = \sum_{i=1}^{n} c_{i+1} + c \\ c + c + c + c = \sum_{i=1}^{n} c_{i+1} + c \\ c + c + c + c = \sum_{i=1}^{n} c_{i+1} + c \end{array}$	******	16 17 18 19	E	******

Figure 32. Modify a percentage histogram.

SURRY		Modify	Hist	ogram	
		Option M E	xit / Mo	dify	
		Range F	ormat www		
Name	RANGERT	st			
Descript	ion example	of a range hist	ngram		
Starting	Prob +0.000E	+000			
Rin	End Prob	Heicha	Rin	End Droh	Naish+
1	2.000E-001	1.000.001	11	LING FLOD	nergine .
2	6.000E-001	7.000E+001	12	ARCAR- ELSA	
3	8.000E-001	2.000E+001	13		
-4	1.000E+000	5.000E+000	16		·····
5	$\alpha_1(\alpha) = \alpha_1(\alpha) + \sum_{i=1}^{n} \alpha_i(\alpha) = \alpha_i(\alpha)$	$(x,w) \in \mathcal{X} \setminus \{w_i\}_{i=1}^{N} \in \mathcal{X} \setminus \{w_i\}_{i=1}^{N}$	15	$(\sigma, \varphi) = (\sigma, \varphi) = \sum_{i=1}^{n} (\sigma, \varphi) = (\sigma, \varphi)$	(x,y,y) = (y,y) = (y
6	$\psi_{i}(x) = \psi_{i}(x) $	$x_1 = x_1 + \dots + \sum_{i=1}^{n} (x_i + x_i)^{-1}$	16	$x,y,w,w,y,y\in \bigoplus_{i=1}^{m} x_i^i,y_i \in [0,\infty)$	$A_{i}(t) = A_{i}(t) + \frac{1}{2} \sum_{i=1}^{n} \left(1 - \frac{1}{2} \right) + \frac{1}{2} \sum_{i=1}^{n} \left($
7	+ + + + + = = = + +	$(x,y) = (x,y) + (y,y) = \sum_{i=1}^{n} (x,y) = \sum_{i=1}^$	17	******	7.737.47E+++2
8	$\widehat{\alpha} = [\alpha, \beta] = [\alpha, \beta] = [\alpha, \beta] = [\alpha, \beta] = [\alpha, \beta]$	$(a,b,a) = (a,b,a) \in \prod_{i=1}^{n} (a,b,a) = (a,b,a)$	18	$w \Rightarrow w = v = w \bigoplus_{i=1}^{n} x_i = w \in$	$\label{eq:alpha} (A,A) = (A,A) + \prod_{i=1}^{n} (A,A) + \prod_{i=1}^{n} (A,A) + \sum_{i=1}^{n} (A,A) + \sum_{i=1}^{n}$
		$\sigma + \gamma + \sigma + \sum_{i=1}^{n} \gamma + \gamma + \gamma$	19		2 * * * * * * E * * * *
10			20	a serve for as	$\dots = A \times A \times A \times B$

Figure 33. Modify a range histogram.

4.7.4 Delete Histograms

To delete an existing histogram from the data base, type a < D > (Delete) in the option field of the Edit Histograms screen, highlight the histogram to be deleted, and press < Enter >.

This action will result in displaying the selected histogram in the appropriate format on the Delete Histograms screen. To delete the histogram, type a $\langle D \rangle$ (Delete) in the option field and press $\langle Enter \rangle$. To exit the Delete Histograms screen without deleting the histogram, press the $\langle Esc \rangle$ key or enter an $\langle E \rangle$ (Exit) in the option field and press $\langle Enter \rangle$. Figure 34 and Figure 35 show examples of the Delete Histograms screen for percentage histograms and range histograms, respectively.

SURRY	PER ANNE DE AL CONTRACTOR DE LA CONTRACTORIZICA DE LA CONTRACTOR DE LA CON	Delete	HISTO	gram		
		Option [E]	Exit / Del	ete		
		memory Percentag	ie Format +	alan na birana tana ana	and the second second second	
Name	DEDFENT	u107				. 1
Descript	tion percent	ade histooram e	a lame			11
waren dea	trant percentia	nde suiscoltain e	in an an a			- I
Area	Probability	Percent	Area	Probability	Percent	: 1
1.11	4.0006-002	15.00	-11		and some state	. 1
1.2	1.200E-001	46.00	12	arrandform.		· . I
- 3	2.000E-002	36.00	13 -	NURTHFAIRS		- 1
	8.0008-001	3.00	- 74	Antoini Eleven		
5	************	****	15		100000	- 1
6	ANTONE FART	A REPORT OF A	16	marked a Barris	a second and	1.1
7	· · · · · · · · E · · · · ·	*******	- 17	ensingerin.	history .	- 1
8	"HEART Erres	*****	18			- 1
9	ARTERA FARA	A REPORT AND A REP	19	and a second second		1.1
10	and the factor of	XXXXXXXX	20	ARRAN Errer	energy enclosed	1

Figure 34. Delete a percentage histogram.



SURRY	Deiete	Hist	ogram	
	Option E E	xit / De	lete	
	Danna E	ormat		
ame RANGERI	ST			
Description example Starting Prob +0.000E	f of a range hist +000	ogram .		
Bin End Prob	Height	Bin	End Prob	Height
1 2.000E-001	1.000E+001	11		
2 6.000E-001	7.000E+001	12		
3 8.000E-001	2.000E+001	13	$x + x = x + x = \sum_{i=1}^{n} x_i + x + x = 1$	$- x + x + y \in \bigcup_{i=1}^{n} x + x + y$
4 1.000E+000	5.0000+000	14	$\cdots \wedge \wedge \wedge \wedge \wedge \wedge \wedge \in \mathbb{E} \times \wedge \wedge \wedge \cdots$	$ \alpha_{i} = \alpha_{i} + \alpha_{i} = \sum_{i=1}^{n} \alpha_{i} + \alpha_{i} = \sum_{i=1}^{n} \alpha_{i} + \alpha_{i} = \alpha_{i} + \alpha_{i} + \alpha_{i} = \alpha_{i} + \alpha_{i} + \alpha_{i} = \alpha_{i} + \alpha_{i} + \alpha_{i} + \alpha_{i} = \alpha_{i} + \alpha_{i} + $
5 · · · · · · · · E · · · ·	$(1,\infty,n) \in [0,\infty,n] \in [0,\infty,n] \times \mathbb{R}$	15	$\mathcal{A}_{-}^{-} = \mathcal{A}_{-}^{-} = \mathcal{A}$	$- (x_i + y_i + y_i) = - \sum_{i=1}^{n} (x_i + y_i) + \sum_{i=1}^{n} (x_i +$
6	***** \$****	14	*********	1999 A 44 E 4 4 7 9
7Ę	(a,a) = (a,a) = (a,b) = (a,b	17	$-a,b,\gamma \in \{1,2,3\}, b\in \{1,2,3\}$	n k a n e v 🖥 e al k n
B E	1.1.7.7.4.8 <u>5</u> .1.7.7.7.	18	1 E	$X_{i}(0,0) = X_{i}(0,0) = \sum_{i=1}^{n} (X_{i}(0,0)) = \sum_{i=1}^{n} (X_{i}(0$
Q assess Banks	$- (x, y) \in [0, 0, 0] \times \sum_{i=1}^{n} (x_i, y_i) \in [0, 0] \times [0, 0]$	19	1 + + + + + E + + + + +	

Figure 35. Delete a range histogram.

4.8 Generate Changes

This option applies the event data modifications s_i ecified by the selected change sets to the basic event data file. This option must be executed prior to any data analysis if you wish the change set modifications to be reflected in the analysis results. The Generate Changes option creates new current event data, which is used when a user wishes to run an uncertainty analysis on a selected sequence.

To activate this function, type $\langle G \rangle$ (Generate Changes) in the option field, use the function key $\langle F2 \rangle$ to mark the change sets to be used during the generation process, and then press $\langle Enter \rangle$. Upon pressing $\langle Enter \rangle$, the default mission time ard propagate event failure flag are displayed at the bottom of the screen (Figure 36). If you do not wish to change the mission time, press $\langle Enter \rangle$. New event data has been generated with changes is disclayed when the changes are successfully generated. If no changes were made, the message New event data has been generated (no changes) is displayed at the bottom of the Change Sets screen.

If the propagate event failure flag is set 'o 'Y' and if an event is failed (i.e., set to house event "T"), then any event with the same "system" and train attribute is set to "T" also. Any event with the same group name is also set to "T." This is a repetitive process. That is, if an event is set to house event "T" because it matches the system and train, any event in the same group will be set to house event "T."

This option transfers the temporary values for the basic events marked with either a probability change or a class change in the marked change set to the current event data values. If an event in the change set has both a class change and a probability change associated with it, the probability change





Figure 36. Set mission time for generate changes.

takes precedence over the class change when generating new current case event data values. If the event does not have an associated probability change or class change, then the current case event data values will be set equal to the base case data for that event.

If you select the Generate Changes option without marking any of the change sets, then the current case data will be initialized to the base case data.

If more than one change set is marked when you enter the Generate Changes option, then the probability and class changes in the change sets marked with the highest number will take precedence over any changes from lower numbered change sets.

For example, three change sets (CS) are marked and Event A has both a probability change and a class change associated with it in CS 2. Event B has a probability change in CS 1, and a class change in CS 3. Event C has only a class change in CS 1. Event D has no probability changes or class changes associated with it in any of the three marked change sets. The outcome of the Generate Changes option on these three marked change sets would be as follows:

The current case data for event A would be set to the temporary values associated with the probability change in CS 2. (Probability changes take precedence over class changes when they occur for the same event within the same change set.)

The current case data for event B would be set equal to the temporary values associated with the class change in CS 3. (The probability or class change in higher numbered change sets take precedence over changes in lower numbered change sets.)

The current case data for event C would be set to the temporary values associated with the class



change in CS 1.

The current case data for event D would be set to the base case values for event D. (There were no probability changes or class changes made to event D in any of the marked change sets.)

4.9 Report Changes

This option allows you to create reports that reflect the event modifications that currently exist within the data base. There are three report types: Unaffected Events, Affected Events, and All Events (see Figure 37). Each of the report types may be sorted by event name, ascending probability, or change sets. In addition, a report may be routed to the console, an attached printer, or a disk file. When a report is routed to the printer or a disk file, the data will be echoed on the console.



Figure 37. Selecting the event report type.

4.9.1 Exit

This option returns you to the Change Sets screen. To invoke the option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press $\langle Esc \rangle$.

4.9.2 Unaffected Events (Base)

This option allows you to generate a report of the events that are not affected by any of the change sets that currently exist in the data base. To invoke this option, type $\langle U \rangle$ in the option field and press $\langle \Sigma_{cuer} \rangle$. The sort selection menu will be displayed, allowing you to sort the data by event name, ascending probability, or change set (see Figure 38).

4-22



Sort By :
Ascending Probabilities Change Sets
And the second



After the sorting method has been selected, you are asked to specify the routing of the report (Figure 39).

	REPORT OFTIONS
	Report litle
Events	Not Affected By Change Sets
NOTE: F	le Name = "CON" - Dutput report to the screen. "PRN" - Output report to the printer. " " - No report is produced. <esc> - No report is produced.</esc>

Figure 39. Unaffected events (base) report routing.

4.9.3 AFfected Events (Current)

This option allows you to generate a report of the events that are affected by the change sets that currently exist in the data base. To invoke the option, type $\langle F \rangle$ in the option field and press $\langle Enter \rangle$. The sort selection menu will be presented allowing you to sort the data by event name, ascending probability, or change set (Figure 38).



After the sorting method has been selected, you are asked to specify the routing of the report (see Figure 39).

۲

4.9.4 All Events (Base, Current)

This option allows you to generate a report of all of the events in the data base. The report will indicate the events affected by a change set. To invoke the option, type < A > in the option field and press < Enter >. The sort selection menu will be presented allowing you to sort the data by event name, ascending probability, or change set (Figure 38).

After the sorting method has been selected, you are asked to specify the routing of the report (see Figure 39).

4.10 Base Case Update

This option allows you to overwrite the base case data values stored in the data base with the current (temporary) data values for each basic event. After executing this option, the original base case data are no longer available.

To invoke this option, type $\langle B \rangle$ (Base Case Update) in the option field, mark the desired change set(s) and press $\langle Enter \rangle$. A confirmation screen (Figure 40) will appear to ensure you wish to update the base case values with the temporary data values from the marked change set(s). To terminate the process type an $\langle N \rangle$ in the option field and press $\langle Enter \rangle$. To initiate the update process, type a $\langle Y \rangle$ in the option field and press $\langle Enter \rangle$. Upon pressing $\langle Enter \rangle$, the default mission time is displayed at the bottom of the screen (Figure 41). If you do not wish to change the mission time press $\langle Enter \rangle$.

WARNING: When you respond with a < Y > on the warning screen, you will overwrite the base case.

In order to run a base case update at least one change set must be marked. This option transfers the temporary values for the basic events marked with either a probability change or a class change in a change set to the base case values. The existing base case values will be lost. Any event in the change set that does not have either a class change or a probability change associated with it will maintain its existing base case data. Upon completion of the base case update, the message **Base case events have been updated with changed values** will appear at the bottom of the screen.

If an event in a marked change set has both a class change and a probability change associated with it, the probability change takes precedence and will be used to update the base case values.

If more than one change set has been marked and is being used in the base case update, then the probability and class changes in the highest numbered change set take precedence over changes from lower numbered change sets. For example, if Event A has a probability change associated with it in Change Set 1, and a class change associated with it in Change Set 2, then the class change from Change Set 2 would override the probability change from Change Set 1.



SURKT	Change Sets
building provident in some district	second and a second sec
WARNING:	This option will transfer the alternate or temporary values for basic events marked with either a class change or a probabilit, change to the base case values in the data base.
	The existing base case values for those marked events will be lost.
	A.e you sure you want to do this? (Y/N) N



Option	[8] Exit / Add / Modify / Delete / Probability / Class Histograms / Generate Changes / Report Changes / Base Case Update
1 AAA 2 OEP HP- P&1	Name Description
-	Enter Mission Tinut for Generation
1	2.4006+001



5. BUILD FAULT TREES

This option allows you to graphically build and edit fault tree models and output them to several hard copy hardware devices. The defaults for the hardware devices are set in the constants screen (Figure 2). When this option is selected, Figure 42 is displayed.

SURRY	I R R A S Fault Tree Graphics System	
	Exit	
	Build Graphic Trees	
	Plot Trees Load Graphic Trees Alpha to Graphics Graphical Pager	
	eXtract Fault Trees Define Plotter Pens	
	Option E	



As shown, the following options are available:

Exit	÷ 11 C.,	Returns you to the IRRAS Main Menu.
Build Graphic Trees		Allows you to construct the actual fault tree diagram.
Plot Trees		Allows you to plot an HP graphics file.
Load Graphic Trees		Allows you to load a .DLS (graphics) file.
Alpha to Graphics		Allows you to convert the alphanumeric logic for a system or subsystem to a graphical format.
Graphical Pager		Allows you to break-up a diagram into multiple pages
eXtract Fault Trees		Allows you to extract fault trees, clear extracted fault trees, and display extracted fault trees.
Define Plotter Pens		Allows you to assign colors to your plotter pens.



5-1

Each of these options is discussed in the following paragraphs.

5.1 Build Graphic Trees

This option allows you to construct the actual fault tree diagram. You may start building from scratch or from an existing file to generate or modify logic. When this option is selected, Figure 43 will be displayed. The editing commands are shown in the left column, while the rest of the screen is the drawing surface. The editing commands shown in upper-case letters have additional pop-up menus associated with them. The cursor is used to position pop-up menus, draw lines, place drawing symbols, and select menu options.

To invoke any of the editing commands (using a mouse), position the cursor over the desired editing command on the active menu. The active menu is the last menu you pulled up or moved. When the editing command box is highlighted (a white line octlines the box), press the left mouse button. The command is now invoked. Each editing command is described in the following paragraphs.

".1.1 EXIT

This option terminates the editing session and returns you to the previous screen. To invoke this option, position the cursor over the EXIT box and press the left mouse button or <Enter>.

5.1.2 Move

The move command, which is represented by \leftrightarrow , allows you to position the editing command menu anywhere or the screen. When you invoke this command, a white line surrounds the entire editing menu. Drag the cursor to position the outline at the new location and press the left mouse button or < Enter >. The meru will be displayed at the new location.

5.1.3 SHOW

This command clears the screen and re-displays the currently defined diagram.

5.1.4 BILD

This command allows you to generate drawing symbols. When you invoke this command, an additional pop-up menu is displayed (Figure 44). This submenu contains all the constructs needed to create fault tree models. These constructs are









Figure 44. BILD pop-up menu options.

+BUILD+

Line

Allows you to move the BUILD pop-up menu to a new location on the screen. When you invoke this command, a white outline box appears. Drag the cursor to position the outline to the desired location and press the left mouse button or < Enter >. The BUILD pop-up menu will be re-displayed at the new location.

Allows you to draw lines. When you invoke this option, you will be prompted with Enter points for line. The cursor will change from an arrow to a cross hair. Position the cross hair at the point where you want the line to begin, press the left mouse button or <Enter>. The cross hair will now change into a small dot. Drag the cursor in the direction you want the line to follow. When satisfied with the length and direction of the line, press the left mouse button or <Enter>. If the multiplek option is turned on (see Section 5.1.16), the prompt Enter points for next line will appear. You may draw another line or press the right mouse button to terminate line mode. When you terminate line mode, the cross hair will return. Press the right mouse button to the normal cursor mode (arrow).

Generates the "And" symbol. Move the symbol to the desired screen location and press the left mouse button or the < Enter > key. If the multipick option is turned on (see Section 5.1.16), another And symbol will appear automatically. You can position this symbol to the desired location or press the right mouse button to terminate And symbol generation.

Generates the "Or" symbol. Move the symbol to the desired screen location and press the left mouse button or the $\langle \text{Enter} \rangle$ key. If the multipick option is turned on (see Section 5.1.16), another Or symbol will appear automatically. You can position this symbol to the desired location or press the right mouse button to terminate Or symbol generation.

Generates the "Bevent" symbol. Move the symbol to the desired screen location and press the left mouse button or the < Enter > key. If the multipick option is turned on (see Section 5.1.16), another Bevent symbol will appear automatically. You can position this symbol to the desired location or press the right mouse button to terminate Bevent symbol generation.

Generates the "Table" symbol. Move the symbol to the desired screen location and press the left mouse button or the < Enter > key. If the multipick option is turned on (see Section 5.1.16), another Table symbol will appear automatically. You can position this symbol to the desired location or press the right mouse button to terminate Table symbol generation.

And Or Bevent Table





Generates the "N/M OR" symbol. Move the symbol to the desired screen location and press the left mouse button or the <Enter > key. The prompt Enter N (out of) M values will appear at the bottom of the screen. Enter the required values (e.g. 2 5) and press <Enter >. If the multipick option is turned on (see Section 5.1.16), another N/M OR symbol will appear automatically. You can position this symbol to the desired location or press the right mouse button to terminate N/M OR symbol generation.

Generates the "Uevent" symbol. Move the symbol to the desired screen location and press the left mouse button or the < Enter > key. If the multipick option is turned on (see Section 5.1.16), another Uevent symbol will appear automatically. You can position this symbol to the desired location or press the right mouse button to terminate Uevent symbol generation.

Generates the "Transin" symbol. Move the symbol to the desired screen location and press the left mouse button or the <Enter > key. If the multiplick option is turned on (see Section 5.1.16), another Transin symbol will appear automatically. You can position this symbol to the desired location or press the right mouse button to terminate Transin symbol generation.

Generates the "House" symbol. Move the symbol to the desired screen location and press the left mouse button or the $\langle \text{Enter} \rangle$ key. If the multipick option is turned on (see Section 5.1.16), another House symbol will appear automatically. You can position this symbol to the desired location or press the right mouse button to terminate House symbol generation.

Generates the "Inhibit" symbol. Move the symbol to the desired screen location and press the left mouse button or the < Enter > key. If the multipick option is turned on (see Section 5.1.16), another Inhibit symbol will appear automatically. You can position this symbol to the desired location or press the right mouse button to terminate Inhibit symbol generation.





Generates the "Not And" symbol. Move the symbol to the desired screen location and press the left mouse button or the < Enter > key. If the multipick option is turned on (see Section 5.1.16), another Not And symbol will appear automatically. You can position this symbol to the desired location or press the right mouse button to terminate Not And symbol generation.

Generates the "Not Or" symbol. Move the symbol to the desired screen location and press the left mouse button or the <Enter>key. If the multiplick option is turned on (see Section 5.1.16), another Not Or symbol will appear automatically. You can position this symbol to the desired location or press the right mouse button to terminate Not Or symbol generation.

Generates the "BBevent" symbol. Move the symbol to the desired recreen location and press the left mouse button or the <Enter > key. If the multipick option is turned on (see Section 5.1.16), another BBevent symbol will appear automatically. You can position this symbol to the desired location or press the right mouse button to terminate BBevent symbol generation.

Generates the "RTrans" symbol. Move the symbol to the desired screen location and press the left mouse button or the $\langle \text{Enter} \rangle$ key. If the multipick option is turned on (see Section 5.1.16), another RTrans symbol will appear automatically. You can position this symbol to the desired location or press the right mouse button to terminate RTrans symbol generation.

Generates the "LTrans" symbol. Move the symbol to the desired screen location and press the left mouse button or the <Enter>key. If the multipick option is turned on (see Section 5.1.16), another LTrans symbol will appear automatically. You can position this symbol to the desired location or press the right mouse button to terminate LTrans symbol generation.

Generates the "UdTrans" symbol. Move the symbol to the desired screen location and press the left mouse button or the <Enter> key. If the multiplick option is turned on (see Section 5.1.16), another UdTrans symbol will appear automatically. You can position this symbol to the desired location or press the right mouse button to terminate UdTrans symbol generation.

Generates the "Horbox" symbol. Move the symbol to the desired screen location and press the left mouse button or the <Enter > key. If the multiplick option is turned on (see Section 5.1.16), another Horbox symbol will appear automatically. You can position this symbol to the desired location or press the right mouse button to terminate Horbox symbol generation.



5-7

Verbox



Generates the "Verbox" symbol. Move the symbol to the desired screen location and press the left mouse button or the <Enter > key. If the multiplick option is turned on (see Section 5.1.16), another Verbox symbol will appear automatically. You can position this symbol to the desired location or press the right mouse button to terminate Verbox symbol generation.

5.1.5 EDIT

This option allows you to modify fault tree diagrams. In addition to modifying the actual diagram, you may use this option to load existing diagrams and modify various attributes of the drawing. When you invoke the EDIT option, a pop-up menu (Figure 45) will be displayed. The first box, *-EDIT-*, is used to position the pop-up menu to a new location on the screen. The remaining editing options are described in the following paragraphs.

5.1.5.1 ATTRIBUTES. This option allows you to specify the actual attributes of the symbols and text used in the fault tree diagram. Attributes include text size, line type, fill, font size, etc. Changing the attributes does not affect the global (default) values. Only the specific objects selected while in the given mode (e.g., Fill Col, Line Col, etc.) will be affected. When you invoke this option an additional pop-up menu is displayed 2s shown in Figure 46. The following attributes may be modified:

Fill Col

This attribute allows you to change the color for the drawing symbols. When you pick this option, the message Pick a new color from the color bar will be displayed at the bottom of the screen. To select a color, position the cursor over the desired color and press the left mouse button. (NOTE: The 14 (color) option is active at this time. This option will display the additional color selections available.) Next, you will be prompted to Pick shapes to be modified. The cursor will change to a cross hair. Box the shapes to be changed by marking opposite corners. To box the symbols, position the cursor on the symbols to change and press the left mouse button. The cross hair is replaced with a small white dot. Drag the cursor over the shapes to be changed. An outline box appears. When the box surrounds the desired shapes completely, press the left mouse button. The box will disappear and the selected symbols will change to the new color. If no shapes change color, then the selected box was not large enough to include any shapes. If the multipick option is turned on (see Section 5.1.16), you will be prompted to Pick next shapes to be modified. At this point you may select more shapes or press the right mouse button to terminate this process.








Figure 45. Edit pop-up menu.



¢



Figure 46. Attributes pop-up menu.

Line Col

Text Col

This attribute allows you to charge the color of the lines in the current diagram. Again, you will be prompted to Pick a new color from the color bar. To select a color, position the cursor over the desired color box and press the left mouse button. (NOT The 14 (color) option is active at this time. This option will display the additional color selections available.) Next, you will be prompted to Pick the line(s) to be modified. The cursor will change from an arrow to a cross hair. Box the lines to be changed by marking opposite corners. To box the lines, position the cursor on the lines whose color you wish to change and press the left mouse button. The cross hair is replaced by a small white dot. Drag the cursor through the lines to be modified. A box appears surrounding the chosen lines. When the box completely surrounds the chosen lines, press the left mouse button. The box will disappear and the selected line(s) will change to the new color. If the multipick option is turned on (see Section 5.1.16), you will be prompted to Pick next line(s) to be modified. At this point, you may choose additional lines to change, or press the right mouse button to terminate the process.

Line Type This attribute allows you to select the line type. When you select this option, a small window appears displaying the three available line types (solid, broken, or dotted line). You will be prompted to Pick line type. The cursor will change to a cross hair. Position the cross hair over the desired line type, and press the left mouse button. Next, you will be prompted to Pick line(s) to be modified. Box the lines to be changed by marking the opposite corners of the region. Choose the lines to be modified by positioning the cursor over the lines to be modified and pressing the left mouse button. Again, the cursor changes to a small dot. Drag the cursor over the lines to be modified. An outline box will appear. When the box completely surrounds the lines to be modified, press the left mouse button. The box will disappear and the selected lines will change to the new line type. If the multipick option is turned on (see Section 5.1.16), you will be prompted to Pick next line(s) to be modified. At this point, you may select more lines or press the right mouse button to terminate the process. Upon termination, the normal cursor will return.

This attribute option allows you to change the color of the text in your fault tree diagram. When you invoke this option, you will be prompted to Pick a new color from the color bar. Position the cursor over the desired color and press the left mouse button. (NOTE: The † 4 (color) option is active at this time. This option will display the additional color selections available.) Next, you will be prompted to Pick text to be modified. The cursor will change from an arrow to a cross hair. Box the text to be changed by marking the opposite corners of the text area. To box the area, position the cursor over the text to be changed and press the left mouse builton. A small dot will appear. Drag the cursor over the text until the outline box surrounds all the desired text. Press



5-11

the left mouse button. All selected text will change to the new color. If the multipick option is turned on (see Section 5.1.16), you will be prompted to **Pick next text to be modified**. At this point you can select more text or press the right mouse button to terminate the process. The color change will only affect the selected text.

Text Size This attribute option allows you to specify the height of the text in your diagram. When you invoke this option, you will be prompted to Enter text size. Text sizes are indicated by a number between 0.01 and 66.00. with 66 being the full 66 lines from the top to the bottom of the screen. This roughly corresponds to the 66 lines on a full sheet of regular paper. For the purpose of writing text in gate blocks, a text size of about 0.5 is appropriate. The size looks too small on the screen, but it is a good size for sending to a laserjet printer. Larger text sizes will be necessary for printers with lower resolution. Next, you will be prompted to Pick text to be modified. Position the cursor on the text to be resized and press the laft mouse button. A small dot will appear. Drag the cursor over the text to be resized. When the outline box completely surrounds the text to be modified, press the left mouse button. The box will disappear and the selected text will be displayed in the new text size. If the multipick option is turned on (see Section 5.1.16), you will be prompted to Pick next text to be modified. At this point you can select more text, or press the right mouse button to terminate the process. This new text size will only affect the selected text.

Text Just - This attribute allows you to justify selected portions of the text in your diagram. When you invoke this option, you will be prompted to Pick text to be modified. Position the cross hair over the text to be justified and press the left mouse button. A small dot will appear. Drag the cursor over the text to be justified. When the outline box completely surrounds the desired text, press the left mouse button. The outline box will disappear and the selected text will be justified (see Section 5.1.11 for a complete discussion on setting the justification). If the multipick option is turned on (see Section 5.1.16), you will be prompted to Pick next text to be modified. At this point you may select additional text to be justified or press the right mouse button to terminate the process. Only the selected text will be justified. The remaining text will be unchanged.





Text Font

Name Size

This attribute option allows you to select the font type for selected text. When you select this option an additional pop-up menu will be displayed (Figure 47). Select the desired font type by positioning the cursor over the font and pressing the left mouse button. You must select a font (or cancel) in order to continue. Next, you will be prompted to **Pick text to be modified**. Position the cursor over the text to be changed and press the left mouse button. A small dot will appear. Drag the cursor over the text to be modified, when the box co-npletely surrounds the text to be modified, press the left mouse button. The selected text will be displayed in the new font size. If the multipick option is turned on (see Section 5.1.16), you will be prompted to **Pick next text to be modified**. At this point you may select more text, or press the right mouse button to terminate the process. Only the selected text will carry the new font type. The remaining text (unselected) will remain unchanged.

Name Col This attribute option allows you to change the color of the default or given name of the symbol/shape. When you select this option, you will be prompted to Pick a new color from the color bar. Position the cursor over the desired color and press the left mouse button. (NOTE: The 14 (color) option is active at this time. This option will display the additional color selections available.) Next, you will be asked to Pick shapes to be modified. Position the cross hair over the shapes to be changed and press the left mouse button. A small dot will appear. Drag the cursor over the shapes to be modified. You must be sure to place the box completely around the shape and not just the name. When the box surrounds the desired shapes, press the left mouse button. The selected symbol name(s) will change to the new color. If the multipick option is turned on (see Section 5.1.16), you will be prompted to Pick next shapes to be modified. At this joint you may select additional shapes or press the right mouse button to terminate the process.

> This attribute option allows you to specify the height of the shape name in your diagram. When you invoke this option, you will be prompted to Enter text size. Text sizes are indicated by a number between 0.01 and 66.00, with 66 being the full 66 lines from the top to the bottom of the screen. This roughly corresponds to the 66 lines on a full sheet of regular paper. For the purpose of writing text in gate blocks, a text size of about 0.5 is appropriate. The size looks too small on the screen, but it is a good size for sending to a laserjet printer. Larger text sizes will be necessary for printers with lower resolution. Next, you will be prompted to Fick shapes to be modified. Position the cursor on the shape to be modified and press the left mouse button. A small dot will appear. Drag the cursor over the text to be changed. When the box surrounds the shapes to be modified, press the left mouse button. If the multipick option is turned on (see Section 5.1.16), you will be prompted to **Pick next shapes to be modified**. At this point you can select more





XIT + HEDIT ATTRA 4-7 ATT F SHOW AaBb Line í, Line T AaBb DUP TEXT Text 4aBb P Text VIEW S AaBb Co ILE Text 1 AdBb Mo NAME Text Fo Ed Name Anth. Þ Text Clo Name S AnBb Name Fo ontr As BL Shape 1.00 AaBb 24.0-AaBb Line AaBb type Mult Assy 1+ AaBb AaBb AaBb

Figure 47. Font selection screen.

shapes, or press the right mouse button to terminate the process. This new text size will only affect the selected shapes.

Name Font This attribute option allows you to select the font type the shape name will have in your diagram. When you select this option an additional pop-up menu will be displayed (Figure 47). Select the desired font type by positioning the cursor over the font and pressing the left mouse button. Next, you will be prompted to **Pick shapes to be modified**. Position the cursor over the shapes to be changed and press the left mouse button. A small dot will appear. Drag the cursor over the text to be modified. When the box surrounds the shapes to be modified, press the left mouse button. If the multipick option is turned on (see Section 5.1.16), you will be prompted to **Pick next shapes to be modified**. At this point you may select more shapes, or press the right mouse button to terminate the process. Only the selected shapes will carry the new font type. The remaining shapes (unselected) will be unchanged.

This attribute allows you to change an existing symbol in the current Shape Type diagram to another symbol. When you select this option an additional pop-up menu will be displayed showing a menu similar to the build popup menu shown in Figure 44. You will be prompted to Select the new shape type, or < cancel > to quit. On this menu, you must select the desired shape. This is the symbol with which you are replacing the existing symbol. Highlight the desired symbol and press the left mouse button on the < Enter: key. You will be prompted with Select shapes to be changed to new type. Position the cursor over the shape to be changed to the new shape and press the left mouse button or the < Enter > key. A small dot appears. Drag the cursor over the shape(s) to be changed. When the box completely surrounds the desired shape(s), press the left mouse button. The box will disappear and the symbols will be changed to the new symbols. If the multipick option is turned on (see Section 5.1.16), you will be prompted to Select next shape(s) to be changed or < cancel > to reselect type. At this point you may select more shapes to change, or press the right mouse button to select a new replacement symbol.

5.1.5.2 Move. This editing option allows you to move a portion of the diagram to a new location. When you invoke this option, you will be prompted to Box region to be moved - press CANCEL to quit. Position the cross hair over the region to be moved and press the left mouse button. A small dot will appear. Drag the cursor over the region until the box completely surrounds the region to be moved, and press the left mouse button. Next, you will be prompted to Pick reference point - press CANCEL to reselect. The reference point is used to give you some indication of the position of the object being moved relative to the box. Position the cross hair at the location where you want the selected region to be moved. Next, you will be prompted to Position Box at new location - press CANCEL to reselect. Use the cursor to move the box to the exact position where you want the box to appear. When you are satisfied with the new position press the right mouse button. The selected region will be moved to this new location. If the multipick option is turned on (see Section 5.1.16), you will



be returned to the **Position box at new location** ... prompt. At this point you may select a new loca on to move the selected symbols or press the right mouse button to return to the "box region" prompt. At this point you may select another region to move or press the right mouse button to terminate the process.

5.1.5.3 Copy. This edicing option allows you to copy a portion of the diagram and move it to a new location. This option does not create an exact copy, but rather copies the structure of the objects. When you invoke this option, you will be prompted to Box region to be copied, arest CANCEL to quit. To box the region to be copied, mark the opposite corners of the region. How the cross hair on the region to be copied and press the left mouse button. A small dot will appear. Drag the cursor over the region to be copied until the box completely surrounds the region. Press the left mouse button. You will be prompted to Pick reference point - press CANCEL to reselect. Position the cursor at the location you want the region to be copied to and press the left mouse button. The selected region will be copied and displayed at the chosen location. If the multipick option is turned on (see Section 5.1.16), you will be returned to the Position Box at new location on the diagram, or press the right mouse button to terminate the process. If the process is terminated, you will be returned to the "box region" press the right mouse button to terminate the process.

5.1.5.4 Duplicate. This editing option allows you to duplicate a portion of the fault tree diagram. This option makes an exact copy. Duplicate replicates the structure, logic, and even the names used. When you invoke this option, you will be prompted to Box region to be duplicated -press CANCEL to quit. To box the region to be duplicated, mark the opposite corners of the region. Position the cross hair at the area to be duplicated and press the left mouse button. A small dot appears. Drag the cursor over the area to be duplicated until the outline box completely surrounds the desired area. When complete, press the left mouse button. The box will disappear and you will be prompted to Pick reference point - press CANCEL to reselect. Position the cursos at the desired point and press the left mouse button. An outline box the size of the selected area will appear. You will be prompted to Position Box at new location - press CANCEL to reselect. Position the cursor at the desired location and press the left mouse button. The selected area will be duplicated (an exact copy) and displayed at the new location. If the multipick option is turned on (see Section 5.1.16), another outline box will appear and you will be prompted to Position Box at new location - press CANCEL to reselect. At this point, you can duplicate the region again, or press the right mouse button to terminate the process. You will be returned to the Box region to be duplicated prompt. You may select another region to be duplicated or press the right mouse button to cancel the process.

5.1.5.5 Delete. This editing option allows you to delete any portion of the displayed fault tree diagram. When you invoke this option, you will be prompted to Pick region to be deleted. Position the cursor at the place you want to delete and press the left mouse button. A small dot will appear. Drag the cursor over the area to be deleted until the outline box completely surrounds the area. Press the left mouse button. Next, you will be prompted Delete region? Left = delete, Right = cancel. If the boxed area is the area you want to delete, press the left mouse button. If it is not the desired region, press the right mouse button. If the multipick option is turned on (see Section 5.1.16), you will be prompted to Pick next region to be deleted. At this point you may select another region to delete, or press the right mouse button to terminate the process.

5.1.5.6 Copy Txt. This option allows you to copy text from one area of the diagram to another. When you invoke this option, you will be prompted to Pick text to be copied - press CANCEL to quit.







Position the cross hair on the text to be copied. A small dot will appear. Drag the cursor over the text until the outline box completely surrounds the entire text. Press the left mouse button. Next, you will be prompted to Pick reference point - press CANCEL to reselect. The reference point is used to give you some indication of the position of the object being moved relative to the box. Position the cursor at the desired point and press the left mouse button. You will be prompted to Position Box at new location - press CANCEL to reselect. The selected text will now be displayed at the new location. (The text still remains at its original location). If the multipick option is turned on (see Section 5.1.16), you will be prompted to Position Box at new location. At this point you may copy the text to yet another location or press the right mouse button to terminate the process. You will be returned to the Pick text to be copied - press CANCEL to quit prompt. You may select another piece of text to copy or press the right mouse button to terminate the copy process.

5.1.5.7 Move Txt. This editing option allows you to move text from one area of the diagram to another. When you invoke this option, you will be prompted to Pick text to be moved - press CANCEL to quit. To mark the text to be moved, box the opposite corners of the text region. Position the cross hair at the start of the text to be moved, and press the left mouse button. A small dot will appear. Drag the cursos over the text to be moved until the outline box completely surrounds the desired text. Press the left mouse button. Next, you will be prompted to Pick reference point - press CANCEL to reselect. The reference point is used to give you some indication of the position of the object being moved relative to the box. Position the cursor at the desired location and press the left mouse button. You will be prompted to Position Box at new location - press CANCEL to reselect. Move the cursor until the box is positioned at the point where you want to move the text. Press the left mouse button again. The text will be moved to the new location. If the multipick option is turned on (see Section 5.1.16), the box will appear again and you will be prompted to Position Box at new location - press CANCEL to reselect. At this point you can move the same text to yet another location or press the right mouse button to terminate the process. If you choose to terminate, you will be returned to the Pick text to be moved prompt. You may choose more text to move or press the right mouse button to cancel the move text option. NOTE: When you move text, the selected text will be removed from its original position.

5.1.5.8 Edit Txt. This editing option allows you to edit text. When you invoke this option, you will be prompted to **Box text to be edited**. To box the text to be edited, mark the opposite corners of the text region. Position the cursor at the text to be modified and press the left mouse button. A small dot will appear. Drag the cursor over the text to be modified until the outline box surrounds all the desired text. Press the left mouse button. A large window will appear displaying the selected text in a readable format. The text is displayed one line at a time. To edit the text, simply type over the existing text. You may use the <Ins> and keys to add and delete characters as necessary. In addition, you may use the backspace and end keys. Backspace moves all characters from the right of the cursor to the left one space. The End key positions the cursor at the end of the current line. When complete, press <ESC> and the next line of text will appear. When complete $\frac{1}{2}$ will be prompted to **Box text to be edited** (if the multipick option is turned on -see Section 5.1. At this point you can select additional text to modify or press the right mouse button to terminat.

5.1.5.9 Global Txt. This editing option allows you to replace a specified word or group of words with a new word(s) at each and every occurrence in the fault tree diagram. When you invoke this option, you will be prompted to Enter search string: Enter the word or phrase to be replaced (up to 42 characters are allowed for one search) and press < Enter >. Next, you will be prompted to Enter



replacement string: Enter the replacement word or phrase (again, up to 42 characters are allowed) and press < Enter >. Every occurrence will be replaced. When complete, the normal cursor (arrow) will return. NOTE: All occurrences of the strings will be replaced. For example, if you have a string in a text aree you wish to change and that string also happens to be part of a NAME, both instances will be replaced with the new string.

5.1.6 TEXT

The TEXT command allows you to create titles, labels, descriptions, and names for you; fault tree diagram. When you invoke this option, Figure 48 will be displayed. Each of the text submanu options is described in the following paragraphs.

The first option shown, \leftarrow TEXT \rightarrow , is used to move the options menu to a new location. Position the cursor on the \leftarrow TEXT \rightarrow box and press \rightarrow : left mouse button. An outline box will appear. Position the box to the desired location and press the left mouse button. To remove the text suboptions menu, position the cursor anywhere in the text suboptions column and press the right mouse button.

The remaining TEXT subcptions are discussed in the following paragraphs.

5.1.6.1 Write Text. This text suboption allows you to add text anywhere on the fault tree diagram. When you invoke this option, you will be prompted to **Pick location for text**. Position the cursor where you want to write text and press the left mouse button. A window will appear. This window can hold up to 10 lines of text. Type in the new text as you want it to appear on the diagram. When finished, press $\leq Esc >$. When you return to the diagram, the newly added text will be included. If the multipick option is turned on (see Section 5.1.16), you will be prompted to **Pick next location for text**. At this point, you can choose another location to add text, or press the right mouse button to terminate the process.

5.1.6.2 Edit T \rightarrow This text suboption allows you to edit any of the text appearing on the diagram. When you in, this option, you will be prompted to **Box text to be edited**. Position the cross har at the beginning of the text you wish to modify and press the left mouse button. A small dot will appear. Drag your cursor over the text until the outline surrounds the text you wish to modify. When the text is surrounded, press the left mouse button. A window will appear displaying the first line of the selected text. To modify the text, simply type over the existing text. Use the < Ins > and < Del > keys to add and delete characters as necessary. In addition, you may use the Backspace and End keys. Backspace moves all characters from the right of the cursor to the left one space. The End key positions the cursor at the end of the current line. When complete, press < Esc > . If the multipick option is turned on (see Section 5.1.16), you will be returned to the **Box text to be edited**, prompt. At this point, you may box more text to edit or press the right mouse button to terminate the process.

5.1.6.3 FONT. This text soption allows you to change the FONT type of the text. When you invoke this option, Figure 49 will be displayed. To select a font, position the cursor on the desired font and press the left mouse buttor. The font is now selected. New text will now appear in the new font style. This option updates the global font default. **NOTE:** The currently selected font is shown in a different color.





EX1T +TEXT+ 4-4 White Tex SHOW Edit Text Aasb BILD FONT EDIT AaBb Rep Name TEXT ALBO Show Name UIEW AaBb Find Name TLE AaBb Clear Fin Thie Entr NAME Ale St. Edit Table Text Antib Rep Ceven ontr Aa 188. Sho Ceven 1.00 AaBb AaBb Line AaBb type Mult] Aush ++ AaBb AaBb AaBb

Figure 49. FONT selection menu.

.

-

5.1.6.4 Rep Name. This text suboption allows you to name or replace a name. When you invoke this option you will be prompted to **Pick shape to be named**. Position the cursor on the shape to be named and press the left mouse button. The prompt **Enter new name or CR for** = > will be displayed. Enter a new label and press < Enter >. The shape(s) will be renamed. If the multiplick option is turned on (see Section 5.1.16), you will be returned to the **Pick next shape to be named** prompt. At this point you may choose another shape to label or press the right mouse button to terminate the process.

5.1.6.5 Show Name. This text suboption allows you to display (in readable characters) a shape name. When you invoke this option, you will be prompted to **Pick the shape**. Position the cross hair on the desired shape and press the left mouse button. The name for the selected shape will be displayed at the bottom of the screen. If the multipick option is turned on (see Section 5.1.16), the cross hair returns and you may select another shape or press the right mouse button to terminate the process.

5.1.6.6 Find Name. This text sub "ption allows you to locate a specific shape name on your diagram. When you invoke this option, yo will be prompted with **Enter name** >. Enter the shape name you wish to locate and press < Enter > \land broken dotted line will outline the shape containing the specified label. This outline will appear un ?, ou invoke the Clear Find option.

5.1.6.7 Clear Find. To invoke this option, position the cursor over the Clear Find menu option, and press the left mouse button. All highlight outlines will be cleared from the screen.

5.1.6.8 Tble Entry. This option allows you to add entries to an existing table. When you invoke this option, you will be prompted to **Pick Table**. Position the cross hair over the table to be modified and press the left mouse button. You will be prompted to **Enter name (terminate with CR)** >. Enter the new table entry and press < Enter >. If the multipick option is turned on, you will be prompted with **Next name (terminate with CR)** >. At this point you may enter another table entry or press the < Enter > key to terminate the process. Upon pressing < Enter > you will be returned to the **Pick Next Table** prompt. At this point, you may mark another table symbol for entry or press the right many button to terminate the process.

S.1.6.9 Edit Table. This option allows you to edit individual table entries. When you invoke this option, you will be prompted to **Pick gate**, event, or tab. Position the cross hair on the desired gate, event, or table and press the left mouse button. A window will appear displaying the names. You are prompted to **Pick the name to modify,-use** "@" in 1st col to delete. Position the cursor over the name to modify and press the left mouse button. You will be prompted to enter a new name. Enter the name and press < Enter >. To cancel or terminate this process, press the right mouse button.

5.1.6.10 Rep Cevent. This option allows you to rename or change the name (label) given to the event part of an inhibit gate. When you select this option you will be prompted to Pick INHIBIT gate.

5.1.6.11 Sho Cevent. This option allows you to view the Cevent name associated with the inhibit gate. When you invoke this option, you will be prompted to Pick the INHIBIT gate.

5.1.7 VIEW

This option allows you to change the position and size of the displayed diagram. You may move the drawing up, down, right, left, zoom in, zoom out, or restore the drawing to its original size and/or position. You may also use a toggling method to display/not display names and text, and display a grid. When you select this option, Figure 50 will be displayed.

The first box, \leftarrow VIEW \rightarrow , is used to move the VIEW menu to a new location on the screen. To invoke this option, position the cursor on the \leftarrow VIEW \rightarrow box and press the left mouse button. An outline box will appear. Move the outline to the desired location on the screen and press the left mouse button. In addition, you can remove the VIEW option box by positioning the cursor on any command in the VIEW menu and pressing the right mouse button or pressing the <Esc> key. The VIEW submenu consists of the following options:

- Page 1: Invoking this option allows you to shift the figure up one page (previous page).
 To invoke this option, position the cursor in the Page 1 box and press the left mouse button or < Enter >. The diagram's previous page will be displayed and the message
 View changed will appear at the bottom of the screen. This message will remain on the screen until the view option is terminated.
- Page 4: Invoking this option allows you to shift the figure down one page (next page). To invoke this option, position the cursor in the Page 4 box and press the left mouse button or < Enter >. The diagram's next page will be displayed and the message View changed will appear at the bottom of the screen. This message will remain on the screen until the view option is terminated.
- Page •: Invoking this option allows you to shift the figure to the left one page (one screen). To invoke this option, position the cursor in the Page box and press the left mouse button or < Enter >. The diagram will shift to the left one screen at a time and the message View changed will appear at the bottom of the screen. This message will remain on the screen until the view option is terminated.
- Page \rightarrow : Invoking this option allows you to shift the figure to the right one page (one screen). To invoke this option, position the cursor in the Page \rightarrow box and press the left mouse button or < Enter >. The diagram will shift to the right one screen at a time and the message **View changed** will appear at the bottom of the screen. This message will remain on the screen ' til the view option is terminated.
- Scruit: Invoking this option allows you to move the diagram to another location on the screen. To invoke this option, position the cursor in the Scroll box and press the left mouse button. A white outline box appears, with a cross hair placed in the center of the outline. Position the cursor at the desired location and press the left mouse button. The cross hair serves as a reference point for placing the drawing. The reference point (+) is used to give you some indication of the position of the object being moved relative to the screen.



Figure 50. VIEW main menu.



Zoom in: Invoking this option allows you to fill the screen with a small portion of the original display (magnifies the selected portion of the screen). To invoke this option, position the cursor in the Zin (zoom in) box and press the left mouse button or < Enter >. The message Pick two corrers of box will be displayed. Move the cursor to the start of the portion of the diagram to be enlarged and press the left mouse button. A small dot appears. Drag the cursor across the desired area until it is completely surrounded by the outline box. Press the left mouse button. The portion of the original display enclosed by the box will now fill the entire screen. The display can be restored to its original size by invoking the Zres (zoon) restore) option.

- Zoom out: Invoking this option allows you to shrink the displayed diagram by approximately 50%. To invoke this option, position the cursor in the Zout (zoom out) box and press the left mouse button or < Enter >. The entire display will be reduced. To restore the display to its original size, invoke the Zres option.
- Zoom Restore: This option restores any display created by Zin (zoom in), page up, page down, page left, page right, or Zout (zoom out) to the original display size or to the last saved file. To invoke this option, position the cursor in the Zres box and press the left mouse button or < Enter >.
- Sh Name: This option allows you to toggle the setting to display or not display names. Sh Name displays all the names in your diagram. No Name does not display any of the names in your diagram.
- Sh Text: This option allows you to toggle the text setting from Sh Text to No Text. Sh Text displays all defined text. No Text does not display the text.
- GRID: This option displays a grid behind your diagram to allow you to line up symbols and text. This is a toggle switch. To turn the grid on position the cursor in the GRID option box and press the left mouse button or < Enter >. To turn the grid off, repeat the same steps.

5.1.8 FILE

This option allows you to perform various file maintenance utilities including loading, saving, merging files, and printing hard copies of your diagrams. When you invoke this option, Figure 51 will be displayed. Each FILE option is discussed in the following paragraphs.

The first box. \leftarrow FILE \Rightarrow , is used to move the FILE menu to a new location on the screen. To invoke this option, position the cursor on the \leftarrow FILE \Rightarrow box and press the left mouse botton. An outlide box will appear. Move the outline to the new location on the screen and press the left mouse outton. The menu will be placed at the new location. In addition, you can remove the FILis menu x by positioning the cursor on any command in the FILE menu and pressing the right mouse button or pressing the <Esc> key.







5.1.8.1 Load. This FILE suboption allows you to load any Fault Tree File (graphics) with the extension ".DLS" onto the screen. When you invoke this option, you will be prompted to Enter file name >. Enter the name of the file you wish to load and press < Enter >. The diagram will appear on the screen. If you do not remember your file name, use the LIST suboption to display all existing files in the current directory.

5.1.8.2 Save. This FILE suboption allows you to save all changes made to the diagram. When you invoke this option you will be prompted to Enter file name or CR for File ...>. The default file name will consist of the currently displayed diagram and an extension of .DLS. Enter a new file name if desired, or press < Enter > to save the diagram under the default name. If the default file name already exists (i.e., you have saved your diagram previously), the message File already exists. Do you wish to replace? Y/N > will be displayed. If you wish to write the changes over the existing file, type a Y and press < Enter >; otherwise, enter an N.

5.1.8.3 Logic. Th' "ILE suboption allows you to save the logic of the diagram. When you invoke this option, you are prompted to Enter file name or CR for file XXX >, where XXX is the file name of the currently displayed diagram. Enter the new file name and press < Enter >. When complete, the message Logic saved will be displayed at the bottom of the screen.

5.1.8.4 New. This FILE suboption allows you to essentially cancel your current editing session. All changes made since the last save will not be applied. When you invoke this option, the screen will be cleared and you will be returned to Figure 43.

5.1.8.5 LIST. This FILE suboption displays the list of files currently residing in the default directory. When you invoke this option, a pop-up menu will be displayed showing all the files contained in the current directory. You will be prompted to Pick the file to load. Position the cursor over the file to load and press the left mouse button or < Enter>. The file will be loaded and displayed on the screen.

5.1.8.6 Merge. This FILE suboption allows you to merge the contents of two files into a single file. When you invoke this option, a pop-up menu will be displayed showing all the files contained in the current directory. You will be prompted to Pick the file to load. Position the cursor over the file to merge into the currently displayed diagram and press the left mouse button or < Enter >. Next, you will be prompted to Pick location for the top center of merged file. Position the cross hair at the desired location and press the left mouse button or < Enter >. The screen will be cleared and then reshown with the merged files displayed.

5.1.8.7 Tran→. This FILE suboption allows you to view the drawing defined in the file given in the transfer name.

5.1.8.8 **Tran.** This FILE suboption allows you to view the drawing that define(s) the transfer logic. Transfers allow you to create fault trees consisting of many pages.

5.1.8.9 Epson. This option formats the diagram for an Epson printer and sends it to the local Epson printer.

5.1.8.10 Laser. This option formats the diagrams for a laser printer and sends it to the local laser printer to be printed.

5.1.8.11 FName. This option displays the current file name of the diagram being displayed. When you invoke this option, the file name will be displayed at the bottom of the screen.

5.1.9 NAME

This option allows you to change the defaults for the name symbols. When you invoke this option, Figure 52 will be displayed. Each option is described below.

- Name Col: This option allows you to change the color of the name. When you invoke this option, you will be prompted to **Pick a new color from the color bar**. Position the cursor on the desired color and press the left mouse button. (**NOTE**: The **f** 4 (color) option is active at this time. This option will display the additional color selections available.) The NAME box will change to the new default color.
- Name Size: This option allows you to change the text size of the name. When you invoke this option, you will be prompted to Enter text size >. As mentioned earlier, text sizes are indicated by a number between 0.01 and 66.00, with 66 being the full 66 lines from the top to the bottom of the screen. This roughly corresponds to the 66 lines on a full sheet of regular paper. Enter the desired text size and press < Enter >.
 - Dflt Gate: This option allows you to assign a default name to the gate. When you invoke this option you will be prompted to Enter gate default name >. Enter the name (up to 11 characters are allowed) and press < Enter >.
- Dflt Event: This option allows you to assign a default name for an event. When you invoke this option you will be prompted to Enter event default name >. Enter the desired name (up to 11 characters are allowed) and press < Enter >.

5.1.10 Text

This option allows you to set a default color for the text in your diagram. To invoke this option position the cursor over the Text box and press the left mouse button. You will be prompted to **Pick a new color from the color bar**. Position the cursor on the desired color box and press the left mouse button. (**NOTE:** The $\dagger \ddagger$ (color) option is active at this time. This option will display the additional color selections available.) All text will now be displayed in this color. If the diagram already exists, only the new text will be displayed in the new color. You must return to the ATTRIBUTES option to change the existing text color (if desired).





January 1992

5-28

5.1.11 cntr/left/right

This option allows you to set the justification for your diagrams. Justification is where the text will be placed offset from the placement point. This works as a toggle switch. To change justification, position the cursor over the cntr (default) box and press the left mouse button or the <Enter > key. You will see the box change from cntr to rght (right). Press the left mouse button or <Enter > key again, and the box will change from rght to left.

Left justification means the text will be anchored at the left bottom corner, or the text will flow to the right of where it was placed. Center justification means the text will be centered about the placement point. Right justification means the text will be placed to the left of the placement point. Set the toggle for the desired justification.

5.1.12 Text Size (0.50)

This option allows you to set a default text size for your diagrams. The default value is .5. Text size ranges from .001 to 9.0. For the purpose of writing text in gate blocks, a text size of about 0.5 is appropriate. The size looks too small on the screen, but it is a good size for sending to a laser printer. **REMEMBER**, here you are setting the default text size. You may always change the text size for special text by invoking the ATTRIBUTES option, under the EDIT command.

5.1.13 Fill

This option allows you to select a color for the drawing symbols. When you invoke this option you will be prompted to **Pick a new color from the color bar**. Position the cursor over the desired color and press the left mouse button. (**NOTE:** The †4 (color) option is active at this time. This option will display the additional colors available.) The Fill box will change to the newly selected color. All drawing symbols created in your new diagram will be displayed in this selected color. If you modify an existing drawing, the symbol colors won't change. To change the color of existing symbols you must invoke the ATTRIBUTES suboption from the EDIT option.

5.1.14 Line

This option allows you to select a color for the lines that connect the text and symbols in the diagram. When you invoke this option you will be prompted to Pick a new color from the color bar. Position the cursor over the desired color and press the left mouse button. (NOTE: The $\uparrow \downarrow$ (color) option is active at this time. This option will display the additional color selections available.) The Line box will change to the selected color. All lines generated in your new diagram will be this selected color. Again you are modifying an existing drawing the lines won't change to this color. To change existing lines to the new color you must invoke the ATTRIBUTES suboption for the EDIT option.

5.1.15 type

This option allows you to set a default line type for your diagrams. When you invoke this option,



a small window appears displaying the three line types (solid, broken, and dotted). To select a line type, position the cross hair over the desired line type and oress the left mouse button. All ones drawn in your new diagram will be of this type. If you have an existing drawing that you are modifying, remember the existing lines do not change. To change existing lives you must access the ATTRIBUTES suboption under the EDIT option.

5.1.16 Mult

This option allows you to toggle between multiple pick and single pick. The multiplek option allows you to continue with a given process until terminated. For example, when creating an "AND" gate, the symbol will reappear after each placement until you cancel the process by pressing the right mouse button. If single 'sngl' is turned on, the user must return to the mend and select "AND" again after each placeme.

5.1.17 Scroll Color Bar († 4)

Daris

This option allows you to scroll the color bar to display the additional color selections available. Position the cursor on the color bar scroll box (up and down arrow, and press the left mouse button or <Enter > key. The next series of available colors will be displayed.

5.2 Plot Trees

This option allows you to plot graphics. When you invoke this option, Figure 53 will be displayed. All available plot files are displayed. The following options are available:

Patiene way to Figure 42

LAII	Returns you to rigure 42.
Plotter	Sends the specified file to the plotter.
Hpgl File	Generates an HPGL formatted file.
Raster File	Rasterizes a selected fault tree diagram for printing (highes) quality - 300 dpi).
Low Resolution Raster	Rasterizes a selected fault tree for printing (lower quality - 150 dpi).

In addition, the following function keys are available for the remaining options presented in this section:

< Esc >		Returns you to the previous screen.
<f1></f1>	11.08	Displays on-line help messages.
< F2 >	걸려가	Marks a file for further processing. When you mark a file, an asterisk will appear in front of the file name



Option [E] Exit / Plotter / Hpgl File / Raster File / Low Resolution Raster
 CCS ECS
the second s
an approximation of a sector point to a more a sector in a sector of an an approximation when the sector of a sect
Note : Use <pgup> and <pgdn> to display more Plot Files.</pgdn></pgup>

Figure 53. HP plot files display screen.

<f3></f3>	. *	Marks all the displayed files for further processing.
< F4 >	*	Allows you to mark a range of files for further processing.
< F5>		Allows you to locate a specific file for further processing. When you invoke this function, Figure 54 will be displayed. Enter the file name to locate and press < Enter >. The specified file will be highlighted. If the file name coes not exist, then the next
		name in alphabetical order will be highlighted.

5.2.1 Plotter

This option takes the .DLS files generated in the HP Plotter option in the graphics editor (Bulld) and sends it to the HP Pen Plotter (7475). If a plotter is not defined, you will receive the message **Plotter not attached**.

5.2.2 Hpgl Files

This option creates .HPG files. An .HPG file may be printed using the batch program PLOTHPG. Once the * HPG file is created, exit IRRAS and return to the PRADATA.B1\XXXXX subdirectory. At this point, you must type the following command:

where

option [E] EXIT / Plotter / Hpg	File / Roster File / Low Resolution Roster
CCS ECS	e Plot Files
Please enter the name to Lucate	ethen press <enter>.</enter>
L	

Figure 54. Locate a specific file prompt.

wild card specification used to spool all .HPG files contained in the specified directory. If you do not wish to print all *.HPG files, you may specify an individual file.

NOTE: PLOTHPG will only work on an HP LaserJet III or IIIP.

5.2.3 Raster File

This option allows you to rasterize fault tree diagrams for printing. Obtaining a "laser printed" output of a fault tree diagram is done in two steps. First, you must select the drawings to rasterize. Once the drawings have been rasterized, you must exit the program and return to the PRADATA B1\XXXXX subdirectory. At this point, you use the PLOTRAS batch program to send the files to the laser printer. The command to invoke this batch program is:

... PLOTRAS *. RAS.

(See 5.2.2 for where list definition).

NOTE: PLOTRAS will work on any laser printer.

This option rasterizes the selected fault tree to 300 dots per inch (dpi). This option provides you with the highest quality of output available. To invoke this option, enter an $\langle R \rangle$ in the option field, highlight or mark the file(s) to rasterize, and press $\langle Enter \rangle$. A percent complete message will be displayed so you can track the progress of the rasterizing. When complete, a message will be displayed notifying you that the raster image has been created.

5.2.4 Low Resolution Raster

This option rasterizes the selected fault tree to 150 dpi. To invoke this option, enter an <L> in the option field, highlight the file(s) to be rasterized, and press < Enter >. A message will be displayed telling you that the rasterizing is being performed. A percent complete will also be displayed so you can track the progress of the rasterizing. When complete, a message will be displayed notifying you that the raster image has been created.

5.3 Load Graphic Trees

This option allows you to load graphic files. When you invoke this option, Figure 55 will be displayed. As shown, two options are available: Exit and Load.

GRADHICS GRADHICS AFW14A AFW15 AFW178 AFW178

Figure 55. Load graphics tree screen.

5.3.1 Exit

This option returns you to the previous screen (Figure 42). To invoke this option, enter an $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

5.3.2 Load

This option allows you to load graphics into the system. A conversion of the selected file will be performed. To invoke this option, enter an <L> in the option field, highlight the file or mark the



files to be loaded using the function keys, and press < Enter >. A message will be displayed telling you that the selected file(s) is being converted.

5.4 Alpha to Graphics

This option allows you to convert the alphanumeric logic for a system or subsystem to a graphical format. Use this option when you define the fault tree logic with an ASCII file or use the logic editor to change the logic for a fault tree. The graphics and logic information for a fault tree are two separate entities, therefore, a conversion must be made if any changes are made to the alphanumeric logic. However, if changes are made to the graphics representation, the changes are automatically reflected in the alphanumeric logic. When you invoke this option, Figure 56 will be displayed. As showr, two options are available. Exit and Convert.



Figure 56. Alpha to graphics screen.

5.4.1 Exit

This option returns you to the previous screen (Figure 42). To invoke this option, enter an $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

5.4.2 Convert

This option allows you to convert an alpha file(s) to a graphical format. To invoke this option.

enter a < C > in the option field, highlight the file (using the arrow key) or mark the files (using the function keys) you wish to convert. A small window will appear at the top of the screen (as shown in Figure 57). Two prompts are displayed asking you whether you want to use tables and/or boxed events in the newly created graphic file. Enter a Y or N. Press < Enter > when complete. A these are will appear notifying you that the graphics image has been created



Figure 57. Set table/box toggle for conversion.

5.5 Graphical Pager

This option allows you to break up larger drawings into many pages or smaller drawings. When you invoke this option, Figure 58 is displayed. Each menu option is discussed in the following paragraphs.

5.5.1 Exit

This option returns you to the previous screen (Figure 42). To invoke this option, enter an $\langle E \rangle$ in the option field and press $\langle Enter \rangle$ or press the $\langle Esc \rangle$ key.

5.5.2 Page

This option invokes the paging options. To invoke this option, enter a $\langle P \rangle$ in the option field, highlight the file to page, and press $\langle Enter \rangle$. Figure 59 will be displayed. As shown, seven options are available. Each option is discussed in the following paragraphs.





Figure 58. Fault tree pager menu.

5.5.2.1 \leftarrow PAGE \rightarrow . This option allows you to move the menu to another location on the screen. To invoke this option, position the cursor on the \leftarrow PA $^{-r_2} \rightarrow$ box and press the left mouse button. An outline box will appear. Move the box to the desired location and press the \sim mouse button again. The pop-up menu will be displayed at the new location.



Figure 59. Graphical pager pop-up menu.

5.5.2.2 . It. This option returns you to the previous screen (Figure 58) To invoke this option, enter an $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

5.5.2.3 Find. This option allows you to locate a specific event or label on the diagram. When you invoke this option, you will be prompted to **Enter name** >. Enter the name you wish to locate and press < Enter >. The symbol containing the entered name will be highlighted with a broken dotted line. The dotted line will remain until you invoke another option.

5.5.2.4 Show. This option allows you to display the labels in a readable format. When you invoke this option, you will be prompted to **Pick a shape**. Position the cross hair on the desired shape and press the left mouse button. The event or label will be displayed at the bottom of the screen. At this point you may position the cross hair on another symbol or shape or press the right mouse button to terminate the process.

5.5.2.5 Page. This option allows you to insert or transfer a page break in your diagram. When you invoke this option you will be prompted to **Select a Gate for the desired page break**. Position the cross hair at the appropriate gate and press the left mouse button. The page break will be created. At this point you may create another page or press the right mouse button to terminate the process.

5.5.2.6 Table/No Tab. This option toggles generation of "tables" rather than "basic events." To toggle, position the cursor on the Table box and press the left mouse button. The menu will be redisplayed and the Table box will now read No Tab (basic events).

5.5.2.7 Boxed/No Box. This option toggles generation of "boxed basic events" rather than "basic events." To toggle, position the cursor on the Boxed option and press the left mouse button. The menu will be redisplayed and the Boxed option will now read No Box (basic events).

5.6 eXtract Fault Trees

This option allows you to extract fault trees from the data base. When you create a fault tree diagram and save it, the .DLS file is saved in the data base as well as in a temporary .DLS file. When you delete the temporary .DLS file, the file still exists in the data base. This option allows you to extract the .DLS file from the data base. When you invoke this option, Figure 60 will be displayed. On this screen all the fault trees residing in the current directory are displayed. Four options are available: Exit, eXtract Trees, Clear Extracted Trees, and Display Extracted Trees.

5.6.1 Exit

This option returns you to the previous screen (Figure 42). To invoke this option, enter an $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

5.6.2 eXtract Trees

This option allows you to remove specific trees from the data base. To invoke this option, enter





Figure 60. Extract fault trees screen.

an <X> in the option field, highlight the tree to be extracted or mark the trees to be extracted (using the function keys) and press < Enter>. A message will be displayed notifying you that the tree(s) was successfully extracted.

5.6.3 Clear Extracted Trees

This option allows you to clear all extracted trees from a file. To invoke this option, enter a <C> in the option field, highlight the file to be cleared or mark the files using the function keys, and press < Enter>. A warning screen will appear (Figure 61) telling you that all existing *.DLS files will be deleted. Enter a <Y> to delete or an <N> to terminate the process. Remember, this deletion only deletes the extracted picture file. The picture file will still reside in the data base.

5.6.4 Display Extracted Trees

e

This option allows you to display all the extracted trees to date. To invoke this option, enter a <D> in the option field and press < Enter>. A screen similar to the one shown in Figure 62 will be displayed. Press < Enter> to return to the previous screen.









5.7 Define Plotter Pens

This option allows you to assign colors to the plotter pens. When you invoke this option, a screen will be displayed showing 16 colors. Select the color to be mapped by positioning the cross hair over the desired color and pressing the left mouse button. You will then be prompted to Enter number of pen >. Enter the number of the pen that will contain the selected $c_{\rm eff}$. Continue this process until all pens have been defined. To terminate the process, position the cursor on the STOP symbol and press the left mouse button.





6.0 FAULT TREE ANALYSIS

The Cault Tree Analysis option consists of the following four functions.

- Analyze Systems includes updating the cut sets, quantifying the cut sets, and running uncertainty analyses.
- 2. Display Results presents the analyses in various report forms
- The Cut Set Editor provides the means to modify the fault tree cut sets.

Logic Editor provides a means for editing the fault tree logic in an alphanumeric format.

Keys that you will frequently use are

< Esc >	Escape cancels your last choice and returns you to the previous screen.
< F1 >	Help briefly explains the function of a field and may show you examples of data entered.
<f2></f2>	Mark/Unmark tags items for use in the selected option
<f3></f3>	Clear All Marked events removes the marks (*) from the listed items. If no items are marked, this option will mark all of the items.
<f4></f4>	Mark/Unmath range of items tags large numbers of items for processing.
<f5></f5>	Locate an item displays a blank field in the center of the screen, and a message
	the name to be located and then press < Enter >. This feature will place the
	name in alphabetical order will be highlighted

To invoke this option, highlight FAULT Tree Analysis and press < Enter >. The fault tree analysis main coefficient is shown in Figure 63. The options available from the fault tree analysis main menu are: Exit, Analyze Systems, Display Results, Cut Set Editor, and Logic Editor. These options and their functions are discussed in the following paragraphs.

6.1 Exit

The Fault Tree Analysis screen appears with Exit $\langle E \rangle$ as the default choice in the command line (see Figure 63). Press $\langle Enter \rangle$ to return to the IRRAS main menu.

6.2 Analyze Systems

This option provides the means to recalculate system values after events and/or cut sets have been modified. To invoke the option, highlight Analyze Systems or type < A > in the option field, and press < Enter >. Figure 64 shows the main screen for system analysis that lists the systems defined for the current family. The letters c, q, and u (in any combination) may precede a system name and are defined



SURRY	Fault Tree Analysis	
	Exit	1
	Analyze Systems	1
	Display Results	
	Cut Set Editor	
	Logic Editor	1
	Option [D]	



anal da sami in antinena alanda a	
Option E	Exit / Generate Cut Sets / Cut Set Update / Quantification Monte Carlo Uncertainty / Latin Hypercube Uncertainty
C.U.C. N.a.m.	e containment spray
CS	CONTAINMENT SYSTEMS
c CV	CORE VULNERABLE TO CD
D1	HIGH PRESSURE INJECTION - AUTOMATIC
c u D2	HIGH PRESSURE INJECTION - MANUAL
cqu pa	MIGH PRESSURE INJECTION + RCP SEALS
COU DA	HIGH PRESSURE INJECTION * EMERGENCY BURATION
EQU DA	IND DEFECTIOE IN TERTION
cqu F1	INSIDE SPRAY RECIRCULATION
and the second sec	control entered a subset of the second
Exit	Kolm Mark Line Mark Ell Mark Bakes Lineste
EXIL	neip mark Line Mark All mark Kange Locate



as follows:

- c flags the system as having cut sets that must be recalculated
- q flags the system as having cut sets that must be requantified
- u flags the system as needing uncertainty distributions recalculated.

Once these functions have been executed the corresponding letter is removed from the display. The options available from the Analyze Systems screen are Exit. Generate Cut Sets, Cut Set Update, Quantification, Monte Carlo Uncertainty, and Latin Hypercube Uncertainty.

IRRAS writes the summary information generated during analysis to a file that the user can tead to determine the results of a batch operation involving several systems or sequences. This file is called "SCREEN.CPY." When either Analyze Systems or Analyze Sequences is executed, IRRAS deletes a file in the current family subdirectory called "SCREEN.OLD." The file "SCREEN.CPY" is then renamed to "SCREEN.OLD." The summary screens for any operations performed during this session (using the Analyze option) are then written to the file "SCREEN.CPY." The user may exit IRRAS, change to the specified subdirectory, and print, edit, or review this file.

6.2.1 Exit

To return to the Fault Tree Analysis menu, type an $\langle E \rangle$ (Exit) in option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

6.2.2 Generate Cut Sets

This option allows you to generate the cut sets for a selected system based on cut set generation cutoff values. You are given the opportunity to specify several cut set generation cutoff values that will be used to determine if a cut set is to be retained or discarded from the selected system.

You are given the ability to generate cut sets for a selected system, a group of systems, or all systems within the current family. To invoke the Generate Cut Sets for a single selected system, type $\langle G \rangle$ (Generate Cut Sets) in the option field, highlight the desired system, and press $\langle Enter \rangle$. To invoke this process for a group of systems, mark the desired systems using the function keys F2, F3, and F4, type a $\langle G \rangle$ in the option field, and press $\langle Enter \rangle$. To invoke this option for all systems in the current family, clear all marked entries with the F3 key, type $\langle G \rangle$ in the option field and press $\langle Enter \rangle$. A message **Process all records?** (Y/N) will appear at the bottom of the screen. Type a $\langle Y \rangle$ to continue the Generate Cut Sets for all records that need processing (e.g., for $\langle G \rangle$ the systems must have a "C" in front of the name, etc.), or type an $\langle N \rangle$ to terminate the update for all systems.

Whether you are generating cut sets for a single system, a group of systems, or for all systems the Cut Set Generation Cetoff Values screen shown in Figure 65 will be displayed. You may change any of the data fields on this screen. Each field is described below. The default values that appear on this screen may be reset to new values by selecting the Utility Options from the IRRAS main menu and then invoking the Define Constants option.

Perform Cut Set Prob Truncation? If you enter a < Y >, then the only cut sets whose product for all of its event probabilities is greater than or equal to the value in the Cutoff V-lue field will be kept. All other cut sets will be removed.

If you enter an <N>, then the probability for the cut set will not be relevant for determining if the cut set should be retained or discarded.



SURRY	Analyze	Syste	m s		
					1
an an a fair an a fair an	CUT SET GENERATIO	n cutoff va	LUES merenanismus		
Perform Cut Set Prot	Truncation? (Y/N)	X s	Cutoff Value	1.000E-015	
Perform Event Prot	Truncation? (Y/N)	N Min <	Cutoff Value	1.0008-003	
Perform Size or Zone	Truncation7 (Y/N/	zi x >	Size Cutoff	6	
	Starting Gale Name				
Note: To perform	Event Probability	truncation	you must als.	specify	

Finure 65. Cut set generation cutoff values.

Perform Event Prob Truncation	If you enter a $\langle Y \rangle$, then you must also enter $\langle Y \rangle$ for "Perform Cut Set Prob Truncation." This option will check all cut sets that are below the probability cutoff (Min < Cutoff Value field) and remove them only if they contain an event whose probability is below this value.
Perform Size or Zone Truncation?	If you enter a $\langle Y \rangle$, then only the cut sets whose number of events is less than or equal to the value specified in the Size Cutoff field will be kept in the cut sets for that system. All other cut sets will be removed. If you enter an $\langle N \rangle$, then the number of events in a cut set will be irrelevant for determining if the cut set should be retained or discarded. If you enter a $\langle Z \rangle$, then only zone flagged events will be checked.
Starting Gate Name	If you specify a gate in this field, then this gate will be the gate assumed to be the top gate of the tree. If you do not specify a name, the top gate will be determined by finding the gate that is not referenced by any other gate. This is usually the same as the system or subsystem name.

During processing the screen shown in Figure 66 is displayed and updated as the calculations proceed. Upon completion of the cut set generation, the results are displayed as shown in Figure 67.
- Family -SURRY Analyze Systems annesse Currently Processing manness stem Name D2 System Name Surrent Gate Name D2 Total Number of Gates 71 Current Gate Number 45 Total # of Cut Sets 440 Press <ESC> to terminate.

Figure 66. Status screen for cut set update.

L		L A D A L Y	2.6	5 7 5	 # 5	
D2 Cut Se 1 2 3 4 5 6 7 8 9 10 20 7 10 20 7 10 20 7 10 7 0 10 7 0 10	et Size 9 616 1483 714 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Elapsed Time D0:11:13.660 UpperBound 9.093E-004 1.513E-004 1.385E-006 1.461E-006 E E E E 1.062E-003				





There is a limit on the number of cut sets that can be stored for a given system. If the number of cut sets generated by this option exceeds this limit, then a message is displayed indicating that cut sets were truncated beyond what the user specified. The screen showing the summary information for all generated cut sets is displayed. When complete, you are seturned to Figure 66 where the "c" in front of the processed cut sets will be removed.

6.2.3 Cut Set Update

This option will update the alternate cut sets for a selected system based on cut set generation cutoff values. You are given the opportunity to specify several cut set generation cutoff values that will be used to determine if a cut set is to be retained or discarded from the selected system. You are also given the choice of using either the base case cut sets or the alternate case cut sets as the starting set of cut sets to be updated. The updated version of the cut sets will be saved as the new alternate cut sets for the system.

You are given the ability to update the alternate cut sets for a selected system, the alternate cut sets for a group of systems, or the alternate cut sets for all of the systems within the current family. To invoke the Cut Set Update process for a single selected system, type < C > (Cut Set Update) in the option field, highlight the desired system, and press < Enter >. To invoke this process for a group of systems using the function keys F2, F3, and F4, type a < C > in the option field, and press < Enter >. To invoke this option for all systems in the current family. clear all marked entries with the F3 key type < C > in the option field and press < Enter >. A message **Process all records?** (Y/N) will appear at the bottom of the screen. Type a < Y > to continue the Cut Set Update for all of the systems, or type an < N > to discontinue the update for all systems.

Whether you are updating *c*^{trannate} cut sets for a single system, a group of systems, or for all systems the Cut Set Generation Cutoff Values screen shown in Figure 68 will be displayed. You may change any of the data fields on this screen. The default values that appear on this screen may be reset to new values by selecting Utility Options on the IRRAS main menu and then invoking the Define Constants option.

If you type a < Y > in the Perform Proform all of its event probabilities is greater than or , the value in the Cutoff Value field will be kept. All other cut sets will be removed from alternate case cut sets for that system. If you type an < N > in this field, then the probability for the cut set will not be relevant for determining if the cut set should be retained of discarded.

If you type a < N > in the Perform Cut Set Size Cutoff field, then only the cut sets whose number of events is less than or equal to the value in the Size Cutoff field will be kept in the alternate case cut sets for that system. All other cut sets will be removed. If you type an < N > in this field, then the number of events in a cut set will be irrelevant for determining if the cut set should be retained or discarded.

Is you enter a $\langle Y \rangle$ in the Use Base Case Cut Sets field, then the base case cut sets will be used as the cut sets to be updated and then stored in the alternate case cut sets. However, if an $\langle N \rangle$ was entered in this field, the alternate cut sets will be used as the cut sets to be updated and then resaved in the alternate case cut sets



1	SURRY Analyze Systems
1	
1	
	Cut Set Generation Cutoff Values
	Perform Probability cutoff? (T/N) Y Cutoff Value 1.000E-015
	Perform Cut Set Size Cutoff? (Y/N) Y Size Cutoff 6
1	Use Base Case Cut Sets? (Y/N) N

Figure 68. Cut set generation cutoff values.

During processing the screen shown in Figure 69 is displayed and updated as the calculations proceed. Upon completion of the cut set update, the results are displayed as shown in Figure 67. When finished viewing the results, press < Eme \sim to return to Figure 64.

	an an an an ann an an an an an an an an	Currently Processing
		Name D2
		PERMITTER CONTRACTOR CONT
		Total Number of Cut Sets 2517 Current Cut Set Number 53C

Figure 69. Status screen for cut set update.



The cut set update flag 'c' will then be removed from every system on the Analyze Systems screen in which the cut set update process has been performed. If an error of some kind occurs during the update process then the message **Error in Cut Set analysis** will be displayed at the bottom of the screen. Once the cut sets are updated, they are also automatically quantified and the appropriate "q" flag is cleared.

6.2.4 Quantification

The quantification process will calculate a new minimum cut set upper bound for the system cut sets using the current data values (event change sets and alternate case cut sets). The new minimum cut set upper bound is saved with the alternate case cut sets for the selected system.

You are given the ability to requantify the alternate cut sets for a selected system, for a group of systems, or for all of the systems within the current family. To invoke the Quantification process for a single selected system, type $\langle Q \rangle$ (Quantification) in the option field of the Analyze Systems screen, highlight the desired system, and press $\langle Enter \rangle$. To invoke this process for a group of systems, mark the desired systems using the function keys F2, F3, and F4, type a $\langle Q \rangle$ in the option field and press $\langle Enter \rangle$. To invoke this option for all systems in the current family, type $\langle Q \rangle$ in the option field and press $\langle Enter \rangle$. A message **Process all records (Y/N)**? will appear c the bottom of the screen. Type a $\langle Y \rangle to$ continue the Quantification process for all of the systems, or type an $\langle N \rangle$ to discontinue this process.

During processing the r essage Quantification in progress appears at the bottom of the screen. If an error occurs, the message Error quantifying cut sets will be displayed at the bottom of the screen.

Upon completion of the quantification process, the results are displayed as shown in Figure 70. The requantify flag 'q' will then be removed from every system on the Analyze Systems screen for which the quantification process has been performed.

6.2.5 Uncertainty Analysis

This option allows you to run an uncertainty analysis on a system using one of the two sampling techniques provided. The two sampling techniques are the Monte Carlo simulation technique and the Latin Hypercube simulation technique.

To use the Monte Carlo sampling technique to obtain an uncertainty analysis for a system or a group of systems, you must type an < M > the option field of the Analyze Systems screen and press < Enter >. However, if you wish to use the Latin Hypercube sampling technique, then type an < L > in the option field and press < Enter >.

You are given the option of running an uncertainty analysis on the alternate cut sets for either a single system, for a group of systems, or for all of the systems within the current family. To invoke this process for a single selected system, type <L> or <M> (Latin Hypercube / Monte Carlo) in the option field, highlight the desired system, and press < Enter>. To invoke this process for a group of systems using the function keys F2, F3, and F4, type a <L> or <M> in the option field and press < Enter>. To invoke this process for all systems in the current family, type <L> or <M> in the option field and press < Enter>. A message **Frocess all records?** (Y/N) will appear at the bottom of the screen. Type a <Y> to continue the uncertainty analysis for all of the

Quantifi	cation Results	 	
Name D4			
Min Cut 5.1	UpperBound 55E-003		
Elaps 00:00	ed Time 0:00.170		

Figure 70. Results of the quantification process

systems, or type an < N > to discontinue the analysis.

Once the sampling technique for the uncertainty analysis has been selected and the desired system(s) selected, the Uncertainty Calculation Values screen will be displayed (see Figure 71 for Monte Carlo sampling). This screen prompts you to enter the number of samples to be generated during the simulation and the initial value of the seed for the random number generator. There will be a default seed in the field for the random seed. You may use this value or enter a new value for the seed. To obtain a random seed from the system clock, you must enter a zero in this field. There will also be a default value in the field for the number of samples. You may use this value or enter another value. You may change the default values for both the number of samples and the random seed in the Utility Options, Define Constants suboption.

When using the Monte Carlo sampling technique for the uncertainty analysis, if the number of samples entered is less than ten, then the number of samples will be increased to ten before the uncertainty analysis process will continue. Any number of samples greater than or equal to ten will be allowed, but a number of at least 1000 is probably a better value for improving the reliability of the Monte Carlo results.

When using the Latin Hypercube Sampling (LHS) technique, if the number of samples entered is less than twice the total number of unique events in the system, then the number of samples will be increased to two times the total number of unique events before the analysis will continue. The LHS technique gives its best results if the number of samples is at least twice the total number of unique events.

SU	RRY Anelyze Systems
	양부가 잘 하는 것 같은 것 같은 것 같은 것 같은 것 같은 것 같이 많이 했다.
1	Monte Carlo Uncertainty Calculation Values
	Number of samples to use in Monte Carlo simulation 1000
	Seed for random number generator 0
	NOTE: Use "O" as the seed to get a random seed from the clock.
-	Press (Enter> to continue

Figure 71. Monte Carlo calculation values.

Once the number of samples has been accepted and a seed obtained from the system clock (if necessary), checks will be run to ensure the events with the same correlation classes have consistent failure data, uncertainty data, and distribution types. If any events with inconsistencies exist, an error message will be displayed and the uncertainty analysis process will be terminated so that the inconsistent values may be corrected.

If an error of some type occurs during the uncertainty analysis process, the process is terminated and the message **Error in Uncertainty analysis** is displayed at the bottom of the Analyze Systems screen.

If all of the events successfully pass the correlation class checks, then the distribution parameters for the events will be checked to ensure that they are valid. If any of the parameters are invalid, error messages will be displayed and the process will be terminated so the distribution parameters may be corrected.

After both of these checks have been passed, a point estimate will be calculated for the selected system. At this point the samples for each event will be generated using the selected sampling technique, either Monte Carlo Sampling technique or the Latin Hypercube Sampling technique. The uncertainty analysis function provides you with eight different distribution types for both sampling techniques. The distribution types include normal, lognormal, beta, gamma, chi-squared, exponential, uniform and the user-defined histograms.

During processing the current status screen will be displayed and updated as the samples are generated. Figure 72 illustrates the current status screen for the Monte Carlo sampling technique. When the requested number of samples has been generated or the user has terminated the process of generating the samples by pressing the < Esc > key, statistical information will be calculated using the generated samples. A sample mean, median, and standard deviation will be calculated for the selected system. Coefficients of skewness and kurtosis, and quantile values will also be calculated for the system. This data will be saved in the data base for the selected system.



Family SURRY Analyze Systems - Currently Processing Total Number of samples.... Current Sample..... 382 Running Mean Value 1.264E-001 Press Esc to terminate.

Figure 72. Current status of the Monte Carlo sampling.

Upon completion of these calculations, the following values will be displayed on the Uncertainty Results screen for viewing: the system name, random seed used, the number of samples generated in this process, the total number of events and cut sets in the system being processed, the point estimate, the mean, the median, the 5th and 95th percentile values, the minimum and maximum generated sample values, the standard deviation, the skewness and kurtosis, and the time involved to perform the analysis. Figure 73 illustrates the Uncertainty Results screens for the Monte Carlo sampling techniques, respectively.

If only one system was selected (highlighted) for the uncertainty analysis process, then you will need to press <Enter > to return to the Analyze Systems screen from the Uncertainty Results screen. However, if more than one system is being processed, the Uncertainty Results screen will be displayed for each system, and when all of the selected systems have been processed, you will automatically be returned to the Analyze Systems screen.

6.3 Display Results

To display the results of your system analysis, highlight Display Results or type $\langle D \rangle$ on the Fault Tree Analysis screen and press $\langle Enter \rangle$. The Display System Results screen will be displayed showing a list of the systems contained in the data base (Figure 74). The following options are available: Exit, Cutsets, Importance, and Uncertainty.













6.3.1 Exit

This option terminates the process and roturns to the Fault Tree Analysis screen (Figure 63). To invoke the option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key

6.3.2 Cutsets

To invoke this option, enter a $\langle C \rangle$ in the option field, highlight the desired system, and press $\langle Enter \rangle$. This option displays the system's cut sets (Figure 75), their percent of contribution to the system, the frequency, and the event names that make up the cut sets. The system minimum upper bound, the number of cut sets that make up the system, the current partition upper bound, the percentage that the partition contributes to the system, and the number of cut sets in the partition are displayed at the bottom of the screen. In addition, five suboptions are available. Each option is discussed in the following paragraphs.

	Option	B Exit /	Partition / Report / Basic Events / Complement
Num	2	Frequency	Event Names
1	24.49	2.600E-004	HP1-CCF-FT-1158D
3	9.42	2.500E-004	HP1-CCF+FT-867CD HP1-CKV-FT-CV225
4	9.42	1.000E-004	HP1-CKV-FT-CV25
5	9.62	1.000E-004	HPI-CKV-FT-CV410
7	3.77	4.000E-005	UPU-CUF-LF-STRAB HPI-XVM-PG-XV24
8	2.89	3.0728-005	CPC-MDP-F5-SW108 CPC-MDP-FR-SW10A
9	1.39	1.475E-005	CPC-MDP-FR-SW10B CPC-MDP-FR-SW10A
.11/	0.03	A 1006-000	NPI-MOV-FI-TIISC HPI-MOV-FT-1115E

Figure 75. Cut sets display.

6.3.2.1 Exit. This option terminates the process and returns to the previous screen. To invoke this option, type $\langle E \rangle$ and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

6.3.2.2 Partition. The partition option allows you to redefine a system as a subset of the original cut sets. This is accomplished by defining a set of events to be used to determine whether a cut set belongs to a partition. The functions available to perform this task are: Exit, Include, Exclude, Complement, Reset, and View Events.

To invoke this option, type $\langle P \rangle$, highlight a system, and press $\langle Enter \rangle$. This will bring up the screen shown in Figure 76.





Figure 76. System partitioning menu.

Exit: This option terminates the process and returns to the previous screen. To invoke this option, type $\langle E \rangle$ and press $\langle Enter \rangle$, or press $\langle Esc \rangle$.

Include: To establish a partition via this option, type $\langle I \rangle$ in the option field, then fill in the entry fields that are to be used to qualify the events that may be used in the new partition and press $\langle Enter \rangle$. The application proceeds to qualify the events and when complete will update the Number of Qualified Events field that appears at the top of the screen. In this case, qualified events are those events that contain the included attributes. Returning to the Cut Sets screen via the Exit option, you will see that the system cut set list contains only those cut sets that are made up of qualified events. Figure 77 was the result of specifying the event name "HPI-CKV-FT-CV410" for the "Include" option. If the system cut sets do not contain any of the qualified events, then the message **No cutsets qualify** is displayed at the bottom of the screen.

EXclude: To establish a partition via this option, type $\langle X \rangle$ in the option field, then fill in the entry fields on the screen that are to be used to remove events from the list of qualified events. The application proceeds to remove the events. When complete, the Number of Qualified Events field that appears at the top of the screen will be updated accordingly. Returning to the Cut Sets screen via the Exit option you will see that the system cut set list contains only those cut sets that are made up of qualified events. Figure 78 was the result of specifying the event name "HPI-CKV-FT-CV410" for the "EXclude" option. If the system cut sets do not contain any of the qualified events, then the message No cutsets qualify is displayed at the bottom of the screen.



	Option	P Exit /	Partition / Report / Basic Events / Complement
Num	x	Frequency	Event Names









Complement: To establish a partition via this option, type < C> in the option field, and press < Enter>. This causes all currently qualified events to be disqualified, and all unqualified events to become the set of qualified events. The Number of Qualified Events field at the top of the screen will change accordingly. Returning to the Cut Sets screen via the Exit option you will see that the system cut set list contains only those cut sets that are made up of qualified events. If the system cut sets do not contain any of the cualified events then the message **No Cut Sets Qualify** is displayed at the bottom of the screen.

Reset: This option sets all events in the data base to qualified. This, of course, removes all partitioning from the current system cut sets. To activate this option, type $\langle R \rangle$ in the option field and press $\langle Enter \rangle$. (Assume the partition shown in Figure 77). The original cut set list is the result, as shown in Figure 75.

View Events: This option displays the list of family events and allows you to mark those events that are to be considered qualified events (Figure 79). To activate this option, type $\langle V \rangle$ in the option field and press $\langle Enter \rangle$. Using $\langle F2 \rangle$, $\langle F3 \rangle$, or $\langle F4 \rangle$ keys, mark the events that are considered qualified and press $\langle Enter \rangle$. You will be returned to the Partition screen where the number of qualified events will be updated accordingly (Figure 80). If the system cut sets do not contain any of the qualified events, then the message **No cutsets qualify** is displayed at the bottom of the screen.

SURRY	View Events	D1
	Option E Exit	
* ACC-CKV-FT-CV107 CHEC ACC-CKV-FT-CV109 CHEC ACC-CKV-FT-CV128 CHEC ACC-CKV-FT-CV130 CHEC * ACC-CKV-FT-CV130 CHEC * ACC-CKV-FT-CV145 CHEC * ACC-CKV-FT-CV147 CHEC * ACC-MOV-PG-1865A ACC ACC-MOV-PG-1865B ACC ACC-MOV-PG-1865C ACC ACC-MOV-PG-1865C ACC ACP-BAC-ST-1H1 480V	D e s c r i p t K VALVE CV107 FAILS TO OPEN K VALVE CV109 FAILS TO OPEN K VALVE CV128 FAILS TO OPEN K VALVE CV130 FAILS TO OPEN K VALVE CV145 FAILS TO OPEN K VALVE CV147 FAILS TO OPEN MOTOR OPERATED VALVE 1865A P MOTOR OPERATED VALVE 1865B P MOTOR OPERATED VALVE 1865C P AC BUS 1H1 BUSWORK FAILURE	1 0 N
<esc> <f1> Exit Help</f1></esc>	<f2> <f3> Mark Line - Mark All</f3></f2>	<f4> <f5> Mark Range Locate</f5></f4>





	Total Number of Events 776 Number of Qualified Events 6
Option V Exit	/ Include / EXclude / Complement / Reset / View Events
	Comp. Id. Sup. Topic. Turne E/Mode Location Init?
Names <p></p>	Event Attributes Comp Id Sys Train Type F/Mode Location Init? N

Figure 80. Results of marking events to view.

6.3.2.3 Report. The Report option allows you to generate a report of the data that is displayed on the screen. The report may be displayed on the console, sent to an attached printer, or saved in a disk file for later processing (Figure 81). To invoke this option, type < R > in the option field, and press < Enter >. Upon pressing < Enter >, Figure 81 is displayed. This screen shows a default title and file name. You may change these defaults to meet your needs.

Accession in the second	EPORT OPTIONS
Partition Cut Set	Report Title
COM	Utput File Name war and and a set of the set
NOTE: File Name =	"CON" - Output report to the screen. "PRN" - Output report to the printer. " " - No report is produced.

Figure 81. Partition report type selection.



6.3.2.4 Basic Events. The Basic Event option provides the following detailed information about the events that make up a cut set: name, description, probability, component ID, system, component type, failure mode, and location (Figure 82). To invoke this option, type $\langle B \rangle$ in the option field, highlight a cut set, and press $\langle Enter \rangle$.

Friend Name	nen ariantes ana erranza de inse					NUMPERSON AND ADDRESS OF TAXABLE STREET,
event wame	Duckshill	Desc	Suntac	Turn	100	
and a second state of the second	Probability	comp 10	system	Type	1.14	Location
OF L MAY FF MERIN	1.0008-004		HPI	CKV	FT	

Figure 82. Details of a cut sets basic events.

6.3.2.5 Complement. The Complement option operates on the current system partition. Its function is to set all currently qualified cut sets to unqualified. This allows you to split a system into two partitions and then switch between the two partitions to view the results. To invoke this option, type <C> in the option field, and press <Enter>. (Assume the partition shown in Figure 77). The resulting display is shown in Figure 78.

6.3.3 Importance

This option calculates and displays the following three important measures for each event in the system:

- Fussell-Vesely importance an indication of the percentage of the minimal cut set upper bound contributed by the basic event.
- Risk Reduction Ratio an indication of how much the minimal cut set upper bound would decrease if the basic event was made perfect (never fail).
- Risk Increase Ratio an indication of how much the minimal cut set upper bound would increase if the basic event was always failed.



If the Intervals flag is set in the Constants Menu, the Birnbaum measure (the partial derivative), the Risk Reduction Interval, and the Risk Increase Interval will be displayed.

To invoke this option, type $\langle I \rangle$ in the option field, highlight a system (or you may process all records by not specifying a system), and press $\langle Enter \rangle$. By default, when the importance data are first displayed it is sorted, high to low, by Fussell-Vesely importance value (Figure 83).

SURRY		Import	ance	01	
Oution 1	ni svir	/ Description	/ Dartition	/ Pennet / S	ort
option	DI CAIL	. 7 peace prior	ey yarereisii	7 Report 7 W	
		Dechebility	ананананананананананананананананананан	Dick Bachic	Rick Incon
tvent Name	-Occur-	Probability		Ratio	Ratio
HP1-CCF-FT-867CD	1	2.600E-004	2.447E-001	1.324E+000	9.4188+002
HP1-CCF-FT-11580	1.1.1	2.600E-004	2.447E-001	1.324E+000	9.41BE+002
P1-CKV-F1-CV225	1	1.000E-004	9.409E-002	1.104E+000	9.4188+002
API-CKV-FT-CV25	1	1.000E-004	9.409E-002	1.104E+000	9.418E+002
HP1-CKV-F1-CV410	1.1	1.0006-004	9.409E-002	1.104E+000	9,418E+002
CPC-MDP-FR-SW10A	32	3.8408-003	5.8468-002	1.0628+000	1.608E+001
CPC-CCF-LF-STRAB	1 A.	4.700E-005	4.422E-002	1.045F+000	9.418E+002
HP1-XVM-PG-XV24	1	4.000E-005	3.763E-002	1,039E+000	9,418E+002
CPC-MDP-FS-SW108	20	8.000E-003	3.469E-002	1.036E+000	5.2988+000
and and the many	22	1 6000-003	1 ARGE-002	1.0178+000	1.149F+001





6.3.3.1 Exit. Returns you to the previous screen.

6.3.3.2 Description. Displays the full description for the highlighted event on the bottom of the screen. To invoke this option, enter a $\langle D \rangle$, highlight the desired event, and press $\langle Enter \rangle$. If no event is highlighted, the message An event must be selected first... is displayed.

6.3.3.3 Partition. This option invokes the same process as described in Section 6.3.2.2, except the effect is to limit which events are displayed/reported.

6.3.3.4 Report. This option invokes the same process as described in Section 6.3.2.3.

6.3.3.5 Sort. When you invoke this option, Figure 84 will be displayed. As shown, the data can be sorted by the following: Name, Occurrence, Probability, F-V, Reduction, and Increase.



	has					
Sort Options :	Exit Sort Reduction	/ Name / I	Occurrence /	Probability /	F-V	
			-			
Event Name	PI PI	obability	F - V	Risk Reduc.	Risk Incre.	
HPI-CKV-FT-CV410	1 1	.000E-004	9.4098-002	1.1048+000	9.4188+002	

Figure 84. Importance measures sorted by probability.

6.3.4 Uncertainty

The Uncertainty option displays the distribution and confidence limits of a system for both base and current data values. These values were calculated using either the Latin Hypercube or the Monte Carlo simulation technique. To invoke this option, type $\langle U \rangle$ (Uncertainty) in the option field of the Display System Results screen, highlight the desired system, and press $\langle Enter \rangle$. Figure 85 shows the base and current case uncertainty data for a highlighted system.

From this screen you may either return to the System Display screen or view the quantile values associated with the current case data or the base case data. To return to the System Display screen, type an $\langle E \rangle$ (Exit) in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key. To view the quantile values for the current case, type a $\langle C \rangle$ in the option field and press $\langle Enter \rangle$. A screen showing the quantile values will appear (Figure 86). To return to the previous screen, press $\langle Enter \rangle$. If you wish to view base case quantile values, type a $\langle B \rangle$ in the option field and press $\langle Enter \rangle$.

6.4 Cut Set Editor

The cut set editor provides you with the means to edit the base case/alternate system cut sets. IRRAS provides room for two sets of cut sets in the data base: base case and alternate, or current case. Whenever IRRAS creates cut sets, they are stored in the current case location. The user may save a set of cut sets to the base case location by performing a Base Case Update. The user can choose to edit either base case or alternate cut sets; however, the results are always stored in the alternate case location.



an and the local discount of the second second second		Loursenances	Contract States	- L	-
Option	ICI Exit / C	urrent Quar	tile Values /	Base Quanti	Le Values
ht : H					
Mean	4.6998-004	Median	4.237E-004	Mincut	2.687E-001
Std. Dev	1.212E-004	Skewness	1.657E+000	Kurtosis	4.822E+000
5th %	3.6548-004	Minimum	3.654E-004	Seed	53990
95th %	7.919E-004	Maximum	7.9198-004	Samples	15
	Size Cutoff	6	Probability (utoff 1.0	00E - 015
		Cu	irrent	-	material and a second second second
Mean	4.699E-004	Median	4.237E-004	Mincut	2.687E-001
Std. Dev	1.212E-004	Skewness	1.657E+000	Kurtosis	4.822E+000
5th %	3.654E-004	Miniskan	3.654E-004	Seed	53990
95th %	7.919E-004	Maxinum	7.919E-004	Samples	15
	Siza Cutoff		Probability P	utoff 1.0	005-015

Figure 85. Uncertainty data display.

Distribution Quantile Level	95% Confidence Interval On Quantile Level		CURRENT 95% Cor Interval c	CASE fidence n Quantile
(in per cent)	1n % (*/*)	Quantile Value	Lower Bound	Upper Bound
0.5	6.9	3.6546E-004	3.65468-004	3.6546E-004
1.0	8.4	3.6546E-004	3.6546E-004	3.6546E-004
2.5	11.2	3.65468-004	3.6546E-004	3,7003E-004
5.0	14.4	3.6546E-004	3.6546E-004	3.7021E-004
10.0	18.5	3.7003E-004	3.6546E-004	3.9247E-004
20.0	23.6	3.9247E-004	3.6546E-004	4.2324E-004
25,0	25.2	3.9247E-004	3.6546E-004	4.23758-004
30.0	26.5	3.9301E-004	3.7003E-004	4.2375E-004
40.0	28.1	4.2324E-004	3.7021E-004	4.6749E-004
50.0	28.6	4.2375E-004	3.9247E-004	4.9533E-004
60.0	28.1	4.6749E-004	4.0875E-004	4.9592E-004
70.0	26.5	4.9265E-004	4.23758-004	6.9206E-004
75.0	25.2	4.9533E-004	4.2375E-004	7.9197E-004
80.0	23.6	4.9592E-004	4.6644E-004	7.9197E-004
90.0	18.5	6.9206E-004	4.95338-004	7.9197E-004
95.0	14.4	7.9197E-004	4.9592E-004	7.91978-004
97.5	11.2	7.9197E-004	6.9206E-004	7.9197E-004
99.0	8.4	7.9197E-004	7.9197E-004	7.9197E-004
99.5	6.9	7.9197E-004	7.9197E-004	7,9197E-004

Figure 86. Quantile values display.



Any event name entered during cut set editing may be preceded by a "/" to indicate that it is to be treated as a complemented event. The probability of a complemented event is 1 - the failure probability.

To invoke this option, highlight Cut Set Editor or type $\langle C \rangle$ in the option field and press $\langle Enter \rangle$. Figure 87 shows the Cut \langle et Editor screen listing the system names and descriptions. Whether the system has associated by a case cut sets and/or alternate cut sets is indicated by the letter B and A preceding the system name. To edit the base case cut set, select $\langle B \rangle$, highlight a system showing a letter B, and press $\langle Enter \rangle$. Likewise, to edit an alternate cut set select $\langle A \rangle$, highlight a system showing a letter A, and press $\langle Enter \rangle$. If a system name is not flagged with a B or A it may still be edited.

Annese services destriction and	annal has a second
Opti	on A Fxit / Base Case Cut Sets / Alternate Cut Sets
op	and but which were save and a second second second
paBaAaaa N a m	e Description
BAC	CONTAINMENT SPRAY
BACS	CONTAINMENT SYSTEMS
BACV	CORE VULNERABLE TO CD
B A D1	HIGH PRESSURE INJECTION + AUTOMATIC
B A D2	HIGH PRESSURE INJECTION - MANUAL
B A D3	HIGH PRESSURE INJECTION - RCP SEALS
BAD4	HIGH PRESSURE INJECTION - EMERGENCY BORATION
B A D5	ACCUMULATORS
BAD6	LOW FRESSURE INJECTION
BAFI	INSIDE SPRAT RECIRCULATION
	<esc> <f1> <f5></f5></f1></esc>
	Exit Help Locate

Figure 87. System selection for cut set editor.

In any case, a screen similar to the one shown in Figure 88 is displayed. This screen shows 14 editing options that are activated by a single key stroke. The 14 options include: Exit, Add, Modify, Delete, Locate, Next, Previous, Search, Options, Insert Events, Replace Events, Copy Cutset and Replace Events, Undo, and Find and Delete Cut Set. The Add, Modify, and Delete functions will perform their function on either the entire cut set or on a single event in a cut set depending upon where the cursor is positioned at the time the key to activate the function is pressed. If the cursor is in the first column of the screen, then the function will be performed on the entire cut set. However, if the cursor is in one of the other four columns, then the function will be performed on the event where the cursor is positioned.

256		£
		в.
		8
285		£ -
120	087	÷.,

	CUTSETEDITOR BASE	CASE CUT SETS
Exit / Add / Modify / Insert Event / Replac Find and Delete Cut Se	Delete / Locate / Next / Previous / S e Events / Cony Cutset and Replace Ev t	earch / Options ents / Undo
Set #Event Names		
1 LOSP	DEP-CCF-FS-DG13	
CSS-CCF-FS-CS1AR		
4 CLS-ACT-FA-CLS28	CSS-MOV-FT-101A	
5 ACP-CR8-CO-15J7	CSS-MOV-FT-101A	
6 ACP-BAC-ST-4KV1J	CSS-MOV-FT-101A	
7 CLS-ACT-FA-CLS2A	CLS-ACT-FA-CLS28	
8 CLS-ACT-FA-CLS2A	CSS-MDP-FR-181HR	
9 CLS-ACT-FA-CLS2A	CSS-FLT-PG-CS18	
10 CLS-ACT-FA-CLS2A	CSS-MOV-PG-100B	
TT CLS-ACT-FA-CLS2A	CSS-MDP-MA-CS1B	
1 1 X X X X X X X X X X X X X X X X X X	CLN-RUI-FA-CLNZR	



To move the cursor about the editing window, use the keyboard cursor pad. The arrow keys move one field in each respective direction. The <Home> key places the cursor in the upper right corner of the window. The <PgUp> key moves the window up one page (12 lines). The <PgDn> key moves the window down one page. <Ctrl-PgUp> and <Ctrl-PgDn> moves to the top and bottom of the file. Pressing the function key <F1> toggles between two cut set editor screens. One screen shows the editing options and the other shows the available cursor movements.

6.4.1 Exit

This option returns you to the Cutset Editor screen (Figure 87). After pressing $\langle E \rangle$ (Exit) you are asked if the changes are to be saved or discarded (Figure 89). If the changes are saved, the sequence and plant frequencies must be recalculated to reflect the new cut set configurations.

6.4.2 Add

This option allows you to add a cut set to a system or an event to a cut set. To add a new cut set, move the cursor into the set number column and press <A>. The cut set display is cleared, and the cursor is placed in the event name column (Figure 90). The event names that make up the cut set may be entered using the <Tab> key to move between fields, and the <Enter> key to end the addition. The window is then updated and the cursor is positioned on the new cut set (Figure 91). The <Insert> key also invokes the Add option.



Do you want to SAVE the changes made to the Cut Sets (Y/N) | j



C	CUIS	ET EDITOR	BASE CASE CUT S	ETS
	New Cut	Set Addition		
Use <tab> to</tab>	move between field	ents use the "Add s, <enter> when</enter>	done, <esc> to abor</esc>	t
Set # Event	Names			
			-	

Figure 90. Adding a new cut set to a system.

To add an event to a cut set place the cursor in the row of the cut set you wish the event to be in, press <A>, and enter the event name. Figure 92 shows the editor screen ready to add an event name. Figure 93 shows the results of adding a new event.





C.	CUT SET E	DITOR BA	SE CASE CUT SETS
Exit / Add / Modify / Insert Event / Replac	Delete / Locate / e Events / Copy C	Next / Previous / Sut Set and Replace	Search / Dptions Events / Undo
Find and Delete Cut Se	t		
t #Event Names			-
1837 ACP-CRB-CO-15H8	ACP-CR8-CO-15J8	DEP-DON-FS-DO03	DEP-DON-MA-DOD1
1838 ACP-CR8-CO-15H8	ACP-CRB-CO-15JB	OEP CRB-FT-15K3	DEP-DON-FR-DG03
1839 ACP-CR8-CO-15H8	ACP-CRB-CO-15J8	OEP-CRB-F1-15H3	DEP-DGN-KA-DG03
1840 ACP-CR5-CO-15H8	ACP-CR8-CO-15J8	OEP-CRB-FT-15H3	OEP-CRB-FT-15J3
	ACP-CPR-CO-15.IR	OEP-CRB-FT-15H3	OEP-DON-FC-DO3U2
1841 ACP-CR8-CO-15H8	WAL HUD DO INDO		
1841 ACP-CR8-CO-15H8 1842 ACP-CR8-CO-15H8	ACP-CR8-CO-15J8	OEP CRB-F1-15H3	DEP-DGN-FS-DG03
1841 ACP-CR8-CO-15H8 1842 ACP-CR8-CO-15H8 1843 ACP-CR8-CO-15H8	ACP-CR8-CO-15J8 ACP-CR8-CO-15J8	OEP - CRB - F1 - 15H3 OEP - DGN - FR - DG03	OEP-DGN-FS-DG03 OEP-DGN-FS-DG01
1841 ACP-CR8-CO-15H8 1842 ACP-CR8-CO-15H8 1843 ACP-CR8-CO-15H8 1844 ACP-CR8-CO-15H8	ACP-CR8-CO-15J8 ACP-CR8-CO-15J8 ACP-CR8-CO-15J8 ACP-CP8-CO-15J8	OEP - CRB - F1 - 15H3 OEP - DGN - FR - DGO3 OEP - DGN - FS - DGO1	OEP-DGN-FS-DG03 OEP-DGN-FS-DG01 OEP-DGN-MA-DG03
1841 ACP-CR8-CO-15H8 1842 ACP-CR8-CO-15H8 1843 ACP-CR8-CO-15H8 1844 ACP-CR8-CO-15H8 1844 ACP-CR8-CO-15H8	ACP - CR8 - CO - 15J8 ACP - CR8 - CO - 15J8 ACP - CR8 - CO - 15J8 ACP - CP8 - CO - 15J8 ACP - CR8 - CO - 15J8	0EP-CR8-F1-15H3 0EP-DGN-FR-DG03 0EP-DGN-FS-DG01 0EP-CR8-F1-15J3	0EP - DGN - FS - DG03 0EP - DGN - FS - DG01 0EP - DGN - MA - DG03 0EP - DGN - FS - DG01
1841 ACP-CR8-CO-15H8 1842 ACP-CR8-CO-15H8 1843 ACP-CR8-CO-15H8 1844 ACP-CR8-CO-15H8 1844 ACP-CR8-CO-15H8 1845 ACP-CR8-CO-15H8	ACP-CR8-CO-15J8 ACP-CR8-CO-15J8 ACP-CR8-CO-15J8 ACP-CR8-CO-15J8 ACP-CR8-CO-15J8 ACP-CR8-CO-15J8	OEP-CR8-F1-15H3 OEP-DGN-FR-DG03 OEP-DGN-FS-DG01 OEP-CR8-F1-15J3 OEP-DGN-FC-DG3U2	0EP-0GN-FS-0G03 0EP-0GN-FS-0G01 0EP-0GN-MA-0G03 0EP-0GN-FS-0G01 0EP-0GN-FS-0G01
1841 ACP-CR8-CO-15H8 1842 ACP-CR8-CO-15H8 1843 ACP-CR8-CO-15H8 1844 ACP-CR8-CO-15H8 1845 ACP-CR8-CO-15H8 1846 ACP-CR8-CO-15H8 1846 ACP-CR8-CO-15H8 1846 ACP-CR8-CO-15H8 1846 ACP-CR8-CO-15H8 1846 ACP-CR8-CO-15H8	ACP-CR8-CO-15J8 ACP-CR8-CO-15J8 ACP-CR8-CO-15J8 ACP-CR8-CO-15J8 ACP-CR8-CO-15J8 ACP-CR8-CO-15J8 ACP-CR8-CO-15J8	0EP-CR8-F1-15H3 0EP-DGN-FR-DG03 0EP-DGN-FS-DG01 0EP-CR8-FT-15J3 0EP-DGN-FC-DG3U2 0EP-DGN-FC-DG01	0EP-DGN-FS-DG03 0EP-DGN-FS-DG01 0EP-DGN-MA-DG03 0EP-DGN-FS-DG01 0EP-DGN-FS-DG01 0EP-DGN-FS-DG01



G	CUT SET I	DITOR BA	SE CASE CUT SETS
And a second			
Enter the	Event name to be	added to this Cut	Set
Use .	Enter> when done,	and <esc> to abort</esc>	the second sector beaution and
of #Event Name			
1801 ACP-CR8-CO-15J8	OFP-BAC-ST-FORF	OEP-CRB-FT-15H3	DEP -DGN -FC-E-3U2
1802 ACP-CR8-CO-15J8	OEP-BAC ST-FDRF	OEP - CRB - FT - 15H3	DEP-DGN-FS-DG03
1803 ACP-CRB-CO-15J8	OEP-BAC-ST-FDRF	OEP-DGN-FR-DG03	OEP-DGN-FS-DG01
1804 ACP-CR8-CO-15J8	OEP BAC-ST-FDRF	DEP-DGN-FS-DG01	OEP-DGN-MA-DG03
1805 ACP-CR8-CO-15J8	OEP-BAC-ST-FDRF	OEP-CR8-FT-15J3	OEP-DGN-FS-DG01
1806	OEP . BAC - ST - FDRF	DEP-DGN-FC-DG3U2	OEP-DGN-FS-DG01
1807 ACP - CRB - CO - 15J8	OEP - BAC - ST - FDRF	OEP-DGN-FS-DG01	OEP-DGN-FS-DG03
1808 ACP-CR8-CO-15H8	OEP-BAC-ST-FDRD	OEP-DON-FR-DG01	OEP-DGN-FR-DG03
1809 ACP - CRB - CO - 15H8	OEP-BAC-ST-FDRD	DEP-DOM-FR-DGO1	OEP-DGN-MA-DG03
1810 ACP-CRB-CO-15H8	OEP - BAC - ST - FORD	OEP - CRU - FT - 15.13	DEP-DGN-FR-DG01
1811 ACP-CR8-CO-15H8	OEP-BAC-ST-FORD	DEP-DGN-FC-DG3U2	OEP-DGN-FR-DG01
	OPE DED OF FRIER	DED DOM CB DODA	050-06N-55-0003





114

C		CUI SET	EDITOR BA	SE CASE CUT SETS
Exit	/ Add / Modify	/ Delete / Locate	/ Next / Previous /	Search / Options
Inser	t Event / Replai	ce Events / Copy	Cut Set and Replace	Events / Undo
rina	and perete cut a	es.		
1.1	Sector Sector			
100×	ACD-COD-CO-15 (0	OCD DAG DT LINK	OFF OFF FT AF-17	APA ANY PA ANY
1001	AUF-UKB-UU-1200	OCD DAP OT CODE	DEP-UKB-F1-12H2	DEP-DON-FC-DOSUE
1002	ACP-CRB-CO-1510	DEP-BAU-ST-FORF	DEP-LEB-FI-1585	OEP-DGN-TS-DG03
1000	APP ORD OD SE ID	DEP BAG ST FORF	DEP-DON-TR-DGUS	DEP-DUN-FS-DUOT
1004	AUP-URB-UU-1208	DED-DAC-ST-FDRF	DEP-DUN-15-DUUT	DEP-DGN-MA-DG03
1804	ACD COD CO 1200	DEP BAL ST FURF	DEF-LEB-FI-1243	DEF-DUN-FS-DUUT
1000	ALP-GRB-LO-1220	UEP-DAL-SI-FURF	DEP DON-FC-DUDUE	DEP-DON-YS-DGUT
1807	ADD-DDD-CD-ADKF	DED DAD ST. CODE	000	NER NON TO NAME
1808	ACD_C00.00.1500	DED BAC OT END	OED -DON- FR- 50001	000-000-75-0003-
1800	ACD - COD - CO- 12HO	DEP-DAU-DI-FURD	OFF DON TR DOUT	DEP-DON-FR-DOUD
1810	ACD .000 .00 1000	DED. BAC. DT. EDED	OEP OUN FR DGUI	DEP-DON-MA-DOUS
1012	ACD_CDD.CD. 1000	DEP DAL ST FDRU	DEP DEP TO DOD	DEP DUN - FR DUUT
	ALF LEB LU IDHO	DEF BAU SI FORD	THEK DOW - NO STORE	UEP-DGN-FK DGU1

Figure 93. A new event added to a cut set.

6.4.3 Modify

To modify events of an entire cut set, place the cursor in the first column of the screen and press the <M> key. This will display a screen containing the first 20 events (if available) of the cut set to be modified (Figure 94). After making modifications to the events of the cut set, press < Enter > to save the modifications or press < Esc > to abort the modification process.

To change a single event in a cut set, position the cursor on the desired event and press the <M> key. This will display the Change Event Name screen (Figure 95) to allow you to modify the current event. When modifying an event, you may use all of the alphanumeric keys, in addition to the <ins> and keys.

6.4.4 Delete

This option allows you to delete an entire cut set or an individual event. Placing the cursor on a cut set number and pressing < D > causes the entire cut set to be deleted. If the cursor is placed on an event name, the event at that location only is deleted from the cut set.

When the delete option is invoked, the option list on the display is replaced by a veto menu. For deletions, the Veto option is always turned on unless you specifically turn it off. The veto options function as follows:



1				k
- 8				8
- 1				8
1	Υų		85	e

2

0

C		CUT SET E	DITOR BA	SE CASE CUT SETS
They	e first 20 events rent names and pre Use s	Modify Entir of the cut set are ss enter to save th Enter> when done, i	re Cutset e being displayed, ne changes made to and ~Esc* to abort	Modify the the cutset.
et # 803	Event Names	OEP-BAC-ST-FDRF	DEP-DGN-FR-DG03	DEP-DGN-F\$-DGD1

8

Figure 94. Modifying a cut set.

Loss contractor and		
and the state of the	Change Event Name	
Event names which	h do not exist will be added to the events list	
Use <e< th=""><th>nter> when done, and <esc> to abort</esc></th><th></th></e<>	nter> when done, and <esc> to abort</esc>	
Set # www.Event Names-	(a) and (a)	
1 LOSP	OEP - CCF - FS - DG13	
2 RW1-1NK-LF-RWS1 2 Dec. CCL. CC. CCLAD		
2 CLC ACT - FA - CLC 70	CEC. HOU. ET. 1014	
S APD PPR-CO-1517	CSS-MOV-F7-101A	
A ACP-RAC-ST-GEVIS	CSS-MOV-FT-1014	
7 CLS-ACT-FA-CLSPA	CLS+ACT-FA-CLS2B	
B CLS-ACT-FA-CLSZA	CSS-MDP-FR-1B1HR	
9 CLS-ACT-FA-CLS2A	CSS-FLT-PG-CS18	
10 CLS-ACT-FA-CLS2A	CSS-MOV-PG-1008	
11 CLS-ACT-FA-CLSZA	CSS-MDP-MA-CS1B	
12 ACP-CR8-CO-14.01	CLS-ACT-FA-CLS2A	
And the second s	and been debalanced. This was been as an an experiment to some or a for stranger or some and the second	-





Stop. Typing $\langle S \rangle$ aborts the delete operation for the highlighted cut set or event.

<u>Continue the process</u>. Typing < C> and pressing < Enter> results in deleting a cut set or event.

6.4.5 Locate

The locate option allows you to search the list of all existing events and the list of all the aewly added events for events that meet the desired search criteria. Events that meet the desired search criteria are considered to be qualified events. You are then able to locate the first occurrence of one of these qualified events from the current set of cut sets.

To invoke this option, press the <L> key. This will display the locate menu (Figure 96). From this screen, the following four options are available: Exit, Locate, Next, and Previous. The number of unique events that exist in the current set of cut sets is listed on this screen, along with the number of currently gualified events.

Anner and the second second	any sector and the se
	Total Number of Events 84 Number of Qualified Events 0
	Antine 111 Exit / Length / Next / Descious
<p>DCP*</p>	es Compild Sys Train Type F/Mode Location Init?

Figure 96. Locate an event in the cut set list.

6.4.5.1 Exit. If the locate function has been performed and an event name exists in the name field of this screen, then upon entering an $\langle E \rangle$ in the option field, the cursor will be placed on the first occurrence of that event name within the current set of cut sets. If the event does not exist within the current cut sets, then a message stating the event was not found will be displayed.

If the locate function has not been performed then entering an $\langle E \rangle$ in the option field will simply terminate the locate function. This may also be terminated by pressing the $\langle Esc \rangle$ key.

6.4.5.2 Locate. This option searches the list of all existing events and the list of all newly added events for events that meet the desired search criteria. This option allows wild card search characters for the two name fields on the screen. A question mark (?) matches any single character in its specified



position, and an asterisk (*) matches a character string of any length. This option will use any of the filled event attribute fields on the screen as search criteria. Any events that meet the selected search criteria are considered to be qualified events. The number of currently qualified events will then be updated on the screen. The first qualified event will be displayed on the screen (see Figure 97).

NOTE: It is possible for an event to be a qualified event and still not exist in the current set of cut sets, because the entire data base is searched in the qualification process.



Figure 97. Results of a locate request.

6.4.5.3 Next. This option will display the next qualified event from the list of qualified events.

6.4.5.4 Previous. This option will display the previous qualified event from the list of qualified events.

6.4.6 Next

This option allows you to go to the next occurrence of the event name that was specified in the locate option. To invoke the option, type < N >.

6.4.7 Previous

This option allows you to go back to the previous occurrence of the event name specified in the locate option. To invoke the option, type $\langle P \rangle$.

6.4.8 Search

This option provides a fast locate feature for a single event name. As each letter of the event name is typed the cut set list is scanned for the first occurrence of the letter. To invoke the option, type $\langle S \rangle$, which will replace the option list with the search request on the screen display (Figure 98). The



search starts with the top cut set of the current display. Type the desired event name and press <Enter>. The cursor will be positioned on the first occurrence of the requested event. At this point, <N> (Next) and <P> (Previous) will allow you to move through the cut set list positioning the cursor on the next or previous occurrence of the requested event name.

RPS		CUT SET	EDITOR	ALTERNATE CUT SETS
		Speed	Search	
-	Enter the	event name to be Use <enter< th=""><th>search for => 1 > when done</th><th>RA1</th></enter<>	search for => 1 > when done	RA1
Dat H	Europe Names			
11	JLB	RTMCCF		
2 1	CHANNEL1	CHANNELS		
3 (CHANNEL 1	CHANNEL 3		
5	CHANNELC	CHANNELS		
61	RTBMCCF			
7.1	RTBMB	1RA2	1RA3	
8 1	RTBMA	1882	1883	
10	IRAZ	IRAD	1885	1805
11	TRMB	IRA1	18A2	
12 1	RTBMb	IRA1	1RA3	
CONTRACTOR OF STREET	Contraction and the state of th	antinana ana ana ana ana ana ana ana ana an	erine on the transmission of the set	

Figure 98. Search for a specific event.

6.4.9 Options

The Options command provides the capability of setting Cut Set Editor defaults. To invoke this option, type $\langle O \rangle$. The normal option list will be replaced with the "default" option list on the screen display (Figure 99). The default options are:

- Range This option sets the maximum number of cut sets with the specified string that will be located. Range is used by the Replace, Insert, Copy and Replace, and Find and Delete commands. By default, Range is set to 9999.
- Veto Veto tells each operation to ask again if you wish to modify the cut set data. By default, this is turned off.
- Delete Veto Functions the same as Veto. However, it applies only to the Delete command. By default, this is turned on.





C	CUTSETEDITOR BASE CASE CUTSE	ΥS
	Ontions	
kange value ≈> 9	999 Veto «DN» => N Delete Veto «DN» => Y	
-Use «Tab» to move be	tween fields, <enter> when done, and <f1> for help</f1></enter>	proved a
et # weimeEvent Namesen		
1 LOSP	DEP-DCF-FS-DU13	
Z RWI-INK-LF-RWSI		
A CIS-ACT-FA-CIS2R	CSS-MOV-FT-101A	
4 CLS-ACT-FA-CLS28 5 ACP-CR8-CO-15J7	CSS-MOV-FT-101A CSS-MOV-FT-101A	
4 CLS-ACT-FA-CLS2B 5 ACP-CRB-CO-15J7 6 ACP-BAC-ST-4KV1J	CSS-MOV-FT-101A CSS-MOV-FT-101A CSS-MOV-FT-101A	
4 CLS-ACT-FA-CLS2B 5 ACP-CRB-CO-15J7 6 ACP-BAC-ST-4KV1J 7 CLS-ACT-FA-CLS2A	CSS-MOV-FT-101A CSS-MOV-FT-101A CSS-MOV-FT-101A CLS-ACT-FA-CLS2B	
4 CLS-ACT-FA-CLS28 5 ACP-CR8-CO-15J7 6 ACP-BAC-ST-4KV1J 7 CLS-ACT-FA-CLS2A 8 CLS-ACT-FA-CLS2A	CSS-MOV-FT-101A CSS-MOV-FT-101A CSS-MOV-FT-101A CLS-ACT-FA-CLS2B CSS-MDP-FR-1B1HR	
3 CSS-CCF-FS-CSTAB 4 CLS-ACT-FA-CLS2B 5 ACP-CRB-CO-15J7 6 ACP-BAC-ST-4KV1J 7 CLS-ACT-FA-CLS2A 8 CLS-ACT-FA-CLS2A 9 CLS-ACT-FA-CLS2A	CSS-MOV-FT-101A CSS-MOV-FT-101A CSS-MOV-FT-101A CLS-ACT-FA-CLS2B CSS-MDP-FR-1B1HR CSS-FLT-PG-CS1B	
3 CSS-CCF-FS-CSTAB 4 CLS-ACT-FA-CLS2B 5 ACP-CRB-CO-15J7 6 ACP-BAC-ST-4KV1J 7 CLS-ACT-FA-CLS2A 8 CLS-ACT-FA-CLS2A 9 CLS-ACT-FA-CLS2A 10 CLS-ACT-FA-CLS2A	CSS-MOV-FT-101A CSS-MOV-FT-101A CSS-MOV-FT-101A CLS-ACT-FA-CLS2B CSS-MDP-FR-1B1HR CSS-FLT-PG-CS1B CSS-MOV-PG-100B	
3 CSS-CCF-FS-CSTAB 4 CLS-ACT-FA-CLS2B 5 ACP-CRB-CO-15J7 6 ACP-BAC-ST-4KV1J 7 CLS-ACT-FA-CLS2A 8 CLS-ACT-FA-CLS2A 9 CLS-ACT-FA-CLS2A 10 CLS-ACT-FA-CLS2A 11 CLS-ACT-FA-CLS2A	CSS-MOV-FT-101A CSS-MOV-FT-101A CSS-MOV-FT-101A CLS-ACT-FA-CLS2B CSS-MDP-FR-1B1HR CSS-FLT-PG-CS1B CSS-MOV-PG-100B CSS-MDP-MA-CS1B	

Figure 99. Change the cut set default options.

6.4.10 Insert Event

This option allows you to execute a global insertion of an event. You specify a search string consisting of up to five event names and the name of the event to be inserted. In all cut sets containing the "search string" the specified event to be inserted will be added. The search starts with the cut set at the top of the current display. To invoke this option, type <1>. The option list will be replaced on the screen with the Find/Insert request as shown in Figure 100.

6.4.11 Replace Events

This option allows you to specify a search string and replace every occurrence of that string with a specified event name. The search string may consist of up to five event names. The replacement is a global function in that all occurrences of the search string will be replaced by the specified event name. The replacement event name may be left blank, which will delete all occurrences of the search string from the cut set list. The search starts with the cut set at the top of the current display. To invoke the option, type $\langle R \rangle$. The option list on the screen will be replaced with the Find/Replace request (Figure 101).

6.4.12 Copy Cutset and Replace Events

This option allows you to search for cut sets containing a specific string, copy the cut sets (minus the events specified in the search string), and add a specific event to the new cut set. You may specify up to five event names as a search string. If the event name to be added is left blank, the process is terminated. The search starts with the cut set at the top of the current display. To invoke this option, type < C >. The option list on the display will be replaced with the Copy/Replace request (Figure 102).



ACD DAD OT UNIT	And Inser	t many
ACP-BAC-ST-4KV1J www.	ACP-BAC-ST-4	KKLM
(1) • 12 (1) (1) (1) (1) (2) (1)		
-Use <tab> to move be</tab>	tween fields, «Enter» when done and, «F1» for	holp
Set # memerent Names-	A second s	Note that the second
1 LOSP	OEP-CCF-FS-DG13	
2 RWT TNK - LF-RWST		
3 CSS-CCF-FS-CS1AB		
4 CLS-ACT-FA-CLS28	CSS-MOV-FT-101A	
5 ACP - CRB - CO - 15 J7	CSS-MOV-FT-101A	
6 ACP BAC-ST-4KV1J	CSS-MOV-FT-101A	
7 CLS-ACT-FA-CLS2A	CLS-ACT-FA-CLS2B	
8 CLS-ACT-FA-CLS2A	CSS-MDP-FR-1B1HR	
9 CLS-ACT-FA-CLS2A	CSS-FLT-PG-CS1B	
10 CLS-ACT-FA-CLS2A	CSS-MOV-PG-100B	
11 CLS-ACT-FA-CLS2A	CSS-MDP-MA-CS18	
12 ACP-CRB-CO-14J1	CLS-ACT-FA-CLS2A	









C CUT BASE CASE CUT SETS SET EDITOR Find And Copy & Replace-->ACP-AAA-LM-15N4 ACP-CRB-CO-15J7 *ACP-BAC-ST-6KV1J --Use <Tab> to move between fields, <Enter> when done and, <FT> for help--Set #Event Names-DEP-CCF-FS-DG13 2 RWT . TNK - LF - RWST 3 CSS-CCF-FS-CS1AB 4 CLS-ACT-FA-CLS28 CSS-MOV-FT-101A 5 ACP-CR8-CO-15J7 CSS-MOV-FT-101A 6 ACP-BAC-ST-4KV1J CSS-MOV-FT-101A 7 CLS-ACT-FA-CLS2A CLS-ACT-FA-CLS2B 8 CLS-ACT-FA-CLS2A CSS-MDP-FR-181HR 9 CLS-ACT-FA-CLS2A CSS-FLT-PG-CS18 10 CLS-ACT-FA-CLS2A CSS-MOV-PG-1008 11 CLS-ACT-FA-CLS2A CSS-MDP-MA-CS18 12 ACP-CRB-CO-14J1 CLS-ACT-FA-CLS2A

Figure 102. Copy and replace a cut set.

This option allows you to recover the last ite.n deleted. The item may be an entire cut set or a single even A deletion may be undone only if the Undo Delete command is used immediately after the deletion has occurred. To invoke the option, type < U >.

6.4.14 Find and Delete Cut Set

This option allows you to delete all cut sets that contain a specified string. You may specify up to five event names as a search string. The search begins at the cut set at the top of the current display. To invoke this option, type $\langle F \rangle$. The option list will be replaced with the find and delete request (Figure 103).

6.5 Logic Editor

This option allows you to modify the logic of a system or subsystem. When you invoke this option, Figure 104 will be displayed. You may toggle the display to list systems or subsystems (see Section 6.5.3). Three options are available: Exit, Logic Editor, and Sub-System.

6.5.1 Exit

6.4.13 Undo

This option terminates the process and returns you to the Fault Tree Analysis screen (Figure 63). To invoke this option, enter an $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.



C	4 U 1 SET EDITOR BASE CASE CUT SETS
- Use crabs to pove be	Ind And Delrte LOSP *ACP-CRB-CO-16J1 * *
Set £ 37 ACF - TFM - NO- 1J 38 ACP - BAC - ST - 4B01J 39 CSS - MDP - FR - 1A1HR 40 CSS - MDP - FR - 1A1HR 40 CSS - MCP - FR - 1A1HR 42 CSS - MCP - FR - 1A1HR 43 CSS - FLT - PG - CS1A 44 CSS - FLT - PG - CS1A 45 CSS - FLT - PG - CS1A 46 CSS - FLT - PG - CS1A 47 ACP - CRB - CO - 14J1 48 ACP - 1FM - NO - 1J	CSS-MDP-FR-1A1HR CSS-MDP-FR-1A1HR DCP-BDC-ST-BUS1B CSS-MDP-FP-1A1HR CSS-XVM-RE-XV15 CSS-MDP-FR-1B1HR CSS-MDP-FR-1B1HR CSS-MDP-FR-1B1HR CSS-MDP-FR-1B1HR CSS-MDP-FR-1B1HR CSS-MDP-FR-1B1HR CSS-MDP-FR-CS1B CSS-MDP-MA-CS1B CSS-FLT-PQ-CS1A CSS-FLT-PQ-CS1A









5.5.2 Logic Editor

To invoke this option, enter an <L> in the option field, highlight the desired system. I press < Enter >. If you do not highlight a system, the message **A record must be h'ghlighted first** will be displayed. After highlighting a system and pressing < Enter >, Figure 105 will be displayed. All gates, types, and inputs contained in the selected system or subsystem will be displayed. Thirteen editing options are availabil. Each option will be discussed in the following paragraphs.

SURRY	1	Logic E	8110F	01
Exit / A Insert / Re	dd glece	Modify / Delete / Search / Options	/ Locate / Next / U.do Delete / P	/ Previous 1 Help
Gote Name	Type TRAN TRAN DR TRAN OR OR	HPI-CKV-FI-CV225 E18 HPI-MDP-FS-CH18 HPI21 HPI6	Inputs HP12 CPCB HP1-MDP-FR-1B6HR HP16	HP13 GKV1J HP1-MDP-MA-CH18 HP15
HP121 HP13 HP14 HP15	TEAN AND TRAN OR TRAN	HP17 SIS-ACT-FA-SISP HP1-MOV-PG-1250A HP19	HP18 HP117 HP1-MOV-PG-12868	HPI-CCF-F6-CH1BC HPI-CKV-F1-CV267

Figure 105. Logic editor display screen.

6.5.2.1 Exit. This option terminates the process and returns you to the System Logic Editor screen. To invoke this option, type an $\langle E \rangle$ and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

6.5.2.2 Add. This option allows you to enter a new gate name or input into the selected system or subsystem. To invoke this option, enter an < A > while positioned in the gate name column or at the desired input row or column. When you invoke this option, Figure 106 will be displayed.

On this screen, you enter the following fields:

Gate Name	Enter the new gate name. If you enter a name that already exists, the message Duplicate gate names are not allowed is displayed. You are returned to the gate name prompt to re-enter a valid name.
Туре	Enter the gate type. Valid gate types include AND, OR, and TRAN.
Inputs	Enter any inputs (event names) for the new gate. After you enter an input, a window will be displayed asking you to identify the input as a



SURRY	Logic	Editor	D1	anni an an
Now names will	Add 1 be automatically be -Use «Enter» when do	nput Name added to the gate ne, and <esc> to a</esc>	or event (ist, bort	
Gate Kame	- Type	in,uts		

Figure 106. Add input to logic editor.

 $<\!G\!>$ (gate), $<\!E\!>$ (event), or $<\!A\!>$ (to abort the process). Press $<\!Enter\!>$ to terminate the process.

If you are adding input to an existing gate, position the cursor at the desired location and press <A>. A blank highlight line will appear. Enter the input name and press < Enter >. The new input will appear on the screen.

6.5.2.3 Modify. This option allows you to modify any of the data displayed on the screen. To invoke this option, highlight the data you wish to update using the arrow or tab keys, and type an < M >. When you invoke this option, Figure 107 will be displayed. Use the insert, delete, or arrow keys to modify the highlighted text. You may only modify the highlighted text. Press < Enter > to exit and save the changes, or press the < Esc > key to exit without saving the changes.

6.5.2.4 Delete. This option allows you to delete any gate or event displayed on the screen. To invoke this option, highlight the gate or ever ι you wish to delete using the arrow or tab keys and type a < D>. When you invoke this option, Figure 108 will be displayed. The following options are available:

Modify		Deletes the current event or gate name but first prompts you with the veto options.
Skip	¥	Skips the selected delete.
Continue to end	•	Deletes without asking if you wish to delete.
End		Terminates the delete process.

宅			ð

PART I		10910 1	ditor	01
		Modia	w	
Use in	sert, Dele	te, and arrow keys	to modify high-li	ghted text.
And the second second second second	Use <	Enter> when done,	and '«Esc> to abort	
mene Gate Name	come Type		inputs	procession and a second se
6KV1J	TRAN			
CPCB	TRAN	while many the monthly		
0.7 5.5 B	OR	HP1-CKV-FT-CV225	HP12	HP13
C10 W0137	1 KAN	F16	0000	
a.r.o.	UK	HP1-MNP-FE-PHID	NEL-MOD. ED. SEAMS	WEY MODE WE CLOB
		HP121	RET MUT TE INDIA	ULL MDP MAYCHIN
HP12	GWA	HP16	HP14	HP15
HP121	TRAN			
HP13	AND	HP17	HP18	
HPIA	TRAN			
HP15	OR	SIS-ACT-FA-SISB	HP117	HP1-DCF-FS-CH1BC
		HP1-MOV-PG-1269A	HPI-MOV-PG-12868	HP1-CKV-FT-CV267
and a second sec	- Andrews	HP19		
HP16	TRAN			

Figure 107. Modify logic editor data.

6.5.2.5 Locate. This option allows you to locate a specified gate or event. When you invoke this or 'ion, Figure 109 will appear at the top of the screen prompting you to enter the name you wish to locate. The input is case-sensitive, so you must enter the name exactly as it appears. Enter the gate or event name and press < Enter>. If found, the entered name or event will be highlighted and the message Name Found will be displayed at the bottom of the screen. If the event or name is not located, the message Not found will be displayed.

6.5.2.6 Next. This option highlights the next record in the display.

6.5.2.7 Previous. This option locates the previous record in the display.

6.5.2.8 Insert. This option allows you to insert a gate or event name. When you invoke this option, Figure 110 is displayed. This option uses two names: the find and the insert/replace name. The find name is the name to be matched. It may contain any number of "*" or "?" characters. These wild card characters have the same meaning as they do in DOS file names. The insert field cannot contain any wild card characters, and will be used exactly as typed.

6.5.2.9 Replace. This option allows you to replace a gate or event name. When you invoke this option, Figure 111 is displayed. This option uses two names: the find name and the replace name. The find name is the name to be matched. It may contain any number of wild card characters. The 'replace with' field cannot contain any wild card specifications.



and assisted to be a second second second	1	rogic r	a) tor	
Option			/ Continue to en	d / End
Seluct an	option	and press «Enter»	when done, or <f1< td=""><td>> for help.</td></f1<>	> for help.
mine Cath Kills and	a Yuna		anna 1999. 19 annai	
4KV1J	TRAN		subara a	the second states and states to
CPCB	TRAN			
01	OR.	HP1-CKV-F1-CV225	HP12	HP 13
E18	TRAN			
HP117	OR	E1B	CPC8	4KV17
		HP1-MDF-FS-CH1B HP121	相臣1-#6.臣。下居。〕最趋相民	HPI-MOP-MA-CHIB
HP12	AND	HP16	HP14	NP15
HP121	TRAN			
HP13	AND	HP17	HP18	and the shares
HP14	TRAN			
HP15	0/R	SIS-AC1-FA-SISB	HP117	HPI-COF-FS-CH18C
		API-MOV-PG-1269A	HP1-MOV-P0-12868	HPI-CKV-FT-CV267
1		HPIY		
MP10	IRAN			

Figure 108. Delete a gate or event from a system or subsystem.

the second se	- 644	and the second s	and the spectrum of the spectr	Construction of the second
		Locat	· ····	
Enter t	he Gate Use <i< td=""><td>or input name to ESC> to abort, and</td><td><pre>locate => </pre> <pre>sEnter> when done</pre></td><td></td></i<>	or input name to ESC> to abort, and	<pre>locate => </pre> <pre>sEnter> when done</pre>	
			Annual	
4KV1J	ITRAN	And the party of a life of the party of the	decomposition (1005/12) described	
CPCB	TRAN			
01	OR .	HP1-CKV-F1-CV225	HP12	HP13
618	TRAN			Contra 1
HH-117	OR	E18 HP1-MDP-FS-CH18 HP121	HPI-MDP-FR-186HR	HP1-MDP-MA-CH1B
HP12	AND	HP16	HP1w	HP15
HP121	TRAN			
HP13	AND	HP1?	HP18	
HP14	TRAN			
NP15	OR	SIS-ACT-FA-SISB HPI-MOV-PG-1269A HPI9	HP117 HP1-MOV-P6-12868	HP1-CCF-FS-CH1BC HP1-CKV-FT-CV267
HP16	TRAN			

Figure 109. Locate a specified gate or event.

SURRY		logic E	44101	D1
Find Use «Tab» to m	ove be	Find and 1 and 1 tween fields, <ent< th=""><th>nsert nsert er> when done, and</th><th>efts for help.</th></ent<>	nsert nsert er> when done, and	efts for help.
man Gate Name	- Type		inputs	
4KV1J	TRAN			
CPCB	TRAN			
01	OR	HP1-CKV-F1-CV225	HP12	HP13
£18	TRAN			
MP117	OR	ETB	CPCB	4KV1J
		HP1-MDP-FS-CH18 HP121	HP1-MDP+FR+1B6HR	HP1-MDP-MA-CH1B
HP12	AND	HP16	HP14	HP15
HP121	TRAN			
HP13	AND	HP17	HPIB	
HP14	TRAN	Contract of the second		
HP15	OR	SIS-ACT-FA-SISE	HP117	HP1-CCF-FS-CH192
		HP1-MOV-P0-1269A	HP1 - MOV - PG - 12868	HP1-CKV-FT-CV267
LINE &	10.44			

Figure 110. Find and insert a gate or event.







6.5.2.10 Search. This option allows you to perform a speed search. When you invoke this option, Figure 112 will be displayed. Enter the name to be located. As you start entering characters, the search will begin. If no record is found, the message **Input not found**!! will be displayed, and you will be returned to the Logic Editor Display.

SURF 1	mand	Fredra 1	ditor	01
En	ter the i	spred So rput name to be so Use «Enter» a	earch earch for => when done	
ume Gote Name	mention Type	-	manager Inputs and	
4KV1J	TRAN			
CPCB	TRAN		100.8.5	Las reserves a
U1 818	OR TURN	MPT CKY-FT-CY225	We IS	MP 13
HP117	DR	516	CPCR	Levit 1
		HP1-MDP-FS-CH18 HP121	HP1 - MDP - FR - 186HR	HP1 MDP MA-CH18
HP12	AND	HP16	HP16	HP15
HP121	TRAN			
HP15	AND	HP17	HP18	
MF14 UD15	TRAM	515.AFT. FA. 5155	46117	UNI - FEE - EE - FUILE
nr 1.2	SUR	HP1-MOV-PD-1269A	HP1-MOV-PG-12868	AP1-CKV-FT-CV267
HP16	TRAN			

Figure 112. Speed search option.

6.5.2.11 Options. This option allows you to set the veto options. When you invoke this option, Figure 113 will be displayed. The following options may be set:

Veto	Veto tells each operation to ask again if you wish to modify the data. By default this is turned off.
Delete Veto	Functions the same as Veto. However, it applies only to the Delete command. By default this is turned on.
Help	Activates the on-line general help messages. When this is turned off, the help messages will not appear

6.5.2.12 Undo Delete. This option allows you to retrieve your last delete. If you have accidentally deleted an event or gate, you may invoke this option to retrieve it. Only your last deletion may be retrieved.


SURK 1	1	robic t	altor	E.1
		Opt ions		
Veto	<on> =></on>	Dolete Veto «	ON> => Help <on></on>	8.2
he Use (Tabe to	move be	stween fields, «En	ters when done, an	d stir for help
men Gate Name ma	····· Type		monome light to come	provide the second s
4EV1J	TRAN			
CPCB	TRAN			
0.1	OR	HP1-CKV-F1-CV225	HP12	HP13
618	TRAN			
KF117	CIR	£18	CPCR .	6KV13
		HP1-MDP-FS-CH1B	HPT-MDP-FR-186HR	HP1-MDP-MA-CH1R
		HP121	The solution of the second	
HP12	AND	WP16	HPTL	HP15
HP121	TRAN	1.44	acta .	1.1.1.
HPIN	AND	HP17	HP18	
HP14	TRAN	ac est		1
HP15	CIP	SIS-ACT-FA-SISB	HP117	HP1-CCF-FR-CHARC
	1.00	HP1-MOV-PC-12604	HP1-MOV-PD-12868	HP1-CEV-F1-CV267
		4010	The second issue	THE ARE TO BEEN
HD1A	TRAN	111-1-2		
PSP X 40	15.68		1	

Figure 113. Set veto options screen.

6.5.2.13 F1 Help. This option displays the cursor movements that allow you to easily move through the Logic Editor Display (see Figure 114). To move the cursor about the editing window, use the keyboard cursor pad. The arrow keys move one field in each respective direction. The <Home> key places the cursor in the upper right corner of the window. The <PgUp> key moves the window up one page (12 lines). The <PgDn> key moves the window down one page (12 lines). The <Ctrl-PgUp> and <Ctrl-PgDn> keys move to the top and bottom of the file, respectively. Pressing the <F1> key toggles between two cut set editor screens. One screen shows the editing options and other shows the cursor movements.

NOTE:

When you press the $\langle Esc \rangle$ key to terminate the Logic Display, Figure 115 is displayed.

6.5.3 Sub-System

Toggies the System Logic editor display to list systems or subsystems on the screen. To invoke this option, enter an $\langle S \rangle$ in the option field, and press $\langle Enter \rangle$. To toggle back to the previous display, simply enter an $\langle S \rangle$ in the option field and press $\langle Enter \rangle$.

Contraction of the second seco		x + y + c = c	97197	Real Property and Property and Property of State
<f>Cursor <end>End of <ctrl-pgu< th=""><th>Up * Page p>Top of</th><th>i>Cursor Down < Home>Top of Page f File <ctrl-pgd< th=""><th><pre>coursor Left c- cPgUp>Page Up r>End of File cf</pre></th><th>*>Cursor Right «PgDn>Page Down 1> Help</th></ctrl-pgd<></th></ctrl-pgu<></end></f>	Up * Page p>Top of	i>Cursor Down < Home>Top of Page f File <ctrl-pgd< th=""><th><pre>coursor Left c- cPgUp>Page Up r>End of File cf</pre></th><th>*>Cursor Right «PgDn>Page Down 1> Help</th></ctrl-pgd<>	<pre>coursor Left c- cPgUp>Page Up r>End of File cf</pre>	*>Cursor Right «PgDn>Page Down 1> Help
perm Gate Name	- Type		quantum inguite means	
4KV1J	TRAN			1. J. C. C. C. C. L. J.
CPCB	TRAN	and once of oursel	444.5	10.17
21.	UK TEAN	WEL-CKA-EL-CASSO	NP 16	RP10
HP117	OR	8.16	ICPCR .	GRV14
The state	- Con	HFI-MOP-FS-CHIE	HP1-MDP+FR-186HR	HP1-MDP-MA-CH1B
		nP121		
HP12	AND	HP16	HP14	HP15
HP121	TRAN		and the second second	NASCE 12 10 - 14 - 1
HP13	AND	MP17	HP18	
MP16 MD16	TRAN	STE. ATT. 54. 2165	40147	NET. COT. 12. CUIDE
40.75	UR.	HP1-MOV-PC-12AQA	HP1-MOV-PG-12868	WP1-CEV-F1-CV267
		HPID	in this is come	THE STREET STREET
4016	TRAN			the first state in the last

Figure 114. Cursor movements help display.

C - Check data, if ok then save and exit, if not ok then continue editing. R - Return to editing after checking data. E- Exit WITHOUT saving changes. (C, R, or E) [C]



7.0 CREATE EVENT TREES

This option allows you to graphically build and edit event trees and output them to several hard copy hardware devices. To select this option, highlight the "CREATE Event Trees" option on the IRRAS main menu or enter a < C> and press < Enter. When you invoke this option, Figure 116 will be displayed. As shown, the following options are unsplayed:

Exit	Returns you to the IRRAS main menu.
Create Event Trees	Allows you to build and edit event tree diagrams.
Link Event Trees	Allows you to define linkage rules and generate sequence logic.
Plot Trees	Allows you to plot graphics files or generate HPGL formatted files.
Graphics Load	Allows you to load event tree graphic files (i.e., .ETG files).
eXtract Event Trees	Allows you to extract event trees, clear extracted event trees, and display extracted event trees.
Define Plotter Pens	Allows you to assign colors to your plotter pens.

Each of these options is discussed in the following paragraphs.

1

SURRY	1 R R A S Event Tree Graphics System
	Exit
	Create Event Trees Link Event Trees
	Plot Trees Graphics Load
	extract Event Trees Define Plotter Pens
	Option E

Figure 116. Event tree graphics system menu.

7.1 Exit

This option terminates the event tree option and returns you to the IRRAS main menu. To invoke this option, highlight the Exit option and press < Enter >.

7.2 Create Event Trees

This option allows you to construct or edit an event tree diagram. You may start building the diagram from scratch or from an existing file to generate or modify logic. When this option is selected, Figure 117 is displayed. The editing commands are shown in the left column, while the rest of the screen is the drawing surface. The commands shown in all upper-case letters (excluding EXIT) have additional pop-up menus associated with them. The cursor '* used to position pop-up menus and select menu options.

To invoke any of the editing commands (using a mouse), position the cursor over the desired editing command on the active menu. The active menu is the last menu you pulled up or moved. When the editing command box is highlighted (a white line outlines the box), press the left mouse button. The command is now invoked. Each editing command is described in the following paragraphs.

7.2.1 EXIT

This option terminates the editing session and returns you to the previous screen (Figure 116). To invoke this option, position the cursor over the EXIT box and press the left mouse button or <Enter >.

7.2.2 Move (*--->)

The move command, which is represented by $\ast \rightarrow \ast$, allows you to position the editing command menu anywhere on the screen. When you invoke this command, a white outline surrounds the entire editing column. Drag the cursor to position the outline at the desired location and press the left mouse button or < Enter >. The menu will be displayed at the new location.

7.2.3 Show

This command clears the screen and re-displays the currently defined diagram.

7.2.4 EDIT

This command allows you to make changes to the event tree. When you invoke this option, an additional pop-up menu is displayed (Figure 118). As shown, six options are available: EDIT, Add, Del, Copy, Pass, and Page.



Figure 117. Create event tree editing commands.



Figure 118. Event tree EDIT pop-up menu.

l

0

7.2.4.1 \leftarrow EDIT \leftarrow . This option allows you to move the pop-up menu to a new location on the screen. To invoke this option, position the cursor on the \leftarrow EDIT \rightarrow box and press the left mouse button. An outline will appear. Move the outline to the new location on the screen and press the left mouse button. The menu will be moved to the new location.

7.2.4.2 Add. This option allows you to add a branch to the event tree. When you invoke this option, you will be prompted to Pick the intersecting point of the new branch. The cursor will change from a filled cursor arrow to an 'empty' cursor arrow. Position the cursor over the place where the new branch is to be added and press the left mouse button. A small box will appear. Next, you will be prompted to Pick the vertical position of the new branch. The vertical position determines whether the new branch will be added above or below the existing branch. Position the cursor to the desired vertical position and press the left mouse button. The diagram is redrawn with the new branch added. The prompt to Pick the intersecting point of the new branch will return. At this point you can select another location to add a branch or press the right mouse button to terminate the add process.

7.2.4.3 Del. This option allows you to delete branches from the tree. When you invoke this option, you will be prompted to Pick the branch to be deleted. Position the cursor on the branch to be deleted and press the left mouse button. The selected branch will be highlighted and the message Delete highlighted branch? Left = YES, Right = NO will be displayed. If the highlighted branch should be deleted, press the left mouse button; if not, press the right mouse button. If you responded yes, the diagram will be redrawn to reflect the deletion. In either case, the prompt Pick the branch to be deleted will return. At this point you may select another branch to delete, or press the right mouse button to terminate the deletion process.

7.2.4.4 Copy. This option allows you to copy existing branches of the tree to new locations. When you invoke this option, you will be prompted to Pick beginning of the branch to copy. Position the cursor at the start of the branch to be copied and press the left mouse button. Next, you will be prompted to Pick copy location. There are three ways to use the copy command. In each case you select the existing branch to be copied. Then you may either place the cursor at the start of a branch and replace the existing branch, or place the cursor at the start of a branch and add the copied logic to the branch. Depending on the copying method you use, you will be prompted with **Replace = left button**. If this branch is an addition, press the left mouse button. If this branch is an addition, you will be prompted to Pick vertical location of the start of new sub-tree. Pick the new location and press the left mouse button.

7.2.4.5 Pass. This option allows you to change a branch to a pass. When you invoke this option you will be prompted to Pick the branch that is to be made a pass. Position the cursor on the branch to be converted and press the left mouse button. The entire branch is highlighted and you will be prompted with Make highlight branch a pass? Left = YES, Right = NO. If this is the branch you wish to change into a pass press the left mouse button; otherwise, press the right mouse button. If you respond yes, the branch is converted to a pass. If you respond no, the process is terminated. In either case, the prompt Pick the branch that is to be made a pass will return. At this point you may select another branch, or press the right mouse button to terminate this procedure.

7.2.4.6 Page. This option allows you to separate a large drawing into separate pages. When you invoke this option, you will be prompted to Pick beginning of the branch to page. Position the

7.5



cursor at the branch where the new page is to start and press the left mouse button. The message Create Transfer Here? [Y] will appear. If this is the point where the page should appear, press Y; otherwise enter an "N."

When you highlight a section of the drawing and enter a "Y" to transfer, the block will be moved to the next page and a transfer will be created automatically by the system. If you respond "N" to the transfer the diagram will not be broken up and a transfer will not be created. In either case, the prompt **Pick beginning of the next branch to page** will appear. At this point, you may select another location to page or press the right mouse button to terminate the paging process.

7.2.5 TOPS

This option allows you to make changes to the top events. When you invoke this option, an additional pop-up menu will be displayed (Figure 119). As shown, the following options are available: TOPS, Add, Del, Edit, Name?, Width. Size, FONT, and DESC.

7.2.5.1 ••TOPS•. This option allows you to move the pop up menu to a new location on the screen. To invoke this option, position the cursor on the **•.TOPS•** box and press the left mouse button. An outline will appear. Move the outline to the new location and press the left mouse button. The menu will be moved to the new location.

7.2.5.2 Add. This option allows you to add a new event. For example, an event tree might have the existing events 1, 2, 3, and 4. A new event is needed between events 2 and 3. You select this event and will be prompted to Pick the event that will follow the new event. The cursor will change from an arrow to a cross hair. Position the cross hair at the desired event (in this example, we would pick event 3) and press the left mouse button. The prompt Enter new event name: will be displayed. Enter the event name (up to 16 characters are allowed) and press < Enter >. The diagram will be redrawn and displayed showing the newly added event and the event list will be updated.

7.2.5.3 Del. This option allows you to delete an event. To delete an event all branch points that logically fall below the event must be converted to passes. When you invoke this option you will be prompted to Pick the event that is to be deleted. The cursor will change from an arrow to a cross hair. Position the cursor on the event to be deleted and press the left mouse button. If the branch points have not been converted to passes, the message The event picked is currently being used in the tree will be displayed. If you want to delete the event you must $g \circ$ back and convert the branch points using the PASS option under the EDIT command. When the deletion is successful, the tree will be updated and redisplayed.

7.2.5.4 Edit. This option allows you to change an event name. When you invoke this option you will be prompted to Pick the event name to be edited. The cursor changes from an arrow to a cross hair. Position the cursor on the event in the diagram or the event name displayed at the top to be renamed and press the left mouse button. The prompt Enter new event name for xxx: will be displayed. Enter the new name (up to 16 characters are allowed) and press < Enter >. The event name will be changed at the top of the diagram.







Figure 119. TOPS pop-up menu.



7.2.5.5 Name?. To conserve space on the diagram, a limited number (default is 5) of characters of the event name are displayed on the diagram. Use this option to display the complete event name (all 16 characters). When you invoke this option you will be prompted to Pick an event name. The cursor will change from an arrow to a cross hair. Position the cursor on the desired event name at the top of the diagram and press the left mouse button. The entire event name will be displayed on the prompt line at the bottom of the screen. The cross hair will remain so you may select additional event names to display or press the right mouse button to terminate the process.

7.2.5.6 Width. This option dlows you to change the displayed width of the event names. When you invoke this option, you will be prompted to Enter top width size < old size >. Enter the new top width size and press < Enter >. You may enter a value greater than 1 (default is 5). If you press < Enter > without entering a value, the width does not change. If the width is changed, the tree will be redisplayed with the new width.

7.2.5.7 Size. This option allows you to change the text size of the event names. When you invoke this option, you will be prompted to Enter new top name text size < old value >. Enter the new text size and press < Enter >. You may enter a value of 0.01 to 9.00. If you press < Enter > without entering a new value, the text size does not change. If the text size is changed, the tree will be redisplayed with the new top text size.

7.2.5.8 FONT. This option allows you to select the font type for the event names and headers. When you select this option, an additional pop-up menu will be displayed (see Figure 120). Select the desired font type by positioning the cursor over the font and pressing the left mouse button. You must select a font (or cancel) in order to continue. If the font type is changed, the tree will be redisplayed with the new event name font type.

7.2.5.9 DESC. This option allows you to edit event description text and set various attributes of the descriptions including text size, font type, and number of lines of description to display. When you invoke this option, an additional pop-up menu will be displayed (see Figure 121).

7.2.5.9.1 \rightarrow DESC \rightarrow — This option allows you to move the pop-up menu to a new location on the screen. To invoke this option, position the cursor on the \leftarrow DESC- \rightarrow box and press the left mouse button. An outline will appear. Move the outline to the new location and press the left mouse button. The menu will be moved to the new location.

7.2.5.9.2 Edit—This option allows you to edit or add any event name descriptions contained in the displayed diagram. When you select this option, you will be prompted to Pick the event description to be edited. To select the event description to edit, move the cursor over the event or event description and press the left mouse button. A window will appear displaying the selected description and the prompt Edit Description - Press < Esc> when done will appear. If no description exists, enter a description. To modify the description, simply type over the existing description. Use the < Ins> and < Del> keys to add and delete characters as necessary. In addition, you may use the Backspace and End keys. The Backspace key deletes the character to the left of the cursor and moves the rest of the characters to the left one space. The End key positions the cursor at the end of the current line. When complete, press < Esc>. You will be prompted to Pick the event description to be edited. At this point you may select another description to edit or press the right mouse button to terminate the process.





Figure 120. Font type menu for event names and headers.





Figure 121. DESC pop-up menu.

7.2.5.9.3 Size—This option allows you to specify the height of the event descriptions in your diagram. The prompt Enter new top description text size: <old size > will be displayed. At this prompt, enter the desired sext size and press < Enter > . You may enter a value of 0.01 to 9.0. The event descriptions will be displayed in the new size.

7.2.5.9.4 Line—This option allows you to specify the number of lines in event descriptions. When you invoke this option, you will be prompted to Enter new top description line count: < old count > . At this prompt, enter the desired line count and press < Enter > . You may enter a value of 0 or greater. The event descriptions will be displayed with the new line count.

7.2.5.9.5 FONT—This option allows you to select the font type for the event descriptions. When you select this option, an additional pop-up menu will be displayed (see Figure 122). Select the desired font type by positioning the cursor over the font and pressing the left mouse button. You must select a font (or cancel) to continue. If the font type is changed, the event descriptions will be redisplayed.

7.2.6 ENDS



 $i.2.6.1 \leftarrow ENDS \rightarrow$. This option allows you to move the pop-up menu to a new location on the screen. To invoke this option, position the cursor on the $\leftarrow ENDS \rightarrow$ box and press the left mouse button. An outline will appear. Move the outline to the new location and press the left mouse button. The menu will be moved to the new location.

7.2.6.2 End States. This option allows you to display information about the end state. When you invoke this option you will be prompted to **Pick the leaf corresponding to the desired endstate information.** The cursor is placed on the tree at the end of the sequences. The cursor will only move in the vertical section allowing you to pick only leaf branches. Position the cursor at the desired end state and press the left mouse button. A window will appear displaying the sequence names, end state names, frequencies, and the data contained in the two user-defined fields for the selected end state. Next, you will be prompted to **Pick the box to be edited**. Position the cross hair over the box to be edited and press the left mouse button. If you choose to edit the sequence names, end state names, or frequency values you will be prompted to **Enter Text**: Enter the desired text and press <Enter >. If a name already exists, it will be overwritten with the new name you entered.









Figure 122. Font selection for event descriptions.



0

-1

7-13

January 1992

C. S. C. F. R.

The two remaining fields are toggle switches. The toggles allow you to specify whether or not to display the item.

IRRAS provides four columns of information for the user to add additional information about each sequence in an event tree. IRRAS requires that the first column be the sequence name, the second must be the end state, the third is used to store the sequence frequency, and the fourth is user defined. The column names and positions may be changed by the user, but the information stored in each column must be as explained above.

To terminate this process, press the right mouse button. The window will disappear and you will be returned to the **Pick the leaf corresponding to the desired endstate information** prompt. At this point, you may select another end state or press the right mouse button to terminate the process.

7.2.6.3 Edit Header. This option allows you to edit header information. When you invoke this option, a window appears. This window contains sequence names, end state names, frequencies, and two user defined fields contained in the header. Next, you will be prompted to **Pick the box to be edited**. Position the cross hair over the box to be edited and press the left mouse button. If you selected to edit sequence names, end state names, or frequency values you will be prompted to **Enter text:**. Enter the desired text and press < Enter >. If the box contained data, it will be replaced by the text you just entered. To terminate this process, press the right mouse button.

7.2.6.4 Place Header. This option allows you to move a header to another location on the screen. When you invoke this option you will be prompted to Pick the header that is to be relocated. Position the cross hair on the header contained in the listing on the right side of the screen or on the title to the left of the status/tops line at the top of the screen to be moved and press the left mouse button. You will be prompted to Pick the placement point for the header. Position the cursor on the desired location and press the left mouse button. The headers will be moved to the new location. The prompt Pick the header that is to be relocated will return. At this point, you may select another header to relocate or press the right mouse button to terminate the process.

7.2.6.5 Edit Height. This option allows you to specify the height or distance between branches (nodes) of the diagram. When you invoke this option, you will be prompted to Enter new node height < old height > . At this prompt, enter the desired node height and press < Enter >. You may enter a value greater than 0.0. The tree will be redisplayed with the new node height.

7.2.6.6 Edit Size. This option allows you to specify the height of the end state text in your diagram. When you invoke this option, you will be prompted to Enter new end state text size: < old value >. At this prompt, enter the desired text size and press < Enter >. You may enter a value of 0.01 to 9.0. The tree will be displayed with the new end state text size.

7.2.6.7 FONT. This option allows you to select the font type for the end state text. When you select this option, an additional pop-up menu will be displayed (see Figure 124). Select the desired font type by positioning the cursor over the font and pressing the left mouse button. You must select a font (or cancel) in order to continue.





7.2.7 TEXT

This option allows you to add text to the diagram at any location, in any size and color. When you invoke this command, an additional pop-up menu is displayed (Figure 125). As shown, seven options are available: TEXT, FONT, Write, Move, Copy, Erase, and EDIT.

7.2.7.1 \leftarrow TEXT \rightarrow . This option allows you to move the pop-up menu to a new location on the screen. To invoke this option, position the cursor on the \leftarrow TEXT \rightarrow box and press the left mouse button. An outline will appear. Move the outline to the new location and press the left mouse button. The menu will be moved to the new location.

7.2.7.2 FONT. This option allows you to select the font type for the text. When you select this option an additional pop-up menu will be displayed (Figure 126). Select the desired font by positioning the cursor over the font and pressing the left mouse button. You must select a font (or cancel) in order to continue. All new text will be displayed in the selected font. Remember, you are changing the default font type using this option. No existing text will be changed to this new font type. To change existing text, select the TEXT option, the EDIT suboption, and invoke the FONT command. You may then mark existing text to be changed to the default font.

7.2.7.3 Write. This option allows you to write text at any location on the screen. When you invoke this option you will be prompted to Pick text placement location. Position the cursor at the desired location and press the left mouse button. A window will appear in the top left corner of the screen. Enter the desired text. When complete, press < Esc > . If the show text option is turned on (See VIEW option), the newly added text will be displayed on the screen. Next, you will be prompted to Pick next text placement location: At this point you may select another location to write text or press the right mouse button to terminate the process.

7.2.7.4 Move. This option allows you to move the selected text to a new location on the screen. When you invoke this option you will be prompted to Pick region to be moved - press CANCEL to quit. You select the region by marking the opposite corners of the text to be moved. Position the cross hair at the text you want to move and press the left mouse button. A small dot appears. Drag the cursor across the desired text until the box totally surrounds the text. Press the left mouse button again. The message Pick reference point - press CANCEL to quit will be displayed. The reference point is used to give you some indication of the position of the text to be moved to location the cross hair at the location where you want the selected text to be moved to. Next, you will be prompted to Pick placement point - press CANCEL to respect. Use the mouse to move the box to the exact position where you want the box to appear. When you are satisfied with the new position press the left mouse button. The selected text will be moved to the new location. The prompt Pick placement point - press CANCEL to the new location. The prompt Pick placement point - press CANCEL to the new location. The prompt Pick placement point - press CANCEL to respect to the new location. The prompt Pick placement point - press CANCEL to respect to the new location. The prompt Pick placement point - press CANCEL to respect to the new location. The prompt Pick placement point - press CANCEL to respect to the new location. The prompt Pick placement point - press CANCEL to respect to the new location. The prompt Pick placement point - press CANCEL to move the location to move the text, or press the right mouse button or < Esc > key to terminate the process.







January 1992

· · (* .)





Figure 126. FONT selection screen.



7.2.7.5 Copy. This option allows you to copy the selected text to a new location on the screen. When you invoke this option you will be prompted to Pick region to be copied - press CANCEL to quit. You select the region by marking the opposite corners of the text to be copied. Position the cross hair at the text you want to copy and press the left mouse button. A sit, that appears. Drag the cursor across the desired text until the box totally surrounds the text. Press the teft mouse button again. The message Pick reference point - press CANCEL to quit will be displayed. The reference point is used to give you some indication of the position of the object being copied relative to the box. Position the cross hair at the location where you want the selected region to be copied to. Next, you will be prompted to Pick placement point - press CANCEL to reselect. Use the mouse to move the box to the exact position where you want the text to appear. When you are satisfied with the new position press the left mouse button. The selected text will be moved to the new location. The prompt Pick placement point - press CANCEL to reselect with the new position press the left mouse button. The selected text will be displayed again. At this point you may select another location to copy the text to, or press the right mouse button or < Esc > key to terminate the process.

7.2.7.6 Erase. This option allows you to delete selected text. When you invoke this option you will be prompted to Fick region to be deleted. You select the region by marking the opposite corners of the text to be deleted. Position the cross hair at the text you want to move and press the left mouse button. A small dot appears. Drag the cursor across the desired text until the box totally surrounds the text. Press the left mouse button again. The message Delete this region? left = YES, Right = NO will be displayed. If this is the text to be deleted, press the left mouse button; otherwise press the right button. Next, you will be prompted to Pick next region to be deleted. At this point you may select more text to delete or press the right mouse button or < Esc > key to terminate the process.

NOTE:

If the text is not deleted, check the size of the outline box used to mark the region. It must be large enough to encompass all the text desired; otherwise, no text will be deleted.

7.2.7.7 EDIT. This option allows you to edit text and set various attributes of the text including color, size, font, and justification. When you invoke this option, an additional pop-up menu will be displayed (Figure 127).

7.2.7.7.1 \leftarrow EDIT TXT \rightarrow — This option allows you to move the pop-up menu to a new location on the screen. To invoke this option, position the cursor on the \leftarrow EDIT TXT \rightarrow box and press the left mouse atton. An outline will appear. Move the outline to the new location and press the left mouse button. The menu will be moved to the new location.

7.2.7.7.2 Text—This option allows you to edit any of the text contained in the displayed diagram. When you select this option you will be prompted to Box text to be edited. You select the text to be edited by boxing the opposite corners of the text region. Position the cross hair at the beginning of the text you wish to modify and press the left mouse button. A small dot will appear. Drag the cursor over the text until the outline surrounds the text you wish to modify. When the text is completely surrounded, press the left mouse button. A window will appear displaying the selected text. To modify the text, simply type over the existing text. Use the < Ins > and < Del > keys to add and delete characters as necessary. In addition, you may use the Backspace and End keys. The Backspace key deletes the character to the left of the cursor and moves the rest of the characters to the left one space. The End key positions the cursor at the end of the current line. When complete, press < Esc >. You will be prompted to Pick next text to be edited. At this point you may select additional text to edit





Figure 127. EDIT pop up menu.



or press the right mouse button to terminate the process.

7.2.7.7.3 Color—This option allows you to change the color of selected text in your event tree diagram. When you invoke this option you will be prompted to Box the text to be changed. Position the cursor at the beginning of the text you wish to change and press the left moute button. A small dot will appear. Drag the cursor over the text until the outline box surrounds all the desired text. Press the left mouse button. Next, you will be prompted to Pick the new text color from the color bar. To select a color, position the cursor over the desired color and press the left mouse button. The text will immediately change to the new color. The prompt Box next region to be changed will be displayed. At this point you may select more text to change or press the right mouse button to terminate the process.

7.2.7.7.4 Size—This option allows you to specify the height of selected text in your diagram. When you invoke this option, you will be prompted to Box the text to be changed. Position the cross hair at the beginning of the text you wish to change and press the left mouse button. A small dot will appear. Drag the cursor over the text until the outline surrounds all the desired text. When all the desired text is surrounded, press the left mouse button. Next, the prompt Enter the new text size > will be displayed. At this prompt enter the desired text size and press < Enter >. You may enter a value of .001 to 9.0. The selected text will be displayed in the new size. Next, you will be prompted to Box next region to be changed. At this point you may select more text to be changed or press the right mouse buttor the text to be changed or press the right mouse buttor this process.

Just—This option allows you to justify selected portions of the text in your diagram. Just a addition is othere the text will be placed offset from the placement point. When you invoke this option yo = 0 by p ompted to **Box the text to be changed**. Position the cursor at the beginning of the text to be changed ad press the left mouse button. A small dot will appear. Drag the cursor over the text until the outline completely surrounds it. When the desired text is completely surrounded press the left mouse out. You will be prompted to **Enter text justification** -('L'=Left, 'R'=Right, 'C'=Center) >. Enter the desired justif. "ion and press < Enter >. The prompt **Box next region to be changed** will be displayed. At this point you may select additional text to modify or press the right mouse button to terminate this process. See Section 6.2.12 for "compete discussion on setting the justification.

7.2.7.7.6 FC TF—This option allows you to select the font type for selected text. When you select this option an a solution and pop-up menu will be displayed (Figure 128). Select the desired font type by positioning the curve center of the font and pressing the left mouse button. You must select a font (or cancel) in order to next, you will be prompted to Box the text to be changed. Position the cursor over the text observation and press the left mouse button. A small dot will appear. Drag the cursor over the text observation. The selected text will be displayed in the new font type. Next, you will be prompted to Box next region to be changed. At this point you may select more text or press the right mouse button to terminate the process.





Figure 128. Text font selection menu.

January 1992

7-22

7.2.8 VIEW

This option allows you to change the position and size of the displayed diagram. You may move the drawing up, down, right, left, zoom in, zoom out, or restore the drawing to its original size and/or position. You may also toggle to display/not display text and turn on and off the grid. When you select this option, Figure 129 will be displayed. The VIEW submenu consists of the following options:

- Page 1: Invoking this option allows you to shift the diagram up one page (previous page). To invoke this option, position the cursor in the Page 1 box and press the left mouse button < Enter >. The diagram's previous page will be displayed. If no previous page exist, only the header information will appear.
- Page 4: Invoking this option allows you to shift the diagram down one page (next page). To invoke this option, position the cursor in the Page 4 box and press the left mouse button or < Enter >. The diagram's next page will be displayed. If no next page exist, only the header information will appear.
- Page →: Invoking this option allows you to shift the diagram to the right one page (one screen). To invoke this option, position the cursor in the Page → box and press <Enter>. The diagram will shift to the right one screen. If no figure exists on this new screen, only the header information will appear.
- Page -: Invoking this option allows you to shift the diagram to the left one page (one screen). To invoke this option, position the cursor in the Page box and press < Enter >. The diagram will shift to the left one screen. If no figure exists on this new screen, only the header information will appear.
- Scroll: Invoking this option allows you to move the diagram to another location on the screen, To invoke this option, position the cursor in the Scroll box and press the left mouse button. A white outline box appears, with a cross hair placed in the center of the outline. Position the cursor at the desired location and press the left mouse button. The cross hair serves as a reference point for placing the drawing. The reference point (+) is used to give you some indication of the position of the object being moved relative to the screen.

Zoom in: Invoking this option allows you to fill the screen with a small portion of the original display (magnifies the selected portion of the screen). To invoke this option, position the cursor in the Zin (zoom in) box and press the left mouse button or < Enter >. The message Pick first corner will be displayed. Move the cursor at the start of the diagram to be enlarged and press the left mouse button. A small dot appears. Next, you will be prompted to Pick next corner. Drag the cursor across the desired area until it is completely surrounded by the outline box. Press the left mouse button. The portion of the original display enclosed by the box will now fill the entire screen. The display can be restored to its original size by invoking the Zres (zoom restore) option.





Figure 129. VIEW pop-up menu.

- Zoom out: Invoking this option allows you to shrink the screen by approximately 50%. To invoke this option, position the cursor in the Zout (zoom out) box and press the left mouse button or < Enter >. The entire display will be reduced, while the drawing space is increased. To restore the display to its original size, invoke the Zres option.
- Zoom Restore: This option restores any display created by zoom in or zoom out to the
 original display size or to the last saved file. To invoke this option, position the cursor
 in the Zres box and press the left mouse button or < Enter >.
- Sh Text: This option allows you to toggle the display the text setting from Sh Text to No Text. Sh Text displays all defined text. No Text does not display the text.
- Grid: This option displays a grid behind your diagram to allow you to line up symbols and text. This is a toggle switch. To turn the grid on, position the cursor in the Grid option box and press the left mouse button or < Enter >. To turn the grid off, repeat the same steps.

7.2.9 FILE

This option allows you to perform various file manipulation functions including loading, saving, listing and creating event tree files. When you invoke this option, an additional pop-up menu will be displayed (Figure 130). Each of these pop-up commands is discussed in the following paragraphs.

7.2.9.1 ←FILE→. This option allows you to move the pop-up menu to a new location on the screen. To invoke this option, position the cursor on the ←FILE→ box and press the left mouse button. An outline will appear. Move the outline to the new location and press the left mouse button. The menu will be moved to the new location.

7.2.9.2 Load. This is one of the options that can be used to load a file. When you invoke this option, you will be prompted to Enter file name >. At this prompt, enter the file name and press $\langle \text{Enter} \rangle$. You must know the name of the file before initiating this option. The LIST command also allows you to load a file, but in addition it will display a list of available files.

7.2.9.3 Save. This option allows you to save the current file. When you invoke this command you will be prompted to Enter file name or CR for file current file name. At this point you may enter a new file name or choose the default file name provided by pressing < Enter >. The file is then written to disk. If you enter a new file name, do not provide an extension The extension " ETG" is provided by IRRAS.

7.2.9.4 New. This option allows you to create a new event tree (file). When this command is invoked you will be prompted to Enter Initiating Event or Top Name. Enter the event or name and press < Enter >. You will then be asked is this an Initiating Event. Respond with a Y or N. You will then be prompted to Enter Event Name #1. Enter an event name and press < Enter >. This prompt will repeat (with the event number increasing each time) until you press < Enter > without entering a name. The newly created event tree will then be displayed on the screen. The default form of an event tree will contain a success/fail branch at the first event and a don't care or pass extending through the ending event.





Figure 130. FILE pop up menu.

7.2.9.5 LIST. This is one of the two options used to load an event tree diagram. When you invoke this option, an additional pop-up menu will be displayed listing the files available. You will be prompted to Pick the file to load. To select a file to be loaded, position the cursor on the desired file name and press the left mouse button. The selected diagram will be displayed on the screen. You may now proceed with other editing functions.

7.2.9.6 Epson. This option formats the current diagram for an Epson printer and sends it to the attached local Epson printer.

7.2.9.7 Laser. This option formats the current diagram for a laser printer and sends it to the local laser printer to be printed.

7.2.9.8 Family. This option allows you to view and change the name of the family corresponding to the current diagram. The default family name is provided. When you invoke this option you will be prompted to **Type in the New Name or < Return > for family name >**. Enter a new file name if desired, or press < Enter > to accept the default family.

7.2.9.9 Event. This option allows you to view and change the event tree file name. The current file name for the event tree is provided. When you invoke this option, you will be prompted to **Type** in the New Name or < Return > for *current file name* >. Enter a new file name and press < Enter > or leave blank and press < Enter > to keep the current file name.

7.2.9.10 File?. This option allows you to view the current file name. When this option is invoked the current file name is displayed at the bottom left corner of the screen. If no file name is assigned to the current diagram, the default file name of **NONAME.ETG** will be displayed.

7.2.10 TRAN

This option allows you to add, delete, and modify transfer file names as well as transfer to and from files that are added as transfers. When you invoke this option, an additional pop-up menu will be displayed (Figure 131). Each pop-up option is discussed in the following paragraphs.

7.2.10.1 Add. The add option allows you to place a transfer file name in the diagram. This transfer file name can then be selected and you can transfer back and forth between the two event tree diagrams or any number of diagrams (files) that are included in the transfer list. When you invoke this option you will be prompted to **Pick the leaf where the transfer is to be placed**. The cursor will move vertically along the end of the event branches. Position the cursor on the desired transfer point and press the left mouse button. The prompt **Enter transfer file name** will be displayed. Enter the appropriate file name and press < Enter > . A "T" is placed after the selected sequence number and the entered file name is placed in the corresponding end state slot. The prompt **Pick the leaf where the transfer is to be placed** will return. At this point you may select another transfer point or press the right mouse button to terminate the process.





Figure 131. TRAN pop-up menu.

7.2.10.2 Delete. This option allows you to delete transfer file names from the diagram. When you invoke this option you will be prompted to **Pick the leaf corresponding to the transfer to be deleted**. Position the cursor on the leaf branch where a transfer has been previously added and press the left mouse button. The "T" will be deleted and the corresponding sequence name will be removed from the transfer list. You will be returned to the previous prompt. At this point you may select another transfer point to delete or press the right mouse button to terminate the process.

7.2.10.3 Modify. This option allows you to change the transfer file name. When you invoke this option you will be prompted to Pick the leaf corresponding to the transfer to be edited. Position the cursor on the transfer leaf and press the left mouse button. The prompt Type in the New Name or <Return > for current transfer file name >. Enter the new transfer file name or press <Enter > to retain the existing file name.

7.2.10.4 Tran→. This option allows you to work with multiple files simultaneously. When you invoke this option you will be prompted to Pick the leaf corresponding to the desired transfer file name. The selected file is then loaded into the system. This option will allow you to transfer back and forth to various files.

7.2.10.5 - Tran. After a transfer to another file has been made, you use this option to return to the previous file. When you invoke this option you will be prompted to Pick the file to transfer to. A pop-up window will appear showing you a list of available transfer files. Select a file by positioning the cursor on the desired file and pressing left mouse button. If no files exist, the message No transfer files will be displayed.

7.2.11 Text

This option allows you to set a default color for the text in your diagram. To invoke this option, position the cursor over the Text box and press the left mouse button. You will be prompted to **Pick a new color from the color bar**. Position the cursor on the desired color and press the left mouse button. (**NOTE:** The $\dagger \downarrow$ is active. Use this option to display additional color selections.) The Text box will change to the selected color. All text created in your diagram from now on will be displayed in this new color. Any existing text in your diagram will retain the old color. To change the color of the existing text, you must invoke the TEXT option and select the EDIT suboption and invoke the Color command (see Section 7.2.7).

7.2.12 cntr/left/rght

This option allows you to set the justification for your diagram. Justification is where the text will be placed offset from the placement point. This works as a toggle switch. To change justification, position the cursor over the cntr (left or rght) box and press the left mouse button. You will see the box change from cntr to rght (right). Press the left mouse button again and the box will change from rght to left.

Left justification means the text will be anchored at the left bottom corner, or the text will flow to the right of where it was placed. Center justification means the text will be centered about the placement point. Right justification means the text will be placed to the left of the placement point. Set the toggle switch for the desired justification.



7.2.13 Text Size

This option allows you to set a default text size for your diagrams. Text sizes range from .001 to 9.0. For the purpose of writing general text in event trees, a text size of about 0.5 is appropriate. The size looks too small on the screen, but it is a good size for sending to a laser printer. When you invoke this option you are prompted to **Enter new text size** >. Enter the desired size and press <Enter >. The Text Size box will reflect the current default setting. Remember, here you are setting the default text sizes. You may always change the text size for selected text by invoking the TEXT option, selecting the EDIT suboption, and invoking the Size command (see Section 7.2.7).

7.2.14 Line

This option allows you to select a color for the lines in your diagram. When you invoke this option you will be prompted to Pick a new color from the color bar. Position the cursor over the desired color and press the left mouse button. The Line box will change to reflect the newly selected color. (Remember, you may use the color feature $(\uparrow \downarrow)$ to display additional colors). All new lines generated in your diagram will be this default color.

7.2.15 * + (Color)

This option allows you to scroll the color bar to display the additional color selections available. Position the cursor on the scroll box $(\uparrow \downarrow)$ and press the left mouse button or <Enter> key. The next series of colors will be displayed.

NOTE:

When you try to exit the Create Event Trees option without saving your changes, you will be prompted with **ARE YOU SURE? Enter** "**Y**" to **Quit anyway**. At this point you may enter a < **Y**> to quit without saving or press < Enter> to terminate the exit procedure.

7.3 Link Event Trees

This option allows you to define linkage rules and generate sequence logic. When you invoke this option, Figure 132 will be displayed. As shown, four options are available: Exit, Generate Sequences, Link Editor, and Sequence Editor. In addition, the following special function keys are available here and in the remaining functions discussed in Section 7.

< Esc >	Returns you to the Event Tree Graphics System Menu (Figure 116).
<f1></f1>	Displays on-line help messages.
<f2></f2>	Marks a file for further processing. When you mark a file, an asterisk will appear in front of the file name.
<f3></f3>	Marks all the displayed files for further processing.



Allows you to mark a range of files for further processing.

<F5>

<F4>

Allows you to locate a specific file for further processing. When you invoke this function, Figure 133 will be displayed. Enter the file name to locate and press < Enter>. The specified file will be highlighted. If the file name does not exist, then the next file name in alphabetical order will be highlighted.

SURRY	Link Event Tree
Option E	Exit / Generate Sequences / Link Editor / Sequence Editor
Mame.	Description
BA	BRIDGE TREE, LARGE LOCA
BS1	BRIDGE TREE, MEDIUM LOCA
RS2	BRIDGE TREE, SMALL LOCA
853	BRIDGE TREE, VERY SMALL LOCA
BT1S	BRIDGE TREE, STATION BLACKOUT AT UNIT 1
BT1SB	BRIDGE TREE, STATION BLACKOUT AT BOTH UNITS
BT2	BRIDGE TREE, LOSS OF MAIN FEEDWATER
BT7	BRIDGE TREE, STEAM GENERATOR TUBE RUPTURE
FA	LARGE LOCA EVENT TREE
FS1	MEDIUM LOCA EVENT TREE
<esc></esc>	<f1> <f2> <f3> <f4> <f5></f5></f4></f3></f2></f1>
Exit	Help Mark Line Mark All Mark Range Locate

Figure 132. Link event tree screen.

To display additional event trees, use the <PgUp>, <PgDn> and arrow keys.

7.3.1 Exit

This option returns you to the Event Tree Graphics System menu (Figure 116). To invoke this option, enter an $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

7.3.2 Generate Sequences

This option allows you to generate sequence logic for event trees. During sequence generation the sequence logic is created as specified in the event tree logic, and then the linkage rules are applied. To invoke this option for a single selected event tree, enter a < G > in the option field, highlight the desired event tree, and press < Enter >. To invoke this process for a group of event trees, mark the desired event trees using the function keys F2, F3, and F4, enter a < G > in the option field, and press < Enter >. To invoke this option for all event trees in the current family clear all marked entries, type a < G > in the option field, and press < Enter >. A message **Process all records?** (Y/N) will appear at the bottom of the screen. Type a < Y > to generate sequences for all of the event trees, or type an < N > to terminate the sequence generation process.



SURRY	Link Event Tree
Option [E]	Exit / Generate Sequences / Link Editor / Sequence Editor
BA BS1	BRIDGE TREE, LARGE LOCA BRIDGE TREE, MEDIUM LOCA
Please enter	the name to locatethen press <enter>.</enter>
BT7 FA FS1	BRIDGE TREE, STEAM GENERATOR TUBE RUPTURE LARGE LOCA EVENT TREE MEDIUM LOCA EVENT TREE
<esc></esc>	<pre><f1> <f2> <f3> <f4> <f5> </f5></f4></f3></f2></f1></pre>

Figure 133. Locate a specific system using <F5>.

In all three cases, after pressing < Enter >, Figure 134 will be displayed. On this screen you must enter a file name where the report will be written. The following options are provided:

CON		Sends the report to the screen.
PRN		Sends the report to the printer.
filename		Sends the report to the specified hard disk file name. A default name is provided, which can be changed by simply typing over the default name.
blank < Esc >	а. – ¹ а.	No report is generated, but the sequences are generated. Terminates the process without generating the sequences.

In addition, the prompt Use Number for Sequence Names? appears. If you respond < Y >, IRRAS uses the sequence numbers to create a name. If you respond < N >, the sequence name is used. In either case, the value is created by taking the number or name and each time a transfer is encountered, it adds a "-" and an incremented number for that transfer (e.g., K-X, 2-1, 2-2, 2-3 etc.).

Finally, you may specify the mutually exclusive top name. This allows the user to specify a set of events that can be eliminated from the final list of cut sets. You may leave this field blank or specify a top name.

The Sequence Generation report contains the listing of the sequences' names, the systems that make up the sequences, any substitutions of systems (based on the linkage rules), and any transfers to other event trees.

Opt	ion [G] Exit / Generate Sequences / Link Editor / Sequence Editor
BA	Re2 Ist
BS1	0061101
BS2	File Name = "CON" Sends report to the screen.
BS3	"PRN" Sends report to the printer.
DITS	filename Sends report to the hard disk file.
BT2	<pre><esc> EXIT > Do not generate sequences.</esc></pre>
817	and the set printing and and and
FA	Use Numbers For Sequence Names? Y
FSI	Mutually Exclusive Tax Name
<7	mutually exclusive top name

Figure 134. Generate sequences output screen.

After sequence generation begins, the first message displayed is **Reading Logic for event tree:** <event tree name>. As each sequence in that event tree is generated, the message **Processing sequence:** <sequence name> is displayed. If there are any transfers in the event tree, the reading logic message will be displayed again, but this time with the name of the event tree being transferred to. When all the sequences for the event tree have been created, the message **Deleting old/unused sequences for < ent tree name>** will appear. At this time all the sequences for the event tree that are no longer valid will be removed.

Invalid sequences can result in several ways. First, the logic of an event tree may be such as to create sequences that have tops in them that are both failed and successful. These are invalid and are deleted. Second, the user may have modified the event tree logic, reducing the number of sequences for the tree. Sequences that existed previously are invalid, also. Sequences that have an end state value of "OK," "SUCCESS" or begin with an "@" character are never generated by IRRAS.

7.3.3 Link Editor

This option allows you to specify linkage rules for an event tree. A linkage rule is a special case, an exception, or substitution to the normal sequence generation. During sequence generation the sequence logic is created as specified in the event tree logic, and then the linkage rules are applied. For example, event tree A contains a sequence named SEQ-1. According to the strict logic of the event tree, the systems that make up SEQ-1's logic are SYS-1, SYS-2, SYS-3, and SYS-4. A linkage rule would enable you to replace all occurrences of SYS-4 in any sequence in the event tree with SYS-5. Or, you may only want to replace SYS-4 with SYS-5 if and only if SYS-2 and SYS-3 are also present in the sequence logic.



You can only invoke this option for a single event tree (i.e., one event tree at a time). To invoke

this option, enter an <L> in the option field, highlight the desired event tree, and press < Enter >.

When you press $\langle \text{Enter} \rangle$, Figure 135 will be displayed showing the rules (if any) for the event tree shown in the upper right hand corner. The message **Reading in tops and systems names** will be displayed. A top is a system that is used in the selected event tree. Then, the message **Reading in rules** will be displayed. The rules (if any) for the event tree will then be shown. The first # column is the rule number or the order in which the rules will be applied. Rule 2 takes precedence over rule 1 and rule n takes precedence over rule n-1. The second and third columns are the conditional columns. If a top or system must exist before a substitution can take place, those tops will be named here. There is no limit on the number of conditional tops for a rule. The fourth or replaced top column contains the tops to be substituted and the fifth column contains the tops/systems to use in place of the replaced tops. The following options are available: Exit, Add, Insert, Copy, Delete, and Restore.

Option: Exit / Add / Insert / Dopy / Delete / Restore						
	17	And	Then	1.1.1		
per Il marror T	op Event	TOD Event worke	Top Event -	Svst	(M) increases	
10.000		the service of the				
				36	-	
Land to the				1 N N		
1.1.1						
				10 M		
1				*		
1.				74		
be the set						
				3K.		
L						
1						

Figure 135. Link editor screen.

7.3.3.1 Exit. This option returns you to the Link Event Tree menu. To invoke this option type an $\langle E \rangle$ and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key. Figure 136 will be displayed asking you if you want to save the rules and then exit. To save the rules in the data base, type a $\langle Y \rangle$, and press $\langle Enter \rangle$. To exit without saving the rules, type a $\langle N \rangle$ and press $\langle Enter \rangle$.

7.3.3.2 Add. This option allows you to add a new rule or modify an existing rule if the carsor is in the first column, or add/modify a top or system if the cursor is in any other column. To add a rule, move the cursor to the first column and type an < A >. A default rule number will appear in the first column. Press < Enter > and the cursor will be moved to the second column. A line of dashes will appear above the rule that is being added. This is to assist in keeping the rules separated. To add a conditional top, type < A >, enter the name of the conditional top, and type < Enter >. If the name you entered is not a top for this event tree, the message **Name not found...** is displayed. A list of all the tops

7-34
Option: Exit / Add / Insert / Copy / Delete / Restore 1 f And Then 1 s	
lf And Then Is	
m # Top Event Top Event Top Event System	
Save Rules, then Exit? Y (Y/N) =	
	1

Figure 136. Save linkage rules screen.

for this event tree is displayed in a help form on the right side of menu. To select one the tops from the help form, type $\langle F1 \rangle$, move the cursor to the top to be added, and press $\langle Enter \rangle$ (see Figure 137). All conditional tops and replaced tops are added in this same way. The replacement top is added in much the same way with some differences. When you type $\langle A \rangle$, the list of all systems in the family is displayed in a help form on the left hand side of the form (see Figure 138). Any system in the family can be entered here, and the message **Name not "ound...** will be displayed if the name entered is not a system. To modify a rule, position the cursor over the top/system to be changed, type $\langle A \rangle$, and enter the new top/system name just as you do on an add.

7.3.3.3 Insert. This option adds one blank line after the line containing the cursor, regardless of the column the cursor is in.

7.3.3.4 Copy. This option copies one rule after another. Figure 139 contains the copy menu. The rule number of the rule closest to the cursor will be in the first blank, but you may enter any other existing rule in its place. To copy the desired rule, enter an existing rule in the second blank and a copy of the first rule will be placed after the second rule.

7.3.3.5 Delete. This option deletes the entire selected rule if in the first column, or a top or system if in any other column. To delete a rule, place the cursor in the first column of the rule to be deleted and type a < D >. To delete a system or top, place the cursor in the column where the desired system or top is located and type a < D >.

7.3.3.6 Restore. This option restores the last deleted rule if the cursor is in the first column, or a top or system if the cursor is in any other column.



	Option: Exit / Add	d / Insert / Copy / D	elete / Tops	
	1 f A r	id. The	n 06	1.
_ # ī	op Event Top E	vent measurement top Ev	ent men F1	
1 05	D6	D5	CV H1-A-S1	
han an anna an	-	NAMES AND ADDRESS OF A DESCRIPTION OF A	nana amazoa kasaa amazoa men	













NOTE:

Please note there is a one to one relationship between replaced tops and replacement tops/system. In other words, a group of systems cannot replace one top and group of tops cannot be replaced by one system. An example of a multi-branch failure is shown in Figure 140.

7.3.4 Sequence Editor

This option allows you to edit sequence names, end states, and assign frequencies after an event tree has been created. To invoke this option, enter an $\langle S \rangle$ in the option field, highlight the desired event tree, and press $\langle Enter \rangle$. Figure 141 will be displayed showing all the sequence names and event states defined for the selected event tree.

NOTE: Any changes made using this option, will be made throughout the entire system (i.e., the corresponding drawing will be updated).

As shown, eight options are available. Each of these options is discussed in the following paragraphs.

7.3.4.1 Exit. This option terminates the sequence editor and returns you to Figure 132.













7.3.4.2 Frequency. This option will display the frequencies or min cut upper bound values associated with the selected sequence. To invoke this option, enter an < F > in the option field, highlight a sequence (or mark a series of sequences), and press < Enter >. The EXTRA-#1 column will now read FREQUENCY and the corresponding frequencies will be displayed. You may now invoke the Line Edit command to modify frequencies, it desired.

7.3.4.3 @Status. An "@" means that the corresponding sequence will not be generated; it will be ignored. This option allows you to set the current state of a sequence (i.e., sequence will be generated or ignored). To invoke this option, enter an "@" in the option field, highlight the sequence whose status is to be changed, and press < Enter >. Depending on the current status, an "@" will appear or disappear from the STATE column.

7.3.4.4 Line Edit. This option allows you to edit the data displayed one line at a time. To invoke this option, enter an $\langle L \rangle$ in the option field, highlight the line you wish to edit using the arrow keys, and press $\langle Enter \rangle$. The selected sequence will appear in a window. You may change any of the fields displayed by simply typing over the existing data. Press the tab key to move from field to field. When complete, press $\langle Esc \rangle$ to exit without saving the changes made. Otherwise, press $\langle Enter \rangle$. When you exit the Sequence Editor option, you will be asked whether or not you wish to save changes before exiting.

7.3.4.5 Transfer. This option will follow the transfer of the selected sequence. If changes were made to the current event tree, you will be asked if you wish to save those changes before following the transfer.

7.3.4.6 Global Replace. This option allows you to replace a specified string at every occurrence with another string. To invoke this option, type a $\langle G \rangle$ in the option field and press $\langle Enter \rangle$. Figure 142 will be displayed. First, you enter the string for which to search. Press the $\langle Tab \rangle$ key once and enter the replacement string. Press the $\langle Tab \rangle$ key once more, and enter the column (1, 2, 3, or 4) on which to perform the search. If the search string is found, all occurrences of the search string will be replaced by the string specified in the replacement string. If the string is not found, the message **No such string found** will be displayed at the bottom of the screen and you will be returned to the previous screen.

7.3.4.7 Header Edit. This option allows you to change the headers that appear across the top of the columns (note: EXTRA-#1 and EXTRA-#2 are user-defined fields). To invoke this option, type an $\langle H \rangle$ and press $\langle Enter \rangle$. An outline will surround the headers that may be changed. Simply type over the existing header. Use the $\langle Tab \rangle$ key to move from field to field. When complete, press $\langle Esc \rangle$ to terminate without saving the header changes. When you exit the Sequence Editor you will be asked whether or not you wish to save all changes made before exiting.

7.3.4.8 Change Transfer. This works as a toggle switch to turn on and off transfer points. If the sequence was marked as a transfer, this option will turn off the transfer. If the sequence was not marked as a transfer, it turns on the transfer.







7.4 Plot Trees

This option allows you to plot graphics files. This option operates the same as the Plot Tree option for fault trees discussed in Section 5.2, except in this option you will be working with event trees. Refer to Section 5.2 for the detailed user instructions.

7.5 Graphics Load

This option allows you to load event tree graphic files. When you invoke this option, Figure 143 will be displayed. As shown, two options are available: Exit and Load.

7.5.1 Exit

This option returns you to the previous screen. To invoke this option, enter an $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

7.5.2 Load

This option allows you to load event tree graphics into the system. A conversion of the selected file will be performed. To invoke this option, enter an <L> in the c_P tion field, highlight the file or mark the files to be loaded using the function keys, and press < Enter>. A message will be displayed telling you that the selected file(s) is being converted.



	Option L Exit / Load
LOSP	ETG Files
a she water a	



7.6 eXtract Event Trees

This option allows you to extract event trees from the system. When you invoke this option, Figure 144 will be displayed. On this screen all the event trees residing in the current directory are displayed. Four options are available: Exit, eXtract Trees, Clear Extracted Trees, and Display Extracted Trees.

7.6.1 Exit

This option returns you to the Event Tress Graphics System menu (Figure 116). To invoke this option, enter an $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

7.6.2 eXtract Trees

This option allows y_{-} , to remove specific event trees from the data base. To invoke this option, enter an $\langle X \rangle$ in the option field, highlight the tree to be extracted or mark the trees to be extracted (using the function keys) and press $\langle Enter \rangle$. A message will be displayed notifying you that the tree(s) was successfully extracted.

7.6.3 Clear Extracted Trees

This option allows you to clear all extracted trees from a file. To invoke this option, enter a $\langle C \rangle$ in the option field, highlight the file to be cleared or mark the files using the function keys, and press $\langle Enter \rangle$. A warning screen will appear (Figure 145) telling you that all existing *.ETG files will be deleted. Enter a $\langle Y \rangle$ to delete or an $\langle N \rangle$ to terminate the process.



7-41











7.6.4 Display Extracted Trees

This option allows you to display all the extracted trees to date. To invoke this option, enter a < D > in the option field and press < Enter >. A screen similar to the one shown in Figure 146 will be displayed. Press < Enter > to return to the previous screen.

	Option	D Exit /	eXtract Display	Trees / Cle Extracted 1	ear Extra Trees	icted Trees	
Ā	ATWS1	ATWS2	Extracted NONAME	i Event Tree TISB			
	Note : Us	e <pgup> a</pgup>	nd «PgDn»	to display	/ more ev	ent trees.	

Figure 146. Display extracted trees screen.

7.7 Define Plotter Pens

This option allows you to assign colors to the plotter pens. When you invoke this option, a screen will be displayed showing 16 colors. Select the color to be mapped by positioning the cross hair over the desired color and pressing the left mouse button. You will then be prompted to **Enter number of pen >**. Enter the number of the pen that will contain the previously defined color. Continue this process until all pens have been defined. To terminate the process, position the cursor on the STOP symbol and press the left mouse button.



8.0 ANALYZE SEQUENCES

The Analyze Sequences option consists of the following four options:

- Analyze Sequences includes updating the cut sets, quantifying the cut sets, and running uncertainty analyses.
- 2. Display Results presents the analyses in various report forms.
- The Cut Set Editor provides the means to modify the event tree cut sets.
- 4. Link Event Trees allows you to define linkage rules and generate sequence logic.

Keys that you will frequently use are

< Esc >	Escape cancels your last choice and returns you to the previous screen
<f1></f1>	Help briefly explains the function of a field and may show you examples of data entered.
<f2></f2>	Mark/Unmark tags items for use in the selected option.
<f3></f3>	Clear All Marked events removes the marks (*) from the listed items. If no items are marked, this option will mark all of the items.
<f4></f4>	Mark/Unmark range of items quickly tags large numbers of items for processing.
<f5></f5>	Locate an item. This option will display a blank field in the center of the screen, and a message Please enter name to locate will appear. The user should enter the name to be located and then press < Enter >. This feature will place the highlight on the located name. If the required name is not found, then the next name in alphabetical order will be highlighted.

To invoke this option, highlight ANALYZE Sequences and press < Enter >. The event tree analysis main menu is shown in Figure 147. The options available from the event tree analysis main menu are: Exit, Analyze Sequences, Display Results, Cut Set Editor, and Link Event Trees. These options and their functions will be discussed in the following paragraphs.

8.1 Exit

The Event Tree Analysis screen appears with Exit as the default choice in the command line (see Figure 147). Press < Enter > to return to the file & AS main menu.

8.2 Analyze Sequences

This option provides the means to recalculate sequence values after events and/or out sets have been modified. To invoke the option, highlight Analyze Sequences or type $\langle A \rangle$ in the option field, and press $\langle Enter \rangle$. Figure 148 shows the main screen for sequence analysis that lists the sequences defined for the current family. The letters c, q, and u (in any combination) may precede a sequence name and are defined as follows:

hamman	L Analysis F	
	Exit	
	Analyze Sequences	1
	Display Results	
	Cut Set Editor	
	Link Event Trees	
	Option [F]	1



SURRY	An±lyz	e Sequence
Option E Ex	it / Generate Cut	Sets / Cut Set Update / Quantification
Un	certainty Analysis	/ Split Fraction
A-D5	FA	LARGE LOCA - ACCUMULATOR FAILURE
A-06	FA	LARGE LOCA - INJECTION FAILURE
A-H1	FA	LARGE LOCA - RECIRCULATION FAILURE
S1-D1	FS1	MEDIUM LOCA - INJECTION FAILURE
S1-D6	FS1	MEDIUM LOCA - INJECTION FAILURE
S1-H1	FS1	MEDIUM LOCA - RECIRCULATION FAILURE
\$2-01	FS2	SMALL LOCA - INJECTION FAILURE
\$3-01	FS3	VERY SMALL LOCA - INJECTION FAILURE
u SBO-BATT2	FT1SB	SBO (U1, U2) - STUCK OPEN PORV
u \$80-12	FT1\$B	SBO (U1, U2) - AFW FAILURE
<esc> <f1></f1></esc>	<\$2>	<f3> <f4> <f5></f5></f4></f3>
Ewit Halm	Mark 1 ins	Mark Mill Hark Danne Locate







- c flags the sequence as having cut sets that must be recalculated
- q flags the sequence as having cut sets that must be requantified
- u flags the sequence as needing uncertainty distributions recalculated

Once these functions have been executed the corresponding letter is removed from the display. The options available from the Analyze Sequences screen are Exit, Generate Cui Sets, Cui Set Update, Quantification, Uncertainty Analysis, and Split Fraction. You have the choice of using either the Monte Carlo sampling technique or the Latin Hypercube sampling technique for running the uncertainty analysis process.

IRRAS writes the summary information generated during analysis to a file that the user can read to determine the results of a batch operation involving several systems or sequences. This file is called "SCREEN.CPY." When either Analyze Systems or Analyze Sequences is executed, IRRAS deletes a file in the current family subdirectory called 'SCREEN.OLD." The file "SCREEN.CPY" is then renamed to "SCREEN.OLD." The summary screens for any operations performed during this session (using the Analyze option) are then written to the file "SCREEN.CPY." The user may exit IRKAS, change to the specified subdirectory, and print, edit, or review this file.

8.2.1 Exit

To return to the Event Tree Analysis menu, type an $\langle E \rangle$ (Exit) in option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

8.2.2 Generate Cut Sets

This option allows you to generate the cut sets for a selected sequence based on cut set generation cutoff values. You are given the opportunity to specify several cut set generation cutoff values that will be used to determine if a cut set is to be retained or discarded from the selected sequence.

You are given the ability to generate cut sets for a selected sequence, a group of sequences, or all sequences within the current family. To invoke the Generate Cut Sets for a single selected sequence, type $\langle G \rangle$ (Generate Cut Sets) in the option field, highlight the desired sequence, and press $\langle Enter \rangle$. To invoke this process for a group of sequences, mark the desired sequences using the function keys F2, F3, and F4, type a $\langle G \rangle$ in the option field, and press $\langle Enter \rangle$. To invoke this option for all sequences in the current family, clear all marked entries with the F3 key, type $\langle C \rangle$ in the option field and press $\langle Enter \rangle$. A message **Process all records**? (**Y**/**N**) will appear at the bottom of the screen. Type a $\langle Y \rangle$ to continue the Generate Cut Sets for all records that need processing (e.g., for $\langle G \rangle$ the sequences must have a "c" in front of the sequence name, etc.) or type an $\langle N \rangle$ to terminate the update for all sequences.

Whether you are generating cut sets for a single sequence, a group of sequences, or for all sequences, the Cut Set Generation Cutoff Values screen shown in Figure 149 will be displayed. You may change any of the data fields on this screen. Each field is described below. The default values that appear on this screen may be reset to new values by selecting the Utility Options from the IRRAS main menu and then invoking the Define Constants option.

Perform Cut Set Prob Truncation?

If you enter a < Y >, then the only cut sets whose product for all of its event probabilities is greater than or



SCRF *		ΙL	Ana	lyi	z e S	e q	uen	ce			
	o este scorectore		CUT SE	T GEN	ERATION	CUT	OFF VA	LUES -	an tan Kito ayan aya		1
Perfora	Cut Set	Prob	Trunca	tion?	(Y/N)	Y		Cuto	f Value	1,000E-015	
Perform	Event	Prob	Trunca	tion?	(Y/N)	N	Min	Cuto	ff Value	1.000E-003	1.25
Perform	size or	Zone	Trunca	tion?	(Y/N/	2) Y		size	Cutoff	6	1.13
Solve 5	iequence	Wish 1	Fault T	rees?	(Y/N)	۲	Flag	Set N	ame		
Note	t 1. per	form i	Event P	obat	allity meation	trun	cation	you m	ust also	specify	



Perform Event Prep Truncation?

Perform Cut Set Size Truncation

Solve Sequence With Fault Trees equal to the value in the Cutoff Value field will be kept. All other cut sets will be removed.

If you enter an < N >, then the probability for the cut set will not be relevant for determining if the cut set should be retained or discarded.

If you enter a < Y >, then you must also enter < Y > for "Perform Cut Set Prob Truncation." This option will check all cut sets that are below the probability cutofi (MIN < Cutoff Value field) and remove there only if they contain an event whose probability is below this value.

If you enter a < Y >, then only the cut sets whose number of events is less than or equal to the value specified in the Size Cutoff field will be kept in the cut sets for that system. All other cut sets will be removed. If you enter an < N >, then the number of events in a cut set will be irrelevant for determining if the cut set should be retained or discarded.

If you enter a $\langle Z_i \rangle$, then only zone flagged events will be checked.

If you enter a < Y >, fRRAS uses the system fault trees to solve the tree, otherwise the system cut sets are used.



8-4

Cutoff Value

MIN < Cutoff Value

> Size 'lutoff

Flag Set . me

N ue set for probability truncation.

Value set for event probability truncation.

Value set for cut set size truncation.

The flag set name is the name of a change set (see Suction 4.5.2) containing a set of events to be set to either "T," "F," or "I" and flags to be set to "X," "Y," or "I." These events must be individual settings (i.e., not class changes).

If this field is left blank, then IRRAS checks to see if there is a default setting for this sequence. The default can be set using the Event Tree Sequences suboption under the Modify Data Base option (see Section 9.3.6). If there is a default setting, then IRRAS uses it, otherwise, IRRAS uses no flags.

If the flag set name is set to "NONE," then no flags are used.

This option allows the user to specify changes to the flags for each sequence without having to regenerate the failure data such time. The fault tree logic is pruned before it is solved, dependent on these flags.

During processing, the screen shown in Figure 150 is displayed and updated as the calculations proceed. Upon completion of the cut set generation, the results are displayed as shown in Figure 151.

8.2.3 Cut Set Update

This option will update the alternate cut sets for a selected sequence based on cut set gene ution cuto? values. This option operates the same for sequences as it does for systems. Rather than near the instructions, the reader is referred to Section 6.2.3.

8.2.4 Quantification

The quantification process will calculate a new minimum cut set upper bound for the sequence out sets using the current data values (event change sets and alternate case out sets. The new minimum out set upper bound is saved with the alternate case out sets for the selected sequence. This option operates the same for sequences as it does for systems. Rather than repeat the instructions, the reader is referred to Section 6.2.4.

8.2.5 Uncertainty Analysis

This option allows you to run a single uncertainty analysis for a sequence or an overall uncertainty analysis for a group of sequences. When you enter a $\langle U \rangle$ and press $\langle Enter \rangle$, Figure 152 is displayed. As shown, the four types of overall uncertainty analyses available are: an analysis for a single sequence, a group of marked sequences, an analysis for all sequences having a particular end state, or an analysis for all sequences within the current family.



- Family -SURRY Analyze Sequence Currently Processing -Sequence Name A-F1-F2H1 Current System +2 Total Number of Systems 9 Current System Number 2 Press *ESC* to terminate.









Uncertainty Opt	ions [-] Exit / Single	/ Group / ENd State / Famil	Ύ.
N & m & m & m & m & m & m & m & m & m &	T T E LARGE FA LARGE LARGE FA LARGE LARGE FA LARGE LARGE FS1 MEDIL LS1 FS1 MEDIL FS1 FS2 SMALI FS2 FT1SE SB0 FT1SE	D E 6 C T I D S I C A LOCA - ACCUMULATOR FAILURE LOCA - INJECTION FAILURE LOCA - INJECTION FAILURE LOCA - RECIRCULATION FAILURE IM LOCA - INJECTION FAILURE IM LOCA - INJECTION FAILURE IM LOCA - RECIRCULATION FAIL LOCA - INJECTION FAILURE SMALL LOCA - INJECTION FAIL UI, U2) - STUCK OPEN PORV (U1, U2) - AFW FAILURE	RE URE URE
dees of the	<fz> <f3< td=""><td>stás st</td><td>5×</td></f3<></fz>	stás st	5×



There are two different sampling techniques provided to the user for generating the samples that will be used in the uncertainty analysis calculations. The two sampling techniques are the Monte Carlo simulation technique and the Latin Hypercube simulation technique.

8.2.5.1 Exit. This option terminates the uncertainty analysis process and returns you to Figure 148. To invoke this option enter an $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

8.2.5.2 Single. To generate a single uncertainty analysis, enter an $\langle S \rangle$ in the option field. The Sequence Uncertainty menu will be displayed (Figure 153). From this menu you may run a single uncertainty analysis for either a single highlighted sequence, for a group of marked sequences, or for all of the sequences within the current family.

To invoke the uncertainty analysis process for a single selected sequence type < L > or < M >(Latin Hypercube / Monte Carlo) in the option field, highlight the desired sequence, and press < Enter >. To invoke this process for a group of sequences, mark the desired sequences using the function keys F2, F3, and F4, type an < L > or < M > in the option field and press < Enter >. To invoke this process for all sequences in the current family, clear all marked sequences and then type an < L > or < M > in the option field and press < Enter >. To invoke this process for all sequences in the current family, clear all marked sequences and then type an < L > or < M > in the option field and press < Enter >. A message **Process all entries**? (Y/N) will appear at the bottom of the screen. Type a < Y > to continue the uncertainty analysis for all of the sequences, or type an < N > to terminate the analysis.

When you have entered the desired sampling technique for the uncertainty analysis, the Uncertainty Calculation Values screen will be displayed (Figure 154 shows the Monte Carlo calculation values). This screen prompts you to enter the number of samples to be generated during the simulation and the initial value of the seed for the random number generator. A default seed velue for the random



Disting (E) East /	donte Parte Inizi		in Nonerreihen Deren	eraikku
obtion (c) control	HURLE CELES UNDE	complete and	an nypercube once	r sammey
V + A-US	A T T C A MARK	LARGE LOCA	AF A FT G FT 3 8	FL MAN PROVIDENT
00-A + U	A	LARGE LOCA	- INJECTION FAILU	RE, NOT RW
+ A-D6-C-F1	A	LARGE LOCA	- INJECTION FAILU	RE, FM RWS
U + A-H1	A	LARGE LOCA	RECIRCULATION F.	AILURE, NO
U + 51-01	\$1	MEDIUM LOCA	- INJECTION FAIL	URE, NOT R
+ \$1=D1=D=F1	£1 · · · ·	MEDIUM LOCA	 INJECTION FAIL INJECTION FAIL 	URE, FM RW
U + \$1-F1-F2-H1	51	MEDIUM LOCA	RECIRCULATION	FAILURE, F
u + \$1+H1	51	MEDIUM LOCA	- RECIRCULATION	FAILURE, N
<esc» <f1=""></esc»>	<72>	8.1 BX	<142	x15x
Exit Hein	Mark Line	Mars, Ali	Mark Range	Locate

Figure 153. Sequence uncertainty ronu.

seed will be provided. You may use this value or enter a new value for the seed. To obtain a random seed from the system clock, enter a zero in this field. There will also be a default value in the field for the number of samples. You may use this value or enter another value. You may change the default values for both the number of samples and the random seed in the Utility Options, Define Constants suboption.

-	Monte Carlo uncertainty is viation Values
	Number of samples to use in Month' aris simulation 1000
	Secd for random number scherator 0
1.1	MOTE: Has HOR as the court to card a court and then the slock

Figure 154. Monte Colo calculation values.



When using the Monte Carlo sampling technique for the uncertainty analysis, if the number of samples entered is less than ten, then the number of samples will be increased to ten before the uncertainty analysis process will continue. Any number of samples greater than or equal to ten will be allowed, but a number of at least 1000 is probably a better value for improving the reliability of the Monte Carlo results.

When using the Latin Hypercube sampling (LHS) technique, if the number of samples entered is less than twice the total number of unique events in the sequence, then the number of samples will be increased to two times the total number of unique events before the analysis will continue. The LHS technique gives its best results if the number of samples is at least twice the total number of unique events.

Once the number of samples has been accepted and a seed obtained from the system clock if necessary, checks will be run to ensure the events with the same correlation classes have consistent failure data, uncertainty data, and distribution types. If any events with inconsistencies exist, an error message will be displayed and the uncertainty analysis process will be terminated so that the inconsistent values may be corrected.

If an error of some type occurs during the uncertainty analysis process, the process is terminated and the message **Error in Uncertainty analysis** is displayed at the bottom of the Analyze Sequence screen. If all of the events successfully pass the correlation class checks, then the distribution parameters for the events will be checked to ensure that they are valid. If any of the parameters are invalid, error messages will be displayed and the process will be terminated so the distribution parameters may be corrected by the user.

After both of these checks have been passed, a point estimate will be calculated for the selected sequence. At this point the samples for each event will be generated using the selected sampling technique, either Monte Carlo Sampling technique or the Latin Hypercube Sampling technique. The uncertainty analysis function provides the user with eight different distribution types for both of the two sampling techniques. The distribution types include Normal, Lognormal, Beta, Gamma, Chi-Squared, Exponential, Uniform and the user-defined histograms.

During processing the current status screen will be displayed and updated as the samples are generated. Figure 155 illustrates the current status screen for the Monte Carlo sampling technique. When the requested number of samples has been generated or the user has terminated the process of generating the samples by pressing the $\langle Esc \rangle$ key, statistical information will be calculated using the generated samples. A sample mean, median, and standard deviation will be calculated for the selected sequence. Coefficients of skewness and kurtosis, and quantile values will also be calculated for the sequence. This data will be saved in the data base for the selected sequence.

Upon completion of these calculations, the following values will be displayed on the Uncertainty Results screen for viewing: the sequence name, random seed used, the number of samples generated in this process, the total number of events and cut sets in the sequence being processed, the point estimate, the mean, the median, the 5th and 95th percentile values, the minimum and maximum generated sample values, the standard deviation, the skewness and kurtosis, and the time involved to perform the analysis. Figure 156 illustrates the Uncertainty Results screen for the Monte Carlo sampling technique.

- Family men SURRY Seq Uncertainty more Currently Processing movement Name A-H1 Total Number of samples.... Current Sample..... 5.63 Running Mean Value Press Esc to terminaie. Figure 155. Current status of the Monte Carlo sampling. - Family me Event Tree me SURRY Seq Uncertainty A-H1 - Uncertainty Results -----Name A-H1 Random Seed 11050 Events 19 Sample Size 1000 Cut Sets 28 Point estimate 9.852E-005 Mean Value 9.357E-005 Mean Value 9.357E-005 5th Percentile Value 3.474E-006 3.325E-005 Median Value 95th Percentile Value 3.887E-004 Minimum Sample Value 3.607E-007 Maximum Sample Value 2.332E-003 Standard Deviation 1.760E-004 5.812E+000 5.646E+001 Skewness Kurtosis Elapsed Time 00:01:26.120

Figure 156. Monte Carlo uncertainty results.

If only one sequence was selected (highlighted) for the uncertainty analysis process, then you will need to press < Enter > to return to the Analyze Sequence screen from the Uncertainty Results screen. However, if more than one sequence is being processed, the Uncertainty Results screen will be displayed for each sequence, and when all of the selected sequences have been processed you will be returned automatically to the Analyze Sequence screen.

8.2.5.3 Group. To generate an overall uncertainty analysis for a group of sequences, enter a $\langle G \rangle$ in the option field. The Group Uncertainty menu will be displayed (Figure 157).





To invoke the overall uncertainty analysis process for a single group of sequences, type <L > or <M> (Latin Hypercube / Monte Carlo) in the option field, and press < Enter >. To invoke the process for a group of sequences, mark the desired groups using the function keys <F2>, <F3>, and <F4>, type an <L> or <M> in the option field, and press < Enter >. To invoke this process for all groups, clear all marked groups and then type an <L> or <M> in the option field and press < Enter >. A message **Process all entries**? $<\mathbf{Y}/\mathbf{N}>$ will appear at the bottom of the screen. Type a <Y> to continue the uncertainty analysis for all groups, or type an <N> to terminate the analysis.

When you have entered the desired sampling technique for the uncertainty analysis, the Uncertainty Calculation Values screen will be displayed. This screen is similar to the one shown in Figure 154 (refer to Section 8.2.5.2 for details). Once you have entered valid values for the uncertainty calculations, the uncertainty analysis process will begin. During processing the current group status screen will be displayed and updated as the samples are generated. Twenty-five samples will be generated at a time, before the status screen will be updated with a new calculated mean value. Figure 158 illustrates the current group status screens for the Monte Carlo sampling technique. When



the requested number of samples has been generated or the user has terminated the process of generating the samples by pressing the $\langle Esc \rangle$ key, statistical information will be calculated using the generated samples. A sample mean, median, and standard deviation will be calculated for the entire group of sequences. Coefficients of skewness, kurtosis, and quantile values will also be calculated for the group.

SURRY	Group Uncertainty
	The subscription of the subscription of the subscription
	Kame GROUP
	total Number of samples 1000
	Currently Processing Cutset Nome A-D6-CF1
	Current Sample
	Running Mean Value 1.1438-004
	Press Esc to terminate.

Figure 158. Current group status of the Monte Carlo sampling.

Upon completion of these calculations, the following values will be displayed on the Uncertainty Results screen for viewing: the group name, random seed used, the number of samples generated in this process, the total number of events and cut sets in the group of sequences being processed, the point estimate, the mean, the median, the 5th and 95th percentile values, the minimum and maximum generated sample values, the standard deviation, the skewness and kurtosis and the time involved to perform the analysis. The results of overall uncertainty analysis for a group of sequences is not stored in the data base. Figure 159 illustrates the group Uncertainty Results screen for the Monte Carlo sampling technique.

SLIRRY	Group Une	ertainty	
Uncertainty	Results management		
Name Revolute Canal 18080	GROUP		
Sample Size 1000	Cut Sets 18		
Point estimate Mean Value	9.264E-005 1.067E-006		
5th Percentile Value	e 3,310E-006		
95th Percentile Val	ue 4,050E-004		
Minimum Sample Valu Baximum Sample Valu	e 2.409E-007		
Standard Deviation	2.689E-004		
Skewness Kurtosis	9,280E+000 1,204E+002		
Fignered Time	00:01:42.050		

Figure 159. Monte Carlo group uncertainty results.

8.2.5.4 ENd State. To generate an overall uncertainty analysis for all of the sequences within a selected End State, enter an $\langle N \rangle$ in the option field. The End State Uncertainty menu will be displayed (Figure 160). From this menu, mark the sequences using the F2, F3, and F4 function keys that will make up the group. From this menu you may run an overall uncertainty analysis for either a single highlighted End State, for a group of marked End States, or for all of the End States within the current family.

To invoke the uncertainty analysis process for a single selected End State type <L > or <M > (Latin Hypercube / Monte Carlo) in the option field, highlight the desired End State, and press < Enter >. To invoke this process for a group of End States, mark the desired End State using the function keys F2, F3, and F4, type an <L > or <M > in the option field and press < Enter >. To invoke this process for all the End States in the current family, clear all marked sequences and then type an <L > or <M > in the option field and press < Enter >. A message **Process all entries?** (Y/N) will appear at the bottom of the screen. Type a <Y > to continue the uncertainty analysis for all of the End States, or type an <N > to terminate the analysis.

When you have entered the desired sampling technique for the uncertainty analysis, the Uncertainty Calculation Values screen (see Figure 154) will be displayed. Once you have entered valid values for the uncertainty calculations, the uncertainty analysis process will begin. During processing the current End State status screen will be displayed and updated as the samples are generated.



Exit / Monte Carlo Uncertainty / Latin Hypercube Uncertainty
ste monocommune Description remains and
MEDIUM LOCA - SUMP FLUG FAILS LER, CHR
MEDIUM LOCA + LPR FAILURE + RCP BEAL COOLING FAILS MEDIUM LOCA + NET FAILURE - BUSY AND LOT ENPERE
MEDIUM LOCA - RWST FAILS HP1, CHR, AND LP1
MEDIUM LOCA + LPI FAILURE
SMALL LOCA - HP1 FAILURE - RCP SEAL COOLING FAILS
SBD - STUCK OPEN RCS PORVS - NON RECOVERY OF AC POWER
VERY SMALL LOCA . HPJ FAILURE - RCP SEAL COOLING FAILS
VERY SMALL LOCA / RWST FAILS HP1, CHR



Twenty-five samples will be generated at a time, before the status screen will be updated with a new calculated mean value. Figure 161 illustrates the current End State status screen for the Latin Hypercube sampling technique. When the requested number of samples has been generated or the user has terminated the process of generating the samples by pressing the < Esc > key, statistical information will be calculated using the generated samples. A sample mean, median, and standard deviation will be calculated for the entire group of sequences within the current End State. Coefficients of skewness and kurtosis, and quantile values will also be calculated for the End State. This data will be saved in the data base for the current End State.

Upon completion of these calculations, the following values will be displayed on the Uncertainty Results screen for viewing: the End State name, random seed used, the number of samples generated in this process, the total number of events and cut sets in the End State being processed, the point estimate, the mean, the median, the 5th and 95th percentile values, the minimum and maximum generated sample values, the standard deviation, the skewness and kurtosis, and the time involved to perform the analysis. Figure 162 illustrates the End State Uncertainty Results screen for the Latin Hypercube sampling technique.

If only one E(.) State was selected (highlighted) for the overall uncertain'y analysis process, press < Enter> to return to the Analyze Sequence screen from the Uncertainty Results screen. However, if more than one End State is being processed, the Uncertainty Results screen will be displayed for each End State, and when all of the selected End States have been processed you will be returned automatically to the Analyze Sequence screen.



SURRY	End State Uncertainty
	Name Processing End State
0.00 M 10 M	Total Number of samples 100
	Currently Processing Cutset Name
[유럽 모음	current cample
	Running Mean Value 3.6756-001
1.2.2.2.2.2.2.2.2.2.2.2.2.2.2.2.2.2.2.2	Press Esc to terminate.

Figure 161. Current end state status of the Latin Hypercube sampling.

	ind State	Uncerte	iinty	
Uncertainty Rest	ults			and the second s
Random Seed 18070 Ever	nts 43			
Sample Size 100 Cut	Sets 55			
Mean Value	2.1856-002 1.1886-001			1.1.1.1.1.1.1.1
5th Percentile Value	+0.000E+000			
Mediah Value	1.0208-001			
Minimum Sample Value	+0.000€+000			· · · · · · · · · · · · · · · · · · ·
Maximum Sample Value	5.586E-001			1.
Standard Deviation	1.646E+000			
Kurtosiis	6,206E+000			19.00
CINERAL TIME	00+00+23.780			

Figure 162. Latin Hypercube end state uncertainty results.



8.2.5.5 Family. To generate an overall uncertainty analysis for all of the sequences within the current family, enter an $\langle F \rangle$ in the option field. The Family Uncertainty menu will be displayed (Figure 163). From this menu, you select the type of uncertainty analysis to be performed on the family (Monte Carlo or Latin Hypercube).

DEMO Family	Family Uncertainty
Option [E] Exit /	/ Monte Carlo Uncertainty / Latin Hypercube Uncertainty

Figure 163. Family uncertainty selection menu.

When you have entered the desired sampling technique for the uncertainty analysis, the Uncertainty Calculation Values screen (Figure 154) will be displayed. Once you have entered valid values for the uncertainty calculation, the uncertainty analysis process will begin. During processing, the current family status screen will be displayed and updated as the samples are generated.

Twenty-five samples will be generated at a time before the status screen will be updated with a new calculated mean value. Figure 164 illustrates the current family status screen for the Latin Hypercube sampling technique. When the requested number of samples has been generated or the user has terminated the process of generating the samples by pressing the <Esc> key, statistical information will be calculated using the generated samples. A sample mean, median, and standard deviation will be calculated for the entire family. Coefficient of skewness and kurtosis, and quantile values will be calculated for the family. This data will be saved in the data base for the current family.

Upon completion of these calculations, the following values will be displayed on the Uncertainty Results screen for viewing: the Family name, random seed used, the number of samples generated in the process, the total number of events and cut sets in the Family being processed, the point estimate, the mean, the median, the 5th and 95th percentile values, the minimum and maximum generated sample values, the standard deviation, the skewness and kurtosis, and the time involved to perform the analysis. Figure 165 illustrates the Family Uncertainty results screens for the Latin Hypercube sampling technique.

8.2.6 Split Fraction

This option allows you to quantify the sequence cut sets using the mincut upper bound values that have been calculated for each successful or failed system which make up this sequence. This option does not generate cut sets and is usually used to give you a quick approximation of the actual result. The actual results may be obtained by generating cut sets and quantifying these cut sets.



· Family SURRY Femily Uncertainty 100 Total Number of samples.... Currently Processing Cutset Current Sample..... 25 Running Mean Value 3.675E-001 Press Esc to terminate.

Figure 164. Current family status of the Latin Hypercube sampling.

DUNK T	FRMILYUNCERI	ainty
Uncertaint	RUSUITS	
Random Seed 18070 Sample Size 100	Events 43 Cut Sets 55	상태 화가 가 먹는
Point estimate Mean Value	2,1856-002 1,1886-001	사람은 동안은 모르는 동
Median Value 95th Percentile Va	1.020E=001 (ue 3.781E=001	성장 이 것이 이 것 같아?
Minimum Sample Val Maximum Sample Val	ue +0.000E+000, ue 5.586E+001	
Standard Deviation Skewness	1.179E-001 1.666E+000	
Kurtosis	6.2066+000	





To invoke this option, enter an <S> in the option field, highlight the desired sequence, and press <Enter>. The quantification results are shown in a screen similar to the one shown in Figure 166.

SURRY	Analyze s	equence \$1-C	
Presidente Quarter f	cation Results		para anti anti anti anti anti anti anti ant
Name 51			
Min Cut	Upper8ound		
5.3	07E-001		
£laps 00:00	ed Yime :00.760		

Figure 166. Split fraction display screen.

As with previous options, you may specify several sequences or a range of sequences using the <F2> and <F4> function keys, respectively. To process all sequences, enter an <S> in the option field and press < Enter>. The message **Process all entries**? will be displayed. At this prompt, enter a <Y>.

8.3 Display Results

This option allows you to display the results of your sequence analysis. To display the results, highlight Display Results or type a < D > on the Event Tree Analysis menu and press < Enter >. The Display Sequence Results screen will be displayed showing the list of sequences available in the current family. This option operates the same for sequences as it does for systems. Rather than repeat the user instructions, the reader is referred to Section 6.3.



8.4 Cut Set Editor

The cut set editor provides you with the means to edit the base case/alternate sequence cut sets. This option operates the same for sequences as it does for systems. Rather than repeat the user instructions, the reader is referred to Section 6.4.

8.5 Link Event Trees

This option allows you to define linkage rules and generate sequence logic. This option is identical to the Link Trees Option discussed in the Create Event Trees module in Section 7.3. Rather than repeat the user instructions, the reader is referred to Section 7.3.

9.0 MODIFY DATA BASE

This option allows you to modify the base or original family data files for a family, event trees, systems, end states, basic events, attributes, gates, graphics, and histograms. To invoke this option, type < M > in the option field or highlight Modify Data Base and press < Enter >. Figure 167 will be displayed.





In general, each of the options shown in Figure 167 use the same modification functions: Exit, Add, Modify, Delete, and Locate. Some of the options have additional functions such as: Text and Sequences.

The following function keys are available throughout the Modify Database option:

<esc></esc>	Exits the current option and returns you to the Modify Database screen.
<f1></f1>	Displays associated help messages.
<f2></f2>	Mark/Clear tags items for use in the selected option.
<f3></f3>	Clear All Marked events removes the marks (*) from the listed items. If no items are marked, this option will mark all of the items.
<f4></f4>	Mark/Clear range of items quickly tags large numbers of items for processing.
<f5></f5>	Locate an item. This option will display a blank field in the center of

January 1992

9-1

the screen, and a message Please enter name to locate will appear. The user should enter the name to be located and then press < Enter >. This feature will place the highlight on the located name. If the required name is not found, then the next name in alphabetical order will be highlighted.

9.1 Exit

This option returns you to the IRRAS main menu. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.2 Family

This option allows you to add, modify, and delete a family or modify the associated text. To invoke this option, type $\langle F \rangle$ in the option field or highlight Family and press $\langle Enter \rangle$. Figure 168 will be displayed.

Family # SURRY Edit Femily Option [E] Exit / Add / Modify / Delete / Text - Pamily Name - Directory -Description -BROWNT BROWN1 DEMO DEMO Demonstration sample family LEARN LEARN Sample family of data for the SARA 4.0 Tutorial Sample family of data for the SARA 4.0 Tutorial LEARN LEARN2 PRADATA, B1 PRADATA.81 SURRY SURRY SURRY UNIT 1 RELEASE DATE 04-11-91 <Esc> <F1= <\$5× Exit Help Locate

Figure 168. Family editing menu.

9.2.1 Exit

This option returns you to the Modify Database screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.2.2 Add

This option allows you to add a family to the data base. To invoke this option type $\langle A \rangle$ in the option field and press $\langle Enter \rangle$. The Add Family screen (shown in Figure 169) will be displayed. The



only required information to be entered on this screen is the family name. The options of this point are Exit, Add, and Passwords.

	Add Fanily	
	Option (A) Exit / Add / Passwords	
ſ	Name Location Company Type Design ven	dor
	Description free T	ype
	Operational date Qualification date Mission time	

Figure 169. Editing screen for adding a family.

9.2.2.1 Exit. This option returns you to the Edit Family screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.2.2.2 Add. This option performs the actual addition of the family to the data base. To invoke this option, type $\langle A \rangle$ in the option field, enter a family name and any of the other information you wish, and press $\langle Enter \rangle$. At completion of the Add you are returned to the Edit Family screen, where the addition of the new family will be reflected.

9.2.2.3 Passwords. This option not yet available.

9.2.3 Modify

This option allows you to modify the family data record. To invoke this option type $\langle M \rangle$ in the option field, highlight the family you wish to edit, and press $\langle Enter \rangle$. The Modify Family screen is shown in Figure 170. The options at this point are Exit, Modify, and Passwords.

9.2.3.1 Exit. This option returns you to the Edit Family screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.2.3.2 Modify. This option applies the actual modification of the family data to the data base. To invoke this option, type $\langle M \rangle$ in the option field, modify any of the data fields on the screen and press $\langle Enter \rangle$. On completion of the Modify you are returned to the Edit Family screen.

9.2.3.3 Passwords. This option not yet available.



9.3

	L ROOTTY / BRILLY	
	Option [M] Exit / Modify / Passwords	
SURRY	ne Location Company Type Design Vendor WILLIAMSBURG VA VEPCO PWR 3 LOOP W	
SURRY UN	Description Tree Type	
	Operational date Qualification date Mission time 1972/12/22/ 2.400E+001	
	Data Version Data Data Update Data	

Figure 170. Editing screen for modifying a family.

9.2.4 Delete

NOTE: A family that contains sub-families cannot be deleted. The deletion process must proceed up from the lowest to the highest sub-family.

This option allows you to delete family data records from the data base. To invoke this option, type < D > in the option field and press < Euler >. The Delete Family screen is shown in Figure 171. The options at this point are Exit and Delete.

	Decete Family
	Option D Exit / Delete
Name LEARN	Location Company Type Design Vender
Sample family of de	Description Tree Type ta for the SARA 4.0 Tutorial Manual.
Operation	al date Qualification date Mission time



9.2.4.1 Exit. This option returns you to the Edit Family screen. To invoke this option, type <E> in the option field and press <Enter>, or press the <Esc> key.

9.2.4.2 Delete. This option verifies the delete family request. To invoke this option, type <D> in the option field and press < Enter >. A warning screen is superimposed over the Delete Family screen allowing you to cancel the deletion process (Figure 172). Enter a <Y> to delete the family or an <N> to terminate the deletion process. If you respond with a <Y>, the message Deletion completed will be displayed at the bottom of the screen.



Figure 172. Warning screen for a family delete.

9.2.5 Text

This option allows you to view and edit any descriptive text associated with a specific family. To invoke this option, type $\langle T \rangle$ in the option field, highlight a family, and press $\langle Enter \rangle$.

The initial display for this option displays the first 13 lines of the text block. The following keys allow you to display additional lines of text:

PgDn		Presents the next 13 lines of text.
PgUp	987	Presents the previous 13 lines of text.
Ctrl-PgDn	6 A 19	Presents the last 13 lines of text.
Ctri-PgUp	1.5	Presents the first 13 lines of text.
The editing k	eys are	
Ctrl-Z		Exits the text editing feature and saves the text information as it currently exists.
ESC	1000	Exits without saving changes.



Alt-A		Adds a line after the line at the current cursor position
Alt-B		Adds a line before the line withe current curser position
Alt-H		Displays editing keys help een.
Alt-D		Deletes a line at the current arsor position.
Alt-R		Restores the previous deleted text
Del		Deletes a character at current cursor position.
Ins		Inserts a character at current cursor position.
Ctrl-End	*	Deletes all characters from the current cursor position to the end of the cursor line.

The arrow keys are used to move the cursor within the block of text.

The editor does not line wrap, therefore, you must use < Alt-A > to establish each new line of text. If you wish to save your text changes, press < Ctrl-Z >. After you have pressed < Ctrl-Z > you are returned to the Edit 1 amily screen with the message **Text record modified** displayed at the bottom of the screen. If you pressed < Esc >, you will be returned to the family selection screen with the message **Text record not modified** displayed at the bottom of the screen.

9.3 EVent Trees

This option allows you to modify event tree data records. To invoke this option, type $\langle V \rangle$ in the option field or highlight EVent Trees and press $\langle Enter \rangle$. The Edit Event Trees screen listing all of the event trees belonging to the current family will be displayed (Figure 173). The modification options are: Exit, Add, Modify, Delete, Text, Sequences, Base Case Update, and Clear Alternate Case.

RA BS1 BS2 BS3 BT15 BT158 BT2 FA F51	Option [E	Exit / Add / Base Case Upd BRIDGE TREE, L DRIDGE TREE, M BRIDGE TREE, S BRIDGE TREE, S BRIDGE TREE, S BRIDGE TREE, S BRIDGE TREE, S BRIDGE TREE, S LARGE LOCA EVE MEDIUM LOCA EV	Modify / Delete (late / Clear Alter D e s c r i p ARGE LOCA EDIUM LOCA MALL LOCA ERY SMALL LOCA TATION BLACKOUT A TATION BLACKOUT A OSS OF MAIN FEED TEAM GENERATOR TU NT TREE ENT TREE	(Text / Sequences rhate Case t i o n t unit i t BOTH UNITS IATER IBE RUPTURE	
<escs Exit</escs 	<∮1s Help	<f2× Nark/Clear Item</f2× 	<f3> Mark/Clear All</f3>	≪Fá> Mark/Clear Rango	<f5> Locate</f5>





9.3.1 Exit

This option returns you to the Modi. Database main menu. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.3.2 Add

When the user creates event trees using the graphical event tree editor or loads a graphical event tree from another data base, an event tree record is automatically added to the data base. The user need . If use this option if the graphical event tree format is not used. To invoke this option, type < A > in the option field and press < Enter >. The Add Event Tree screen will be displayed (Figure 174). The uptions at this point are Exit and Add.

Lan	No. of the second second second	Commentation of the contraction
		Option [A] Exit / Add
		advisor (al. exist) una
ſ	Name	
	Initiating Event Nam	•
he	Carrier and the second state of the second second	and the second

Figure 174. Editing screen for adding an event tree.

9.3.2.1 Exit. This option returns you to the Edit Event Trees screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.3.2.2 Add. This option performs the actual add of a new event tree record. To invoke this option, type < A > in the option field, fill in the requested data fields, and press < Enter > The required information for an event tree add is the name and initiating event.

When you position the cursor in the Initiating Event Name field, a window will appear listing all initiating events for the current family. Press $\langle F1 \rangle$ to position the cursor in the window. Use the arrow, tab, or space bar keys to scroll through the list of events. When the desired event is highlighted, press $\langle Enter \rangle$. The selected event will be placed in the corresponding field. Upon pressing $\langle Enter \rangle$, the message **Record added** will be displayed at the bottom of the screen.

9.3.3 Modify

This option allows you to modify an event tree record. To invoke this option, type $\langle M \rangle$ in the option field, highlight an event tree name, and press $\langle Enter \rangle$. The Modify Event Tree screen is shown in Figure 175. The options at this point are Exit and Modify.





Figure 175. Editing screen for modifying an event tree.

9.3.3.1 Exit. This option returns you to the Edit Event Trees screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.3.3.2 Modify. This option performs the actual modification of the event tree record. To invoke this option, type < M > in the option field, modify any of the data fields on the Modify Event Trees screen, and press < Enter >.

To modify the initiating event field, position the cursor in the Initiating Event Name field to display a window listing all initiating events for the current family. Press <F1> to position the cursor in the window. Use the arrow, tab, or space bar keys to scroll through the list of events. When the desired event is highlighted, press <Enter>. The selected event will be placed in the corresponding field. Upon pressing <Enter>, the message **Record modified** will be displayed at the bottom of the screen.

9.3.4 Delete

This option allows you to delete an event tree record and associated sequence records from the data base. To invoke this option, type $\langle D \rangle$ in the option field, highlight an event tree, and press $\langle Ente; \rangle$. The delete event tree record is shown in Figure 176. The options at this point are Exit and Delete.

9.3.4.1 Exit. This option returns you to the Edit Event Trees screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.3.4.2 Delete. This option performs the actual deletion of the event tree record. To invoke this option, type $\langle D \rangle$ in the option field and press $\langle Enter \rangle$. A warning screen will appear, allowing you to cancel the deletion at this point (Figure 177). If you respond Y (yes), all sequence records associated with the event tree will be deleted from the data base. You will be returned to the Edit Event Trees screen with the message **Record deleted** displayed.


1.1.1	SURRY Delete Event Tree
	Option [E] Exit / Delete
	Name ATWS1 Description CORE DAMAGE TREE, ANTICIPATED TRANSIENT WITHOUT SCRAM
1.11	
Figure 1	76. Editing screen for deleting an event tree.
-	
i	SURRY Delete Event Tree
	Option [D] Exit / Delete
	Option [D] Exit / Delete Name ATWE1 Description CORE DAMAGE TREE, ANTICIPATED TRANSIENT WITHOUT SCRAM
	Option [D] Exit / Delete Name ATWE1 Description CORE DAMAGE TREE, ANTICIPATED TRANSIENT WITHOUT SCRAM
	Option [D] Exit / Delete Nome ATWE1 Description CORE DAMAGE TREE, ANTICIPATED TRANSIENT WITHOUT SCRAM W A R N I N G Deleting an Eventree will also delete all sequences associated with the Eventree H1

Figure 177. Warning screen for an event tree deletion.

9.3.5 Text

This option allows you to view and edit any descriptive text associated with a specific event tree. This option operates the same as the Text option discussed in the Family section. Refer to Section 9.2.5.

9.3.6 Sequences

This option allows you to modify the sequences associated with an event tree. To invoke this option, type $\langle S \rangle$ in the option field, highlight an event tree name, and press $\langle Enter \rangle$. If an event tree was not highlighted before pressing $\langle Enter \rangle$, the message **An event tree must be highlighted first** will be displayed. After highlighting an event tree and pressing $\langle Enter \rangle$, the Edit Sequences screen shown in Figure 178 will be displayed. The editing options for sequences are: Exit, Add, Modify, Delete, Text, Base Case Update, and Clear Alternate Case.





Figure 178. Sequence selection screen for editing.

9.3.6.1 Exit. This option returns you to the Edit Event Trees screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.3.6.2 Add. This option allows you to add a sequence record to the data base. To invoke this option, type <A> in the option field and press < Enter> The Add Sequence screen is shown in Figure 179. The options at this point are Exit and Add.

	Law of the second second	1 (1 5 B
	Option [A] Exit / Add	
Name Description End State		

Figure 179. Editing screen for adding a sequence.



9.3.6.2.1 Exit—This option returns you to the Edit Sequences screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.3.6.2.2 Add—This option performs the actual add of a new sequence record. To invoke this option, type < A > in the option field, fill in the requested data fields, and press < Enter >. The only required information for a sequence add is the name.

When you position the cursor in the End State Field, a window will appear listing all end states for the current family. Press <F1> to position the cursor in the window. Use the arrow, tab, or space bar keys to scroll through the list of end states. When the desired end state is highlighted, press <Enter>. The selected end state will be placed in the corresponding field.

The Flag Set Name is the name of a change set containing flags to be used when generating cut sets for this sequence. IRRAS uses this default flag set name to modify or prune the fault tree logic for this sequence before it is solved (see Analyze Sequences). Enter a flag set name or leave blank and press < Enter >. Upon pressing < Enter > , the message **Record added** will be displayed at the bottom of the screen.

9.3.6.3 Modify. This option allows you to modify a sequence record. To invoke this option, type < M > in the option field, highlight a sequence name, and press < Enter >. The Modify Sequence screen is shown in Figure 180. The options at this point are Exit and Modify.



Figure 180. Editing screen for modifying a sequence.

9.3.6.3.1 Exit—This option returns you to the Edit Sequences screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.3.6.3.2 Modify—This option performs the actual modification of the sequence record. To invoke this option, type $\langle M \rangle$ in the option field, modify any of the data fields on the Modify Sequence screen, and press $\langle Enter \rangle$. You will be returned to the Edit Sequences screen with the message **Record Modified** displayed at the bottom of the screen.



9.3.6.4 Delete. This option allows you to delete a sequence record. To invoke this option, type < D > in the option field, highlight a sequence name, and press < Enter >. The Delete Sequence screen is shown in Figure 181. Two options are available: Exit and Delete.

		Option	E Exit / Delet	e	
Name Descr	TK-R-Z iption ATWS -	UNFAVORABLE	MODERATOR TEMPER	ATURE COEFFIC	I ÊN T

Figure 181. Editing screen for deleting a sequence.

9.3.6.4.1 Exit—This option returns you to the Edit Sequences screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.3.6.4.2 Delete—This option performs the actual deletion of the sequence record. To invoke this option, type $\langle D \rangle$ in the option field and press $\langle Enter \rangle$. The message **Record deleted**, will be diaplayed at the bottom of the screen.

9.3.6.5 Text. This option allows you to view and edit any descriptive text associated with a specific sequence. This option operates the same as the Text option discussed in the Family section. Refer to Section 9.2.5.

9.3.6.6 Base Case Update. This option allows you to overwrite all base case (original) data with the current case data. The base case cut sets will be set to the alternate case cut sets; the base case uncertainty data will be set to the current case uncertainty data; and the base case quantile values will be set equal to the quantile values for the current case. The base case minimum cut set upper bound will be initialized to the current case minimum cut set upper bound. WARNING: The original base case data will be lost if this option is (xecuted!

The base case update may be performed on a single sequence, a group of sequences, or on all of the sequences in the current family. To activate this option for a single sequence, type a < B > (Base Case Update) in the option field, highlight the desired sequence, and press < Enter >. To perform a base case update on a group of sequences, mark the desired sequences using the F2, F3, or F4 keys, type a < B > in the option field and press < Enter >. To perform this option on all sequences in the current family, type a < B > in the option field and press < Enter >. A message **Process all records?** (**Y**/**N**) will appear at the bottom of the screen. Type a < Y > to continue the base case update for all of the sequences, or type an <N > to discontinue the update for all sequences.



A warning screen (Figure 182) will then be displayed asking for a (Y/N) confirmation prior to performing the update. To terminate the update, type an <N> in the option field or press the <Esc> key. To initiate the base case update, type a <Y> in the option field. This will cause the current case data to overwrite the base case data. Upon completion of this process, a message **Base case update complete** will be displayed at the bottom of the screen.

	Option [A] Exit / Add / Kodify / Delete / Text Base Case Update / Clear Alternate Case
	mannen Base Case Update mannen and and and and and and and and and an
WARNING	This option will transfer the siternate or temporary values stored in the data base to the base case. The old base case will be lost.
	Are you sure you want to do this? (Y/N) N

Figure 182. Base case update for sequences.

9.3.6.7 Clear Alternate Case. This option clears all alternate case information for the specified sequence(s). To invoke this option, type a < C> in the option field, highlight the desired sequence and press < Enter>. A warning will be displayed (see Figure 183). To continue enter a < Y> and press < Enter>, otherwise enter an < N> and press < Enter> to terminate the process.

9.3.7 Base Case Update

This option operates the same as described in Section 9.3.6.6, except all sequences for the specified event tree are updated.

9.3.8 Clear Alternate Case

This option operates the same as described in Section 9.3.6.7, except the alternate case information for all sequences for the specified event tree is cleared (see Figure 184).



2.13

	Option [C] Exit / Base C	Add / Modify / Delete ase Uprinte / Clear Alt	/ Text ernate Case	
	Clea	r Alternate Case		
WARNING	Thi option will cle s' red in the data b	ar the alternate or te ase.	mporary values	
	Are you sure you wan	t to do this? (Y/N) N		

Figure 183. Clear alternate case for selected sequence(s).

SURRY	Edit Event Trees	
	Dption C Exit / Add / Modify / Delete / Text / Sequences Base Case Update / Clear Alternate Case	
	Clear Alternate Case	-
WARN 1	IG: This option will clear the alternate or temporary values stored in the data base for all the sequences associated with the highlighted/marked event tree.	
	Are you sure you want to do this? (Y/N) N	
1.1.1.1.1		

Figure 184. Clear alternate case for event trees.

9.4 SYstems

This option allows you to modify system data records. To invoke this option, type $\langle Y \rangle$ in t' option field or highlight SYstems and press $\langle Enter \rangle$. The Edit Systems screen inclusive of the system contained in the current family (Figure 185). The modification options are: E. t, Add, Modify, Delet. Text, Base Case Update, and Clear Alternate Case.

				-	
	Op	tion E Exit / Base Ca	Add / Modify / De se Update / Clear	lete / Text Alternate Case	
			- Descrip		
C		CONTAINMENT SP	RAY	C 1 D D manual manual	
CS		CONTAINMENT SY	STEMS		
CV		CORE VULNERABL	E TO CD		
01		HIGH PRESSURE	INJECTION - AUTOM	ATIC	
02		HIGH PRESSURE	INJECTION · MANUA		1
05		HIGH PREDSURE	INJECTION - REP S	EALS ENCY BODATION	
DS		ACCUMULATORS	INVESTION - ENERG	ERCT DUMNITUM	
D6		LOW PRESSURE 1	NJECTION		
F1		INSIDE SPRAY R	ECIRCULATION	The second s	
«Esc>	(F1)	<52>	(13)	5142	4155
	Mar I m	Hark/Clear	Mark/Clear	Mark/Clear	in te
Exit	nelp	TTERT TO J AV L SCHOLT	and the second sec		

Figure 185. Selection screen for system editing.

9.4.1 Exit

This option returns you to the Modify Database menu. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.4.2 Add

This option allows you to add a system record to the current family. To invoke this option, type $\langle A \rangle$ in the option field and press $\langle Enter \rangle$. The Add System screen is shown in Figure 186. The options at this point are Exit and Add.

9.4.2.1 Exit. This option returns you to the Edit Systems screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.



	Option A Exit / Add
Name Description System Code	MinCut Upperbound

Figure 186. Editing screen for adding a system.

9.4.2.2 Add. This option performs the actual add of a new system record. To invoke this option, type $\langle A \rangle$ in the option field, fill in the requested data fields, and press $\langle Enter \rangle$. The only required information for a system add is the name. When complete, you will be returned to the Edit Systems screen with the massage **Record Added** displayed.

9.4.3 Modify

This option allows you to modify a system record. To invoke this option, type $\langle M \rangle$ in the option field, highlight a system name, and press $\langle Enter \rangle$. The Modify System screen is shown in Figure 187. The options at this point are Exit and Modify.

	Option [M] Exit / Modify	
Name CV Description CORE System Code	VULNERABLE TO CD Mincut Upperbound 2.000E-002	

Figure 187. Editing screen for modifying a system.

9.4.3.1 Exit. This option returns you to the Edit Systems screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter_{,} \rangle$ or press the $\langle Esc \rangle$ key.

9.4.3.2 Modify. This option performs the actual modification of the system record. To invoke this option, type < M > in the option field, modify any of the data fields on the Modify System screen, and press < Enter >. When completed, you will be returned to the Edit Systems screen with the message

Record modified displayed.

9.4.4 Delete

This option allows you to delete a system record from the data base. To invoke this option, type < D > in the option field, highlight a system and press < Enter >. The Delete System screen is shown in Figure 188. The options at this point are Exit and Delete.

SURRY	Delete System
	Option E Exit / Delete
Name CV Description CORE V	ULNERABLE TO CD



9.4.4.1 Exit. This option returns you to the Edit Systems screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Es \rangle$ key.

9.4.4.2 Delete. This option performs the actual dele of the system record. To invoke this option, type $\langle D \rangle$ in the option field and press $\langle Enter \rangle$. When complete, you will be returned to the Edit Systems screen with the message **Record deleted** displayed.

9.4.5 Text

This option allows you to view and edit any descriptive text associated with a specific syster. This option operates the same as the Text option discussed in the Family section. Refer to Section 9.2.5.

9.4.6 Base Case Update

This option operates the same as described for sequences in Section 9.3.6.6. Here, the update is performed on an entire system(s).

9.4.7 Clear Alternate Case

This option operates the same as described for sequences in Section 9.3.6.7. Here, the clearing is performed on an entire system(s).



9.5 ENd States

This option allows you to modify the end state data records. To invoke this option, type $\langle N \rangle$ in the option field or highlight ENd States and press $\langle Enter \rangle$. The Edit End State screen, which lists all of the end states belonging to the current family, is shown in Figure 189. The modification options are: Exit, Add, Modify, Delete, and Text.

SURRY	Edit	End Sta	t e	
Oţ	otion E Exit / A	id / Modify / De	lete / Text	
Mame .	and the local division of the state of the s	Descrip	tion	
AINY-YYN	LARGE LOCA - LPI	FAILURE, CHR F	AILS, RCP SEAL CLN	G FAILS
AISY-YYN	LARGE LOCA - LP	FAILS, CHR FAI	LS, RCP SEAL CLNG	FAILS
AIYY-YYN	LARGE LOCA . LPI	R FAILURE, RCP S	EAL COOLING FAILS	
ALNY-YYY	LARGE LOCA - HP	FAILURE, CHR F	AILS	
ALSY-YYY	LARGE LOCA - HP	FAILURE, CHR F	AILS	
ALYY-YYY	LARGE LOCA - HP	FAILURE		
ANNY-NYN	LARGE LOCA - ECH	IS FAILS, CHR FA	ILS, RWST FAILS, R	CP]
ANNY - NYY	LARGE LOCA - ECI	CS FAILS, CHR FA	ILS, RWST FAILS	
ANNY-YYN	LARGE LOCA - EC	IS FAILS, CHR FA	ILS, RCP SEAF COOL	ING FAILS
ANSY - YYN	LARGE LOCA - ECI	CS FAILURE, CHR	FAILS, RCP SEAL CL	NG FAILS
Esc> <f1></f1>	<f2></f2>	<f3></f3>	<f4></f4>	<f5></f5>
xit Help	Mark/Clear Item	Mark/Clear All	Mark/Clear Range	Locate

Figure 189. Selection screen for end state editing.

9.5.1 Exit

This option returns you to the Modify Database menu. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.5.2 Add

This option allows you to add an end state record to the current family. To invoke $\pm i$ option, type <A> in the option field and press < Enter>. The Add End State screen is shown in Figure 190. The options at this point are Exit and Add.

9.5.2.1 Exit. This option returns you to the Edit End State screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.5.2.2 Add. This option performs the actual add of a new end state record. To invoke this option, type < A > in the option field, fill in the requested data fields, and press < Enter >. The only required information for an end state add is the name. When complete, yc: will be returned to the Edit

	SURRY Add End State	
	Option A Exit / Add	
1	Description	

Figure 190. Editing screen for adding an end state.

End State screen with the message Record added displayed.

9.5.3 Modify

This option allows you to modify an end state record. To invoke this option, type $\langle M \rangle$ in the option field, highlight an end state name, and press $\langle Enter \rangle$. The Modify End State screen is shown in Figure 191. The options at this point are Exit and Modify.

SHRRY		Hodify	End	State]	
		Option M	Exit / M	odify		
Name	1LYYYY	N		a lana in Parata an and Ana ana		
Descr	iption MEDIUM	LOCA - HPI FA	LURE - RWS	T AND LP1 S	UCCESS	

Figure 191. Editing screen for modifying an end state.

9.5.3.1 Exit. This option returns you to the Edit End State screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.5.3.2 Modify. This option performs the actual modification of the end state record. To invoke this option, type < M > in the option field, modify any of the data fields on the Modify End State screen, and press < Enter >. When complete, you will be returned to the Edit End State screen with the message **Record modified** displayed at the bottom of the screen.

9-19

9.5.4 Delete

This option allows you to delete an end state record from the data base. To invoke this option,



type < D > in the option field, highlight an end state and press < Enter >. The Delete End State screen is shown in Figure 192. The options at this point are Exit and Delete.

SURRY	Delete End State
	Option E Exit / Delete
Name 11YYY	YYN IM LOCA - LPR FAILURE - RCP SEAL COOLING FAILS



9.5.4.1 Exit. This option returns you to the Edit End State screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.5.4.2 Delete. This option performs the actual deletion of the end state record. To invoke this option, type $\langle D \rangle$ in the option field and press $\langle Enter \rangle$. When complete, you will be returned to the Edit End State screen with the message **Record deleted** displayed at the bottom of the screen.

9.5.5 Text

This option allows you to view and edit any descriptive text associated with a specific end state. This option operates the same as the Text option discussed in the Family section. Refer to Section 9.2.5.

9.6 Basic Events

This option allows you to modify the basic event data records. To invoke this option, type $\langle B \rangle$ in the option field or highlight Basic Events and press $\langle Em_{ef} \rangle$. The Edit Events screen, listing all of the basic events belonging to the current family, is displayed (Figure 193). The modification options are: Exit, Add, Modify, Delete, an⁴ Remove Unused Events.

9.6.1 Exit

This option returns you to the Modify Database menu. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.6.2 Add

This option allows you to add a basic event record to the current family. To invoke this option, type < A > in the option field and press < Enter >. The Add Event screen is shown in Figure 194. The options at this point are Exit and Add.









Figure 194. Editing screen for adding a basic event.

9.6.2.1 Exit. This option returns you to the Edit Events screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.6.2.2 Add. This option performs the actual add of a new basic event record. To invoke this option, type < A > in the option field, fill in the requested data fields, and press < Enter >. The only required information for a basic event add is the name. When complete, you will be returned to the Edit Event screen with the message **Record Added** displayed at the bottom of the screen.

9.6.3 Modify

This option allows you to modify a basic event record. To invoke this option, type < M > in the option field, highlight a basic event name, and press < Enter >. The Modify Basic Event screen is shown in Figure 195. The options at this point are Exit and Modify.

- Family SURRY Modify Event Option [M] Exit / Modify www.event Attributes ---Names Comp 1d Sys Train Type F/Mode Location Init? <P>ACC-CKV-FT-CV109 CV109 ACC CKV FT N <A>ACC-CKV-FT-CV109 <G>ACCCKVCV109 1 2 3 4 5 6 7 8 9 10 11 12 13 14 15 16 Class Attributes N N N N N N N N N N N NNN Description CHECK VALVE CV109 FAILS TO OPEN - Uncertainty Data -------- Failure Data ---Distribution Type 1 Commission Calculation Type ---(press <F1> for list of types) (press <F1> for list of types) Value -----> 3.000E+000 Correlation Class ------ 31 +0.000E+000 Lambda management Process Flag -Flag . Mission Time +0.000E+000

Figure 195. Editing screen for modifying a basic event.

9.6.3.1 Exit. This option returns you to the Edit Events screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.6.3.2 Modify. This option performs the actual modification of the basic event record. To invoke this option, type $\langle M \rangle$ in the option field, modify any of the data fields on the Modify Event screen, and press $\langle Enter \rangle$. When complete, you will be returned to the Edit Events screen with the message **Record Modified** displayed at the bottom of the screen.

9.6.4 Delete

This option allows you to delete a basic event record from the current family. To invoke this

option, type $\langle D \rangle$ in the option field, highlight a basic event and press $\langle Enter \rangle$. The Delete Event screen is shown in Figure 196. The options at this point are Exit and Delete.

SURRY	L	D e	le	t -2		E Y	6	n t	1							
		Opti	on f	E.	Exi	1	De	let	e							
			1000			£ 6										
Names <p>ACC-CKV-FT-CV128 <a>ACC-CKV-FT-CV128</p>	Comp	1 d	Ever Sys ACC	nt A T	ttr raii	i bu	Ty Ck	pe V	1	/Mc F1	de	L	001	atic	2n	Init

Figure 196. Editing screen for deleting a basic event.

9.6.4.1 Exit. This option returns you to the Edit Events screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.6.4.2 Delete. This option performs the actual deletion of the basic event record. To invoke this option, type <D> in the option field and press <Enter>. When complete, you will be returned to the Edit Events screen with the message Record deleted displayed at the bottom of the screen.

9.6.5 Remove Unused Event

This option looks at all references to an event in the current family and deletes any events that are not referenced by anything. When you invoke this option, a warning screen will be displayed telling you that all records that are marked as unused will be deleted. At this prompt, enter a $\langle Y \rangle$ to continue the deletion process, or enter an $\langle N \rangle$ to terminate without deleting the unused events (Figure 197).

9.7 Attributes

This option allows you to edit the six attributes in the current family (Figure 198): Systems, Locations, Failure Modes, Class Attributes, Basic Event Types, and Trains. The options available within each of these attributes are: Exit, Add, Modify, and Delete.

The Edit Attributes screen shown in Figure 198 and succeeding screens (Figure 199 - Figure 202) are consistent throughout the attributes option for each of the six categories. Because this option operates the same for all six attribute categories, a generic write-up is presented.





SURRY		Edit	Events		
O ACP-BA ACP-BA ACP-BA ACP-CR	ption [R] 1 8 m e C-ST-VB11 C-ST-VB111 C-ST-VB111 B-CD-14M1	VITAL BUS 11 BU VITAL BUS 11 BU VITAL BUS 111 I VITAL BUS 11V I AC CIRCUIT BRE	dify / Delete / Re D e s c r i p t JSWORK FAILURE BUSWORK FAILURE BUSWORK FAILURE AKER 1441 TRANSFER	move Unused Event	s
ACP-CR ACP-CR ACP-CR ACP-CR ACP-CR ACP-CR ACP-CR	B-CO-14H13 5-CO-14H14 8-CO-14H15 8-CO-14J1 8-CO-14J1 8-CO-14J11 B-CO-14J14	All record unused	W A R N I N G s that are marked will be deleted. h to proceed? N Y/	as N N N N N	
<esc> Exit</esc>	NOTE: A <f1> Help</f1>	<f2> Mark/Clear Item</f2>	<f3> Mark/Clear All</f3>	sed event. <f4> Mark/Clear Range</f4>	«F5» Locate





Figure 198. Attribute selection for editing.

è?



To invoke this option, type $\langle A \rangle$ in the option field or highlight Attributes and press $\langle Enter \rangle$. This will display a list of attribute categories (see Figure 198). Highlighting a category or typing a category's highlighted letter and pressing $\langle Enter \rangle$ displays the list of attributes available under the selected category. For each attribute category, the following options are available: Exit, Add, Modify, and Delete.

Brones accor		d barren nonerissi cantan		examined homeospectrum	
		option [E] EXI	t / AGG / MOGITY	Delete	
-	N B (T) C assessment		escripti	0 17	Pression
	The NAME col	umn will contain	the list of syst	ems, failure modes	, etc.
	available fo	r the current far	mily.		1000
	available fo The DESCRIPT each attribu	r the current fa 10% column will i te displayed.	mily. contain the corre	sponding descripti	on for
	available fo The DESCRIPT each attribu	r the current far 10% column will i te displayed.	mily. contain the corre	sponding descripti	on for
	available fo The DESCRIPT each attribu	r the current far ICA column will n te displayed.	mily. contain the corre	sponding descripti	on for
	available fo The DESCRIPT each attribu	r the current fam ICW column will n te displayed.	mily. contain the corre	sponding descripti	on for
(Fac)	available fo The DESCRIPT each attribu	r the current far ION column will n te displayed.	mily, contain the corre	sponding descripti	on for



SURRY	Add Attribute	ATTRIBUTE NAME
	Option (A) Exit / Add	
Attribute Name		

Figure 200. Editing screen for adding an attribute.

SURRY	Modify Attribute ATTRIBUTE NAME
	Option (M) Exit / Modify
Attribute Name Description	

Figure 201. Editing screen for modifying an attribute.

SURRY	Delete Attribute ATTRIBUTE NAME
	Option [E] Exit / Delete
Attribute Name	

Figure 202. Editing screen for deleting an attribute.

9.7.1 Exit

This option returns you to the Modify Database menu. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.7.2 Add

The add option allows you to add a new attribute record to the current family. To invoke this option, type <A> in the option field and press < Enter>. The Add Attribute screen is shown in Figure 200. The options at this point are Exit and Add.

9.7.2.1 Exit. This option returns you to the Edit Attributes screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.7.2.2 Add. This option performs the actual add of a new attribute record. To invoke this option, type < A > in the option field, fill in the requested data fields, and press < Enter >. The only required information for an attribute add is the name. When complete, you will be returned to the Edit Attributes screen with the message **Record added** displayed at the bottom of the screen.



9.7.3 Modify

This option allows you to modify an attribute record. To invoke this option, type < M > in the option field, highlight an attribute name, and press < Enter >. The Modify Attribute screen is shown in Figure 201. The options at this point are Exit and Modify.

9.7.3.1 Exit. This option returns you to the Edit Attributes screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.7.3.2 Modify. This option performs the actual modification of the attribute record. To invoke this option, type $\langle M \rangle$ in the option field, modify any of the data fields on the Modify Attribute screen, and press $\langle Enter \rangle$. When complete, you will be returned to the Edit Attributes screen with the message **Record modified** displayed at the bottom of the screen.

9.7.4 Delete

This option allows you to delete an attribute record from the current family. To invoke this option, type $\langle D \rangle$ in the option field, highlight an attribute and press $\langle Enter \rangle$. The Delete Attribute screen is shown in Figure 202. The options at this point are Exit and Delete.

9.7.4.1 Exit. This option returns you to the Edit Attributes screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.7.4.2 Delete. This option performs the actual deletion of the attribute record. To invoke this option, type $\langle D \rangle$ in the option field and press $\langle Enter \rangle$. When complete, you will be returned to the Edit Attributes screen with the message **Record deleted** displayed at the bottom of the screen.

9.8 GaTes

This option allows you to modify gate records. To invoke this option, type $\langle T \rangle$ in the option field or highlight GaTes and press $\langle Enter \rangle$. The Edit Gates screen, listing all of the gates belonging to the current family, is displayed (Figure 203). The modification options are: Exit, Add, Modify, and Delete.

9.8.1 Exit

This option returns you to the Modify Database menu. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.8.2 Add

This option allows you to add a gate record to the current family. To invoke this option, type <A> in the option field and press <Enter>. The Add Gate screen is shown in Figure 204. The options at this point are Exit and Add.



		Edit	Getes		
	Opt	ion E Exit / A	Add / Modify / Del	ete	
GATEO	a m e		Descript	t i o n	
GATE1					
GATE2	<f1></f1>	<f2></f2>	<f3></f3>	<f4></f4>	«F5»

Figure 203. Selection screen for gate editing.

	Option [A] Exit / Add	
Name Description Type		
Type		

Figure 204. Editing screen for adding a gate.

150

9.8.2.1 Exit. This option returns you to the Edit Gates screen. To invoke this option, type <E> in the option field and press <Enter>, or press the <Esc> key.

9.8.2.2 Add. This option performs the actual add of a new gate record. To invoke this option, type < A > in the option field, fill in the requested data fields, and press < Enter >. The only required information for a gate add is the name and type fields. When complete, you will be returned to the Edit Gates screen with the message **Record Added** displayed at the bottom of the screen.

9.8.3 Modify

This option allows you to modify a gate record. To invoke this option, type < M > in the option field, highlight a gate name, and press < Enter >. The Modify Gate screen is shown in Figure 205. The options at this point are Exit and Modify.

Figure 205. Editing screen for modifying a gate.

9.8.3.1 Exit. This option returns you to the Edit Gates screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.8.3.2 Modify. This option performs the actual modification of the gate record. To invoke this option, type < M > in the option field, modify any of the data fields on the Modify Gate screen, and press < Enter >. When complete, you will be returned to the Edit Gates screen with the message **Record** Modified displayed at the bottom of the screen.

9.8.4 Delete

This option allows you to delete a gate record from the current family. To invoke this option,



type $\langle D \rangle$ in the option field, highlight the gate to be deleted, and press $\langle Enter \rangle$. The Delete Gate screen is shown in Figure 206. The options at this point are Exit and Delete.

	Option [E] Exit / Delete	
Description	123GATE	

Figure 206. Editing screen for deleting a gate.

9.8.4.1 Exit. This option returns you to the Edit Gates screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.8.4.2 Delete. This option performs the actual deletion of the gate record. To invoke this option, type a < D > in the option field and press < Enter >. When complete, you will be returned to the Edit Gates screen with the message **Record deleted** displayed at the bottom of the screen.

9.9 Graphics

This option allows you to edit three categories of graphics data records (Figure 207). They are: Fault Tree Graphics, Event Tree Graphics, and P&ID Graphics. The options available within each of these categories are: Exit, Add, Modify, and Delete.

The Edit Graphics screen shown in Figure 208 and succeeding screens Figure 209 through Figure 211 are consistent throughout the graphics option for each of the three categories.

To invoke this option, type $\langle G \rangle$ in the option field or highlight Graphics and press $\langle Enter \rangle$. This will display a list of graphics categories. Highlighting a category or typing a category's highlighted letter and pressing $\langle Enter \rangle$ displays the list of graphics records available under the selected category.









SURRY	Edit G	raphics		
	Option [E Exit /	Add / Modity / D	elete	
Name -		Descript:	0 n	
4KV1H	FAILURE OF 4KV AC	BUS 1H	()	FT 4KV1H)
4KV1J	FAILURE OF 4KV AC	BUS 1J	(1	FT 4KV1J)
ACC4	INSUFFICIENT FLOW	THRU PIPE SEGMEN	T PS120 (GATE AL	CC1/FT D5
ACCS	INSUFFICIENT FLOW	THRU PIPE SEGMEN	T PS121 (GATE A	CC3/FT D5
ACC6	INSUFFICIENT FLOW	THRU PIPE SEGMEN	T PS122 (GATE AD	CC3/FT D5
AFWI	INSUFF FLOW THRU I	PIPE SEG. P\$93 TO	SG A (GATE L2-2	2/FT L2)
AFW13	INSUFFICIENT FLOW	THRU PIPE SEGMEN	T PS83 (GATE AF)	14/FT L)
AFW13A	INSUFFICIENT FLOW	THRU PIPE SEGMEN	T PS83 (GATE AF	J7/FT AFW
AFW138	INSUFFICIENT FLOW	THRU PIPE SEGMEN	T PS83 (GATE AF)	19/FT L3)
AFW14	INSUFFICIENT FLOW	THRU PIPE SEGMEN	T PS84 (GATE AF)	J8/FT L)
<esc> <f1></f1></esc>	<f2></f2>	<f3></f3>	<f4></f4>	<f5></f5>
Exit Help	Mark/Clear	Mark/Clear	Mark/Clear	Locate





SURRY	Add Graphic Picture
	Analyze (a) and a set
	uption A EXIT / Add
Name Description	



	Option [M]	Exit / Modify	
Name 4KV1J	100011420129201410101999999999		

Figure 210. Editing screen for modifying a graphics picture.

	Option E Exit / Delete
Name AFW1 Description	





9.9.1 Exit

This option returns you to the Edit Graphics main menu. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.9.2 Fault Tree Graphics

This option allows you to modify the Fault Tree Graphics records. To invoke this option, type $\langle F \rangle$ in the option field or highlight Fault Tree Graphics and press $\langle Enter \rangle$. The screen shown in Figure 208, listing the fault tree graphics (pictures), will be displayed. The option available are Exit, Add, Modify, and Delete.

9.9.2.1 Exit. This option returns you to the Edit Graphics main menu. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.9.2.2 Add. This option allows you to add a graphics picture record to the data base. To invoke this option, type < A > in the option field and press < Enter >. The Add Graphics Picture screen is shown in Figure 209. The options at this point are Exit and Add.

9.9.2.2.1 Exit—This option returns you to the Edit Graphics screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.9.2.2.2 Add—This option performs the actual add of a new graphics picture record. To invoke this option, type < A > in the option field, fill in the requested data fields, and press <Enter >. The only required information for a graphics picture add is the name. When complete, you are returned to the Edit Graphics screen with the message **Record added** displayed at the bottom of the screen.

9.9.2.3 Modify. This option allows you to modify a graphics picture record. To invoke this option, type $\langle M \rangle$ in the option field, highlight a graphics picture name, and press $\langle Enter \rangle$. The Modify Graphics Picture screen is shown in Figure 210. The options at this point are Exit and Modify.

9.9.2.3.1 Exit—This option returns you to the Edit Graphics screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.9.2.3.2 Modify—This option performs the actual modification of the graphics picture record. To invoke this option, type $\langle M \rangle$ in the option field, modify any of the data fields on the Modify Graphics Picture screen, and press $\langle Enter \rangle$. When complete, you will be returned to the Edit Graphics screen with the message **Record modified** displayed at the bottom of the screen.

9.9.2.4 Delete. This option allows you to delete a graphics picture from the data base. To invoke this option, type $\langle D \rangle$ in the option field, highlight a graphics picture and press $\langle \exists nter \rangle$. The Delete Graphics Picture screen is shown in Figure 211. The options at this point are Exit and Delete.

9.9.2.4.1 Exit—This option returns you to the Edit Graphics screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.9.2.4.2 Delete—This option performs the actual deletion of the graphics picture record. To invoke this option, type $\langle D \rangle$ in the option field and press $\langle Enter \rangle$. When complete, you will be returned to the Edit graphics screen with the message **Record deleted** displayed at the bottom of the screen.

9.9.3 Event Tree Graphics

This option allows you to modify the Event Tree Graphics records. To invoke this option, type $\langle V \rangle$ in the option field or highlight Event Tree Graphics and press $\langle Enter \rangle$. The screen shown in Figure 208, listing the event tree graphics pictures, will be displayed. The option available are Exit, Add, Modify, and Delete.

9.9.3.1 Exit. This option returns you to the Edit Graphics menu. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$ or press the $\langle Esc \rangle$ key.

9.9.3.2 Add. This option allows you to add a graphics picture record to the data base. To invoke this option, type < A > in the option field and press < Enter >. The Add Graphic Picture screen is shown in Figure 209. The options at this point are Exit and Add.

9.9.3.2.1 Exit—This option returns you to the Edit Graphics screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.9.3.2.2 Add—This option performs the actual add of a new graphics picture record. To invoke this option, type $\langle A \rangle$ in the option field, fill in the requested data fields, and press $\langle \text{Enter} \rangle$. The only required information for a graphics add is the name. When complete, you will be returned to the Edit Graphics screen with the message **Record added** displayed at the bottom of the screen.

9.9.3.3 Modify. This option allows you to modify a graphics picture record. To invoke this option, type $\langle M \rangle$ in the option field, highlight a graphics picture name, and press $\langle Enter \rangle$. The Modify Graphic Picture screen is shown in Figure 210. The options at this point are Exit and Modify.

9.9.3.3.1 Exit—This option returns you to the Edit Graphics screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.9.3.3.2 Modify—This option performs the actual modification of the graphics picture record. To invoke this option, type $\langle M \rangle$ in the option field, modify any of the data fields on the Modify Graphic Picture screen, and press $\langle Enter \rangle$. When complete, you will be returned to the Edit Graphics screen with the message **Record modified** displayed at the bottom of the screen.

9.9.3.4 Delete. This option allows you to delete a graphics picture from the data base. To invoke this option, type $\langle D \rangle$ in the option field, highlight a graphics picture and press $\langle Enter \rangle$. The Delete Graphic Picture screen is shown in Figure 211. The options at this point are Exit and Delete.

9.9.3.4.1 Exit—This option returns you to the Edit Graphics screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9-34

9.9.3.4.2 Delete—This option performs the actual deletion of the graphics picture record. To invoke this option, type < D> in the option field and press < Enter>. When complete, you will be returned to the Edit Graphics screen with the message **Record deleted** displayed at the bottom of the screen.

9.9.4 P&ID Graphics

This option allows you to modify the P&ID Graphics records. To invoke this option, type $\langle P \rangle$ in the option field or highlight P&ID Graphics and press $\langle Enter \rangle$. The screen shown in Figure 208, listing the P&ID graphics pictures, will be displayed. The option available are Exit, Add, Modify, and Delete.

9.9.4.1 Exit. This option returns you to the Edit Graphics main menu. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.9.4.2 Add. This option allows you to add a graphics picture record to the data base. To invoke this option, type $\langle A \rangle$ in the option field and press $\langle Enter \rangle$. The Add Graphic Picture screen is shown in Figure 209. The options at this point are Exit and Add.

9.9.4.2.1 Exit—This option returns you to the Edit Graphics screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.9.4.2.2 Add—This option performs the actual add of a new graphics picture record. To invoke this option, type $\langle A \rangle$ in the option field, fill in the requested data fields, and press $\langle Enter \rangle$. The only required information for a graphic add is the name. When complete, you will be returned to the Edit Graphics screen with the message **Record added** displayed at the bottom of the screen.

9.9.4.3 Modify. This option allows you to modify a graphics picture record. To invoke this option, type $\langle M \rangle$ in the option field, highlight a graphics picture name, and press $\langle Enter \rangle$. The Modify Graphic Picture screen is shown in Figure 210. The options at this point are Exit and Modify.

9.9.4.3.1 Exit—This option returns you to the Edit Graphics screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.9.4.3.2 Modify—This option performs the actual modification of the graphics picture record. To invoke this option, type $\langle M \rangle$ in the option field, modify any of the data fields on the Modify Graphic Picture screen, and press $\langle Enter \rangle$. When complete, you will be returned to the Edit Graphics screen with the message **Record modified** displayed at the bottom of the screen.

9.9.4.4 Delete. This option allows you to delete a graphics picture from the data base. To invoke this option, type $\langle D \rangle$ in the option field, bighlight a graphics picture and press $\langle Enter \rangle$. The Delete Graphic Picture screen is shown in Figure 211. The options at this point are Exit and Delete.

9.9.4.4.1 Exit—This option returns you to the Edit Graphics screen. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

9.9.4.4.2 Delete—This option performs the actual deletion of the graphics picture record. To invoke this option, type < D > in the option field and press < Enter >. When complete, you will be returned to the Edit Graphics screen with the t. ssage **Record deleted** displayed at the bottom of the screen.

9.10 Histograms

This option allows you to create, modify, and delete user-defined histograms. This is a useful option for allowing you to input your own distribution for a variable that can not be expressed with one of the predefined distribution types. The Edit Histograms screen (Figure 212) displays the names of all the currently existing histograms with their associated format type. As shown, four options are available: Exit, Add, Modify, and Delete.

To activate this option type an $\langle H \rangle$ (Histograms) in the option field or highlight Histograms and press $\langle Enter \rangle$. Figure 212 will be displayed.

SURRY	Edit	Histogram	n s	
	Option [E] Ex	it / Add / Modify	/ Delete	
	m e	Descr	iption	
1 P E1				
10 P H10				
11 P H11				1.1.1.1.1.1.1.1
12 P H12				
13 P H13				
14 P H14				
15 P H15				
16 P H16				
17 P H17				
18 9 110				
<esc> <f1></f1></esc>	<12>	<f3></f3>	<\$4>	<f5></f5>
Exit Help	Mark/Clear	Mark/Clear	Mark/Clear Range	Locate

Figure 212. Edit histograms menu.

9.10.1 Exit

This option returns you to the Modify Database screen. To invoke this option, type an $\langle E \rangle$ (Exit) in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.



9.10.2 Add Histograms

This option allows you to create a user-defined distribution type. To add a new histogram to the data base, type an <A> (Add) in the option field and press < Enter >. At this point, Figure 213 is displayed and you are given the choice of adding the histogram data in either a percentage format or range format.

	Option Perce	intege Format / Rang	e format	
	K & E C annicipation and	energy and P B S C .	Stion	
1 P E1 10 P H10				
11 P H11				
13 P H13				
14 P H14 15 P H15				ST. 11
16 P H16				
18 P H18				
And the Real Property lies and the real Property	The statement of the second	And the second se	A DESCRIPTION OF A DESC	The second second second second

Figure 213. Select percentage or range format for the histogram.

If you wish to add a percentage histogram to the data base, enter a $\langle P \rangle$ (Percentage) in the option field. Entering a $\langle P \rangle$ in the option field will cause a Percentage Format Histogram screen to appear. From this screen you should ty in a name and a description for the new histogram. Enter the percentages for the histogram along with the corresponding probabilities. Figure 214 demonstrates how to enter a percentage histogram, given that 15% of the data points have a probability of 0.04, 46% of the data points have a probability of 0.02, and the remaining 3% of the data points have a probability of 0.8. "The sum of the percentages entered must total 100%, in order for the histogram to be accepted as a valid percentage histogram (Figure 215). In the upper right-hand area of the screen is a box that shows the current sum of the percentages that have been input and the remaining percentage needed to reach the 100% total.



SUKKT		Add His	togra	111			
		Option A E	xit / Add Format =		Percen Percen	t Sum f Left	97.00 3.00
Name Descript	PERCENT ion percent	HIST age histogram ex	ample				
Area 1 2 3 4 5 6 7 8 9 10	Probability 4.000E-D02 1.200E-D01 2.000E-002 E E E E E E E E	Percent 15.00 46.00 36.00	Area 11 12 13 14 15 16 17 18 20	Probab	i (i ty E E E E	Fercer	1









Another way to input a histogram \odot to use the range format. To add a range histogram to the octa base, enter an $\langle R \rangle$ (Range \rangle in the option field of the Add Histogram screen. This action will bring up a Range Format Histogram screen (Figure 216). On this screen, type in a name and description for the range histogram. Then, enter the starting probability point, the ending probability point, and the height associated with the first bin of the histogram. Next, for each successive bin of the histogram, an ending probability point and a height should be entered. There is a maximum of 20 bins allowed for each range histogram. Figure 216 is an example of inputting a range histogram whose data points lie on the closed interval of 0.0 and 1.0. The height associated with the data points on the sub-interval of 0.0 to 0.2 is 10.0 (Bin 1), the height for the sub-interval of 0.2 to 0.6 is 70.0 (Bin 2), the height for the sub-interval of 0.6 to 0.8 is 20.0 (Bin 3), and the height for the last sub-interval of 0.8 to 1.0 is 5.0 (Bin 4).

SURRY	Add His	tog	a m	
	Option A E	kit / Add		
anna ann an channa tha 'art a tha bar ta tha an	Range Fi	ormat	anne e certene con de constant e constant	1
Name RANGE	(151			
Starting Prob +0.000	le of a range hist DE+000	ogram		1
			ward the second s	
Bin End Prob	Height	610	End Prob Height	- 1
1 2002-00	1 1.000E+001	42	·····	
8 0000 00 8 0000 00	2 0006+001	12	entered and an entered at the	- 1
4 1.000F+00	5 000F+000	34	and a free and a second free for	
5	e areas France	15	A A A A A A A A A A A A A A A A A A A	
6 month in		16	Andrew Frank Andrew Errer	-1
7 in mere	A. TANKA AND AREA	17	AN ANTENNA COMPANY FROM	
8 Samara Ease	e and a family and	18	and the same and the same	- 1
9E	a and a second sec	19	1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.	
		10.00		



The code calculates the midpoint of each bin, finds the area of each bin, and normalizes each area so the sum of the areas equals 1.0. The midpoint is the probability for each bin and the normalized area corresponds to the percent in the Percent Histogram format. The basic event mean probability should correspond to the mean of the histogram.

Once you by e typed in the histogram data, enter an <A> in the option field and press < Enter> to save the newly created histogram. When complete, you will be returned to the Edit Histograms screen with the message **Record Added** displayed at the bottom of the screen. However, if you wish to exit the Add process without saving the new histogram, enter an <E> in the option field or press the < Esc> key.

9.10.3 Modify Histograms

To modify a currently existing histogram, type an < M > (Modify) in the option field of the Edit Histograms screen, highlight the desired histogram, and press < Enter >.

If you selected to modify a percentage histogram, then the histogram data will be displayed in percentage format. You may make modifications to the nistogram's name, description, or any of the probabilities or percentages. The percentages must still total 100% before it will be accepted as a valid percentage histogram. Figure 217 demonstrates the screen for modifying histograms in percentage format.

SURRY		Modify	Histo	gram	
		Option M	Exit / Mod	lify Percer Percer	it Sum 100.00 it Left .00
erector inte		minimum Percenta	ige Format ~	or no more resident to the second	
Descript	tion percent	age histogram	example		
Area	Probability	Percent	Area	Probability	Percent
	4.000E-002	15.00	11		
2	2 0005-003	90,00	12		
4	B 000E-002	3.00	14		
5	FILLER ELTER	11111111	15		*******
6	A E		16		
7	E	*******	17		******
8	****	$\mathbf{u} \leftarrow \mathbf{v}(\mathbf{u}) + \mathbf{v}(\mathbf{u})$	18	······································	******
9	*****	******	19		******
10		******	20	"annie fanne."	

Figure 217. Modify a percentage histogram.

If you selected , modify a range histogram, then the histogram data will be displayed in range format. You may change the histogram's name, description, and any of the bins' starting points, ending points or probabilities. Figure 218 demonstrates the screen for modifying histograms in range format.

If you wish to save the modifications made to the selected histogram, type an $\langle M \rangle$ (Modif⁽⁾) in the option field of the Modify Histograms screen and press $\langle Enter \rangle$. When complete, you will be returned to the Edit Histograms screen with the message **Record modified** displayed at the bottom of the screen. If you wish to evit this screen without modifying the histogram, enter an $\langle E \rangle$ in the option field and press $\langle Enter \rangle$ or press the $\langle Esc \rangle$ key.

SURRY		Modify	Hist	ogram	
		Option [M]	Exit / Mo	dify	
		Range	Format		
	PANOPU1				
Descript	ion example	of a canno bir	togram		
Starting	Prob +0.000E	+000	er ogram		
Bin	End Prob	Height	Gin	End Prob	Height
1	2.000E-001	1.000E+001	11	(X,Y,Y,Y,Y,Y,Y,Y,Y,Y,Y,Y,Y,Y,Y,Y,Y,Y,Y,Y	*****E
2	6.000E-001	7.000E+001	12	******E****	1. 1. 1. 1. 1. 1. E. 1. 1. 1. 1.
3	8.000E-001	2.000E+001	15		
4	1.0008+000	5.0008+000	14	***********	******
2		seres. E. se	15	******E****	A A A A A A A A A A A A A A A A A A A
0			10		
			1.0	and a second second	
0			10		
		and a start of the	1.24	and a set for search and a set of the second s	and a second second

Figure 218. Modify a range histogram.

9.10.4 Delete Histograms

To delete an existing histogram from the data base, type a $\langle D \rangle$ (Delete) in the option field of the Edit Histograms screen, highlight the histogram to be deleted, and press $\langle Enter \rangle$.

This action vill result in displaying the selected histogram in the appropriate format on the Delete Histograms screen. To delete the histogram, type a $\langle D \rangle$ (Delete) in the option field and press $\langle Enter \rangle$. When complete, you will be returned to the Edit Histograms screen with the message **Record Deleted** displayed at the bottom of the screen. To exit the Delete Histograms screen without deleting the histogram being displayed, press the $\langle Esc \rangle$ key or enter an $\langle E \rangle$ (Exit) in the option field and press $\langle Enter \rangle$. Figure 219 and Figure 220 show examples of the Delete Histograms screen for percentage histograms and range histograms, respectively.



SURRY	land a state of the state of th	DELEE Commencement (April 10	HISTO	g r a m		
		Option [E]	Exit / Del	ete		
-		Percent	age Format a	and the second	Alexandro de constituir barrantes destas as	
Descript	ion percent	HIST Age histogram	esuir-Le			
Area 1	Probab [[ty 4.000E-002	Percent 15.00	Aren 15	Probability	Percent	
3	2.0008-002	46.00 30.00	12	**************************************	2+++x++	
5		1179191 1179191 22110235	15	**************************************	*****	
7 8	****** * ****	******	17 18	$\begin{array}{c} \mathbf{x} \in \mathcal{T} \times \mathcal{T} \in \mathcal{T} \\ \mathbf{y} \in \mathcal{T} \times \mathcal{T} \in \mathcal{T} \\ \mathbf{y} \in \mathcal{T} \times \mathcal{T} \times \mathcal{T} \in \mathcal{T} \\ \mathbf{y} \in \mathcal{T} \times \mathcal{T} \times \mathcal{T} \\ \mathbf{y} \in \mathcal{T} \\ \mathbf{y} \in \mathcal{T} \times \mathcal{T} \\ \mathbf{y} \in \mathcal{T} \\ \mathbf{y} \in \mathcal{T} \times \mathcal{T} \\ \mathbf{y} \in \mathcal{T} \\ $	******** ******	÷
	******		12/	************	$(X,Y) \in \mathcal{M}, (Y,X,Y)$	







10.0 REPORT ON DATA BASE

This option allows you to obtain information about the selected family. Reports are available for family, basic events, attributes, systems, event trees, sequences, end states, and user information. To invoke this option, type an < R > or highlight REPORTS on the IRRAS main menu and press < Enter >. The REPORTS main menu is shown in Figure 221. Each report is discussed in the following paragraphs.

Data Types		
Exit		
Family Basic Event Attributes SYstem EVent Tree Sequence ENd States User Info.		



The following special function keys are available:

Esc	Escape exits the Family Reports option and returns you to the IRRAS Reports meral.
<f1></f1>	Help briefly explains the function of a field and may show examples of data entered.
< F2 >	Mark Family tags items for use in the selected option.
<f3></f3>	Mark All removes the mark (*) from the listed items. If no items are marked, this option will mark all of the items.
< F4 >	Mark Range quickly tags large numbers of items for processing.
< F5 >	Locate will display a blank field in the center of the screen, and a message Please enter the name to locate will appear. The user should enter the name to be located and then press < Enter >. This feature will place the highlight on the located name. If the required name is not found, then the next name in alphabetical order will be highlighted.



These function keys are available in all report options.

10.1 Exit

This option returns you to the IRRAS main menu. To invoke this option, type $\langle E \rangle$ in the option field or highlight Exit and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

10.2 Family

This option allows you to generate a family summary report listing all the families and associated descriptions contained in the current data base or a report containing any descriptive text associated with the family. When you invoke this option, Figure 222 will be displayed. As shown, two options are available: Summary and Text. When you invoke the family report option by pressing <Enter>, Figure 223 will be displayed. This screen lists all the families currently contained in the data base.



Figure 222. Family summary option.

10.2.1 Exit

This option returns you to the IRRAS reports menu. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$.

10.2.2 Summary

This option generates a summary report listing all the families and associated descriptions contained in the data base. When you invoke this option, Figure 224 will be displayed. On this screen you must specify the output destination for the report. The following output destinations are provided:
Family + SURRY J. Y Reports 01.1 / Exit / Summary / Text - Family Nome www Directory wanted Description -DEREI DEMO Demonstration sample fomily LEARN LEARN Sample family of data for the SARA 4.0 Tutori LEARN LEARN? Sample family of data for the ERRA 6.0 Tutori PRADATA, 81 PB.ATAGASS SURRY UNIT 1 RELEASE DATE 04-91-91 SURRY SURRY *Esc> 化产生分 <F2> KF3× 4854 1845 Exit Help Mark Family Mark All Mark Range Locate

Figure 223. Select family and report option menu.

CON		Sends the report to the screen.
PRN	×	Sends the report to the printer.
blank	1.1	Terminates the option and returns you to the Reports Main Menu without generating the report.
< Esc >		Terminates the process and returns you to the main menu (same as blank).
other	*	Writes the report to the specified file name.
NOTE:	These	output options are available throughout the Report module.

10.2.3 Text

This option allows you to view and edit any descriptive text associated with a specific falling. To invoke this option, enter a < T > in the option field, nighlight or mark a family(s), and press < Enter >. The output destination screen will appear. Select the desired destination and press < Enter >. The report contains all descriptive text associated with the selected family.

10.3 Basic Event

The basic event report option allows you to generate an overview, probabilities, uncertainty data, and cross reference reports (Figure 225). You indicate the basic event report y_{c} want to generate by highlighting the desired report using the arrow keys and pressing < Enter >.



	REPORT OPTIONS
FAMILY SU	MEMARY REPORT
harmonic restriction of	
	Dutput File Name
	LON
NOTE: File	Name = "CON" - Output report to the screen.
NOTE: File	<pre>Name = "CON" - Output report to the screen. "PRN" - Output report to the printer. " " - No report is produced."</pre>
NOTE: File	e Name = "CON" - Output report to the screen. "PRN" - Output report to the printer. " " - No report is produced. <esc> - No report is produced.</esc>

Figure 224. Output destination screen.



10.3.1 Basic Event Overview

This option allows you to generate a basic event summary report. The overview report includes the basic event number, primary and secondary name, component type and ID, system location, and attribute fail mode. To invoke this option, highlight Overview and press < Enter >. Figure 226 will be displayed. Two options are available: Exit and Overview. Exit terminates the process and returns you to the previous menu. Overview generates the report based on your event selections. On this screen, you must do one of the following:

- Press < Enter > to report all basic events.
- Highlight a basic event and press < Enter >.
- Mark a group of basic events and press < Enter >.

	Option 0	Exit / Overview		
N D D D		Descripti	0 n	providence
ACC-CKV-FT-CV107	CHECK VALVE CV107	FAILS TO OPEN		1.1.1
ACC-CKV-F1-CV109 ACC-CKV-F1-CV128	CHECK VALVE CV109 CHECK VALVE CV108	FAILS TO OPEN		1.000
ACC-CKV-FT-CV130	CHECK VALVE CV130	FAILS TO OPEN		
ACC-CKV-FT-CV145	CHECK VALVE CV145	FAILS TO OPEN		
ACC-CKV-FT-CV147	CHECK VALVE CV147	FAILS TO OPEN		
ACC-MOV-PG-1865A	ACC MOTOR OPERATE	VALVE 1865A PLUGGE	D	1.1.1.1.1.1.1.1.1
ACC-MOV-PG-18650	ACC MOTOR OPERATES	VALVE TRASC PLUGGE	9	
ACP-BAC-ST-111	480V AC BUS 1H1 B	ISWORK FAILURE		the second s
ACP-BAC-ST-110	480V AC BUS 1H1 B	ISWORK FAILURE		

NOTE: For most report options, you must perform the selection process described above.

Figure 226. Basic event overview screen.

After a basic event(s) has b on selected, the output destination screen will be displayed. Type in the desired destination and press < Fnter >.

10.3.2 Basic Event Probability Report

This option allows you to generate a basic event probability report. The probability report shows the event number, primary name, failure calculation type, mean probability, and event lambda and tau values. To invoke this option, highlight Probability and press < Enter >. Figure 227 will be displayed. Two options are available: Exit and Probability. Exit terminates the process and returns you to the previous menu. Probability generates the report based on the event selections. Select the events as described in Section 10.3.1.

After a basic event(s) has been selected, the output destination screen will be displayed. Type in the desired destination and press < Enter >



SURRY	Basic Event Probability
	Option (P) Exit / Probability
N a m e ACC-CKV-FT-CV10 ACC-CKV-FT-CV10 ACC-CKV-FT-CV12 ACC-CKV-FT-CV12 ACC-CKV-FT-CV14 ACC-CKV-FT-CV14 ACC-CKV-FT-CV14 ACC-MOV-PG-1865 ACC-MOV-PG-1865 ACC-MOV-PG-1865 ACC-BAC-ST-1H1	D e s c r i p t i o n 7 CHECK VALVE CV107 FAILS TO OPEN 9 CHECK VALVE CV109 FAILS TO OPEN 8 CHECK VALVE CV128 FAILS TO OPEN 0 CHECK VALVE CV128 FAILS TO OPEN 5 CHECK VALVE CV145 FAILS TO OPEN 7 CHECK VALVE CV147 FAILS TO OPEN A ACC MOTOR OPERATED VALVE 1865A PLUGGED 8 ACC MOTOR OPERATED VALVE 1865C PLUGGED C ACC MOTOR OPERATED VALVE 1865C PLUGGED 480V AC BUS 1H1 BUSWORK FAILURE
<esc></esc>	<pre>xf1> <f2> <f3> <f4> <f5></f5></f4></f3></f2></pre>

Figure 227. Basic event probability screen.

10.3.3 Basic Event Uncertainty Report

This option allows you to generate a basic event uncertainty report. The report shows the event number, primary name, distribution type, mean probability, uncertainty value, and correlation class. To invoke this option, highlight Uncertainty and press < Enter >. Two options are available: Exit and Uncertainty. Exit terminates the process and returns you to the previous menu. Uncertainty generates the report based on your event selections. Select the events as described in Section 10.3.1.

After a basic event(s) has been selected, the output destination screen will be displayed. Select the desired destination and press < Euter >. When the report is complete, you will be returned to the Basic Event Uncertainty screen. At this point, you may select another event or enter an < E> in the option field and press < Enter > to return to the IRRAS Report Menu.

10.3.4 Basic Event Cross Reference Report

This option allows you to generate a basic event cross reference report. To invoke this option, highlight X-Reference and press < Enter >. Figure 228 will be displayed. As shown, there are three cross-reference reports available: Sequence Cut Set, System Cut Set, and System Logic. Select the events as described in Section 10.3.1.

10.3.4.1 Sequence Cut Set. This option produces a sequence cut set cross reference. The report provides an event to se, sence cross reference. The report will include the event number, event name and associated sequence names.

option [E] Exit	/ Sequence Cut Set / 1	SYstem Cut Set / Sys	tem Logic
ACC. NEW EX CURAT DU		cription	
ACC-CKV-FT-CV107 CH	CK VALVE CV107 FAILS I	TO OPEN	
ACC-CKV-FT-CV128 CH	CK VALVE CV128 FAILS 1	O OPEN	
ACC-CKV-FT-CV130 CH	ECK VALVE CV130 FAILS 1	O OPEN	
ACC-DEV-FT-CV145 CHI ACC-DEV-FT-CV147 CHI	CK VALVE CV165 FAILS 1	FO OPEN	
ACC-MOV-PG-1865A AC	MOTOR OPERATED VALVE	1865A PLUGGED	
ACC-MOV-PG-18658 AC	MOTOR OPERATED VALVE	18658 PLUGGED	
ACC-MOV-PG-1865C ACI ACP-BAC-ST-1K1 48	C MOTOR OPERATED VALVE	1865C PLUGGED	
the commission of the company relation of the		and the second second second	

Figure 228. Basic event cross reference screen.

To invoke this option, mark the desired event(s), enter an $\langle S \rangle$ in the option field, and press $\langle Enter \rangle$. You may include all sequences by entering an $\langle S \rangle$ in the option field and pressing $\langle Enter \rangle$. The message **Process all records** will be displayed. Enter a $\langle Y \rangle$ to include all sequences in the report or enter an $\langle N \rangle$ to terminate the process. In any case, upon pressing $\langle Enter \rangle$, the output destination screen will be displayed. Select the desired destination and press $\langle Enter \rangle$.

10.3.4.2 System Cut Set. This option produces a system cut set cross reference. This report provides an event to system cross reference. The report includes the event number, the event name and the corresponding system name.

To invoke this option, mark the desired event(s), enter a < Y > in the option field, and press < Enter >. You may include all systems by entering a < Y > in the option field and pressing < Enter >. The message **Process all records** will be displayed. Enter a < Y > to include all systems in the report or enter an < N > to terminate the process. In any case, upon pressing < Enter >, the output destination screen will be displayed. Select the desired destination and press < Enter >.

10.3.4.3 System Logic This option procedures a system logic cross reference. This report provides an event to system logic cross reference. The report includes the event number, the event name and the corresponding system names.

To invoke this option, mark the desired event(s), enter an < L > in the option field, and press < Enter >. You may include all system logic by entering an < L > in the option field and pressing < Enter >. The message **Process all records** will be displayed. Enter a < Y > to include all system logic in the report or enter an < N > to terminate the process. In any case, upon pressing < Enter >, the output destination screen will be displayed. Select the desired destination and press < Enter >.



This option is not yet available.

10.5 SYstem

This option allows you to generate a variety of system reports. These include summary, logic, cut sets, importance, and cross reference reports. When you invoke this option, Figure 229 will be displayed. You indicate the system report you want to generate by highlighting the desired report using the arrow keys and pressing < Enter >.



Figure 229. System report selection screen.

10.5.1 System Summary Report

This option allows you to generate a system summary that can be based on current or base case values. When you invoke this option, Figure 230 will be displayed. Three system summary reports are available: Summary, Combination, and Uncertainty. Select the desired system option and press <Enter>.

10.5.1.1 Summary. This option allows you to generate the System Brief Summary Report. This report contains the names of the systems residing in the current family, the associated minimum cut set upper bound, and the system description. When you invoke this option, the output destination screen will be displayed. On this screen, you must indicate the desired output destination (see Section 10.2.2).

10.5.1.2 COmbination. This option allows you to generate the System Combination Report. This report contains all systems in the current family, the minimum cut set, mean, and number of cut sets in the system. When you invoke this option, the output destination screen will be displayed. On this

January 1992



Figure 230. System summary report selection screen.

screen, you must indicate the desired output destination (see Section 10.2.2).

10.5.1.3 Uncertainty. This option allows you to generate the System Uncertainty Value Report. This report contains all systems in the current family, the mean and median values, standard deviation, the 5th and 95th percentile, the minimum and maximum values, and the seed size. When you invoke this option, the output destination screen will be displayed. On this screen, you must indicate the desired output destination (see Section 10.2.2).

10.5.2 System Logic Report

This option allows you to generate system logic reports. When you invoke this option, Figure 231 will be displayed. On this screen, all systems names or subtree names contained in the current family are displayed. You may use the Subtrees option to toggle the display from system names to subtree names. If you toggle to subtrees, the report will be based on subtrees not systems. On this screen, you must select the desired systems or subtrees (see Section 10.3.1) to include in the report.

As shown, three report options are available: Logic, Expanded, and Modified.

10.5.2.1 Logic. This option allows you to generate the System Logic Report consisting of the gate names, types, and inputs for the specified system(s) or subtree(s). When you invoke this option, the output destination screen will be displayed. On this screen, you must indicate the desired output destination (see Section 10.2.2).

To invoke this option, mark the system(s) or subtrees to include, enter an <L> in the option field and press < Enter >.

10.5.2.2 eXpanded. This option allows you to generate the System Expanded Logic Report consisting of the gate names, types, and inputs for the specified system(s). When you invoke this option, you will be prompted to enter a starting gate for the report. You may specify a gate or press < Enter >



	Option E Exit / Logic / expanded / Modified SubTrees
N a M	CONTAINMENT SPRAY
CS	CONTAINMENT SYSTEMS
CV	CORE VULNERABLE TO CD
D1	HIGH PRESSURE INJECTION - AUTOMATIC
D2	HIGH PRESSURE INJECTION - MANUAL
D3	HIGH PRESSURE INJECTION - RCP SEALS
D4	HIGH PRESSURE INJECTION - EMERGENCY BORATION
D5	ACCUMULATORS
D6	LOW PRESSURE INJECTION
F1	INSIDE SPRAY RECIRCULATION
<esc></esc>	kF1> KF2> KF3> KF4> KF5>
Exit	Help Mark Line Mark All Mark Range Locate

Figure 231. System logic report screen.

to include all gates. Next, the output destination screen will be displayed. On this screen, you must indicate the desired output destination (see Section 10.2.2).

To invoke this option, mark the system(s) or subtree(s) to include, enter an $\langle X \rangle$ in the option field, and press $\langle Enter \rangle$.

10.5.2.3 Modified. This option allows you to generate the System Modified Logic Report consisting of the gate names, types, and inputs for the specified system(s). When you invoke this option, you will be prompted to enter a starting gate for the report. You may specify a gate or press < Enter > to include all gates. Next, the output destination screen will be displayed. On this screen, you must select the desired output destination (see Section 10.2.2).

To invoke this option, mark the system(s) or subtree(s) to include, enter an < M > in the option field, and press < Enter >.

10.5.3 Cut Sets Report

This option allows you to generate cut set reports based on alternate or base case values. Use the <T> oggle Alternate Cut Set option to set the values to alternate or base case. The Word ALTERNATE or BASE will appear in the upper right corner. When you invoke this option, Figure 232 will be displayed. On this screen, all names of the systems residing in the current family will be displayed. On this screen, you must select the systems to include in the report (see Section 10.3.1).

January 1992

10-10

Option [E] E	xit / Cut Set / Quantified Cut Set / Toggle Alternate Cut Set
	Description
C C	CONTAINMENT SPRAY
CV	CORE VULNERABLE TO CD
01	HIGH PRESSURE INJECTION - AUTOMATIC
02	HIGH PRESSURE INJECTION - RCP SEALS
D4	HIGH PRESSURE INJECTION . EMERGENCY BORATION
D5	ACCUMULATORS
F1	INSIDE SPRAY RECIRCULATION

Figure 232. Cut sets report screen.

As shown, two report options are available: Cut Set or Quantified Cut Set.

10.5.3.1 Cut Set. This option allows you to generate the System Cut Sets Report which consists of the cut set number, size, and all corresponding cut sets (alternate or base). When you invoke this option, Figure 233 will be displayed. You can modify any of the default values provided by simply typing over the existing data. Upon pressing < Enter >, the output destination screen will be displayed. On this screen, you must select the desired output destination (see Section 10.2.2).

To invoke this option, mark the system(s) to include, enter a < C> in the option field, and press < Enter >

10.5.3.2 Quantified Cut Set. This option allows you to generate the System Cut Set (Quantification) Report for quantified cut sets only. The report consists of the cut set number, percent of total, percent of the cut set, orobability/frequency and all associated cut sets (alternate or base) for the selected system(s). When you invoke this option, Figure 233 will be displayed. You can modify any of the default values provided by simply typing over the existing data. Upon pressing < Enter >, the output destination screen will be displayed. On this screen, you must select the desired output destination (see Section 10.2.2).

To invoke this option, mark the system(s) to include, enter a $\langle Q \rangle$ in the option field, and press $\langle Enter \rangle$.

Option [C]	Exit Total number of Cut Sets: 9999 Max size of each Cut Set: 99
U	Puantified value cutoff : +0.0E+000
CS	Percent of total value : 100.0
01	HIGH PRESSURE INJECTION - AUTOMATIC
02	HIGH PRESSURE INJECTION - MANUAL
03	HIGH PRESSURE INJECTION - RCP SEALS
04	HIGH PRESSURE INJECTION - EMERGENCY BORATION
05	ACCUMULATORS
06	LOW PRESSURE INJECTION
F1	INSIDE SPRAY RECIRCULATION

Figure 233. Cut set default report settings.

10.5.4 System Importance

This option allows you to generate importance reports based on alternate or base case values. Use the < T > oggle Alternate Cut Set option to set the values to alternate or base case. The Word ALTERNATE or BASE will appear in the upper right corner. When you invoke this option, Figure 234 will be displayed. On this screen, all names of the systems residing in the current family will be displayed. On this screen, you must select the systems to include in the report (see Section 10.3.1).

As shown, two report options are available: Importance and Sort Criteria.

10.5.4.1 Importance. This option allows you to generate the System Importance Measures report consisting of event names, number of times the event occurs, probability of failure, Fussell-Vesely importance value, risk reduction interval, and risk increase interval for the selected system(s). When you invoke this option, the output destination screen will be displayed. On this screen, you must select the desired output destination (see Section 10.2.2).

To invoke this option, mark the system(s) to include, enter an <1> in the option field, and press < Enter >.

10.5.4.2 Sort Criteria. This option allows you to select the sort order in which to display the System Importance Measures Report. When you invoke this option, Figure 235 will be displayed.

As shown, the following sort options are available:

Name

Sorted by event name

10-12





Option E	Exit / Importance / Sort Criteria / Toggle Alternate Cut Set
C N B M	CONTAINMENT SPRAY
CV	CORE VULNERABLE TO CD
D1	HIGH PRESSURE INJECTION - AUTOMATIC
D2	HIGH PRESSURE INJECTION - MANUAL
D3	HIGH PRESSURE INJECTION - CCP SEALS
D4	HIGH PRESSURE INJECTION - EMERGENCY BORATION
06	LOW PRESSURE INJECTION
F1	INSIDE SPRAY RECIRCULATION









Occurrence Count	-kr	Sorted by the number of occurrences (most to fewest)
Probability of Failure		Sorted by the probability failures
F-V or Birnbaum	4	Sorted in Fussell-Vesely or Birnbaum order
Risk Reduction	*	Sorted in risk reduction ratio order
Risk Increase		Sorted in risk increase ratio order

10.5.5 X-Reference

This option allows you to generate cross reference reports for either systems or subsystems. When you invoke this option, Figure 236 will be displayed. You may use the Level option to toggle between systems and subsystems. Two types of reports are available: Uses and Used By.



Figure 236. System cross reference report selection screen.

10.5.5.1 Uses. This type of report allows you to list either the subsystems or basic events that a subsystem references, or uses.

10.5.5.1.1 Subsystems—This option allows you to generate the System Hierarchy Report consisting of a system/subsystem and a list of all subsystems it uses. When you invoke this option, a list of systems or subsystems will be displayed. On this screen you must select the desired systems or subtrees (see Section 10.3.1). After a subsystem(s) has been selected, the output destination screen will be displayed. Type in the desired destination and press < Enter >.

10.5.5.1.2 Basic Events—This option allows you to generate the System/Basic Event Reference Report consisting of a system/subsystem and a list of all basic events that it uses. When you invoke this option, a list of systems or subsystems will be displayed. On this screen you must select the desired systems or subtrees (see Section 10.3.1). After a subsystem(s) has been selected, the output destination screen will be displayed. Type in the desired destination and press < Enter >.

10-14

10.5.5.2 Used By. This type of report allows you to list all the systems/subsystems, event trees, or sequences that use a particular system/subtree.

10.5.5.1.2 Subsystems—This option allows you to generate the SubTree Cross Reference Report consisting of a subsystem and a list of all systems/subtrees that reference or use it. When you invoke this option, a list of systems or subsystems will be displayed. On this screen you must select the desired systems or subtrees (see Section 10.3.1). After a subsystem(s) has been selected, the output destination screen will be displayed. Type in the desired destination and press < Enter > .

10.5.5.1.3 Event Trees-This option not yet available.

10.5.5.1.4 Subsystems-This option not yet available.

10.6 EVent Tree

This option allows you to generate a variety of event tree reports. When you invoke this option, Figure 237 will be displayed. You indicate which event tree report you want to generate by highlighting the desired report using the arrow keys and pressing < Enter >. As shown, the following reports may be generated: Logic/Rules, Initiating Events, Cross Reference, and System.



Figure 237. Event tree report selection screen.

10.6.1 Logic/Rules

This option allows you to generate the Event Tree Logic Report and the Event Tree Rules report. When you invoke this option, Figure 238 will be displayed. On this screen all event trees contained in



the current family are displayed. On this screen, you must select the event trees to include in the report (see Section 10.3.1).

NAME AND ADDRESS OF ADDRE	mean year and an
	Particular 191 Forder a south of Andrew
	option [E] EKIT / LOgic / Rules
A ATWS1 ATWS2 S1 S2 S3 T1 T1S1 T1SB	BRIDGE TREE, 6 TO 29 INCH LOSS OF COOLANT ACCIDENT CORE DAMAGE TREE, ANTICIPATED TRANSIENT WITHOUT SCRAM CORE DAMAGE TREE, ANTICIPATED TRANSIENT WITHOUT SCRAM BRIDGE TREE, 2 TO 6 INCH LOSS OF COOLANT ACCIDENT BRIDGE TREE, ONE HALF TO 2 INCH LOSS OF COOLANT ACCIDENT BRIDGE TREE, LESS THAN ONE HALF INCH LOCA LOSS OF OFFSITE POWER EVENT TREE BRIDGE TREE, STATION BLACKOUT AT UNIT 1 ALONE BRIDGE TREE, STATION BLACKOUT AT UNITS 1 AND 2
Not	e : Use <pgup> and <pgdn> to display more event tree. An event tree must be selected for options L and R.</pgdn></pgup>
<esc> Exit</esc>	<f1> <f2> <f3> <f4> <f4></f4></f4></f3></f2></f1>

Figure 238. Logic/Rules selection screen.

As shown, two report options are available: Logic and Rules.

10.6.1.1 Logic. This option produces the Event Tree Logic Report. This report lists all sequence names and the associated logic (pass/fail) associated with each selected event tree. To invoke this option, mark the event trees, enter an <L> in the option field, and press < Enter>. When you invoke this option, the output destination screen will be displayed. On this screen, you must select the desired output destination (see Section 10.2.2).

10.6.1.2 Rules. This option produces the Event Tree Rules Report. This report lists all linkage rules (exceptions) associated with each selected event tree. To invoke this option, mark the event trees, enter an $\langle R \rangle$ in the option files, and press $\langle Enter \rangle$. The same output options are available as described in Section 10.2.2.

10.6.2 Initiating Event Report

This option allows you to generate the Event Tree Initiating Events Report. This report contains the event tree name and the corresponding event tree description and initiating event for all event trees in the current family. When you invoke this option, the output destination screen will be displayed. On this screen, you must select the desired output destination (see Section 10.2.2).

10.6.3 System

This option is not yet available.

10.7 Sequence

This option allows you to generate a variety of sequence reports. When you invoke this option, Figure 239 will be displayed. You indicate which sequence report you want to generate by highlighting the desired report using the arrow keys and pressing < Enter >. As shown, the following reports may be generated: Summary, Logic, Cut Sets, and Importance.



Figure 239. Sequence report selection screen.

10.7.1 Summary

When you invoke this option Figure 240 will be displayed. As shown four different summary reports may be generated: Brief, Summary, Combination and Uncertainty. Each of these reports can be generated using the Current or Base case values.

10.7.1.1 Brief Summary. This option allows you to generate the Sequence Brief Summary Report. The report lists each sequence name contained in the current family, the minimum cut set upper bound, and the sequence description. When you invoke this option, the output destination screen will be displayed. On this screen you must select the desired output destination (see Section 10.2.2).

10.7.1.2 Summary. This option allows you to generate the Sequence Summary Report. The report lists each sequence number and name contained in the current family and the minimum cut set upper bound. When you invoke this option, the output destination screen will be displayed. On this screen, you must select the desired output destination (see Section 10.2.2).





Figure 240. Sequence summary report selection menu.

10.7.1.3 COmbination. This option allows you to generate the Sequence Combination Report. The report lists each sequence number and name contained in the current family, the minimum cut set value, the mean cut set value, and the number of cut sets contained in each sequence. When you invoke this option, the output destination screen will be displayed. On this screen, you must select the desired output destination (see Section 10.2.2).

10.7.1.4 Uncertainty. This option allows you to generate the Sequence Uncertainty Values Report. The report lists each sequence number and name contained in the current family, the mean and median values, standard deviation, 5th and 95th percentile, minimum and maximum range values, and the seed size. When you invoke this option, the output destination screen will be displayed. On this screen, you must select the desired output destination (see Section 10.2.2).

10.7.2 Logic

This option allows you to generate the Sequence Logic Report. This report contains every sequence number and name contained in the current family, and the logic paths (pass/fail) for the event tree, initiating event, and logic. When you invoke this option, the output destination screen will be displayed. On this screen, you must select the desired output destination (see Section 10.2.2).

10.7.3 Cut Sets

This option allows you to generate two sequence cut set reports using alternate or base cut set values. Use the Toggle Alternate Cut Set option to set the desired values to alternate or base case. The word ALTERNATE or BASE will appear in the upper right corner. When you invoke this option, Figure 241 will be displayed. All sequences contained in the current family are displayed. On this screen, you must indicate the sequences to include in the report.

As shown, two options are available: Cut Set and Quantified Cut Set.

Event Event A-C-F1-CV A-C-F1-CV A-C-H1 A-D5 A-D5-C A-D5-C A-D5-CF1 A-D5-F1 A-D5-F1 A-D5-F1 A-D5-F1 A-D5-F1 A-D5-F1 A-D5-F1 A-D5-F1 A-D5-CF1 A-D5-CF1 A-D5-F	Aug. 101 . 0	di chiaka s	August start for the start structure for the
Event A-C-F1-CV A A-C-H1 A A-D5-C A A-D5-C A A-D5-C-F1 A A-D5-F1 A A-D5-F1 A A-D5-F1 A A-D5-F1 A A-D5-F1 A A-D5-F1 A A-D5-F1 A A-D5-F1 A A-D5-C A LARGE LOCA - INJECTION FAILURE, NOT RWST A-D6-C A	oprion [e] e	KIR A PAR SAL 1	waantitied out bet 7 loggre kiternete out set
A-C-F1-CV A A-C-F1 A A-D5 A A-D5 A A-D5-C A A-D5-C-F1 A A-D5-F1 A A-D5-F1 A A-D5-F1 A A-D5-F1 A A-D5-F1 A A-D5-F1 A A-D5-F1 A A-D5-F1 A A-D5-C A LARGE LOCA - INJECTION FAILURE, NOT RWST A-D6-C A		Even	t.
A-C-F1-CV A A-C-H1 A A-D5 A LARGE LOCA - SCOUMULATOR FAILURE A-D5-C A A-D5-C-F1 A A-D5-F1 A A-D5-F1 A A-D6-C A LARGE LOCA - INJECTION FAILURE, NOT RWST	nome N n n e -	ancerentation 7 P & B	managementeries Descripticn enterester
A-C-HI A A-D5 A LARGE LOCA - SCOUMULATOR FAILURE A-D5-C A A-D5-C-FI A A-D5-FI A A-D5-FI A A-D5-FI A A-D6-C A LARGE LOCA - INJECTION FAILURE, NOT RWST	A-C-F1-CV	A	
A-D5-C A A-D5-C-F1 A A-D5-F1 A A-D5-F1 F2 A A-D6-C A LARGE LOCA - INJECTION FAILURE, NOT RWST	A-C-H1	A	LARGE LOOK . CONTRACT AVAIL REALLING
A-D5-C-F1 A A-D5-F1 A A-D5-F1 A A-D5-F1-F2 A A-D6 A LARGE LOCA - INJECTION FAILURE, NOT RWST A-D6-C A	A-02-		LARGE LOCA . MESOMOLATOR PRICORE
A-D5-F1 A A-D5-F1-F2 A A-D6 A LARGE LOCA - INJECTION FAILURE, NOT RWST A-D6-C A	A-05-0-F1	1. 2	
A-D5-F1-F2 A A-D6 A LARGE LOCA - INJECTION FAILURE, NOT RWST A-D6-C A	A-05-F1	- A	
A-D6 A LARGE LOCA - INJECTION FAILURE, NOT RWST A-D6-C A	A-05-F1-F2	A	
A-D6-C A	A-06	A .	LARGE LOCA - INJECTION FAILURE, NOT RWST
	A-06-C	A	
A-D6-C-F1 A LARGE LOCA - INJECTION FAILURE, FM RWST	A-D6-C-F1	A	LARGE LOCA - INJECTION FAILURE, FM RWST



10.7.3.1 Cut Set. This option allows you to generate the Sequence Cut Sets Report, which consists of the cut set number, size and cut set name for each selected sequence. To invoke this option, mark the sequence(s) to include, enter a < C> in the option field, and press < Enter>. When you invoke this option, Figure 242 will be displayed. You can modify any of the default values provided by simply typing over the existing data. Upon pressing < Enter>, the output destination screen will be displayed. On this screen, you must select the desired output destination (see Section 10.2.2).

10.7.3.2 Quantified Cut Set. This option allows you to generate the Sequence Cut Sets (Quantification) Report, which consists of the cut set number, percent of total, percent of cut set, probability/frequency and all associated cut sets (alternate or base) for the selected sequence(s). To invoke this option, mark the sequence(s) to include, enter a $\langle Q \rangle$ in the option field, and press $\langle Enter \rangle$. When you invoke this option, a screen similar to the one shown in Figure 242 will be displayed. You can modify any of the default values provided by simply typing over the existing data. Upon pressing $\langle Enter \rangle$, the output destination screen will be displayed. On this screen, you must select the desired output destination (see Section 10.2.2).

10.7.4 Importance

This option allows you to generate reports based on alternate or base case values. Use the Toggle Alternate Cut Set to set the values to alternate or base case. The word ALTERNATE or BASE will appear in the upper right corner (Figure 243). When you invoke this option, Figure 243 will be displayed. On this screen, all sequence names residing in the current family will be displayed. On this screen, you must select the sequences to include in the report (see Section 10.3.1).



10-19



Figure 242. Cut set report options.



Figure 243. Sequence importance selection screen.

As shown, two options are available: Importance and Sort Criteria.

10.7.4.1 Importance. This option allows you to generate the Sequence Importance Measures Report which consists of event names, number of times the event occurs, probability of failure, Fussell-Vesely importance value, risk reduction interval, and risk increase interval for the selected sequence(s). To invoke this option, mark the sequence(s) to include, enter an <1> in the option field, and press < Enter >. When you invoke this option, the output destination screen will be displayed. On this screen, you must select the desired output destination (see Section 10.2.2).

10.7.4.2 Sort Criteria. This option allows you to select the sort order in which to display the Sequence Importance Measures Report. When you invoke this option, Figure 244 will be displayed.



Figure 244. Sort criteria selection screen.

As shown, the following sort options are available:

Name	14.11	Sorted by event name
Occurrence Count	1.00	Sorted by the number of occurrences (most to fewest)
Probability of Failure	18.13	Sorted by probability failures
F-V or Birnbaum		Sorted by Fussell-Vesely or Birnbaum values
Risk Reduction		Sorted in risk reduction ratio order
Risk Increase		Sorted in risk increase ratio order



10-21

10.8 ENd States

This option allows you to generate two end state reports. When you invoke this option, Figure 245 will be displayed. You indicate the end state report you want to generate by highlighting the desired report using the arrow keys and pressing < Enter >. As shown, the following reports may be generated: Brief Summary and Uncertainty. Each of these reports can be generated using the Current or Base Case values. The Case option (<C>) is a toggle that allows you to set the case to current or base case.



Figure 245. End state report selection screen.

10.8.1 Brief Summary

This option allows you to generate the End State Brief Summary Report. This report lists each end state name contained in the family, the minimum cut set upper bound, and the end state description. When you invoke this option, the output destination screen will be displayed. On this screen, you must select the desired output destination (see Section 10.2.2).

10.8.2 Uncertainty

This option allows you to generate the End State Uncertainty Values Report. This report lists each end state number and name contained in the current family, the mean and median values, standard deviation, 5th and 95th percentile, minimum and maximum range values, and the seed size. When you invoke this option, the output destination screen will be displayed. On this screen, you must select the desired output destination (see Section 10.2.2).

10.9 User Info

User Information Reports not yet available.





11. UTILITY OPTIONS

The IRRAS Utility Options allow you to perform routine functions that are required by IRRAS such as defining constants, recovering the data base, and MAR-D data exchange. When you invoke this option, Figure 246 will be displayed.

Assessment and a second property of the second seco	d homosonacionencompany announcempany	
	EXIT	
	Define Constants	
	Load MAR-D Data Files Extract MAR-D Data Files	
	Version 1.0/4.0 Interface	
	Recover Data Base	
	The submitted of the su	

Figure 246. Utility options main menu.

11.1 Exit

Type < E> in the command line, or highlight Exit, and press < Enter>, or press the < Esc> key to return to the IRRAS main menu.

11.2 Define Constants

The Define Constants $\langle D \rangle$ option allows you to define what hardware the system uses as well as defining uncertainty settings, cut set constants, and default values for the graphics editor. After choosing this option, the User Information Constants screen shown in Figure 247 is displayed. Table 1 provides a brief description of each of the fields in Figure 247.



Figure 247. User information constants screen.

FIELD	DESCRIPTION		
User Name	36 character user identification (optional) field		
Monitor type	 C - Enhanced graphics monitor. 1 · DEFAULT - Standard color graphics monitor 2 · Video graphics monitor (2640 x 480) 3 · Video graphics monitor plus (800 x 600) 4 · B5144 		
Card type	0 - IBM 1 - Paradise 2 - ATI 3 - TECMAR 4 - TSENG 5 - VIDEO-7 6 - PLASMA		
Printer type	0 - Other 1 - Epson 7 - HP Laser (DEFAULT)		
Printer has complete IBM character font?	Y - Yes. N - No (DEFAULT)		
Plotter communications port	 No Plotter connected to part (DEFAULT). Platter connected to Com1 part. Plotter connected to Com2 part. Plotter connected to Com3 part. 		
Use alternate bacic event names?	Y \sim Alternate name will be used. N \sim Primary name will be used (DEFAULT).		
Importance Measurement Type	R - Ratios (DEFAULT) 1 - Intervals U - Uncertainty		
Varify when loading MAR-D Gate?	Y - File dependency will be checked. N - No file cross checking will done.		
Random number wead for uncertainty calculations	5 digit numeric field indic/ting the first random number in the seed to be used in the Monto Carlo calculation. 0 (DEFAULT) indicates that the random number will be the current value of the real clock.		
Sample size	6 digit numeric field indicating the default number of Monte Carlo samples to be run in the uncertainty analyses. Sample size may range from 1 to 999999 (DEFAULT = 1,000).		

Table 1. User Information field descriptions for constants option

1





January 1992

ف

11-3

Table 1. (continued)

FIELD	DESCRIPTION
Cutoff by size?	Y - Do not generate fault tree or stydence cut sets containing more basic events than indicated in the Size cutoff field (DEFAULT).
	N - Generate all cut sets for the fault tree or sequence that meet the probability cutoff criteria (if in effect)
Size cutoff	The default maximum number of basic eventa allowed in cut set generation when size cutoff is in effect. DEFAULT = 6
Cutoff by probability?	Y - Do not generate fault tree or sequence cut sets that have a probability less than the cutoff indicated in the probability cutoff field (DEFAULT)
	N - Generate all cut sets that meet the size cutoff criteria (if in effect) regardless of the cut set probability.
Probability cutoff	The default minimum cut set probability allowed in cut set generation when probability sutoff is in effect. (DEFAULT = DODE-015)
Mission time (hours)	The default mission time to be used in the calculation of basic event probabilities (when appropriate). DEFAULT = 2.200E+001
Scratch drive and directory	36 character field indicating the drive and path to the scratch diractory where files will be stored (DEF* AT set to blanks)
Halo drive and directory path	36 character field indicating the drive and path to the Hulo graphics that IRRAS 4.0 should use (\HALO88\)



After setting the User Information constants, press < Enter >. The next screen displays the Fault Tree Graphics Constants (Figure 248). Table 2 provides a brief description of each of the fields in Figure 248.







FIELD	DESCRIPTION
Event Name	10-character field for the fault tree basic event default / me. The default name is followed by a sequential number for each separate event, e.g., EVENT1, EVENT2, etc. (DEFAULT < TVENT)
Gste Name	10-characte, for the fault tree gate default name. The default name is followed by a sequential number for each separate event, e.g., GATE7, GATE2, etc. (DEFAULT = GATE)
Name Height	The height of the event and gate names. This is a number between 0.01 and 66.00, where 66.00 represents the full 66 lines from the top of the screen to the bottom. (DEFAULT = 0.50)
Line Type	1 - Solid line (DEFAULT) 2 - Deshed line. 3 - Dotted line.
Text Height	The height of the text to be written. This is a number between 0.01 and 66.00 (see Name Height). DEFAULT = .50
N/M Height	The h , this of the numbers on an N/M OR gate showing the N and \approx values. This is a number between 0.0 and 66.00. (DEFAULT = 1.50)
Justification	Text justification. Left justified. C - Centered (DEFAULT). R - Right justified.
Space factor	The spacing between lines of text written consecutively. This is a number between 1.00 and 10.00. A value of 1.00 will cause the top of the following line of text to touch the bottom of the preceding line. (DEFAULT = 1.40)

Table 2. Fault tree graphics information field descriptions





Table 2. (continued)

F1E1.D	DESCRIPTION
Fill	Yes/No toggle turning the fill on/off for plotting. DEFAULT = No
Grid	Yes/No toggle turning a reference grid on/off. DEFAULT = No
Blank	 Y - The immediate area surrounding a gate or event name will be blanked out, (DEFAULT)
	N - The gate and event names will be written over any lines drawn in the areas for the names.
Show Name	Yes/No toggle turning on/off the display of event and gate names when the SHOW command is used. DEFAULT = Yes
Show Text	Yes/No toggle turning on/off the display of descriptive text when the SHOW command is used. DEFAULT = Yes.
MultiPick	When building trees, multipick will generate multiple gates for each pick of a jate type. DEFAULT = Yes.
Fill Color	An X under the desired color selects the default color for filling in shapes. DEFAULT color = blue.
Wanne Color	An X under the desired color selects the default color for displaying names. DEFAULT color = white.
ine Color	An X under the desired color selects the default color for drawing lines. DEFAULT color = white.
Text Color	An X under the desired color selects the default color for writing text. DEFAULT color = white.
Cursor Color	An X under the desired color selects the default color for the cursor. DEFAULT color = white.

After setting the Fault Tree graphics information constants press <Enter>. The next screen displays Event Tree graphics information (Figure 249). Make any changes needed and press <Enter>. Table 3 provides a brief description of each of the fields in Figure 249.



	Event					
Co Ma	lors . 0 1 2 3 In Menu Text Color	Tree Graphics	Information 8 9 10 11		5 . 15	
Ma Zn Zn 3r 3r	in Menu Background J Level Menu Text d Level Menu Backg d Level Menu Text d Level Kenu Backg	Color	· · · · · · ·		. 1 . 14 . 5 . 12	
Cu Li Te Te	rsor Colur ne Color xt Color xt Height				. 15 . 10 . 14	
Hi Te Ma	de Text xt Justification in Menu Side	<y =="" hide,="" n<br=""><l =="" c<br="" left,=""><l =="" left,="" r<="" td=""><td>don't hide center, R right></td><td>= right> .</td><td>. N . L</td><td></td></l></l></y>	don't hide center, R right>	= right> .	. N . L	

Figure 249. Event tree graphics information.





Table 3. Event tree graphics information field descriptions

FIELD	DESCRIPTION
Colors	Colors that can be used in event tree graphics and their associated reference numbers.
Main Menu Text Color	Color of main menu's text. DEFAULT = 15 (white)
Main Menu Background Color	Background color upon which text is written. DEFAULT = 1 (blue).
2nd Level Menu Text Color	Calor of text in second level menus. DEFAULT = 14 (yellow).
2nd Level Menu Background Color	Background color of second level menus upon which text is written. DEFAULT = 5 (purple).
3rd Level Menu Text Color	Color of text in third level menus. (DEFAULT = 1.7 (cyan).
3rd Level Menu Background Color	Background colo: of third level menus upon which text is written. DEFAULT = 9 (light blue).
Cursor Color	Default color o, cursor. DEFAULT ~ 15 (white).
Line Color	Default color of lines. DEFAULT = 15 (white).
Text Color	Default color of text. DEFAULT = 14 (yellow).
Text Height	Default text height. DEFAULT = 1.00.
Hide Text	Y/N Hide text when displaying tree. DEFAULT = N.
Text Justification	L = Left (DEFAULT) C = Center R = Right.
Main Menu Side	L/R - Side of screen to place main menu. (DEFAULT = left).
File Compacting	Y/N - Compact file when leaving editor. DEFAULT = No.





11.3 Load MAR-D Data Files

The Load MAR-D Data Files option facilitates loading of Probabilistic Risk Assessment (PR_{c}) data from the Models and Results Data Base (MAR-D). The process converts information from the generic format found in the MAR-D data base to the IRRAS data base format. When you select this option, Figure 250 will be displayed.

Select DATA SOURCE Exit MAR-D IRRAS SARA 3ETS	SURRY	Load
Exit MAR-D IRRAS SARA SETS	Select DATA SOURCE	
MAR-D IRRAS SARA SETS	Exit	
	MAR-D IRRAS SARA SETS	

Figure 250. Load MAR-D main menu.

Before loading any data, a family must be created through the MODIFY Database Family option. Copy the files you wish to load into that family's directory (use the DOS copy command or the File Copy option under Select Family). Select that family and verify that the family name displayed in the top left corner of each menu is where the data should be loaded. If it is not correct, use the SELECT Family option again to choose the desired family.

Upon entering the LOAD module, Figure 250 will be displayed. As you cursor down through the data tools, autohelp menus will appear to the right, listing the types of data that can be loaded (Figure 251). The right arrow or < Enter > will take you to this second menu (Figure 252). Again, as you cursor down through the data types, menus will appear listing the specifics of what types of data can be loaded (Figure 253). After choosing the data to load (Figure 254), the form in Figure 255 will appear listing the files that contain that data and have the proper extension.

Each of the subsections below contains additional information on loading various types of data. In general, new data loaded into the data base overwrites old data of the same name.



Select DATA SOURCE		
Exit MAR-D IRRAS SARA SETS	Family Attributes Basic Event SYstem Event Tree ENdstate	
• Option [M]	Sequence Gate Change Sets	





à.









Figure 254. Actual data files that can be loaded.

Ð



Figure 255. List of files with the selected extension.

11.3.1 Load Verification

Consistency checking for names contained in multiple files can be done by setting "Verify when loading MAR-D data?" to "Y" using the Utility Constants option (Figure 247). Data files that have no prerequisite loaded files (*.CTD, *.FTD, *.BED, etc.) are assumed to be correct and are loaded directly into the data base without any field checking taking place. Data files having prerequisite loaded files contain dependent fields referenced in those data tiles (e.g., the *.BEI file contains event names also contained in the *.BED file). See Table ℓ for a listing of files and their prerequisite files and dependent fields.

Table 4. MAR-D files load order dependencies

Group	File	Prereq. files	Dependent fields
			2.通信的设数方式影响建设在建筑市场建设,这建筑地域和路路运用的双边市东市市方的在市场交流来说这些市场和港市市市市和新有市场市场。
Family:	.FAD	****	
	FAT	****	and the second
	FTT		And the second
NOTE: consi	stency	checking for fam	ity names is active for all files.
	******		***************************************
Attributes:	.CAD	8.898	8.8.5.X
	.CTD	6.44	4.244
	, FMD		A A A A
	.LCD	10.00	1 K K K K K K K K K K K K K K K K K K K
	.STD		14.4.4.4.
$\cdots + x + x + x + \cdots + x + x$	******	**************	* * * * * * * * * * * * * * * * * * * *
BasicEvents:	.BED	16 A A A	
	.8E1	.BED	Event Name
	.BEA	.BED	Event Name
		.CAD	Class Attr
		.CTD	Coop Type
		, FMD	Zail Mode
		LCD	Location
		.stp	System Type



11-13

.

Contra-

Systems: .FT0 .DLS .FT1 .FT0 .FT0 .FT0 .FT0 .FTA .FTA .FTA .FTA .FTA .FTA .FTA .FTA .FTA .FTO System Name .FTA NOTE: .DLS and .FTL flies are put into graphic relation instead of system relation. No cross relational cross checking available for graphics name. .EVentTrees: .ETD .ETA .ETA .ETA .ETA .ETA .ETA .ETA .ETO Event Tree Name .ETI .ETO Event Tree Name .ETI .ETI .ETI .ETI .ETI .ETI .ETI .ETI .ETI .ETI .ETI .ETI .ETI .ETI .ETI .ETI .ETI .ETI .ESI .ESI 	**********	***********		
.DLS .FTL .BED Event Name .BED Event Name .BED Event Name .FTA .FTD .FTT .FTD .FTT .FTD .FTT .FTD .FTA .FTD .FTD System Name .FT .FTD .FTA .FTD	Systems:	.FTD	5.4.8.4	MAN -
 FTL		.DLS	4444	
 FTC .FTC .FTC BED Event Name .FTA .FTD System Name .FTT .FTD Event Stional cross checking available for graphics name. EventTrees: .ETDETA .ETD Event Tree Name .ETA .ETD Event Tree Name .ETI .ETD Event Tree Name .ETI .ETD Event Tree Name .ETI .ETD Event Tree Name .ETT .ESD End State Name .SQA .ETD Event Tree Name 		.FTL	, BAD	Event Name
.BED Event Name .FTA .FTD System Name .FTT .FTD System Name .FTT .FTD System Name NOTE: .DLS and .FTL flies are put into graphic relation instead of system relation. No cross relational cross checking available for graphics name. EventTrees: .ETD .ETA .ETD Event Tree Name .ETA .ETD Event Tree Name .ETA .ETD Event Tree Name .ETI .ETO Event Tree Name .ESI ???? ???? .ESI ???? ???? .ESI ???? ???? .ESI ?ESD End State Name .SQC .ETD Event Tree Name .SQC .ETD Event Tree Name .SQL .ETD Event Tree Name		.FTC	.FTD	System Name
 FTA .FTD System Name FTT ;FTD System Name Sequences: .SQD .ETD Event Tree Name .ESC .SQA .ETD Event Tree Name .ETA .ETD Event Tree Name .ETA .ETD Event Tree Name .ETT .ETB .ETT .ETB .ETT .ETD Event Tree Name .ETT .ETD Event Tree Name .SQD Sequence Name .SQD Sequence Name .SQT .ETD Event Tree Name .SQD Sequence Name .SQD SeqUen			BED	Evens Name
.FTT .FTD System Rame NOTE: .DLS and .FTL files are put into graphic relation instead of system relation. No cross relational cross checking available for graphics name. EventTrees: .ETD .ETA .ETD .ETA .ETD .ETA .ETD .ETA .ETD Event Tree Name .ETA .ETD Event Tree Name .ETI .ETD Event Tree Name .ETI .ETD Event Tree Name .ETT .ETD Event Tree Name .ETT .ETD Event Tree Name NOTE: .ETG is currently loaded as text. .ETL should look exactly like .ETG file. NOTE: .ESI .ESI .ESI .ESI .ESI .ESI .ESI .ESI <th></th> <th>.FTA</th> <th>.FTD</th> <th>System Name</th>		.FTA	.FTD	System Name
NOTE: .DLS and .FTL files are put into graphic relation instead of system relation. No cross relational cross checking available for graphics name. EventTrees: .ETD		.FTT	FTD	System Name
EventTrees: .ETD .ETA ETD Event Tree Name .ETA .ETD Event Tree Name .ETG .ETD Event Tree Name .ETL .ETD Event Tree Name .ETI .ETD Event Tree Name NOTE: .ETG is currently loaded as textETL should look exactly like .ETG file. EndStates: .ESD .EST .ESD End State Name Sequences: .SQD .ETD Event Tree Name .SQC .ETD Event Tree Name .SQC .ETD Event Tree Name .SQL .ETD Event Tree Name	NOTE: .DLS inste check	and .FTL fli and of system (ing availab)	les are put into & relation. No cu le for graphics r	graphic relation ross relational cross name.
.ETA .ETD Event Tree Name .BEI Init Event .ETG .ETD Event Tree Name .ETL .ETO Event Tree Name .ETL .ETO Event Tree Name .ETT .ETD Event Tree Name NOTE: .ETG is currently loaded as textETL should look exactly like .ETG file. EndStates: .ESD .ESI ???? ???? .EST .ESD End State Name .SQD .ETD Event Tree Name .SQD Sequence Name .SQA .ETD Event Tree Name	EventTrees:	ETD		****
.ETG .ETG .ETD Event Tree Name .ETL .ETD Event Tree Name .ET1 .ETD Event Tree Name NOTE: .ETG is currently loaded as text. .ETL should look exactly like .ETG file. EndStates: .ESD .ESI ???? ???? .ESI ???? ???? .ESI ???? ???? .ESI End State Name Sequences: .SQD .ETD Event Tree Name .SQC .ETD Event Tree Name .SQC .ETD Event Tree Name .SQC .ETD Event Tree Name .SQA .ETD Event Tree Name .SQD Sequence Name <		ETA	FTD	Event Tree Name
.ETG .ETD Event Tree Name .ETL .ETD Event Tree Name .BEI Init Event: unless Top .FTD System Name for TOPS .ETT .ETD .ETT .ETT .ETT .ETT .ETT .ETT .ETT .ETT .ETT .ETT .ESD .EST .ESD EndStates: .SQD .SQD .ETD Event Tree Name .SQC .ETD .SQA .ETD .SQU Sequence Name .SQU Sequence Name .SQD Sequence Name .SQT .ETD .SQD Sequence Name <t< td=""><td></td><td></td><td>REI</td><td>Init Event</td></t<>			REI	Init Event
.ETL .ETD Event Tree Name .ETL .ETD Event Tree Name .ETT .ETD Event Tree Name NOTE: .ETG is currently loaded as textETL should look exactly like .ETG file. EndStates: .ESD .ESI ???? ???? .EST .ESD End State Name .SQC .ETD Event Tree Name .SQC .ETD Event Tree Name .SQL Sequence Name		ETG	ETD	Event Trae Name
.BEI Init Event: unless Top .FTD System Name for TOPS .ETT .ETD Event Tree Name NOTE: .ETG is currently loaded as textETL should look exactly like .ETG file. EndStates: .ESD .ESI ???? .EST .ESD .ETD Event Tree Name Sequences: .SQD .SQL .ETD .SQL .SQL .SQL .SQL .SQL .SQL .SQL .SQL .SQL <td></td> <td>ETL</td> <td>ETD</td> <td>Event Tree Name</td>		ETL	ETD	Event Tree Name
.ETT .ETD System Name for TOPS .ETT .ETD Event Tree Name NOTE: .ETG is currently loaded as text. .ETL should look exactly like .ETG file. EndStates: .ESD .ESI .ESI .ESI .ESI .ESI .ESI .SQD End State Name Sequences: .SQD .SQC Event Tree Name .SQC .SQD Sequence Name .SQA .SQU Sequence Name .SQ1 .SQ2 Sequence Name .SQ1			BEI	Init Event: unless Top
.ETT .ETD Event Tree Name NOTE: .ETG is currently loaded as text. .ETL should look exactly like .ETG file. EndStates: .ESD .ESI ???? ???? .ESI ???? ???? .EST .ESD End State Name Sequences: .SQD .ETD Event Tree Name .SQC .ETD Event Name .SQA .ETD Event Name .SQU Sequence Name .SQU Sequence Name .SQU Sequence Name .SQT .ETD Event Tree Name .SQU Sequence Name .SQT .ETD Event Tree Name .SQU Sequence Name .SQT .ETD Event Tree Name .SQU Sequence Name .SQD Sequence Name .SQD <td></td> <td></td> <td>FTD</td> <td>System Name for TOPS</td>			FTD	System Name for TOPS
NOTE: .ETG is currently loaded as textETL should look exactly like .ETG file. EndStates: .ESD .ESI ???? ??? .EST .ESD End State Name Sequences: .SQD .ETD Event Tree Name .SQC .ETD Event Tree Name .SQC .ETD Event Tree Name .SQL .ETD Event Name .SQL .ETD Event Tree Name		ETT	ETD	Event Tree Name
EndStates: .ESD .ESI ???? ???? .EST .ESD End State Name Sequences: .SQD .ETD Event Tree Name .SQC .ETD Event Tree Name .SQC Sequence Name .SQD Sequence Name .SQA .ETD Event Tree Name .SQU Sequence Name .SQU Sequence Name .SQU Sequence Name .SQT .ETD Event Tree Name .SQT .ETD Event Tree Name .SQD Sequence Name	NOTE: .ETG	is currently	loaded as text	ETL should look exactly like .ETG file.
Endstates: LED FILE LEST LESD FILE Sequences: LSQD LETD Event Tree Name SQC LETD Event Tree Name SQD Sequence Name BED Event Name SQA LETD Event Tree Name LSQU Sequence Name LSQU Sequence Name LSQU Sequence Name LSQU Sequence Name LSQU Sequence Name LSQU Sequence Name LSQT LETD Event Tree Name SQD Sequence Name	**************************************		***************	***************************************
.ESI 7777 .EST .ESD End State Name Sequences: .SQD .ETD Event Tree Name .SQC .ETD Event Tree Name .SQD Sequence Name .SQA .ETD Event Tree Name .SQU Sequence Name .SQU Sequence Name .SQU Sequence Name .SQU Sequence Name .SQT .ETD .SQD Sequence Name .SQT .ETD .SQD Sequence Name .SQT .ETD .SQD Sequence Name .SQD Sequence Name .SQD Sequence Name	endstates:	.ESD	0000	2010
Sequences: .SQD .ETD Event Tree Name .SQC .ETD Event Tree Name .SQC .ETD Event Tree Name .SQD Sequence Name .SQA .ETD Event Name .SQU Sequence Name .SQU Sequence Name .SQT .ETD Event Tree Name .SQT .ETD Event Tree Name .SQD Sequence Name		.ESI	1111	
Sequences: .SQD .ETD Event Tree Name .SQC .ETD Event Tree Name .SQD Sequence Name .SQA .ETD Event Name .SQA .ETD Event Tree Name .SQU Sequence Name .SQU Sequence Name .SQT .ETD .SQD Sequence Name .SQT .ETD .SQD Sequence Name .SQD Sequence Name .SQD Sequence Name		.651	.ESD	End State Name
.SQC .ETD Event Tree Name .SQD Sequence Name .BED Event Name .SQA .ETD Event Tree Name .SQU Sequence Name .ESP End State Name .SQT .ETD Event Tree Name .SQD Sequence Name	Sequences:	. SQD	ETD	Event Tree Name
SQD Sequence Name BED Event Name SQU Event Tree Name SQU Sequence Name ESP End State Name SQT ETD Event Tree Name SQD Sequence Name	3. S. A. A. 7	.SQC	.ETD	Event Tree Name
BED Event Name SQA .ETD Event Tree Name SQU Sequence Name .ESP End State Name .SQT .ETD Event Tree Name .SQD Sequence Name			. SQD	Sequence Name
SQA .ETD Event Tree Name .SQU Sequence Name .ES ^D End State Name .SQT .ETD Event Tree Name .SQD Sequence Name			, BED	Event Name
.SQU Sequence Name .ES ⁿ End State Name .SQT .ETD Event Tree Name .SQD Sequence Name		, SQA	.ETD	Event Tree Name
.ES ^D End State Name .SQT .ETD Event Tree Name .SQD Sequence Name			. 500	Sequence Name
.SQT .ETD Event Tree Name .SQD Sequence Name			.ES?	End State Name
.SQD Sequence Name		.SQT	.ETD	Event Tree Name
			.500	Sequence Name

If a dependent field does not exist in the database at load time, that field will be displayed in a confirmation/add menu. Thus, if an event name contained in a *.BEI file does not match any event name previously loaded by the *.BED file, the new name will be displayed with the option to add the event name or simply abort the load process. Note that the dependent field name cannot be edited at this point. If the name is not correct then exit, fix the mistake, and reload.

However, if you suspect your data files have multiple cross-reference errors, load the files and allow the load procedure to find the errors for you. As an error is displayed on the screen, write down the line number and file name and continue with the load. When the load is finished, make the necessary corrections to the data files, delete the database files for that family (i.e., the *.DAT, *.IOX, and *.BLK files), and restart the load procedure.

11.3.2 Fam'y

Family descriptions, attributes, and text can be loaded into the database. This information is currently stored in only the MAR-D format (see Appendix B). Note that each file contains data for only one family, and that a change in the data file family name will not change the database family name.

11.3.3 Attributes

Basic event attributes descriptions (focations, failure modes, class attributes, system types and component types) can be loaded using the MAR-D attributes option. The file format is described in

Appendix B.

11.3.4 Basic Event

Event descriptions, failure rates, and attributes can be loaded into the data base. The alternate name of the event defaults to the value of the primary name if an alternate name is not specified. The file formats are described in Appendix B.

11.3.5 SYstem

System descriptions, graphics, logic, cut sets, attributes, text and piping and instrumentation diagrams (P&IDs) can be loaded into the database. System cut sets and attributes data can be loaded into either the base case (permanent) or alternate (temporary) field areas. MAR-D database applications will normally use the base case load option because only permanent data should be logical into the database. Analysis software such as IRRAS or SARA will use alternate fields for comparisons of changed value results. The file formats are described in Appendix B.

Any basic event found within the system logic or cut sets will be added to the Event relation.

11.3.6 EVent Tree

Event tree descriptions, graphics, logic, attributes, rules, and text can be loaded into the database. The file format is described in Appendix B.

11.3.7 ENdstate

End state names, descriptions, and text can be loaded into the database. Formats for end state information have not yet been determined.

11.3.8 Sequence

Sequence cut sets for MAR-D (IRRAS, SARA) file formats can be loaded. In the MAR-D format, sequence descriptions, attributes, text, and logic can be loaded as well. With cut sets and attributes, data can be loaded into either the base case or alternate field areas.

The event tree initiating event must have been entered into the database prior to loading sequence cut sets and, therefore, does not need to be included; any found will be removed from the cut set data. The event tree initiating event will be included in each cut set term for SETS *.DNF format output.

11.3.9 Gate

Gate names and descriptions and types can be loaded into the database for use in graphics conversion. The file format is described in Appendix B.



All change set information used in the GENERATE change set option can be loaded via this file.

11.4 EXtract MAR-D Data Files

Data can be out ut in MAR-D (IRRAS, SARA) format using the EXtract MAR-D Data Files option. The extracted file is created in the current family subdirectory. The default output file name for description, information, and attribute data is the first eight characters of the family name plus the appropriate extension. For MAR-D cut set, logic, graphic, and textual information, if all files are selected for extraction, the file name will be the family name with the extension. Otherwise, the file name will be the first six characters of the name plus a 2-character ID number.

NOTE: If a file with this name already exists it will be overwritten.

The EXTRACT menus are identical with the LOAD menus (see Figure 256 through Figure 260). A data tool type is selected, then a data type, and finally the specific data to be output. In outputting some data, a fourth menu appears. Read the note at the bottom of the screen as well as the appropriate section on extracting that data type.

11.4.1 Family

Family descriptions, attributes, and text can be output from the database. Note that each file will contain information for the selected family.

11.4.2 Attributes

Event attributes (locations, failure modes, class attributes, system types, and component types) can be output from the data base. Event attributes will be output for the entire family.

11.4.3 Basic Event

Event descriptions, failure rates, and attributes can be output for all the events in a family. If alternate basic event names are used, the constant is set to "Y", and the alternate name will be used instead of the primary name for descriptions and failure rates. Primary names will be output.

11.4.4 SYstem

Descriptions and attributes can be output for all the systems in a family. System logic, graphics, and cut sets are selected from an output menu. If all logic or graphics is selected, a single file will be created for each of the names; for all cut sets and text, the family name will be used, with systems separated by ^EOS.




11.4.5 EVent Tree

Descriptions and attributes can be output for all the event trees in a family. Event tree logic, graphics, rules, attributes, and text are selected from an output menu.

11.4.6 ENdstate

End state descriptions can be output for the whole family. Text is selected from an output menu.

11.4.7 Sequence

Sequence descriptions and attributes can be output for an entire family. Sequence cut sets, logic, and text must first have an event tree selected, then that event tree's sequences are displayed and can be output. If all event trees are selected, the sequences for each event tree will be output to a file with that event tree's name. If a single event tree is selected, the output sequence menu containing all of the sequences associated with that event tree will be displayed. If all of the sequences are selected, those sequences will be output to a single file with the event tree's name. If a single sequence is selected, it is output to a file with that sequence's name.

11.4.8 Gate

Gate names and descriptions and types can be output for an entire family.

11.4.9 Change Sets

All change set information used in the GENERATE option can be output for an entire family.

For further details see Models and Results Data Base User's Guide Ver. 2.0, October 1989, EGG-CATT-8249.

Lawrence	
DATA to EXTRACT	
Exit	
MAR-D IRRAS SARA SETS	
Option E	

Figure 256. Extract main menu.

DATA to EXTRACT		
Exit		
EAIL		
MAR-D	Family	
SARA	Basic Event	
SETS	SYstem	
house a subscription of the subscription of th	Event Tree Endstate	
Option M	Sequence	
	Gate	
	enange sers	
	An open and a second second second	

Figure 257. Data types that can be extracted.













Figure 260. Selection of the data file type to extract.

11.5 Version 1.0/4.0 Interface

This option allows you to convert data files created using IRRAS Version 1.0 software to IRRAS Version 4.0 or from Version 4.0 to Version 1.0. When you invoke this option, Figure 261 will be displayed. As shown, three options are available: Exit, 1 to 4, and 4 to 1. In addition, the following keys have special functions:

< ESC > Noturns to the Ornities Menu (Figure 240)	<esc></esc>	Returns to th	ne Utilities M	enu (Figure 246)
---	-------------	---------------	----------------	------------------

<+> Returns to the Utilities Menu (Figure 246).

<→> Selects the option currently highlighted (works like < Enter >).

11.5.1 Exit

This option returns you to the Utility Menu. To invoke this option, enter an $\langle E \rangle$ in the option field or highlight Exit and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

11.5.2 Convert 1.0 to 4.0

This option allows you to convert data files created with IRRAS Version 1.0 into IRRAS Version 4.0 data files. To invoke this option, type <1> in the option field, or highlight <1 to 4> and press <Enter> or <tight arrow>. When you select this option, Figure 262 will be displayed. On this





Figure 261. Version 1.0/4.0 selection screen.

screca you must select what type of data to convert from Version 1.0 to Version 4.0. Five types of data files can be converted: Basic Event Failure Rates, Fault Tree Logic, System Cut Sets, Sequence Cut Sets, and Fault Tree Graphics.

11.5.2.1 Failure Rates. This option allows you to convert basic events and their failure rates generated in Version 1.0 to Version 4.0. Version 1.0 of IRRAS stored this information in files with a ".RAT" extension, one ".RAT" file per family. When you invoke this option, Figure 263 will be displayed, listing all 1.0 files with the ".RAT" extension. To invoke this option for a single ".RAT" file, enter an <L> in the option field, highlight the desired file name, and press < Enter>. To invoke this process for a group of files, mark the desired files using the function keys F2, F3, and F4, enter an <L> in the option field, and press < Enter>. To invoke this option for all ".RAT" files, clear all marked entries, enter an <L> in the option field, and press < Enter>. A message Process all records? (Y/N) will appear at the bottom of the screen. Type a <Y> to convert all files, or type an <N> to terminate the conversion.

11.5.2.2 Fault Trees. This option allows you to convert fault tree logic generated in Version 1.0 to Version 4.0. Version 1.0 of IRRAS stored this information in files with a "TRE" extension, one "TRE" file per fault tree. When you invoke this option, Figure 264 will be displayed, listing all 1.0 files with the "TRE" extension. To invoke this option for a single logic file, enter an <L> in the option field, highlight the desired file name, and press < Enter >. To invoke this process for a group of logic files, mark the desired files using the function keys F2, F3, and F4, enter an <L> in the option field, and press < Enter >. To invoke this option for all logic files, clear all marked entries, enter an <L> in the option field, and press < Enter >. A message Process all records? (Y/N) will appear at the bottom of the screen. Type a < Y> to convert all files, or type an < N> to terminate the conversion.



Figure 262. Load selection screen.

an a substantia di Kada na manin' para da arat da			.0 mm/ 4.0]	Failure Rates
		Option E Ex	it / Load Da	ita	
SURRY	_	Version	1.0 Files		
<esc></esc>	<f1></f1>	<f2></f2>	<f3></f3>	<f4></f4>	<f5></f5>
<esc> Exit</esc>	<f1> Help</f1>	<f2> Mark Line</f2>	<f3> Mark All</f3>	<f4> Mark Range</f4>	<f5> Locate</f5>

Figure 263. Family selection screen.



	encond			Lange and the second second		
		Option	h L Exit /	Load Logi	¢	
4KV1H ATWS2 T3	4KV1J S1 T5A	A 52 758	Version 1.0 AFV1 S3 T7	Files AFW14 T1S1	AFW17B T1SB	ATWS1 T2

Figure 264. Fault tree selection screen.

11.5.2.3 Cut Sets (FT). This option allows you to convert system cut sets generated in Version 1.0 to Version 4.0. Version 1.0 of IRRAS stored this information in files with a ".CUT" extension, one ".CUT" file per system. When you invoke this option, a screen similar to the one shown in Figure 264 will be displayed, listing all files with the ".CUT" extension. To invoke this option for a single cut set file, enter an <L> in the option field, highlight the desired file name, and press < Enter >. To invoke this process for a group of cut set files, mark the desired files using the function keys F2, F3, and F4, enter an <L> in the option field, and press < Enter >. To invoke this option for all cut set files, clear all marked entries, enter an <L> in the option field, and press < Enter >. To invoke this option for all cut set files, clear <L> in the option field, and press < Enter >. A message **Process all records?** (Y/N) will a bear at the bottom of the screen. Type a <Y> to convert all files, or type an <N> to terminate the conversion.

11.5.2.4 Sequence Cut Sets. This option allows you to convert sequence cut sets generated in Version 1.0 to Version 4.0. Version 1.0 of IRRAS stored this information in files with a ".CUT" extension, one ".CUT" file per sequence. When you invoke this option, a screen similar to the one shown in Figure 264 will be displayed, listing all files with the ".CUT" extension. To invoke this option for a single cut set file, enter an <L> in the option field, highlight the desired file name, and press < Enter >. To invoke this process for a group of cut set files, mark the desired files using the function keys F2, F3, and F4, enter an <L> in the option field, and press < Enter >. To invoke this option for all cut set files, clear all marked entries, enter an <L> in the option field, and press < Enter >. To invoke this option for all cut set files, clear all marked entries, enter an <L> in the option field, and press < Enter >. To invoke this option for all cut set files, clear all marked entries, enter an <L> in the option field, and press < Enter >. To invoke this option for all cut set files, clear all marked entries, enter an <L> in the option field, and press < Enter >. To invoke this option for all cut set files, clear all marked entries, enter an <L> in the option field, and press < Enter >. A message **Process all records?** (**Y**/**N**) will appear at the bottom of the screen. Type a < **Y** > to convert all files, or type an < **N** > to terminate the conversion.

NOTE:

IRRAS Version 1.0 does not distinguish between system and sequence cut set files. They both have the same extension. Only those ".CUT"



files that correspond to a system should be marked and loaded with the Cut Sets option. Also, only those ".CUT" files that correspond to a ...quence should be loaded with the Sequence Cut Sets option.

When you invoke this option you will be prompted to specify the event tree you wish to load the sequence cut sets under.

11.5.3 Convert 4.0 to 1.0

This option allows you to convert data files created with IRRAS Version 4.0 into IRRAS Version 1.0 data files. To invoke this option, type <2> in the option field, or highlight <4 to 1> and press < Enter > or $< \rightarrow>$. When you select this option, Figure 265 will be displayed. On this screen you must select what type of data to convert from Version 4.0 to Version 1.0. Four types of data files can be converted: basic event failure rates, fault tree logic, system cut sets, and sequence cut sets.



Figure 265. Version 4.0/1.0 selection screen.

11.5.3.1 Failure Rates. This option allows you to output basic event failure rates generated in Version 4.0 to a Version 1.0 compatible format (as ".RAT" files). To invoke this option, type an $\langle F \rangle$ in the option field, or highlight "Failure Rates" and press $\langle Enter \rangle$ or $\langle \rightarrow \rangle$ Figure 266 will be displayed. On this screen, you have three options: Exit, Current Probability Unload, or Base Probability Unload.

11.5.3.1.1 I sit—This option returns you to the 4 to 1 menu. To invoke this option type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

11.5.3.1.2 Current Probability Unload—This option unloads or outputs all the basic events of the current family along with the current probability for each of those events into the file specified by the output file name field with the ".RAT" extension. To invoke this option, enter a < C > in the option field, modify the output file name (if desired), and press < Enter >.

SURRY	Version 4.0> 1.0	Failure Rates
Option C Exit	/ Current Probability Unload / Base Prob	ability Unload

Figure 266. Current vs base probability values.

11.5.3.1.3 Base Probability Unload—This option unloads or outputs all the basic events of the current family along with the base case probability for each of those events into the file specified by the output file name field with the ".RAT" extension. To invoke this option, enter a $\langle B \rangle$ in the option field, modify the output file name (if desired), and press $\langle Enter \rangle$.

11.5.3.2 Fault Trees. This option allows you to output fault tree logic generated in Version 4.0 to a Version 1.0 compatible format (as ".TRE" files). To invoke this option, type a <T> in the option field, or highlight "Fault Trees" and press <Enter> or $<\rightarrow>$. Figure 267 will be displayed listing all the fault trees in the current family. On this screen, you have three options: Exit, Logic, or Modified Logic.

11.5.3.2.1 Exit—This option returns you to the 4 to 1 menu. To invoke this option, type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

11.5.3.2.2 Logic—This option outputs the logic as it is stored in the Version 4.0 data base to a file with the ".TRE" extension appended to the first eight characters of the fault tree name. To invoke this option for a single fault tree, enter an < L > in the option field, highlight the desired fault tree, and press <Enter >. To invoke this process for a group of fault trees, mark the desired fault trees using the function keys F2, F3, and F4, enter an < L > in the option field, and press <Enter >. To invoke this arked entries, enter an < L > in the option field, and press <Enter >. To invoke this option for all fault trees, clear all marked entries, enter an < L > in the option field, and press <Enter >. To invoke this option for all fault trees, clear all marked entries, enter an < L > in the option field, and press <Enter >. A message Process all records? (Y/N) will appear at the bottom of the screen. Type a < Y > to convert fault trees, or type an < N > to terminate the conversion.

11.5.3.2.3 Modified Logic—This option outputs the modified logic from the Version 4.0 data base, to a file with the ".TRE" extension appended to the first eight characters of the fault tree name. The modified logic is the fault tree logic as it is modified by IRRAS Version 4.0 to generate minimal cut sets. To invoke this option for a single fault tree, enter an <M> in the option field, highlight the desired fault tree, and press < Enter> To invoke this process for a group of fault trees,





Figure 267. Logic vs modified logic values.

mark the desired fault trees using the function keys F2, F3, and F4, enter an < M > in the option field, and press < Enter >. To invoke this option for all fault trees, clear all marked entries, enter an < M >in the option field, and press < Enter >. A message **Process all records**? (Y/N) will appear at the bottom of the screen. Type a < Y > to convert fault trees, or type an < N > to terminate the conversion.

11.5.3.3 Cut Sets. This option allows you to convert system cut sets generated by IRRAS Version 4.0 to Version 1.0 ".CUT" files. To invoke this option, type a <C> in the option field, or highlight "Cut Sets (FT)" and press <Enter> or $<\rightarrow>$. Figure 268 will be displayed listing all the systems in the current family. On this screen, you have three options: Exit, Basecase, or Alternate.

11.5.3.3.1 Exit—This option returns you to the 4 to 1 menu. To invoke this option type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

11.5.3.3.2 Basecase—This option outputs the base case cut sets for the systems to files with the ".CUT" extension appended to the first eight characters of the system names. To invoke this option for a single system cut set, enter a < B > in the option field, highlight the desired system, and press <Enter>. To invoke this process for a group of system cut sets, mark the desired entries using the function keys F2, F3, and F4, enter a < B > in the option field, and press <Enter>. To invoke this option for all system cut sets, clear all marked entries, enter a < B > in the option field, and press <Enter>. To invoke < S and < B and > B and < B and < B and < B and > B a

11.5.2.3.3 Alternate—This option outputs the alternate or temporary cut sets for the systems to files with the ".CUT" extension appended to the first eight characters of the system names. To invoke this option for a single system cut set, enter an < A > in the option field, highlight the desired



	www.mad has a management and a management
	Option E Exit / Basecase / Alternate
	m
c	CONTAINMENT SPRAY
CS	CONTAINMENT SYSTEMS
CV	CORE VULNERABLE TO CD
D1	HIGH PRESSURE INJECTION - AUTOMATIC
D2	HIGH PRESSURE INJECTION - MANUAL
03	HIGH PRESSURE INJECTION - RCP SEALS
D4	HIGH PRESSURE INJECTION - EMERGENCY BORATION
05	ACCUMULATORS
06	LOW PRESSURE INJECTION
F1	INSIDE SPRAY RECIRCULATION
"Esc>	<f1> <f2> <f3> <f4> <f5></f5></f4></f3></f2></f1>
Exit	Help Mark Line Mark All Mark Range Locate



system, and press $\langle \text{Enter} \rangle$. To invoke this process for a group of system cut sets, mark the desired entries using the function keys F2, F3, and F4, enter an $\langle A \rangle$ in the option field, and press $\langle \text{Enter} \rangle$. To invoke this option for all system cut sets, clear all marked entries, enter an $\langle A \rangle$ in the option field, and press $\langle \text{Enter} \rangle$. A message **Process all records?** (Y/N) will appear at the bottom of the screen. Type a $\langle Y \rangle$ to convert all system cut sets, or type an $\langle N \rangle$ to terminate the conversion.

11.5.3.4 Sequence Cut Sets. This option allows you to convert sequence cut sets generated by IRRAS Version 4.0 to Version 1.0 ".CUT" files. To invoke this option, type an $\langle S \rangle$ in the option field, or highlight "Seq. Cut Sets" and type $\langle Enter \rangle$ or $\langle \rightarrow \rangle$. Figure 269 will be displayed listing all the sequences in the current fi. On this screen, you have three options: Exit, Base Case, or Alternate.

11.5.3.4.1 Exit—This option returns you to the $\pm \infty$ 1 menu. To invoke this option type $\langle E \rangle$ in the option field and press $\langle Enter \rangle$, or press the $\langle Esc \rangle$ key.

11.5.3.4.2 Base Case — This option outputs the base case cut sets for the sequences to files with the ".CUT" extension appended to the first eight characters of the sequence names. To invoke this option for a single sequence cut set, enter a < B > in the option field, highlight the desired sequence, and press < Enter >. To invoke this process for a group of sequence cut sets, mark the desired entries using the function keys F2, F3, and F4, enter a < B > in the option field, and press < Enter >. To invoke this option for all sequence cut sets, clear all marked entries, enter a < B > in the option field, and press < Enter >. To invoke this option for all sequence cut sets, clear all marked entries, enter a < B > in the option field, and press < Enter >. To invoke this option for all sequence cut sets, clear all marked entries, enter a < B > in the option field, and press < Enter >. To invoke this option for all sequence cut sets, clear all marked entries, enter a < B > in the option field, and press < Enter >. To invoke this option for all sequence cut sets, clear all marked entries, enter a < B > in the option field, and press < Enter >. To invoke this option for all sequence cut sets, clear all marked entries, enter a < B > in the option field, and press < Enter >. A message Process all records? (Y/N) will appear at the bottom of the screen. Type a < Y > to convert all sequence cut sets, or type an < N > to terminate the conversion.

11.5.2.4.3 Alternate---This option outputs the alternate or temporary cut sets for the





sequence to files with the ".CUT" extension appended to the first eight characters of the sequence names. To invoke this option for a single sequence cut set, enter an $\langle A \rangle$ in the option field, highlight the desired sequence, and press $\langle Enter \rangle$. To invoke this process for a group of sequence cut sets, mark the desired entries using the function keys F2, F3, and F4, enter an $\langle A \rangle$ in the option field, and press $\langle Enter \rangle$. To invoke this option for all sequence cut sets, clear all marked entries, enter an $\langle A \rangle$ in the option field, and press $\langle Enter \rangle$. A message **Process all records?** (Y/N) will appear at the bottom of the screen. Type a $\langle Y \rangle$ to convert all sequence cut sets, or type an $\langle N \rangle$ to terminate the conversion.

NOTE:

IRRAS Version 1.0 does not distinguish between system and sequence cut set files. They both have the same extension. If a sequence and a system have the same name (through the first eight characters), the last one unloaded will overwrite any of the same name.



11.6 Recover Data Base

NOTE:

Before recovering the data base, it is suggested that you backup the *.DFL files and corresponding data files (*.IDX, *.DAT, and *.BLK) in the directory of the family to be rebuilt.

The Recover Data Base $\langle R \rangle$ option allows you to restructure the data base and re-index the data. Some indications that a data base rebuild is necessary include:

- 1. Data elements such as events/syste ns have been deleted and seem to reappear
- 2. During cut set generation or update, the min cut upper bound seems surprisingly high
- 3. Cross Reference reports show/don't show events being used properly
- 4. Events/systems that don't appear to be referenced cannot be deleted, and
- 5. After a software version update.

You may rebuild the data base anytime because the rebuild process compacts the data and generally helps the software run faster. The screen shown in Figure 270 appears when this option is selected. As shown, several different recovery methods are available. Each method is discussed in the following paragraphs.

houseneess of sourcestary erand and houseness		and a second
Exit		
Recover Everything		
Data Base Recovery	23.0	
Cross Reference Recovery System Logic Events System Cut Set Events Sequence Cut Set Events Fault Tree Sub Trees Sequence Logic Systems Sequence INd States Event Tree Sub Trees		



11.6.1 Recover Everything

This option performs all the recovery methods shown on the screen. This option will recover all key indexes and then recover the cross references. This option will take several minutes to complete



If ; our data base has not been damaged, this option will restructure and optimize your data base.

To invoke this option, highlight Recover Everything or enter an $\langle R \rangle$ in the option field and press $\langle Enter \rangle$. When you invoke this option, a warning screen will be displayed (see Figure 271). At the prompt, enter a $\langle Y \rangle$ and press $\langle Enter \rangle$ to continue with the recovery, or enter an $\langle N \rangle$ and press $\langle Enter \rangle$ to terminate the process.



Figure 271. Recover everything warning screen

11.6.2 Data Base Recovery

This option rebuilds all the files contained in your data base. If you think your data base has been damaged, use this option to recover all data files. If your data base has not been damaged, this option will restructure and optimize your data base.

To invoke this option, enter a < D > in the option field or highlight Data Base Recovery and press <Enter>. When you invoke this option, a warning screen will be displayed (see Figure 272). At the prompt, enter a < Y > and press <Enter> to continue with the recovery, or enter an <N> and press <Enter> to terminate the process.

11.6.3 System Logic Events

This option rebuilds the system logic events cross reference list. Each system's logic record is read and each event used is tracked. If no logic exists for a system, the list is cleared, and a message will be displayed on the screen, and the recovery process will continue. If an event is referenced, but not used, it will be added.

To invoke the option, enter an <L> in the option field or highlight System Logic Events and press <Enter>. When you invoke this option, a warning screen will be displayed (see Figure 273). At the prompt, enter a <Y> and press <Enter> to continue with the recovery, or enter an <N> and press <Enter> to terminate the process.



Figure 272. Data base recovery warning screen.



Figure 273. System logic events warning screen.

11.6.4 SYstem Cut Set Events

This option rebuilds the system cut set events cross reference list. Each system's base and current cut sets are read. The program tracks each event used. If no cut sets exist for a system, a message will be displayed on the screen and the recovery process will continue. If an event is referenced but not used, it will be added.

To invoke this option, enter a $\langle Y \rangle$ in the option field or highlight SYstem Cut Sets Events and press $\langle Enter \rangle$. When you invoke this option, a warning screen will be displayed (see Figure 274). At the prompt, enter a $\langle Y \rangle$ and press $\langle Enter \rangle$ to continue with the recovery, or enter an $\langle N \rangle$ and press $\langle Enter \rangle$ to terminate the process.





Figure 274. System cut set events warning screen.

11.6.5 Sequence Cut Set Events

This option rebuilds the sequence cut set events cross reference list. Each sequence's base and current cut sets are read. The program tracks each event used. If no cut sets exist for a sequence, a message will be displayed on the screen and the recovery process will continue. If an event is referenced but not used, it will be added.

To invoke this option, enter an $\langle S \rangle$ in the option field or highlight Sequence Cut Sets Events and press $\langle Enter \rangle$. When you invoke this option, a warning screen will be displayed (see Figure 275). At the prompt, enter a $\langle Y \rangle$ and press $\langle Enter \rangle$ to continue with the recovery, or enter an $\langle N \rangle$ and press $\langle Enter \rangle$ to terminate the process.



Figure 275. Sequence cut set events warning screen.



11.6.6 Fault Tree Sub Trees

This option rebuilds the fault tree subtree cross reference list. Each fault tree graphic relation's logic is read. The program notes any transfer in that logic. If a subtree is referenced, but does not exist in the data base, that subtree will be added to the data base.

To invoke this option, enter an $\langle F \rangle$ in the option field or highlight Fault Tree Sub Trees and press $\langle Enter \rangle$. When you invoke this option, a warning screen will be displayed (see Figure 276). At the prompt, enter a $\langle Y \rangle$ and press $\langle Enter \rangle$ to continue with the recovery, or enter an $\langle N \rangle$ and press $\langle Enter \rangle$ to terminate the process.





11.6.7 SeQuence Logic Syr

This option rebuilds under logic systems cross reference list. Each sequence's logic is read and each system used is unted. If no logic exists a message will be displayed and the recovery process will continue.

To invoke this option, enter a $\langle Q \rangle$ in the option field or highlight SeQuence Logic Systems and press $\langle Enter \rangle$. When you invoke this option, a warning screen will be displayed (see Figure 277). At the prompt, enter a $\langle Y \rangle$ and press $\langle Enter \rangle$ to continue with the recovery, or enter an $\langle N \rangle$ and press $\langle Enter \rangle$ to terminate the process.





Figure 277. Sequence logic systems warning screen.

11.6.8 Sequence ENd States

This option actually checks that the end states for each sequence are contained in the end state relation. Each sequence is read and each end state is noted. If a referenced end state does not exists, it will be added to the end state relation and the recovery process will continue.

To invoke this option, entry a < Q > in the option field or highlight Sequence ENd States and press < Enter >. When you invox e this option, a warning screen will be displayed (see Figure 278). At the prompt, enter a < Y > and press < Enter > to continue with the recovery, or enter an < N > and press < Enter > to terminate the process.



Figure 278. Sequence end states warning screen.



11.6.9 EVent Tree Sub Trees

This option rebuilds the event tree subtree cross reference list. Each event tree graphic relation's logic is read. The program notes any transfer in that logic. If a subtree is referenced, but does not exists in the data base, that subtree will be added to the data base.

To invoke this option, enter a $\langle V \rangle$ in ,the option field or highlight EVent Tree Sub Trees and press $\langle Enter \rangle$. When you invoke this option, a warning screen will be displayed (see Figure 279). At the prompt, enter a $\langle Y \rangle$ and press $\langle Enter \rangle$ to continue with the recovery, or enter an $\langle N \rangle$ and some $\langle Enter \rangle$ to terminate the process.



Figure 279. Event tree subtrees warning screen.

APPENDIX A

.

HARDWARE REQUIREMENTS AND INSTALLATION PROCEDURES

APPENDIX A

HARDWARE REQUIREMENTS AND INSTALLATION PROCEDURE

The IRRAS 4.0 system requires the following hardware configurations:

IBM-PC/XT/AT PS2 or 100% compatible 640K main memory DOS 3.3 or later 20M hard disk (minimum) Math co-processor (optional) 16 color enhanced monitor (EGA or better) graphics input devices - keyboard or mouse.

If an enhanced graphics adapter is used, it must have the memory expansion option to extend the standard 4 colors to 16. This option is an upgrade to the IBM EGA board, but is usually standard on boards manufactured by other vendors. IRRAS 4.0 does not support the 4-color mode on the EGA adapter.

The recommended configuration contains a VGA color monitor and adapter, a mouse as the graphics input device, and a math co-processor. The keyboard can serve as the graphics input device but is not nearly as user-friendly as the mouse. IRRAS will not run as fast if a math co-processor is not present.

NOTE: The IRRAS 4.0 system requires the above specified amount of random access memory (RAM). This memory must be available for the program and not used by memory resident programs, such as "Side-Kick". Also certain memory resident programs may interfere with the execution of IRRAS 4.0. These should be unloaded before using IRRAS 4.0. The MS/DOS command "CHKDSK" can be used to display the amount of memory available on your machine.

To install the system on your hard drive C, put the #1 diskette in drive A and type A:INSTALL C: < Enter >. For PS2 users with an external 5 1/4" drive, use the MS/DOS Command ASSIGN to assign your external drive as the A drive. You will be prompted when you should insert the remaining disks. Be certain to insert the disks in the order specified in the installation procedure. The directories \PRALATA_B1\DEMO, and \HALOS8 will be created and loaded with the appropriate software.

NOTE: You must have at least 10 megabytes of disk storage on your destination disk before installing IRRAS.



After the installation is completed, you need to locate and modify your CONFIG.SYS file. This file is in the root directory. If you do not have a file by this name, you must create one. In either case, the following parameters must be included in the file, if not already present.

The device driver ANSI.SYS must also be copied from the DOS directory to the root directory if it is not already there.

The installation procedure will create a batch procedure, IRRAS.BAT, for executing IRRAS 4.0 in the \PRADATA.B1 directory. This procedure may be used directly or adapted to meet your specific needs. If you choose not to modify the procedure, the following format is used to execute IRRAS 4.0. Type:

CD\PRADATA.B1 IRRAS

This completes the installation of the IRRAS 4.0 software. The user must now ensure that the proper graphics input device is hooked up and ready for use. When this is done, the IRRAS 4.0 system is ready for use. Refer to Section 11.2 for a discussion on defining constants for your configuration.





APPENDIX B

DATA INTERCHANGE FORMATS



IRRAS

PRA Models and Results Data Base Data Interchange Formats. December 18, 1991

B.1 MAR-D (IRRAS, SARA)

B.1.1 General Format Rules

1.

2.

3.

4.

All name references (family names, event names, etc.) must be upper case alphanumeric. All lower case characters will be converted to upper case. Any alpha fields that are longer than the format specified will be truncated. No spaces are allowed in the middle of names.

Descriptions can have both upper-case and lower-case characters. No character checking will be done. No commas are allowed in the description.

Commas are used as field delimiters in most formats, and can be used as placehoiders for unknown fields. Any number of leading and trailing field spaces can be inserted. Exceptions to this format are detailed as needed.

Text rules:

- File is standard ASCII text, single spaced, upper and lower case.
- First line of paragraph is indented 5 spaces, with a blank line between paragraphs.

 *EOS signals the End of Section so that multiple names in the same family can be collected in one file.

These rules apply to all files unless specifically stated otherwise.



B.1.2 Family (Plant) Information

B.1.2.1 Family Names and Descriptions.

File Name:

xxxxxx.FAD File Format: name,description

where

name - 16 character description - 60 character Family name (first 8 characters must be unique). Family description

B.1.2.2 Family Attribute File.

File Name:

XXXXXX.FAA

File Format:

name, mission, new Sum, co, loc, type, design, vendor, AE, OpDate, QualDate

where

name	- 16 character	Family name
mission	- Floating point	Default mission time in hours
newSum	- Floating point	New sequence frequency sum
co	- 10 character	Company name
loc	- 16 character	Location name
type	- 3 character	Facility type
design	- 10 character	Facility design
vendor	- 5 character	Vendor name
AE	- 10 character	Architectural Engineer
OpDate	 (vvvv/mm/dd) 	Operational date
QualDate	- (vvvv/mm/dd)	Qualification date

B.1.2.3 Family Textual Information.

	family	- 16 character	Family nam
where			
	text		
	family as		
File I	² ormat:		
	xxxxxx.FAT		
File M	Name:		





B.1.3 Basic Event Information

B.1.3.1 Basic Event Names and Descriptions.

```
File Name:
xxxxxx.BED
File Format:
family =
name,description
```

where

family - 16 character name - 16 character description - 60 character Family name Event name Alphanumeric description

B.1.3.2 Basic Event Rate Information.

The basic event failure rates are stored in the Event relation. This information can also be entered through the modify option.

```
File Name:
```

XXXXXX BEI

File Format:

family = name, calc, udC, udT, udV, prob, lambda, tau, mission, init

where

family	- 16 character	Family name				
name	- 16 character	Basic event name				
calc	- 1 character	Calculation type				
	1 - Probability					
	2 - Lambda * Mission Time					
	3 - 1 - Exp(-Lambda * Mission Time)					
	4 - Lambda * Min(Mission Time, Tsu)					
	5 - Operating component with full repair					
	6 - Lambda * Tau / 2.0					
	7 - 1 + (EXP(-Lambda*Tau)-1.0)/(Lambda*Tau)					
	8 - Base Probability * Probability					
	9 - Base Probability * Probability					
	T - Set to House Event (Failed, Prob = 1.0)					
	F - Set to House Event (Successful, Prob = 0.0)					
udC	- 4 characters Uncertainty	v correlation class				
	Events in same class are I	00% correlated				



uđT - 1	character L N B G E U	Uncertainty distribution type - Log normal, error factor - Normal, standard deviation - Beta, b of Beta(a,b) - Gamma, a Gamma(a) - Exponential, none - Uniform, Upper end pt.
udV	- Float	ing point Uncertainty distribution value
prob	- Float	ing point Probability value
lambda	- Float	ing point Basic event failure rate per hr
tau	- Float	ing point Time to repair in hours

Floating point Mission time mission -

Boolean Initiating event flag (Y/N) init - 16

General Rules:

1. The name field is mandatory.

B.1.3.3 Basic Event Attribute Codes.

Basic event attributes are entered through MODIFY--Basic Event and stored in Event.

File Name:

XXXXXX.BEA

```
File Format:
```

family =

name, Aname, type, sys, fail, loc, compID, Gname, train, att 1, ..., att 16

where

family	- 16 character	Family name		
name	- 16 character	Event name		
Aname	- 16 character	Alternate event name		
type	- 3 character	Event component type		
SVS	- 's character	Event component system		
fail	- 2 character	Failure mode		
loc	- 3 character	Component location		
compID	- 7 character	Component ID		
Gname	- 16 character	Event group identifier		
train	- 1 character	Train identifier		
att1att16	- Class attribute flags 16 values of Y			
	or N (yes or no) indicate whether			
	attribute file is at	oplicable.		

General Rules:

1. The name field is mandatory.





B.1.4.1 Failure Mode Descriptions.

File Name:

xxxxxx.FMD File Format: family = fail,description

where

family - 16 character fail - 2 character description - 60 character

Family name Failure mode identifier Failure mode description

B.1.4.2 Component Type Descriptions.

File Name:

XXXXXX.CTD

File Format:

where

family = comp, description

family - 16 character comp - 3 character description - 60 character

Family name Component type identifier Component type description

B.1.4.3 System Type Descriptions.

File Name: xxxxxx.STD File Format:

family = sys,description

where

family - 16 character sys - 3 character description - 60 character

Family name Component system identifier System description





B.1.4.4 Location Descriptions.

File Name: xxxxxx.LCD File Format: family = loc,description

where

family - 16 character loc - 3 character description - 60 character Family name Component location identifier Component location description

B.1.4.5 Class Attribute Descriptions.

File Name:

xxxxxx.CAD File Format: family = Attr#,description

where

family - 16 character Attr# - Integer 1..16 description - 60 character Family name Attribute number Attribute description



B.1.5 Fault Tree Information

B.1.5.1 Fault Tree Names and Descriptions.

- 1 character

```
File Name:
       XXXXXX, FTD
File Format:
       family =
       name.description[,s]
       + + + x + + x + +
where
       family
                       - 16 character
                                               Family name
       name
                       - 16 character
                                               Fault tree name
       description
                      - 60 character
                                               Fault tree description
```

B.1.5.2 Fault Tree Graphics.

Fault tree graphics are stored in the block data file of the Graphics relation. The MAR-D file (.DLS) is a display list sequence for the graphics in a binary format. It is loaded and output as-is with no conversion performed.

If included indicates fault tree is a subsystem

File Name:

xxxxxx.DLS

File Format:

5

IRRAS 2.5/4.0 Fault Tree Graphics file (DLS format)

B.1.5.3 Fault Tree Logic.

Fault tree logic is stored in the block data file of the Graphics relation.

File Name:

```
XXXXXX.FTL
```

1 de Format:

```
family, fault tree ==
```

```
* gatename1,description
```

```
gatename1 gatetype input1 input2 .... inputn
```

```
* gatenamen, description
```

```
Burningergereiter
```

```
gatonamen gatetype input! input2 . . . inputn
```

where

family	1	16	character	Family
fault tree	'n	16	character	Fault tre

Family name Fault tree name





gatenas	me	- 16 0	haracter	Gate name		
gatetyp	e .	- 4 c	haracter	Gate type		
	AND		= logical	AND		
	OR		= logical (= logical OR		
	TBL		- table of	- table of events		
	TRAN		= transfer			
			followed	by a 16-character fault tree name		
	NAND		= logical l	NOT AND		
	NOR		= logic NOT OR			
	N/M		= N out of M logic gate			
	CONT		= continuation of inputs to the previous gate			
input		- 16 0	character	inputs to the gate		
				(event or gate names)		
descrip	tion	- 60 c	haracter	gate name descriptions included as comment		

General Rules:

- 1. A gate definition cannot exceed 255 characters.
- 2. A line beginning with an asterisk ("*") is a comment.
- 3. For each gate name a comment should be included giving the gate description.

B.1.5.4 Fault Tree Cut Sets.

The fault tree cut sets are stored in the System relation in the block data file.

```
File Name:
```

```
xxxxx.FTC

File Format:

family, fault tree =

eventname * eventname +

eventname * eventname *

eventname +

eventname * eventname.

^EOS

family, fault tree2 =
```

where

family	 16	character
fault tree	16	character
eventname	16	character

Family name Fault tree name Event names in the cut set



General Rules:

- 1. An asterisk ("*") separates cut set events. Spaces are ignored.
- 2. A plus sign ("+") separates cut sets.
- 3. A period (".") denotes the end of a sequence.
- 4. A slash ("/") precedes complemented events.
- 5. Event names are a maximum of 16 characters including the "/".
- 6. A line beginning with an asterisk ("*") is a comment.

B.1.5.5 Fault Tree Attributes.

File Name:

XXXXX.FTA

File Format:

family =

ranciny -	
name, level, mission, mincut, proCut, sample, seed,	sizCut, sys, cuts,
events, value1,,value9	

where

amily	- 16 character	Family name
name	- 16 character	Fault tree name
evel	- Integer 2	0 = top level tree
nission	- Floating point	Mission time
nincut	- Floating point	Mincut upper bound
oroCut	- Floating point	Probability cut off value
ample	- Integer 4	Sample size
eed	- Integer 8	Random number seed
izecut	- Integer 2	Size cut off value
VS	- 3 character	System identifier
auts	- Integer 5	Base number of cut sets
vents	- Integer 5	Base number of events
alue	- Floating point	Base uncertainty values

B.1.5.6 Fault Tree Textual Information.

```
File Name:

xxxxxx.FTT

File Format:

family, fault tree =

- text -

^EOS

family, fault tree2 =

where

family - 16 character

fault tree - 16 character
```

Family name Fault tree name



B.1.6 Event Tree Information



B.1.6.1 Event Tree Names and Descriptions.

```
File Name:
```

XXXXXX ETD

File Format:

family = name,description[,s]

where

family- 16 characterFamily namename- 16 characterEvent tree namedescription- 60 characterEvent tree descriptions- 1 characterIf included indicates fault tree is a system

B.1.6.2 Event Tree Attributes.

File Name: xxxxxx.ETA

File Format:

family =

name, init

A. K. K. K. A. K. A.

where

family - 16 character name - 16 character init - 16 character Family name Event tree name Initiating event name

B.1.6.3 Event Tree Graphics.

The IRRAS Event Tree Graphics file (*,ETG) is a display list sequence for the graphics. Its format and contents are the same as the Lyent Tree Logic File.

File Name:

xxxxxx.ETG File Format: See file format for the Event Tree Logic



SAMPLE GRAPHICAL EVENT TREE



B.1.6.4 Event Tree Logic.

File Name: XXXXXX.ETL

File Format: family, event tree, init event [,T] =**^TOPS** * 1 | 2 | 3 | 4 | 5 | this is a comment ABCDE BCDEF CDEFG DEFGH EFGHI ^LOGIC +1 +2 3 +4 +5 -5 5 -4 -2 4 5 +3 -3 +4 +5-5 5 -4 +2 5 -1 3 4 -2 +3 +4 5 -4 5 -4 5 5 -4



-3

4

5

Y/N, header#2,

*SEQUENCES

Y/N, header#1, Y/N, sequence#1, Y/N, sequence#2. Y/N, sequence#3. Y/N, sequence#4. Y/N, sequence#5, Y/N, sequence#6, Y/N, sequence#7. Y/N, sequence#8, Y/N, sequence#9, Y/N, sequence#10, Y/N, sequence#11, Y/N, sequence#12. Y/N, sequence#13,

Y/N, end state#1, Y/N, end state#2, y/N, end state#3. Y/N, end state#4, Y/N, end state#5, Y/N, end state#6, Y/N, end state#7, Y/N, end state#8. Y/N, tran file#9, Y/N, end state#10,

Y/N, end state#11,

Y/N, end state#12,

Y/N, end state#13,

Y/N.xdata2#1 Y/N, xdata1#2, Y/N.xdata2#2 Y/N, xdata1#3, Y/N,xdata2#3 Y/N, xduta1#4, Y/N.xdata2#4 Y/N, xdata1#5, Y/N,xdata2#5 Y/N, xdata1#6, Y/N,xdata2#6 Y/N, xdata1#7, Y/N.xdata2#7 Y/N, xdata1#8, Y/N.xdata2#8 Y/N,xdata2#9, T Y/N, xdata1#9, Y/N, xdata1#10, Y/N,xdata2#10 Y/N, xdata1#11, Y/N.xdata2#11 Y/N, xdata1#12, Y/N,xdata2#12 Y/N, xdata1#13, Y/N,xdata2#13

Y/N.header#4

Y/N, header#3,

Y/N, xdata1#1,

TEXT

SIZE s. JUST j COLOR j XY xvalue, yvalue "60 character line of text" XY xvalue, yvalue "60 character line of text" "60 character line of text"

^PARMS

START yvalue WINDOW x1,y1,x2,y2 HEADER x1,x2,x3,x4 ^EOS family, event tree2 = (additional event trees)

where

family	- 16 character	Family name
event tree	- 16 character	Event tree na
init event	- 16 character	Initiating Eve
[,T]	- 1 character	Optional flag init event nan
mont		event system
TOPS	- 16 character	Top event/sys
Y/N	- Boolean	End state text
header	- 16 character	Sequence hea
sequence	- 16 character	Sequence nan
endstate	- 16 character	End state nan
tran file	- 16 character	Name of tran

me nt indicating ne is a Top stern names displayed? der ne ne. sfer file



xdata1	- 16 character	Information (optional)
xdata2	- 16 character	Information (optional)

General Rules:

- 1. A line beginning with an asterisk ("*") is a comment.
- 2. Literal "^TOPS", "^LOGIC", "^SEQUENCES" labels must be present.
- Logic is built according to the position of the top event in the definition. Plus sign ("+")---the specified top event succeeded. Minus sign ("-")---the specified top event failed. Blank (" ")---the response of the indicated top event did not matter.
- 4. Header, Sequence name, End State name, Xdatai, Xdata fields associated with each sequence. "Y/N" indicates whether the specified field is visible. A "T" at the end indicates the sequence transfers to another tree.
- User text is input following the 'TEXT command. Parameters include the size, justification, color, and location of the text block.
- 6. The "PARMS command allows input of program control parameters.

B.1.6.5 Event Tree Rules.

File Name:

```
xxxxxxxx.ETR
```

File Format:

```
family, event tree =
IF top1 top2 top3
THEN top4 = sys1,
top5 = sys2.
IF top3 top4 top5 THEN top3 = sys4.
```

```
*EOS
```

family, event tree2

where:

family - 16 character event tree - 16 character tops - 16 character Family name Event tree name Top event/system names

B.1.6.6 Event Tree Textual Information.

```
File Name:
xxxxxx.ETT
File Format:
family, event tree =
-- text -
^EOS
family, event tree2 =
-- text --
```


where

family - 16 character event tree - 16 character Family name Event tree name



B.1.7 End State Information

Each sequence can be tied to a single plant damage state. The end state probabilities are currently entered by the user. The name and description data are loaded with the SARA *.PDS file.

B.1.7.1 End State Names and Descriptions.

```
File Name:
```

XXXXXX.ESD

1223 233

```
File Format:
```

family = name, description

where

family	×	16	character
name	÷.	16	character
description		60	character

Family name End state name End state description

B.1.7.2 End State Information.

```
File Name:
```

xxxxxx.ESI

File Format: family =

B.1.7.3 Erd State Textual Information.



B.1.8 Sequence Information

B.1.8.1 Sequence Names and Descriptions.

```
File Name:
```

XXXXXX.SOD File Format: family, eventree = name, description

> 11-2 11 ^EOS

where

family ~ 16 character eventree - 16 character name - 16 character description - 60 character

Family name Event tree name Sequence name Sequence description

B.1.8.2 Sequence Cut Sets.

The sequence cut sets are the minimal cut sets for sequence logic as derived from the fault tree logic. The cut sets are stored in the block data file of the Sequence relation.

The MAR-D sequence cut sets (SQC) are in a format similar to that of the fault tree cut sets described in Section 5.1.5.

File Name:

```
XXXXXXX.SQC
File Format:
       family, event tree, sequence =
       eventname * eventname +
       eventname * eventname * eventname *
       eventname +
       eventname * eventname.
       *EOS
       family, event tree2, sequence2 =
```

where

amily	1.1	16	character	F
event tree		16	character	E
equence		16	character	Se
eventname		16	character	E

amily name vent tree name equence name vent names in the cut set



General Rules:

- 1. An asterisk ("*") separates events in a cut set. Spaces are ignored.
- 2. A plus sign ("+") separates cut sets.
- 3. A period (".") denotes the end of the sequence.
- 4. A slash ("/") precedes complemented events.
- 5. Event names have a maximum of 16 characters including the "/" character for complemented events.
- 6. A line beginning with an asterisk ("*") is a comment.

B.1.8.3 Sequence Attributes.

File Name:

XXXXXX.SQA

File Format:

family, event tree =

name,endstate,mincut,mission,procut,sample,seed,size,cuts, events, value1, ..., value9, default flags, used flags

```
^EOS
```

family, event tree2 ==

where

family		- 16 ch	ara	icter	Family name
event tre	e	- 16 ch	ara	icter	Event tree name
name		- 16 ch	ara	cter	Sequence name
endstate		- 16 ch	ara	icter	End State name
mincut		- Floati	ng	point	Mincut upper bound
mission		- Floati	ng	point	Mission time in hours
procut		- Floati	ng	point	Probability cut off value
sample		- Intege	r	4	Sample size
seed		- Intege	r I	8	Random number seed
size		- Intege	r i	2	Size cut off value
cuts		- Intege	r :	5	Base number of cut sets
events		- Intege	r :	5 C () - C (- Base number of events
value		- Floati	ng	point	Base uncertainty values
1	aluel		4	5th percen	tile
1	/alue2		85	Median	
1	alue3		×.	Mean	
1	alue4		e)	95th perce	ntile
	/alue5		÷.	Minimum	sample
1	alue6			Maximum	sample
1	alue7		÷.	Standard d	leviation
1	alue8			Skewness	
1	alue9			Kurtosis	
í	Default	flags	4	Default fla	ig set for this sequence
1	Jsed fl	ags	*	Flag set us	sed to generate these cut sets





B.1.8.4 Sequence Logic.

0

File Name:

XXXXXXXX SQL

File Format: family, event tree, sequence = sys1 sys2 /sys3 sys4

> *EOS family, event tree2, sequence2=

where

family	- 16 character	Family name
event tree	- 16 character	Event tree name
sequence	- 16 character	Sequence name
sys	- 16 character	System name

General Rules:

1. Complemented systems are prefixed with "/",

B.1.8.5 Sequence Textual Information.

File Name:

XXXXXX SQT

File Format:

family, event tree, sequence = --- text ---*EOS

family, event tree2, sequence2=

where

family - 16 character sequence - 16 character event tree - 16 character Family name Sequence name Event tree name

B.1.9 Piping and Instrumentation Diagrams

B.1.9.1 P&ID.

The piping and instrumentation diagrams is a graphics file in binary format. It will be loaded and output as-is: no conversion will be performed.

File Name: xxxxxx.PID File Format: (P&ID Editor format)



B.1.10 Gate

B.1.10.1 Gate Description.

File Name:

XXXXXX.GTD

File Format:

family = name, description

where

family	- 16 character	Family name
name	- 16 character	Gate name
description	- 60 character	Gate description

B.1.10.2 Gate Attributes.

File Name:

xxxxxx.GTA

File Format

family= name,attribute

where

family	- 16 character	Family name
name	- 16 character	Gate name
attribute	- 4 characters	Gate type

B.1.11 Change Sets

B.1.11.1 Change Set Description.

File Name:

xxxxxx.CSD File Format: family = name,description

where

- family name description
- 16 character
 16 character
 60 character
- Family name Change set name Change set description









B.1.11.2 Change Set Information.

File Name:

XXXXXX CSI

File Format:

family,change = ^PROBABILITY eventname,calc,udT,prop,lan,bda,tau,udV,udC,iaiasion,init ^CLASS eventname,group,compType,compId,system,location,failMode,train,init,att1,...att16 calcType,udT,prob,lambda,tau,udV,udC,mission,init ^EOS family,change2 =

where

hange	- 16 character	change set name
ventname	- 16 characters	name mask
roup	- 16 characters	event group mask
ompType	- 7 characters	component type mask
ompld	- 3 characters	component ID mask
vstem	- 3 characters	system mask
ocation	- 3 characters	location mask
ailMode	- 2 characters	failure mode mask
rain	- 2 characters	train mask
nit	- 1 character	initiating event (Y/N)
tt1att16	- Class attribute flats1	6 values of Y
	or N (yes or no) indica	te whether
	the attribute described	in the class
	attribute file is applicab	ble.
	and the second se	

calc

- 1 character

Calculation type

1 - Probability

2 - Lambda * Mission Time

3 - 1 - Exp(-Lambda * Mission Time)

4 - Lambda * Min(Mission Time, Tau)

5 - Operating component with full repair

6 - Lambda * Tau / 2.0

7 - 1+(EXP(-Lambda*Tau)-1.0)/(Lambda*Tau)

8 - Base Probability * Probability

9 - Base Probability * Probability

T - Set to House Event (Failed, Prob=1.0)

F - Set to House Event (Successful, Prob=9.0)



B-21

udT		1 character Uncertain	ty distribution type
		L - Log norm	al, error factor
		N - Normal, s	standard deviation
		B - Beta, b of	Beta(a,b)
		G - Gamma, a	a Gamma(a)
		E - Exponenti	ial, none
		U - Uniform,	Upper end pt.
prob	18.1	Floating point Probability	value
lambda		 Floating point Bas 	sic event failure rate per hr.
tau	1966	Floating point Time to rep	pair in hours
udV	1992 - P	Floating point Uncertainty	v distribution value
udC	li la S	4 characters Uncertain	ty correlation class
		Events in	same class are 100% correlated.
mission	1.4	Floating point Mission tin	ne
init		Boolean (T/F) Initiating e	vent
init		Boolean (T/F) Initiating e	vent





B.2.1 Sequences

B.2.1.1 Sequence Cut Sets.

File Name:

XXXXXX DNF.

The format of the SETS output cut sets file (.DNF) is dependent upon the command issued within SETS. The factored form is

$$A * (3 + C)$$

The disjunctive normal form is

$$A * B + A * C.$$

ONLY the disjunctive normal form is accepted by the MAR-D at this time.

File Format:

```
sequence-name =
eventName * eventName +
eventName * eventName.
```

where

1 2 3 3.

General Rules:

- 1. An asterisk ("*") separates event names. Spaces are ignored.
- 2. A plus sign ("+") separates cut sets.
- 3. A period (".") denotes the end of a sequence.
- 4. An asterisk ("*") in the first column denotes a comment.

B.2.2 Fault Trees

B.2.2.1 Fault Tree Logic.

File Name:

```
xxxxax.SET.

File Format:

FAULT TREE$ fault-tree-name.

COMMENT$ descriptive material $

gate-type $ gate-name. IN$ input-1, input-2, ..., input-n.

OUT$ output-1, output-2, ..., output-n.
```



where

fault-tree-name	- The name of the fault tree.
gate-type	- The type of gate being defined.
	AG = AND g e
	OG = OR gate
	EOR = Exclusive OR gate (converted to SC)
	EAG = Exclusive AND gate (converted to SG)
	SG = Special Gate
gate-name	- The name of the gate being defined
	(16 characters) input-n
	- The names of the gates or primary
	events that are the immediate inputs
	to the gate being defined (16 characters)
output-n	- The names of the gates that are the
	immediate outputs of the gate or primary
	event being defined (16 characters).
event-type	- The type of primary event being defined.
	BE = Basic Event
	CE = Conditional Event
	UE = Undeveloped Event
	DE = Developed Event
	EE = External Event
COMMENT\$	- Defines a comment. Must follow a "." delimiter.

B.2.2.2 Fault Tree Cut Sets.

The fault tree cut sets are stored in the System relation in the block data file. The format of the cut set file (.DNF) is given above.

B.2.3 Basic Events

B.2.3.1 Basic Event Failure Rates.

File Name:

XXXXXXX VBK

File Format:

VALUE BLOCK\$ value-block-name

```
prob $ name-list$
prob $ name-list$
```

where

prob - point value probability estimate name-list - list of event names separated by commas



B.2.4 Output Report,

Output reports can be converted to other formats or loaded by the user They are not stored intact in the database. The cut sets are stripped from the listing file (.LIS) and stored in the Sequence release. A variable occurrence table is written to file "sequence-name.VOT" in the family directory.

File Name:

xxxxxx.LIS.

File Format:

. . . Header information EXECUTE LDBLK (sequence name, sequence name, . . .)

COMTRMVAL (sequence name)

-- blank line --

/OMEGA means empty cut set

-- 12 blank lines --

Variable Occurrence Table--Output as is

-- 5 lines to cut set table --

41 character leader + 1 space + basic event name

THE MAXIMUM TERM--ends the cut sets

General Rules:

- 1. A plus sign ("+") followed by a blank line separates cut sets.
- 2. Cut set terms can be continued on seps, ate lines.
- 3. An asterisk ("*"), plus sign ("+"), or blank (" ") separates basic event names.
- 4. A period (".") denotes the last cut set.

THA WO BABTINA ECG-2664 Integrated Reliability and Risk Analysis System (IRRAS) ECG-2664 Wersion 4.0 Integrated Reliability and Risk Analysis System (IRRAS) AMPORT Non-Reference Manual Manual Man	(See instruction on the reverse)	MISSION 1. REPORT NUMBER (Assigned by NRC, Add Vol., Supp., Rev., and Addendum Nucobers, if any.) NUREG/CR-5813
Integrated Reliability and Risk Analysis System (IRRAS) Vol. 1 Reference Manual * TWORD MINISTRATION CONTROL CONTROL OF CONTR	2. TITLE AND SUBTITLE	EGG-2664
Integrated 4.0 Integrated 5.0 Integrated 5.0 Reference Manual Integrated 5.0 Integrated 5.0 Antonio Integrated 5.0 Integrated 5.0 Idaho National Engineering Laboratory *U.S. Nuclear Regulatory Commission Integrated 5.0 EQ6.0 Idaho, Inc. Washington, DC 20555 Integrated 7.0 Integrated 5.0 PONELESTINE Nuclear Regulatory Commission Washington, DC 20555 Idaho Salty Issue Resolution Office of Nuclear Regulatory Commission Washington, DC 20555 Submitting a microcomputer. This program provides functions that range from graphical fault tree and acident safety issue sequences using a microcomputer. This program provides functions that range from graphical fault tree construct to cut set generation and quantification. Version 1.0 of the IRRAS program was released in February of 1987. Since that time, man	Integrated Polishility and Dick Analysis System (IRPAS)	Vol. 1
Marin Yes Autrony 1992 Autrony 1992 K. D. Russell, M. K. McKay, M. B. Sattison, N. L. Skinner, 1.1429 Autrony 1.1429 Autrony </td <td>Version 4.0</td> <td>3. DATE REPORT PUBLISHED</td>	Version 4.0	3. DATE REPORT PUBLISHED
Reference Manual January 1992 Arbons First Global Multiple Arbons L1429 Arbons E. The Global Multiple K. D. Russell, M. K. McKay, M. B. Sattison, N. L. Skinner, Technical The Machine Multiple Technical Mathematical Multiple Technical <t< td=""><td></td><td>MONTH YEAR</td></t<>		MONTH YEAR
AUTHORNED AUTHOR	Reference Manual	January 1992
Authonsis K. D. Russell, M. K. McKay, M. B. Sattison, N. L. Skinner, S. T. Wood, D. M. Rasmuson* Improved the second set of the set		1 1420
K. D. Russell, M. K. McKay, M. B. Sattison, N. L. Skinner, S. T. Wood, D. M. Rasmuson*	5. AUTHOR(5)	6. TYPE OF REPORT
S. T. Wood, D. M. Rasmuson* Technical Technica	K. D. Russell, M. K. McKay, M. B. Sattison, N. J. Skinner	
PERCONCENCE OFFICE START OF A DOMESS ((**********************************	S. T. Wood, D. M. Rasmuson*	Technical
REPORTED 000000000000000000000000000000000000		7. PERIOD COVERED (Inclusive Date)
PERCENDINO ORGANIZATION - NAME AND ADDRESS (FWC provide Division, Office or Regins U.J. Name Fegdurer, Community, and willing address. of community, encode (Jadho National Engineering Laboratory EG&O Idaho, Inc. PO. E. Jx 1625 Jadho Palls, Idaho 83415 PORTURATION - NAME AND ADDRESS (FWC, oper "terms or where". (Functional, Office or Regins, U.J. Natione Regulatory Community, Division of Safety Issue Resolution Office of Nuclear Regulatory Commission Washington, DC 20555 Superstanding address, U.J. Natione Regulatory Commission Washington, DC 20555 Superstanding address, Segundary Commission Segundary Commission S		
Intervention and the second method of the the data of the the data of the the data of the the data of the d	PROPERTY OR OR OWNERS AND ADDRESS WARE OF DUCK OF THE DUCK	
Idaho National Engineering Laboratory *U.S. Nuclear Regulatory Commission BG&G Idaho, Inc. Washington, DC 20555 Idaho Falls, Idaho 83415 BRONZONCOMENTATION - NAME AND ADDRUSS (NRC. ope "Semicle and where", (connection, pender NRC Director, Office of National Engineering Connections, Office of National Engineering, Calibration, Office of National Engineering, Calibratica, Office of National Engineering, Calibratica, Calibrati, Calibrati, Calibrati, Calibratica, Calibratica, Cali	name and mailing eldress.)	s, and mailing address, if contractor, provide
EUGCO Idano, Inc. Washington, D.C. 20555 P.O. E. Jx 1625 Idaho Falis, Idaho 83415 Division of Safety Issue Resolution Office of Nuclear Regulatory Research U.S. Nuclear Regulatory Commission Washington, DC 20555 EUPREMEMARY NOTES CSSTRACT (not work or here) The Integrated Reliability and Risk Analysis System (IRRAS) is a state-of-the-art. microcomputer-based probabilistic risk assessment (PRA) model development and analysis tool to address key nuclear plant safety issue Requences using a microcomputer. This program provides functions that range from graphical fault tree construction to cut set generation and quantification. Version 1.0 of the IRRAS program was released in February of 1987. Since that time, many user comments and enhancements have been incorporated into the program providing a much more powerful and user-friendly system This version has been designated IRRAS 4.0 and is the subject of this Reference Manual. Version 4.0 of IRRAS provides the same capabilities as Version 1.0 and adds a relational data base facility for managing the data, improved functionality, and improved algorithm performance. REY WORDS/DESIGNEPTONS (Law works or phrases the well asset researchers is loosing the neost.)	Idaho National Engineering Laboratory *U.S. Nuclear Regulator	y Commission
P.O. L.X 1625 Idaho Falls, Idaho 83415 Stemotion Concentration - NAME AND ADDRUSS (VRC: ope "Time a show", Viewmenter, provide NECDividue, Office of Regime, U.S. Nuclear Regulatory Commission Office of Nuclear Regulatory Research U.S. Nuclear Regulatory Commission Washington, DC 20555 SUPPLEMENTARY NOTES ASSTRACT (not work or law) The Integrated Reliability and Risk Analysis System (IRRAS) is a state-of-the-art. microcomputer-based probabilistic risk assessment (PRA) model development and analysis tool to address key nuclear plant safety issue IRRAS is an integrated software tool that gives the user the ability to create and analyze fault trees and accident sequences using a microcomputer. This program provides functions that range from graphical fault tree construct to cut set generation and quantification. Version 1.0 of the IRRAS program was released in February of 1987. Since that time, many user comments and enhancements have been incorporated into the program providing a much more powerful and user-friendly system This version has been designated IRARS 4.0 and is the subject of this Reference Manual. Version 4.0 of IRRAS program vas released in Status asses facility for managing the data, improved functionality, and improved algorithm performance. NEY WORDSDEECONFIDING Law events or phrases that well asset researchers in loading the speed. NEY WORDSDEECONFIDING Law events or phrases that well asset researchers in loading the speed. NEY WORDSDEECONFIDING Law events or phrases that well asset researchers in loading the speed. 10. NUMERER OF PA	EG&G Idaho, Inc. Washington, DC 2055:	
Idamo Pails, Idamo 83413 Several Construction - NAME AND ADDREES VAC. Open "Same as above". Vision and NEC Division of Safety Issue Resolution Office of Nuclear Regulatory Research U.S. Nuclear Regulatory Commission Washington, DC 20555 BUPPLEMENTARY NOTES Additional Control of Safety Issue Resolution Office of Nuclear Regulatory Commission Washington, DC 20555 BUPPLEMENTARY NOTES Additional Control of Safety Issue Resolution The Integrated Reliability and Risk Analysis System (IRRAS) is a state-of-the-art, microcomputer-based probabilistic risk assessment (PRA) model development and analysis tool to address key nuclear plant safety issue IRRAS is an integrated software tool that gives the user the ability to create and analyze fault trees and accident sequences using a microcomputer. This program provides functions that range from graphical fault tree construct to cut set generation and quantification. Version 1.0 of the IRRAS program was released in February of 1987. Since that time, many user comments and enhancements have been incorporated into the program providing a much more powerful and user-friendly system This version has been designated IrRAS 4.0 and is the subject of this Reference Manual. Version 4.0 of IRRAS provides the same capabilities as Version 1.0 and adds a relational data base facility for managing the data, improved functionality, and improved algorithm performance. NEY WORDS/DESCRIPTIONS (An work control of planes that was assessment in exercise is loading the mood.) IRRAS, PRA, risk assessment 1. Availability Statement Unclassified 1. NUMBER OF PAGES	P.O. E JX 1625	
BIOMSOBLO GRANDATION - FAMILIAND ADDRIDS (VICE, oper "Same ar above", (Vicentrative, periods NRC Direlears, Office or Register, Vice Registery, Commission Office of Nuclear Regulatory Research U.S. Nuclear Regulatory Commission Washington, DC 20555 BUMPLEMENTARY NOTES SUSTRACT, (200 month or head) probabilistic risk assessment (PRA) model development and analysis tool to address key nuclear plant safety issue IRRAS is an integrated Software tool that gives the user the ability to create and analyze fault trees and accident sequences using a microcomputer. This program provides functions that range from graphical fault tree construct to cut set generation and quantification. Version 1.0 of the IRRAS program was released in February of 1987. Since that time, many user comments and enhancements have been incorporated into the program providing a much more powerful and user-friendly syster This version has been designated IRRAS 4.0 and is the subject of this Reference Manual. Version 4.0 of IRRAS provides the same capabilities as Version 1.0 and adds a relational data base facility for managing the data, improved functionality, and improved algorithm performance. NEY WORDBOESCRIPTORS (Lawords or phrases that well asset researches is locating the specify Unclassified 15. NUMBER OF PAGES 11. AVALABENTY STATEMENT Unlimited 16. NUMBER OF PAGES	Idano Falls, Idano 83415	
Division of Safety Issue Resolution Office of Nuclear Regulatory Research U.S. Nuclear Regulatory Commission Washington, DC 20555 AUPPLEMENTARY NOTES ASTRACT (200 works or low) The Integrated Reliability and Risk Analysis System (IRRAS) is a state-of-the-art. microcomputer-based probabilistic risk assessment (PRA) model development and analysis tool to address key nuclear plant safety issue IRRAS is an integrated software tool that gives the user the ability to create and analyze fault trees and accident sequences using a microcomputer. This program provides functions that range from graphical fault tree construct to cut set generation and quantification. Version 1.0 of the IRRAS program was released in February of 1987. Since that time, many user comments and enhancements have been incorporated into the program providing a much more powerful and user-friendly system This version has been designated IRRAS 4.0 and is the subject of this Reference Manual. Version 4.0 of IRRAS provides the same capabilities as Version 1.0 and adds a relational data base facility for managing the data, improved functionality, and improved algorithm performance. NEW WORDB/DEBCR/PPTORS (Lat words or phrases that well asset asset to some the second (The Pege) Unclassified (The	. SFONSORING ORGANIZATION - NAME AND ADDRESS (INRC, type "Same as above"; (I contractor, provide NRC Division, Office or Reg and multing address.)	tion, U.S. Nuclear Regulatory Commission,
Office of Nuclear Regulatory Research U.S. Nuclear Regulatory Commission Washington, DC 20555 BUPPLENTARY NOTES ASTRACT 2000 works or lead The Integrated Reliability and Risk Analysis System (IRRAS) is a state-of-the-art, microcomputer-based probabilistic risk assessment (PRA) model development and analysis tool to address key nuclear plant safety issue IRRAS is an integrated software tool that gives the user the ability to create and analyze fault trees and accident sequences using a microcomputer. This program provides functions that range from graphical fault tree construct to cut set generation and quantification. Version 1.0 of the IRRAS program was released in February of 1987. Since that time, many user comments and enhancements have been incorporated into the program providing a much more powerful and user-friendly syster This version has been designated IRRAS 4.0 and is the subject of this Reference Manual. Version 4.0 of IRRAS provides the same capabilities as Version 1.0 and adds a relational data base facility for managing the data, improved functionality, and improved algorithm performance. REV WORDSDEECRIPTONS (Let words or phrases that well asset researchers is loosing the report) IRRAS, PRA, risk assessment 12. AVALABILITY STATEMENT Unclassified 14. NUALABILITY STATEMENT Unclassified 15. NUMBER OF PAGES	Division of Safety Issue Resolution	
U.S. Nuclear Regulatory Commission Washington, DC 20555 aupPLEMENTARY NOTES ASUPPLEMENTARY NOTES ASUPPLEMENTARY ASUPPLEMENTARY ASUPPLEMENTARY ASUPPLEMENTARY ASUPPLEMENTARY ASUPPLEMENT ASUPPLEMENT ASUPPLEMENTARY AS	Office of Nuclear Regulatory Research	
Washington, DC 20555 BUPPLEMENTARY NOTES ASTRACT (200 words or base) The Integrated Reliability and Risk Analysis System (IRRAS) is a state-of-the-art. microcomputer-based probabilistic risk assessment (PRA) model development and analysis tool to address key nuclear plant safety issue IRRAS is an integrated software tool that gives the user the ability to create and analyze fault trees and accident sequences using a microcomputer. This program provides functions that range from graphical fault tree construct to cut set generation and quantification. Version 1.0 of the IRRAS program was released in February of 1987. Since that time, many user comments and enhancements have been incorporated into the program providing a much more powerful and user-friendly system. This version has been designated IRRAS 4.0 and is the subject of this Reference Manual. Version 4.0 of IRRAS provides the same capabilities as Version 1.0 and adds a relational data base facility for managing the data, improved functionality, and improved algorithm performance. REV WORDBOESCRIPTONS (Let words or phrases that well assist researchers in hosting the report) 13. AVAILABILITY STATEMENT Unlimited IRRAS, PRA, risk assessment 14. NUMBER OF PAGES 15. AVAILABILITY STATEMENT Unlimited Inclassified 10. NUMBER OF PAGES 16. NUMBER OF PAGES	U.S. Nuclear Regulatory Commission	
ASIMACT (200 works or leas) ASIMACT (200 works or leas) The Integrated Reliability and Risk Analysis System (IRRAS) is a state-of-the-art. microcomputer-based probabilistic risk assessment (PRA) model development and analysis tool to address key nuclear plant safety issu IRRAS is an integrated software tool that gives the user the ability to create and analyze fault trees and accident sequences using a microcomputer. This program provides functions that range from graphical fault tree construct to cut set generation and quantification. Version 1.0 of the IRRAS program was released in February of 1987. Since that time, many user comments and enhancements have been incorporated into the program providing a much more powerful and user-friendly system This version has been designated IkRAS 4.0 and is the subject of this Reference Manual. Version 4.0 of IRRAS provides the same capabilities as Version 1.0 and adds a relational data base facility for managing the data, improved functionality, and improved algorithm performance. NEW WORDBORGESCRIPTORS (Lat seeds or phrases that well asset meanschers in locating the moont) IRRAS, PRA, risk assessment	Washington, DC 20555	
ASSTRACT (200 works or level) The Integrated Reliability and Risk Analysis System (IRRAS) is a state-of-the-art. microcomputer-based probabilistic risk assessment (PRA) model development and analysis tool to address key nuclear plant safety issu IRRAS is an integrated software tool that gives the user the ability to create and analyze fault trees and accident sequences using a microcomputer. This program provides functions that range from graphical fault tree construction to cut set generation and quantification. Version 1.0 of the IRRAS program was released in February of 1987. Since that time, many user comments and enhancements have been incorporated into the program providing a much more powerful and user-friendly system. This version has been designated IrRAS 4.0 and is the subject of this Reference Manual. Version 4.0 of IRRAS provides the same capabilities as Version 1.0 and adds a relational data base facility for managing the data, improved functionality, and improved algorithm performance. KEY WORDS/CEE/CRIPTORS (Lat words or phrases that well asset researchers in boating the report) In. AVALABLITY STATEMENT Unlimited In. SEQURITY CLASSEPCATION (The Page) In. AVALABLITY STATEMENT Unclassified (The Report) Unclassified (The Report) In. AVALABLITY STATEMENT	0. BUPPLEMENTARY NOTES	
KEY WORDS/DESCRIPTORS (List words or phrases that will assist researchers in locating the report.) 13. AVAILABILITY STATEMENT IRRAS, PRA, risk assessment 14. SECURITY CLASSIFICATION (This Page) Unclassified (This Report) Unclassified (The Report) Unclassified	probabilistic risk assessment (PRA) model development and analysis tool to add IRRAS is an integrated software tool that gives the user the ability to create and sequences using a microcomputer. This program provides functions that range t	dress key nuclear plant safety issue analyze fault trees and accident
Unclassified (This Report) Unclassified 15. NUMBER OF PAGES	to cut set generation and quantification. Version 1.0 of the IRRAS program was released in February of 1987. Since the enhancements have been incorporated into the program providing a much more This version has been designated IRRAS 4.0 and is the subject of this Reference provides the same capabilities as Version 1.0 and adds a relational data base fact improved functionality, and improved algorithm performance.	rom graphical fault tree construction hat time, many user comments and powerful and user-friendly system e Manual. Version 4.0 of IRRAS ility for managing the data,
(The Report) Unclassified 15. NUMBER OF PAGES	to cut set generation and quantification. Version 1.0 of the IRRAS program was released in February of 1987. Since the enhancements have been incorporated into the program providing a much more This version has been designated IRRAS 4.0 and is the subject of this Reference provides the same capabilities as Version 1.0 and adds a relational data base fac improved functionality, and improved algorithm performance. 2. KEY WORDS/DESCRIPTORS (List words or phrases that will assist researchers in locating the report.) IRRAS, PRA, risk assessment	nom graphical fault tree construction hat time, many user comments and powerful and user-friendly system e Manual. Version 4.0 of IRRAS ility for managing the data, 13. AVALABILITY STATEMENT Unlimited 14. SECURITY CLASSIFICATION (The Page)
16. NUMBER OF PAGES	to cut set generation and quantification. Version 1.0 of the IRRAS program was released in February of 1987. Since the enhancements have been incorporated into the program providing a much more This version has been designated IRRAS 4.0 and is the subject of this Reference provides the same capabilities as Version 1.0 and adds a relational data base fac improved functionality, and improved algorithm performance. 2. KEY WORDS/DESCRIPTORS (List words or phrases that will assist researchers in locating the recort.) IRRAS, PRA, risk assessment	In time, many user comments and powerful and user-friendly system e Manual. Version 4.0 of IRRAS ility for managing the data, In AVAILABILITY STATEMENT Unlimited In SECURITY CLASSIFICATION (This Page) Unclassified
15. NUMBER OF PAGES	to cut set generation and quantification. Version 1.0 of the IRRAS program was released in February of 1987. Since the enhancements have been incorporated into the program providing a much more This version has been designated IRRAS 4.0 and is the subject of this Reference provides the same capabilities as Version 1.0 and adds a relational data base fac improved functionality, and improved algorithm performance. 2. KEY WORDS/DESCRIPTORS (List words or phrases that will assist researchers in locating the report.) IRRAS, PRA, risk assessment	Inclassified
	to cut set generation and quantification. Version 1.0 of the IRRAS program was released in February of 1987. Since the enhancements have been incorporated into the program providing a much more This version has been designated IRRAS 4.0 and is the subject of this Reference provides the same capabilities as Version 1.0 and adds a relational data base fac improved functionality, and improved algorithm performance. A KEY WORDS/DESCRIPTORS (Lat words or phrases that will assist researchers in locating the report.) IRRAS, PRA, risk assessment	In graphical fault tree construction hat time, many user comments and powerful and user-friendly system e Manual. Version 4.0 of IRRAS ility for managing the data, In availability for managing the data, In availability statement Unlimited In SECURITY CLASSIFICATION (The Page) Unclassified (The Report) Unclassified
	to cut set generation and quantification. Version 1.0 of the IRRAS program was released in February of 1987. Since the enhancements have been incorporated into the program providing a much more This version has been designated IRRAS 4.0 and is the subject of this Reference provides the same capabilities as Version 1.0 and adds a relational data base fact improved functionality, and improved algorithm performance. A KEY WORDS/DESCRIPTORS (List words or phrases that will assist measurchers in locating the report.) IRRAS, PRA, risk assessment	In graphical fault tree construction hat time, many user comments and powerful and user-friendly system e Manual. Version 4.0 of IRRAS ility for managing the data, International terms of the data, International terms of the system (This Report) Unclassified (This Report) Unclassified (15. NUMBER OF PAGES



]

